UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

☑ QUARTERLY REPORT PURSUAN	T TO SECTION 13 OR 15(d) OF THE SEC	URITIES EXCHANGE ACT OF 1934
I	For the quarterly period ended June 30, 2023	
	OR	
☐ TRANSITION REPORT PURSUAN	T TO SECTION 13 OR 15(d) OF THE SEC	URITIES EXCHANGE ACT OF 1934
	For the transition period from to	
	Commission File Number: 001-39405	
	MarketWise, Inc.	
(Exac	ct Name of Registrant as Specified in Its Cha	rter)
Delaware		87-1767914
(State or other jurisdiction of incorporation or organization)		(I.R.S. Employer Identification Number)
1125 N. Charles Street Baltimore, Maryla (Address of principal executive offices)	nd	21201 (Zip Code)
	(888) 261-2693	
	(Registrant's telephone number, including area code)	
	N/A	
(Former name	e, former address and former fiscal year, if changed since	e last report)
Securities registered pursuant to Section 12(b) of the Act:		
Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Class A common stock, \$0.0001 par value per share	MKTW	The Nasdaq Stock Market LLC
Indicate by check mark whether the registrant (1) has filed the preceding 12 months (or for such shorter period that the the past 90 days. \square Yes \square No		
Indicate by check mark whether the registrant has submitted Regulation S-T (§232.405 of this chapter) during the precediles). \square Yes \square No		
Indicate by check mark whether the registrant is large accel growth company. See the definitions of "large accelerated f 2 of the Exchange Act.		

Large accelerated filer		Accelerated filer								
Non-accelerated filer	\square	Smaller reporting company	7							
		Emerging growth company								
If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 12 (a) of the Exchange Act.										
Indicate by check mark whether the regist	rant is a shell company (as defined in Rule 12b-2 of the Exc	change Act). □ Yes ☑ No								
As of August 7, 2023, there were 37,480,6 each with a par value of \$0.0001 per share	87 shares of the registrant's Class A common stock and 289, outstanding.	9,842,303 shares of the registrant's Class B co	ommon stock,							
	2									

TABLE OF CONTENTS

		Page
PART I.	1. Financial Statements Condensed Consolidated Balance Sheets as of June 30, 2023 and December 31, 2022 (unaudited) Condensed Consolidated Statement of Operations for Three and Six Months Ended June 30, 2023 and 2022 (unaudited) Condensed Consolidated Statements of Comprehensive Income for the Three and Six Months Ended June 30, 2023 and 2022 (unaudited) Condensed Consolidated Statements of Changes in Stockholders' Deficit for the Three and Six Months Ended June 30, 2023 and 2022 (unaudited) Condensed Consolidated Statements of Cash Flows for the Six Months Ended June 30, 2023 and 2022 (unaudited) Notes to the Condensed Consolidated Financial Statements (unaudited) Management's Discussion and Analysis of Financial Condition and Results of Operations Quantitative and Qualitative Disclosures About Market Risk Controls and Procedures T. II. OTHER INFORMATION	<u>5</u>
Item 1.	<u>Financial Statements</u>	<u>5</u>
	Condensed Consolidated Balance Sheets as of June 30, 2023 and December 31, 2022 (unaudited)	<u>6</u>
		<u>8</u>
		<u>9</u>
		<u>10</u>
	Condensed Consolidated Statements of Cash Flows for the Six Months Ended June 30, 2023 and 2022	<u>13</u>
	Notes to the Condensed Consolidated Financial Statements (unaudited)	<u>14</u>
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>30</u>
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	<u>47</u>
Item 4.	Controls and Procedures	<u>47</u>
PART II.	OTHER INFORMATION	<u>48</u>
Item 1.	<u>Legal Proceedings</u>	<u>48</u>
Item 1A.	Risk Factors	<u>48</u>
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	<u>49</u>
Item 3.	Defaults Upon Senior Securities	<u>49</u>
Item 4.	Mine Safety Disclosures	<u>49</u>
Item 5.	Other Information	<u>49</u>
Item 6.	<u>Exhibits</u>	<u>50</u>
<u>Signatures</u>		51

CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS

This Quarterly Report on Form 10-Q contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, and as such are not historical facts. This includes, without limitation, statements regarding our financial position and business strategy, and the plans and objectives of management for our future operations. Such statements can be identified by the fact that they do not relate strictly to historical or current facts. When used in this report, words such as "anticipate," "believe," "continue," "could," "estimate," "expect," "intend," "may," "might," "plan," "possible," "potential," "predict," "project," "should," "strive," "would," and similar expressions may identify forward-looking statements, but the absence of these words does not mean that a statement is not forward-looking. Forward-looking statements are predictions, projections, and other statements about future events that are based on current expectations and assumptions and, as a result, are subject to risks and uncertainties. Many factors could cause actual future events to differ materially from the forward-looking statements in this report, including, but not limited to:

- our ability to attract new subscribers and to persuade existing subscribers to renew their subscription agreements with us and to purchase additional products and services from us;
- our ability to adequately market our products and services, and to develop additional products and product offerings;
- our ability to manage our growth effectively, including through acquisitions;
- failure to maintain and protect our reputation for trustworthiness and independence;
- our ability to attract, develop, and retain capable management, editors, and other key personnel;
- our ability to grow market share in our existing markets or any new markets we may enter;
- adverse or weakened conditions in the financial sector, global financial markets, and global economy;
- current macroeconomic events, including heightened inflation, rise in interest rates and the potential for an economic recession;
- failure to comply with laws and regulations or other regulatory action or investigations, including the Investment Advisers Act of 1940, as amended (the "Advisers Act");
- our ability to respond to and adapt to changes in technology and consumer behavior;
- failure to successfully identify and integrate acquisitions, or dispose of assets and businesses;
- our public securities' potential liquidity and trading;
- the impact of the regulatory environment and complexities with compliance related to such environment;
- our future capital needs;
- our ability to maintain an effective system of internal control over financial reporting, and to address and remediate existing material weaknesses in our internal control over financial reporting;
- our ability to maintain and protect our intellectual property; and
- other factors detailed under the section of our Annual Report on Form 10-K for the year ended December 31, 2022 entitled "Risk Factors."

These forward-looking statements are based on information available as of the date of this report and current expectations, forecasts, and assumptions, and involve a number of judgments, risks, and uncertainties. Additionally, as a result of a number of known and unknown risks and uncertainties, our actual results or performance may be materially different from those expressed or implied by these forward-looking statements. Accordingly, forward-

looking statements should not be relied upon as representing our views as of any subsequent date, and we do not undertake any obligation to update forward-
looking statements for any reason, except as may be required under applicable securities laws.

PART 1—FINANCIAL INFORMATION

Item 1. Financial Statements.

Condensed Consolidated Balance Sheets (Unaudited)

(In thousands, except share and per share data)

	June 30, 2023	December 31, 2022		
Assets				
Current assets:				
Cash and cash equivalents	\$ 187,022	\$	158,575	
Accounts receivable	3,138		4,040	
Prepaid expenses	10,416		11,725	
Related party receivables	1,008		1,512	
Deferred contract acquisition costs	99,854		99,960	
Other current assets	2,867		3,363	
Total current assets	304,305		279,175	
Property and equipment, net	799		892	
Operating lease right-of-use assets	8,459		9,468	
Intangible assets, net	15,180		16,047	
Goodwill	31,307		31,307	
Deferred contract acquisition costs, noncurrent	77,822		97,658	
Deferred tax assets	7,265		7,332	
Other assets	459		629	
Total assets	\$ 445,596	\$	442,508	
Liabilities and stockholders' deficit		-		
Current liabilities:				
Trade and other payables	\$ 3,127	\$	686	
Related party payables, net	422		1,004	
Accrued expenses	38,358		45,976	
Deferred revenue and other contract liabilities	288,375		315,231	
Operating lease liabilities	1,507		1,484	
Other current liabilities	22,828		21,125	
Total current liabilities	354,617		385,506	
Deferred revenue and other contract liabilities, noncurrent	340,032		348,273	
Derivative liabilities, noncurrent	1,982		1,281	
Tax receivable agreement liability, noncurrent	1,095		_	
Operating lease liabilities, noncurrent	5,121		5,831	
Total liabilities	702,847		740,891	
Commitments and Contingencies	_		_	
Stockholders' deficit:				
Common stock - Class A, par value of \$0.0001 per share, 950,000,000 shares authorized; 32,073,995 and 29,039,655 shares issued and outstanding at June 30, 2023 and December 31, 2022, respectively	3		3	
Common stock - Class B, par value of \$0.0001 per share, 300,000,000 shares authorized; 289,842,303 and 291,092,303 shares issued and outstanding at June 30, 2023 and December 31, 2022, respectively	29		29	
Preferred stock - par value of \$0.0001 per share, 100,000,000 shares authorized; 0 shares issued and outstanding at June 30, 2023 and December 31, 2022, respectively	<u>_</u>		_	
Additional paid-in capital	112,146		106,852	
Accumulated other comprehensive income	30		44	
Accumulated deficit	(126,622)		(128,125)	
	 · · · /	_	` . ,	

Condensed Consolidated Balance Sheets (Unaudited)

(In thousands, except share and per share data)

Total stockholders' deficit attributable to MarketWise, Inc.	 (14,414)	(21,197)
Noncontrolling interest	(242,837)	(277,186)
Total stockholders' deficit	 (257,251)	(298,383)
Total liabilities, noncontrolling interest, and stockholders' deficit	\$ 445,596	\$ 442,508

Condensed Consolidated Statements of Operations (Unaudited)

(In thousands, except per share data)

Three Months Ended June 30, Six Months Ended June 30, 2023 2022 2023 2022 \$ 103,363 \$ 127,466 \$ 228,978 \$ 264,086 Net revenue 899 Related party revenue 281 548 726 103,644 128,014 229,877 264,812 Total net revenue Operating expenses: Cost of revenue (1) 14,635 16,229 29,925 33,846 Sales and marketing (1) 49,033 65,050 97,760 133,287 General and administrative (1) 27,629 20,364 55,662 50,909 Research and development (1) 2,230 2,289 4,693 4,567 Depreciation and amortization 994 613 1,978 1,217 204 97 332 194 Related party expense 94,725 104,642 190,350 224,020 Total operating expenses Income from operations 8,919 23,372 39,527 40,792 238 11,923 625 19,219 Other income, net 1,013 1,551 (389)Interest income (expense), net (218)10,170 35,077 41,703 59,622 Income before income taxes Income tax expense 427 1,040 1,355 2,562 57,060 Net income 9,743 34,037 40,348 Net income attributable to noncontrolling interests 9,707 22,156 38,845 39,354 36 11,881 1,503 17,706 Net income attributable to MarketWise, Inc. 0.53 0.05 0.77 Earnings per share - basic 0.53 0.05 0.77 Earnings per share – diluted 30,526 22,288 29,853 Weighted average shares outstanding - basic 23,065 31,837 22,296 31,162 23,069 Weighted average shares outstanding - diluted

⁽¹⁾ Cost of revenue, sales and marketing, general and administrative, and research and development expenses are exclusive of depreciation and amortization shown as a separate line item

Condensed Consolidated Statements of Comprehensive Income (Unaudited)

(In thousands)

	Three Months	Ended	June 30,	Six Months Ended June 30,				
	 2023	3 2022		2023			2022	
Net income	\$ 9,743	\$	34,037	\$	40,348	\$	57,060	
Other comprehensive loss:								
Cumulative translation adjustment	(35)		(89)		(14)		(106)	
Total comprehensive income	\$ 9,708	\$	33,948	\$	40,334	\$	56,954	

Condensed Consolidated Statement of Stockholders' Deficit (Unaudited)

(In thousands, except share and per share data)

	Common Class		Common Class		Preferi	red Stock	Stockh Dei Accumulated Attrib , Additional Other t			Total Stockholders' Deficit Attributable to MarketWise,	Noncontrolling	Total Stockholders'	
	Shares	Amoun	Shares	Amount	Shares	Amount		deficit	Income (Loss)	Inc.	Interest	Deficit	
Balance at April 1, 2022	22,574,956	\$ 2	291,092,303	\$ 29	_	\$ —	\$ 88,645	\$ (140,290)	\$ (26)	\$ (51,640)	\$ (342,371)	\$ (394,011)	
Equity-based compensation	_	_	_	_	_	_	2,449	_	_	2,449	_	2,449	
Proceeds from issuance of common stock	161,178	_	_	_	_	_	517	_	_	517	_	517	
Repurchase of stock		_	_	_	_	_	(1,563)	_	_	(1,563)	_	(1,563)	
Vesting of restricted stock units	110,240	_	_	_	_	_	_	_	_	_	_	_	
Distributions	_	_	_	_	_	_	_	_	_	_	(947)	(947)	
Foreign currency translation adjustments	_	_	_	_	_	_	_	_	(89)	(89)	_	(89)	
Net income	_	_	_	_	_	_	_	11,881	_	11,881	22,156	34,037	
Balance at June 30, 2022	22,505,103	\$ 2	291,092,303	\$ 29		\$ —	\$ 90,048	\$ (128,409)	\$ (115)	\$ (38,445)	\$ (321,162)	\$ (359,607)	

Condensed Consolidated Statement of Stockholders' Deficit (Unaudited)

(In thousands, except share and per share data)

	Common Class Shares		Common Class Shares	В		red Stock Amount	Additional paid-in capital	Accumulated deficit	Accumulated Other Comprehensive Income (Loss)	Total Stockholders' Deficit Attributable to MarketWise, Inc.	Noncontrolling Interest	Total Stockholders' Deficit
Balance at April 1, 2023	29,913,490	\$ 3	291,092,303	\$ 29	_	\$ —	\$ 109,925	\$ (126,658)	\$ 65	\$ (16,636)	\$ (248,028)	\$ (264,664)
Equity-based compensation			_	_	_	_	3,678	_	_	3,678	_	3,678
Proceeds from issuance of common stock	648,849	_	_	_	_	_	332	_	_	332	_	332
Vesting of restricted stock units	261,656	_	_	_	_	_	_	_	_	_	_	_
Restricted stock units withheld to pay taxes Cash dividends declared	_	_	_	_	_	_	(476)	_	_	(476)	_	(476)
(\$0.01 per share)	_	_	_	_	_	_	(440)	_	_	(440)	_	(440)
Distributions	_	_	_	_	_	_	_	_	_	_	(5,583)	(5,583)
Issuance per redemption of Class B shares for Class A	1,250,000	_	(1,250,000)	_	_	_	(873)	_	_	(873)	1,067	194
Foreign currency translation adjustments	_	_	_	_	_	_	_	_	(35)	(35)	_	(35)
Net income								36		36	9,707	9,743
Balance at June 30, 2023	32,073,995	\$ 3	289,842,303	\$ 29		<u>\$</u>	\$ 112,146	\$ (126,622)	\$ 30	\$ (14,414)	\$ (242,837)	\$ (257,251)
	Common Class Shares		Common Class Shares	В		red Stock Amount	Additional paid-in capital	Accumulated deficit	Accumulated Other Comprehensive Income (Loss)	Total Stockholders' Deficit Attributable to MarketWise, Inc.	Noncontrolling Interest	Total Stockholders' Deficit
Balance at January 1,	24,718,402	ф Э	201 002 202	ф 20		¢	¢ 07.540	Ф (14C 11E)	f (0)	ф (40 F4F)	Ф (ЭБС 717)	Ф (40F 262)
Equity-based	24,710,402	Φ 2	291,092,303	\$ 29	_	\$ —	\$ 97,548	\$ (146,115)	\$ (9)	Ì	\$ (356,717)	Ì
compensation Proceeds from issuance of common	_	_	_	_	_	_	5,037	_	_	5,037	_	5,037
stock Repurchases	161,178	_	_	_	_	_	517	_	_	517	_	517
of stock	(2,484,717)	_	_	_	_	_	(13,054)	_	_	(13,054)	_	(13,054)
Vesting of restricted stock units Distributions	110,240	_	_	_	_	_	_	_	_ _	_ _	(3,799)	(3,799)
Foreign currency translation adjustments	_	_	_	_	_	_	_	_	(106)	(106)	_	(106)
Net income								17,706	(100)	17,706	39,354	57,060
Balance at June 30, 2022	22,505,103	\$ 2	291,092,303	\$ 29		<u>\$</u>	\$ 90,048	\$ (128,409)	\$ (115)	\$ (38,445)	\$ (321,162)	\$ (359,607)

Condensed Consolidated Statement of Stockholders' Deficit (Unaudited)

(In thousands, except share and per share data)

Total Stockholders' **Deficit** Accumulated Attributable Common Stock -Common Stock -Additional Other **Total** to Class A **Preferred Stock** Class B paid-in Accumulated Comprehensive MarketWise, Noncontrolling Stockholders' Shares **Shares Amount** deficit Income (Loss) Shares Amount Amount capital Inc. Interest **Deficit Balance** at January 1, 2023 29,039,655 291,092,303 \$ 29 \$ 106,852 \$ (128, 125)44 (21,197) \$ (277,186) (298,383)\$ 3 \$ Equity-based compensation 7,381 7,381 7,381 Proceeds from issuance of common stock 648,849 332 332 332 Vesting of restricted 1,135,491 stock units Restricted stock units withheld to (1,106)(1,106)(1,106)pay taxes Cash dividends declared (\$0.01 per share) (440)(440)(440)Distributions (5,563)(5,563)Issuance per redemption of Class B shares for Class A 1,250,000 (1,250,000)(873)(873)1,067 194 Foreign currency translation adjustments (14)(14)(14)1,503 1,503 38,845 40,348 Net income Balance at June 30, 2023 32,073,995 3 289,842,303 \$ \$ (126,622) \$ \$ 29 \$ 112,146 30 (14,414) \$ (242,837) \$ (257,251)

${\bf MARKETWISE, INC.}$

Condensed Consolidated Statements of Cash Flows (Unaudited)

(In thousands)

	\$ 40,348 \$ 1,978 7,381 701 1,356 (26) 1,053					
	2023	2022				
Cash flows from operating activities:						
Net income	\$ 40,348 \$	57,060				
Adjustments to reconcile net income to net cash provided by operating activities:						
Depreciation and amortization	1,978	1,217				
Impairment of right-of-use assets	_	287				
Stock-based compensation	7,381	5,037				
Change in fair value of warrant liabilities and other derivative liabilities		(19,654)				
Deferred taxes	1,356	2,454				
Unrealized gains on foreign currency	(26)	(132)				
Noncash lease expense	1,053	906				
Changes in operating assets and liabilities:						
Accounts receivable	902	4,495				
Related party receivables and payables, net	(78)	(479)				
Prepaid expenses	1,309	3,802				
Other current assets and other assets	666	(615)				
Deferred contract acquisition costs	19,942	(5,617)				
Trade and other payables	2,408	(2,804)				
Accrued expenses	(7,618)	(2,586)				
Deferred revenue	(35,097)	(9,120)				
Operating lease liabilities	(731)	(981)				
Other current and long-term liabilities	 (1,648)	(5,408)				
Net cash provided by operating activities	 32,846	27,862				
Cash flows from investing activities:						
Purchases of property and equipment	(46)	(26)				
Capitalized software development costs	(913)	(81)				
Net cash used in investing activities	 (959)	(107)				
Cash flows from financing activities:						
Proceeds from related party notes receivable, net	_	294				
Proceeds from issuance of common stock	332	517				
Repurchases of stock	_	(13,054)				
Restricted stock units withheld to pay taxes	(1,106)	_				
Distributions to noncontrolling interests	 (2,652)	(3,799)				
Net cash used in financing activities	(3,426)	(16,042)				
Effect of exchange rate changes on cash	(14)	(106)				
Net increase in cash, cash equivalents and restricted cash	28,447	11,607				
Cash, cash equivalents and restricted cash — beginning of period	158,575	139,578				
Cash, cash equivalents and restricted cash — end of period	\$ 187,022 \$	151,185				

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

1. Organization

MarketWise, Inc. ("MarketWise," "the Company," "we," "us," or "our") is a holding company that has no material assets other than its ownership in MarketWise, LLC, and operates and controls all of the businesses and operations of MarketWise, LLC and its subsidiaries. The Company provides independent investment research for investors around the world. We believe we are a leading content and technology multi-brand platform for self-directed investors. We offer a comprehensive portfolio of high-quality, independent investment research, as well as several software and analytical tools, on a subscription basis.

While our headquarters are in Baltimore, Maryland, we operate multiple subsidiaries in the United States.

On July 21, 2021, MarketWise, Inc. became a publicly traded company as a result of completing the business combination, recapitalization and other transactions with Ascendant Digital Acquisition Corp. ("ADAC" or "Sponsor"), a special purpose acquisition company, MarketWise, LLC, all of the members of MarketWise, LLC (the "MarketWise Members"), and Shareholder Representative Services LLC (collectively, the "Transactions").

2. Summary of Significant Accounting Policies

Basis of Consolidation

The accompanying condensed consolidated financial statements include the accounts of MarketWise, Inc. and its subsidiary, MarketWise, LLC, a variable interest entity ("VIE") for which MarketWise, Inc. is deemed to be the primary beneficiary.

MarketWise, Inc. is a holding company that owns a minority economic interest in MarketWise, LLC but, through its role as the managing member of MarketWise, LLC, controls all of the business and operations of MarketWise, LLC. Therefore, MarketWise, LLC and its subsidiaries are included in the Company's condensed consolidated financial statements. As of June 30, 2023, MarketWise, Inc. had a 10.0% ownership interest in MarketWise, LLC.

The Company determined that MarketWise, LLC is the primary beneficiary of a VIE, Stansberry Pacific Research, and therefore, the assets, liabilities, and results of operations of that VIE are included in the Company's condensed consolidated financial statements. For more information on Stansberry Pacific Research, see *Note 12 – Variable Interest Entities*.

The condensed consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP"). All intercompany balances and transactions have been eliminated in consolidation.

The accompanying statements of operations include certain expenses incurred by, and charged to us by, a related party for general corporate services. These expenses are based primarily on direct usage when identifiable, direct capital expenditures, or other relevant allocations during the respective periods. We believe the assumptions underlying the accompanying condensed consolidated financial statements, including the assumptions regarding these expenses from this related party, are reasonable. Actual results may differ from these expenses, assumptions and estimates. The amounts recorded in the accompanying condensed consolidated financial statements are not necessarily indicative of the actual amount of such indirect expenses that would have been recorded had we been a separate independent entity.

Unaudited Interim Financial Information

The accompanying unaudited condensed consolidated financial statements and the related footnote disclosures have been prepared by us in accordance with GAAP for interim financial reporting and as required by Rule 10-01 of Regulation S-X. Accordingly, the unaudited condensed consolidated financial statements may not include all of the information and notes required by GAAP for audited financial statements. The year-end December 31, 2022 consolidated balance sheet data included herein was derived from audited financial statements but does not include

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

all disclosures required by GAAP for complete financial statements. In the opinion of management, the accompanying unaudited condensed consolidated financial statements contain all adjustments, consisting of items of a normal and recurring nature, necessary to present fairly our financial position as of June 30, 2023, the results of operations, comprehensive income, stockholders' deficit, and cash flows for the three and six months ended June 30, 2023 and 2022. The results of operations for the three and six months ended June 30, 2023 and 2022 are not necessarily indicative of the results to be expected for the full year. The information contained herein should be read in conjunction with the audited financial statements included in our Annual Report on Form 10-K for the year ended December 31, 2022 filed with the Securities and Exchange Commission ("SEC"). Management considers events or transactions that occur after the balance sheet date but before the financial statements are issued to provide additional evidence relative to certain estimates or to identify matters that require additional disclosure. Subsequent events have been evaluated through the date of issuance of these condensed financial statements.

Use of Estimates

The preparation of condensed consolidated financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities as of the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Significant estimates and assumptions made in the accompanying condensed consolidated financial statements include, but are not limited to, the fair value of common units, derivatives, warrants, valuation of assets acquired and liabilities assumed in business combinations, useful lives of intangible assets with definite lives, benefit period of deferred contract acquisition costs, determination of standalone selling prices, estimated life of lifetime customers, recoverability of goodwill and long-lived assets, valuation allowances on deferred tax assets, the incremental borrowing rates to calculate lease liabilities and right-of-use ("ROU") assets and certain accruals. We evaluate our estimates and assumptions on an ongoing basis using historical experience and other factors and adjust those estimates and assumptions when facts and circumstances dictate. Actual results could differ from those estimates.

Emerging Growth Company

We are an "emerging growth company," as defined in Section 2(a) of the Securities Act of 1933, as amended (the "Securities Act"), as modified by the Jumpstart Our Business Startups Act of 2012 (the "JOBS Act"), and we may take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies including, but not limited to, not being required to comply with the independent registered public accounting firm attestation requirements of Section 404 of the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in its periodic reports and proxy statements, and exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved.

Further, Section 102(b)(1) of the JOBS Act exempts emerging growth companies from being required to comply with new or revised financial accounting standards until private companies (that is, those that have not had a Securities Act registration statement declared effective or do not have a class of securities registered under the Securities Exchange Act of 1934, as amended) are required to comply with the new or revised financial accounting standards. The JOBS Act provides that an emerging growth company can elect to opt out of the extended transition period and comply with the requirements that apply to non-emerging growth companies but any such election to opt out is irrevocable. We have elected not to opt out of such extended transition period, which means that when a standard is issued or revised and it has different application dates for public or private companies, we, as an emerging growth company, can adopt the new or revised standard at the time private companies adopt the new or revised standard. This may make comparison of our unaudited condensed financial statements with another public company that is neither an emerging growth company nor an emerging growth company that has opted out of using the extended transition period difficult or impossible because of the potential differences in accounting standards used.

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

Noncontrolling Interest

Noncontrolling interest represents the Company's noncontrolling interest in consolidated subsidiaries which are not attributable, directly or indirectly, to the controlling Class A common stock ownership of the Company.

Net income for the three and six months ended June 30, 2023 and 2022 was attributable to consolidated MarketWise, Inc. and its respective noncontrolling interests.

As of June 30, 2023, MarketWise, Inc.'s controlling interest in MarketWise, LLC was 10.0% and the noncontrolling interest was 90.0%. For the three months ended June 30, 2023, net income attributable to controlling interests included a \$0.4 million tax provision, and for the six months ended June 30, 2023, net income attributable to controlling interests included a \$1.4 million tax provision, both of which were 100% attributable to the controlling interest.

As of June 30, 2022, MarketWise, Inc.'s controlling interest in MarketWise, LLC was 7.2% and the noncontrolling interest was 92.8%. For the three months ended June 30, 2022, net income attributable to controlling interests included a \$12.1 million gain on warrant liabilities and a \$1.0 million tax provision, and for the six months ended June 30, 2022, net income attributable to controlling interests included a \$18.8 million gain on warrant liabilities and a \$2.6 million tax provision, both of which were 100% attributable to the controlling interest.

3. Revenue Recognition

Disaggregation of revenues

The following table depicts the disaggregation of revenue according to customer type and is consistent with how we evaluate our financial performance. We believe this depicts how the nature, amount, timing and uncertainty of revenue and cash flows are affected by economic factors.

Tiffee World States July 2025									
Subscriptions	Advertising	Revenue Share (Related Party)		Revenue Share (Third- party)			Total		
\$ 103,148	\$	_	\$		\$	_	\$	103,148	
_		169		281		46		496	
\$ 103,148	\$	169	\$	281	\$	46	\$	103,644	
\$	\$ 103,148 —	\$ 103,148 \$ 	Subscriptions Advertising \$ 103,148 \$ — — 169	Subscriptions Advertising \$ 103,148 \$	SubscriptionsAdvertisingRevenue Share (Related Party)\$ 103,148\$ —\$ ——169281	Subscriptions Advertising (Related Party) \$ 103,148 \$ — \$ — \$ — 169 281	SubscriptionsAdvertisingRevenue Share (Related Party)Revenue Share (Third-party)\$ 103,148\$\$16928146	SubscriptionsAdvertisingRevenue Share (Related Party)Revenue Share (Third-party)\$ 103,148\$ —\$ —\$ —-16928146	

		11110	C 141	onidis Ended June 30,	2022			
	 Subscriptions	Advertising		Revenue Share (Related Party)		Revenue Share (Third- party)		Total
Timing of transfer:								
Transferred over time	\$ 127,231	\$ _	\$	_	\$	_	\$	127,231
Transferred at a point in time	_	160		548		75		783
Total	\$ 127,231	\$ 160	\$	548	\$	75	\$	128,014

Three Months Ended June 30, 2022

	Six Months Ended June 30, 2023										
	Subscriptions		Advertising		Revenue Share (Related Party)	Reve	nue Share (Third- party)	Total			
Timing of transfer:											
Transferred over time	\$ 228,560	\$	_	\$	_	\$	— \$	228,560			
Transferred at a point in time	 		316		899		102	1,317			
Total	\$ 228,560	\$	316	\$	899	\$	102 \$	229,877			

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

Six Months Ended June 30, 2022

	Sul	oscriptions	Advertising	Revenue Share (Related Party)	Reve	nue Share (Third- party)	Total
Timing of transfer:							
Transferred over time	\$	263,331	\$ _	\$ _	\$	_	\$ 263,331
Transferred at a point in time			437	726		318	1,481
Total	\$	263,331	\$ 437	\$ 726	\$	318	\$ 264,812

Revenue recognition by subscription type was as follows:

		Three Months	End	ed June 30,	Six Months Ended June 30,				
		2023	2022			2023	2022		
Membership subscriptions	\$	41,409	\$	49,852	\$	89,811	\$	101,304	
Term subscriptions		61,739		77,379		138,749		162,027	
Non-subscription revenue		496		783		1,317		1,481	
Total	\$	103,644	\$	128,014	\$	229,877	\$	264,812	

Revenue for the membership and term subscription types are determined based on the terms of the subscription agreements. Non-subscription revenue consists of revenue from advertising and other revenue from revenue share arrangements and the sale of print products and events, such as webinars and conferences.

Net revenue by principal geographic areas was as follows:

		Three Months	Ende	ed June 30,	Six Months Ended June 30,				
	2023			2022	2023			2022	
United States	\$	103,644	\$	128,001	\$	229,786	\$	264,712	
International		_		13		91		100	
Total	\$	103,644	\$	128,014	\$	229,877	\$	264,812	

Revenue by location is determined by the billing entity for the customer.

Contract Balances

The timing of revenue recognition, Billings, cash collections and refunds affects the recognition of accounts receivable, contract assets and deferred revenue. Our current deferred revenue balance in the condensed consolidated balance sheets includes an obligation for refunds for contracts where the provision for refund has not lapsed. Accounts receivable, deferred revenue and obligation for refunds are as follows:

		As of					
	June 30, 2023			December 31, 2022			
Contract balances							
Accounts receivable	\$	3,138	\$	4,040			
Obligations for refunds	\$	3,866	\$	4,676			
Deferred revenue – current	\$	284,509	\$	310,555			
Deferred revenue – non-current	\$	340,032	\$	348,273			

We recognized \$82,614 and \$83,644 of revenue during the three months ended June 30, 2023 and 2022, respectively, and \$186,649 and \$193,394 during the six months ended June 30, 2023 and 2022, respectively, that was included within the beginning contract liability balance of the respective periods. The Company has collected all amounts included in deferred revenue other than \$3,138 and \$4,040 as of June 30, 2023 and December 31, 2022, respectively, related to the timing of cash settlement with our credit card processors.

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

Assets Recognized from Costs to Obtain a Contract with a Customer

The following table presents the opening and closing balances of our capitalized costs associated with contracts with customers:

Balance at January 1, 2023	\$ 197,618
Royalties and sales commissions – additions	15,049
Revenue share and cost per acquisition fees – additions	19,637
Amortization of capitalized costs	 (54,628)
Balance at June 30, 2023	\$ 177,676

We did not recognize any impairment on capitalized costs associated with contracts with customers for the three and six months ended June 30, 2023 and 2022.

Remaining Performance Obligations

As of June 30, 2023, the Company had \$628,407 of remaining performance obligations presented as deferred revenue in the condensed consolidated balance sheets. We expect to recognize approximately 46% of that amount as revenues over the next twelve months, with the remainder recognized thereafter.

4. Goodwill and Intangible Assets, Net

Goodwill

The carrying amounts of goodwill are as follows:

Balance at December 31, 2022	\$ 31,307
Balance at June 30, 2023	\$ 31,307

Intangible assets, net

Intangible assets, net consisted of the following as of the dates indicated:

		June 30, 2023								
	 Accumulated Cost Amortization		Net Book Value		Weighted-Average Remaining Useful Life (in years)					
Finite-lived intangible assets:										
Customer relationships	\$ 21,718	\$	(11,241)	\$	10,477	5.1				
Tradenames	4,287		(2,499)		1,788	5 . 5				
Capitalized software development costs	3,974		(2,146)		1,828	2.4				
Finite-lived intangible assets, net	29,979		(15,886)		14,093					
Indefinite-lived intangible assets:										
Internet domain names	1,087		_		1,087					
Indefinite-lived intangible assets, net	1,087		_		1,087					
Intangible assets, net	\$ 31,066	\$	(15,886)	\$	15,180					

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

December 31, 2022

	· · · · · · · · · · · · · · · · · · ·							
		Cost		Accumulated Amortization		Net Book Value	Weighted-Average Remaining Useful Life (in years)	
Finite-lived intangible assets:								
Customer relationships	\$	21,718	\$	(9,924)	\$	11,794	5.4	
Tradenames		4,287		(2,265)		2,022	5.8	
Capitalized software development costs		3,002		(1,858)		1,144	2.0	
Finite-lived intangible assets, net		29,007		(14,047)		14,960		
Indefinite-lived intangible assets:								
Internet domain names		1,087		_		1,087		
Indefinite-lived intangible assets, net		1,087		_		1,087		
Intangible assets, net	\$	30,094	\$	(14,047)	\$	16,047		

We recorded amortization expense related to finite-lived intangible assets of \$925 and \$531 for the three months ended June 30, 2023 and 2022, respectively, and \$1,033 for the six months ended June 30, 2023 and 2022, respectively, within depreciation and amortization in the accompanying condensed consolidated statement of operations. These amounts include amortization of capitalized software development costs of \$146 and \$130 for the three months ended June 30, 2023 and 2022, respectively, and \$289 and \$235 for the six months ended June 30, 2023 and 2022, respectively.

We recorded additions to capitalized software development costs of \$590 and \$972 for the three and six months ended June 30, 2023, respectively. We recorded additions to capitalized software development costs of \$50 and \$81 for the three and six months ended June 30, 2022, respectively.

As of June 30, 2023, the total expected future amortization expense for finite-lived intangible assets is as follows:

Remainder of 2023	\$ 1,699
2024	3,983
2025	2,539
2026	2,268
2027	1,647
Thereafter	 1,957
Finite-lived intangible assets, net	\$ 14,093

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

5. Fair Value Measurements

The following tables summarize our financial assets and liabilities measured at fair value on a recurring basis by level within the fair value hierarchy as of the dates indicated:

Level 1

June 30, 2023

Level 3

As of

Level 2

Aggregate Fair Value

Assets:						
Money market funds	\$ 90,560	\$		<u> </u>	\$	90,560
Total assets	90,560		_	_		90,560
Liabilities:						
Derivative liabilities, noncurrent	 _		<u> </u>	1,982		1,982
Total liabilities	\$ 	\$		1,982	\$	1,982
	 	-			-	
			December 3	31, 2022		
	 Level 1	Le	evel 2	Level 3	Aggregat	e Fair Value
Assets:						
Money market funds	\$ 80,327	\$		<u> </u>	\$	80,327
Total assets	80,327		_	_		80,327
Liabilities:						
Derivative liabilities, noncurrent	 			1,281		1,281

The level 3 liabilities relate to certain employee contracts with embedded derivatives, see *Note 7 – Derivative Financial Instruments*.

The following table provides quantitative information regarding Level 3 fair value measurements inputs at their measurement dates:

	June 30, 2023	Decemb	oer 31, 2022
Volatility	39.70 %		37.30 %
Discount rate	22.20 %		21.80 %
Credit spread	2.00 %		0.20 %
Risk-free rate	3.50 %		3.50 %
The following table summarizes the change in fair value of the liabilities during the six months ended June	ne 30, 2023 and 2022:		
Balance at January 1, 2023		\$	1,281
Change in fair value of derivative instruments			701
Balance at June 30, 2023		\$	1,982
Balance at January 1, 2022		\$	31,347
Change in fair value of derivative instruments			(19,654)
Balance at June 30, 2022		\$	11,693

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

6. Balance Sheet Components

Capitalized Implementation Costs

We capitalized cloud computing implementation costs for customer-relationship management, revenue management, and enterprise resource planning systems of \$66 and \$492 for the three months ended June 30, 2023 and 2022, respectively, and \$71 and \$1,032 for the six months ended June 30, 2023 and 2022, respectively. The capitalized implementation costs are capitalized within other current assets and other assets on the condensed consolidated balance sheets. Amortization expense related to capitalized cloud computing implementation costs was \$229 and \$10, for the three months ended June 30, 2023 and 2022, respectively, and \$451 and \$12 for the six months ended June 30, 2023 and 2022, respectively.

Property and Equipment, Net

Property and equipment, net consists of the following:

			As	s of	
	Estimated Useful Lives	June 30, 2023			December 31, 2022
Furniture and fixtures	5 years	\$	960	\$	960
Computers, software and equipment	3 years		1,504		1,458
Leasehold improvements	Shorter of estimated useful life or remaining term of lease		1,278		1,278
			3,742		3,696
Less: Accumulated depreciation and amortization			(2,943)		(2,804)
Total property and equipment, net		\$	799	\$	892

Depreciation and amortization expense for property and equipment was \$69 and \$81 for the three months ended June 30, 2023 and 2022, respectively, and \$139 and \$184 for the six months ended June 30, 2023 and 2022, respectively.

Accrued Expenses

Accrued expenses consist of the following:

	As of						
		June 30, 2023	December 31, 2022				
Commission and variable compensation	\$	20,211	\$	24,207			
Payroll and benefits		4,568		5,258			
Other accrued expenses		13,579		16,511			
Total accrued expenses	\$	38,358	\$	45,976			

7. Derivative Financial Instruments

Prior to the closing of the Transactions, as part of our compensation and employee retention strategy, we entered into contracts with key employees and independent contractors which contain embedded derivatives. These contracts are intended to compensate the employees or independent contractors for services provided and retain their future services. These embedded derivative instruments are issued in the form of phantom interests in Net Income, as defined by our board of directors (the "Board"), that grant the holder value equal to a percentage of Net Income *multiplied* by a price multiple. All derivative instruments are recorded at fair value as derivative liabilities on our condensed consolidated balance sheets.

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

As of June 30, 2023, there is one embedded derivative instrument outstanding. The following table presents information on the location and amounts of derivative instrument gains and losses:

		Three Months	Ended	June 30,		June 30,		
		2023 2022				2023		2022
Derivatives Not Designated as Hedging Instruments	Location of Gain (Loss) Recognized in Income Statement							
Warrants	Other income, net	\$ _	\$	12,082	\$	_	\$	18,799
Phantom Interests in Net Income	General and administrative	310		441		(701)		855
Total		\$ 310	\$	12,523	\$	(701)	\$	19,654

See Note 5 – Fair Value Measurements for more information regarding the valuation of our derivative instruments.

8. Stock-Based Compensation

Included within cost of revenue, sales and marketing, and general and administrative expenses are total stock-based compensation expenses as follows:

		Three months	ende	ed June 30,	Six months ended June 30,					
	2023			2022	-	2023		2022		
Cost of revenue	\$	683	\$	525	\$	1,691	\$	1,059		
Sales and marketing		716		577		1,808		1,142		
General and administrative		2,279		1,347		3,882		2,836		
Total stock based-compensation expense	\$	3,678	\$	2,449	\$	7,381	\$	5,037		

Total stock-based compensation expense for the three and six months ended June 30, 2023 and 2022 includes expenses related to our MarketWise, Inc. 2021 Incentive Award Plan (the "2021 Incentive Award Plan") and our 2021 Employee Stock Purchase Plan ("ESPP"), as follows:

		Three months	ende	ed June 30,	Six months ended June 30,					
	<u> </u>	2023		2022	2023			2022		
2021 Incentive Award Plan	\$	3,587	\$	2,312	\$	7,190	\$	4,742		
Employee Stock Purchase Plan		91		137		191		295		
Total stock-based compensation expense	\$	3,678	\$	2,449	\$	7,381	\$	5,037		

2021 Incentive Award Plan

As of June 30, 2023, The Company has reserved a total of 33,960,802 shares of MarketWise Class A common stock for issuance pursuant to the 2021 Incentive Award Plan.

During the six months ended June 30, 2023, we granted 4,525,609 restricted stock units ("RSUs") and 715,000 fully vested shares of Class A common stock to certain employees and service providers in aggregate under our 2021 Incentive Award Plan.

For employees and service providers, both RSUs and stock appreciation rights ("SARs") are primarily time-based and typically vest ratably over four years, as specified in the individual grant notices. The RSUs may entitle the recipients to dividend equivalents if approved by the Plan Administrator, which are subject to the same vesting terms and accumulate during the vesting period. Upon vesting, the RSU holder will be issued a corresponding number of the Company's Class A common stock. The SARs will be settled in the Company's Class A common

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

stock upon exercise. The shares to be issued upon exercise of a SAR will have a total market value equal to the SAR value calculated as (x) number of shares underlying such SAR, multiplied by (y) any excess of the Company's share value on the date of exercise over the exercise price set in each individual grant notice.

The fair value of the RSU is the same as the Company's share price on the date of grant. The fair value of the SARs was determined using a Black-Scholes model.

The activities of the RSUs and SARs and the related weighted average grant-date fair value of the respective share classes, including granted, exercised and forfeited, from January 1, 2023 to June 30, 2023, are summarized as follows:

	RS	SUs		SA	SARs					
	Units	W	eighted-Average Grant Date Fair Value	Units	Weighted-Average Grant Date Fair Value					
Outstanding at January 1, 2023	6,261,543	\$	3.45	1,747,473	\$	4.05				
Granted	4,525,609		1.84	_		_				
Exercised or vested	(1,488,543)		2.02	_		_				
Forfeited	(567,488)		2.43	(84,226)		4.05				
Outstanding at June 30, 2023	8,731,121	\$	2.89	1,663,247	\$	4.05				
Exercisable at June 30, 2023		\$	_	417,523	\$	4.05				

The stock compensation expense related to the RSU and SAR grants was \$2,443 and \$2,312 for the three months ended June 30, 2023 and 2022, respectively, and \$6,046 and \$4,742 for the six months ended June 30, 2023 and 2022, respectively. As of June 30, 2023, 417,523 of the SARs were exercisable and they have a remaining contractual term of 8.2 years. For the fully vested shares, during the six months ended June 30, 2023, we issued 417,684 shares of Class A common stock after withholding for taxes, resulting in compensation expense of \$1,144.

Employee Stock Purchase Plan

As of June 30, 2023, The Company has reserved for issuance a total of 6,728,300 shares of Class A common stock for the ESPP. The most recent offering period began on January 1, 2023 and ended on June 30, 2023.

The Company recognized \$91 and \$137 of stock-based compensation expense related to the ESPP during the three months ended June 30, 2023 and 2022, respectively, and \$191 and \$295 for the six months ended June 30, 2023 and 2022, respectively. On each purchase date, eligible employees may purchase the shares at a price per share equal to 85% of the lesser of the fair market value of the Company's Class A common stock on (1) the first trading day of the offering period, or (2) the last trading day of the offering period. The Company issued 231,165 shares of Class A common stock for \$332,184 related to employee purchases under the ESPP on June 30, 2023.

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

9. Earnings Per Share

The following tables set forth the computation of basic and diluted earnings per share for the three and six months ended June 30, 2023 and 2022:

	Three months ended June 30,					Six months ended June 30,				
		2023		2022		2023		2022		
Basic earnings per share:										
Numerator:										
Net income	\$	9,743	\$	34,037	\$	40,348	\$	57,060		
Less: Net income attributable to noncontrolling interests	_	9,707		22,156		38,845		39,354		
Net income attributable to Class A common stockholders	\$	36	\$	11,881	\$	1,503	\$	17,706		
Denominator:										
Weighted average shares outstanding (in thousands)		30,526	_	22,288		29,853		23,065		
Basic earnings per share	\$	<u> </u>	\$	0.53	\$	0.05	\$	0.77		
Diluted earnings per share:										
Numerator:										
Net income	\$	9,743	\$	34,037	\$	40,348	\$	57,060		
Less: Net income attributable to noncontrolling interests		9,707		22,156		38,845		39,354		
Net income attributable to Class A common stockholders	\$	36	\$	11,881	\$	1,503	\$	17,706		
Denominator:										
Weighted average shares outstanding (in thousands)		31,837	_	22,296	_	31,162	_	23,069		
Diluted earnings per share	\$	_	\$	0.53	\$	0.05	\$	0.77		

The Company's potentially dilutive securities and their impact on the computation of diluted earnings per share is as follows:

- Public and private placement warrants: the public and private placement warrants were "out of the money" during the three and six months ended June 30, 2022, therefore, net income per share excludes any impact of the 20,699,993 public warrants and 10,280,000 private placement warrants. The warrants were exchanged in September 2022 and there are no warrants outstanding as of June 30, 2023.
- Sponsor and MarketWise management member earnout shares: the 3,051,000 Sponsor earnout shares and the 2,000,000 MarketWise management member earnout shares granted in connection with the closing of the Transactions held in escrow are excluded from the earnings per share computation in both periods since the earnout contingency has not been met.
- Restricted stock units: The basic earnings per share calculation includes the impact of vested RSUs as of June 30, 2023 and June 30, 2022. The diluted earnings per share calculation includes the impact of dilutive RSUs and excludes the impact of antidilutive RSUs for the three and six months ended June 30, 2023 and 2022. The diluted earnings per share calculation for the three and six months ended June 30, 2023 excludes

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

certain RSUs with performance conditions, since the performance conditions have not been met as of June 30, 2022 and such RSUs were forfeited as of June 30, 2023.

- Stock appreciation rights: The diluted earnings per share calculation for the three and six months ended June 30, 2023 and 2022 excludes the impact of SARs since the effect was antidilutive.
- ESPP: The basic earnings per share calculation includes the impact of the shares that were issued under the ESPP as of June 30, 2023 and 2022.

10. Income Taxes

We are subject to U.S. federal and state taxes with respect to our allocable share of any taxable income or loss of MarketWise, LLC, as well as any stand-alone income or loss we generate. MarketWise, LLC is treated as a partnership for U.S. income tax purposes and for most applicable state and local income tax purposes and generally does not pay income taxes in most jurisdictions. Instead, MarketWise, LLC's taxable income or loss is passed through to its members, including us.

Our effective income tax rate was 4.2% and 3.3% for the three and six months ended June 30, 2023, respectively, and 3.0% and 4.3% for the three and six months ended June 30, 2022, respectively. The main driver of the effective income tax rate for the three and six months ended June 30, 2022 was the income allocation to the non-controlling interest. The main driver of the increase in the effective tax rate from the three months ended June 30, 2022 is the change in the income allocation to the non-controlling interest. Our effective tax rate for the three and six months ended June 30, 2023 differs from the U.S. federal statutory rate primarily because we generally do not record income taxes for the noncontrolling portion of pre-tax income.

As a result of the reverse capitalization in 2021, we recorded a deferred tax asset resulting from the outside basis difference in our interest in MarketWise, LLC. The Company considers both positive and negative evidence when measuring the need for a valuation allowance. A valuation allowance is not required to the extent that, in management's judgment, positive evidence exists with a magnitude and duration sufficient to result in a conclusion that it is more likely than not (a likelihood of more than 50%) that the Company's deferred tax assets will be realized.

In evaluating the need for a valuation allowance on the deferred tax asset, the company considered positive evidence related to its historic earnings, forecasted income and reversal of temporary differences. Therefore, the Company recorded a valuation allowance in the amount of \$29,088 for certain deferred tax assets that are not more likely than not to be realized.

As part of the reverse capitalization in 2021, we entered into a Tax Receivable Agreement ("TRA") with certain stockholders. Pursuant to our election under Section 754 of the Internal Revenue Code of 1986, as amended, and the regulations issued thereunder, we expect to receive a step up in the tax basis of our ownership in MarketWise, LLC as exchanges of the LLC Units (as defined herein) occur. These increases in tax basis may reduce the amount we may otherwise pay to various tax authorities in the future. The TRA liability will represent approximately 85% of the calculated tax savings based on basis adjustments and other carryforward attributes assumed that we anticipate being able to utilize in future years. We will retain the remaining 15% of calculated tax savings. The payments contemplated by the TRA are not conditioned upon any continued ownership interest in MarketWise, LLC. The timing and amount of aggregate payments due pursuant to the TRA may vary based on several factors including the timing and amount of future taxable income, as well as future applicable tax rates. As such, significant inputs and assumptions are used to estimate the future expected payments under the TRA. During the three months ended June 30, 2023, Members of MarketWise, LLC exchanged an aggregate of 1,250,000 common units of MarketWise, LLC ("LLC Units") together with an equal number of shares of Class B common stock for 1,250,000 newly-issued shares of Class A common stock. As a result, we have recorded a liability of \$1,095 under the TRA as of June 30, 2023. No payments have been made under the TRA and no payments are expected in the next twelve months.

As of June 30, 2023, we had no unrecognized tax positions and believe there will be no changes to uncertain tax positions within the next 12 months.

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

11. Related Party Transactions

We have certain revenue share agreements with related parties. Accordingly, we recognized revenue from related parties of \$281 and \$548 for the three months ended June 30, 2023 and 2022, respectively, and \$899 and \$726 for the six months ended June 30, 2023 and 2022, respectively.

We incurred costs related to revenue share agreements with related parties which are capitalized within deferred contract acquisition costs. We capitalized \$935 and \$847 for the three months ended June 30, 2023 and 2022, respectively, and \$2,087 and \$1,770 for the six months ended June 30, 2023 and 2022, respectively.

Additionally, a related party provided call center support and other services to the Company for which we recorded an expense within cost of revenue of \$217 and \$289 for the three months ended June 30, 2023 and 2022, respectively, and \$353 and \$494 for the six months ended June 30, 2023 and 2022, respectively.

A related party provided marketing and copywriting services to the Company for which we recorded an expense within cost of revenue of \$332 and \$585 for the three and six months ended June 30, 2023, respectively.

A related party also provided certain corporate functions to the Company and the costs of these services are charged to us. We recorded \$27 and \$22 for the three months ended June 30, 2023 and 2022, respectively, and \$54 and \$43 for the six months ended June 30, 2023 and 2022, respectively, within related party expense in the accompanying condensed consolidated statement of operations. We held balances of \$296 and \$1,043 as of June 30, 2023 and December 31, 2022, respectively, of related party payables related to revenue share expenses, call center support, and the services noted above. The balances with our related party are presented net and are included in related party payables, net in the condensed consolidated balance sheet.

We earned fees and provided certain accounting and marketing services to companies owned by certain of MarketWise's Class B stockholders. As a result, we recognized \$205 and \$86 in other income, net for the three months ended June 30, 2023 and 2022, respectively, and \$507 and \$253 for the six months ended June 30, 2023 and 2022, respectively. Related party receivables related to these services were \$442 and \$403 as of June 30, 2023 and December 31, 2022, respectively.

We lease offices from related parties. Lease payments made to related parties were \$435 and \$528 for the three months ended June 30, 2023 and 2022, respectively, and \$868 and \$919 for the six months ended June 30, 2023 and 2022, respectively, and rent expense of \$595 and \$588 were recognized in general and administrative expenses for the three months ended June 30, 2023 and 2022, respectively, \$1,200 and \$1,141 for the six months ended June 30, 2023 and 2022, respectively, related to leases with related parties. At June 30, 2023 and December 31, 2022, ROU assets of \$8,295 and \$9,210 and lease liabilities of \$6,454 and \$7,041, respectively, are associated with leases with related parties.

In April 2020 we provided a loan to a related party and recognized a related party note receivable from the unitholder of \$1,148. We recognized \$6 and \$10 in interest income for the three and six months ended June 30, 2022, respectively. This loan was repaid in October 2022.

12. Variable Interest Entities

We consolidated a VIE based on our ability to exercise power and being the primary beneficiary of the entity, including directing the operations and marketing campaigns and sharing customer lists and publications, as of June 30, 2023 and December 31, 2022. There have been no reconsideration events during these periods. The assets of consolidated variable interest entities may only be used to settle obligations of these entities. In addition, there is no

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

recourse to MarketWise for the consolidated VIE's liabilities. The following represents financial information for the consolidated VIE included in the condensed consolidated balance sheets:

		As of						
	June 30, 2023			December 31, 2022				
Current assets	\$	1,106	\$	3,678				
Noncurrent assets		196		300				
Total assets	\$	1,302	\$	3,978				
	-							
Current liabilities	\$	269	\$	227				
Total liabilities	\$	269	\$	227				

13. Supplemental Cash Flow Information

Supplemental cash flow disclosures are as follows:

	Six Months Ended June 30,				
	2023	20)22		
Supplemental Disclosures of Cash Flow Information:					
Cash paid for interest	\$ 479	\$	403		
Cash paid for amounts included in the measurement of lease liabilities:					
Operating cash flows from operating leases	(994)		(1,259)		
Operating lease right-of-use assets obtained in exchange for lease obligations	(76)		(795)		
Supplemental Disclosures of Non-Cash Investing and Financing Activities:					
Capitalized software included in accounts payable	59		167		
Recording of deferred tax assets and capital contribution related to exchanges of Class B common stock to Class A common stock	1,289		_		
	As of Ju	ne 30,			

	As of June 30,					
	2023 2022					
Reconciliation of Cash and Cash Equivalents and Restricted Cash:						
Cash and cash equivalents	\$ 187,022	\$	151,185			
Restricted cash	_		_			
Total	\$ 187,022	\$	151,185			

14. Stockholders' Equity

The Company's capital stock consists of (i) issued and outstanding Class A common stock with a par value of \$0.0001 per share, and (ii) issued and outstanding Class B common stock with a par value of \$0.0001 per share.

The table set forth below reflects information about the Company's equity, as of June 30, 2023. The 3,051,000 Sponsor earnout shares held in escrow and the 2,000,000 management member earnout shares are considered contingently issuable shares and therefore excluded from the number of Class A common stock issued and outstanding in the table below.

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

	Authorized	Issued	Outstanding
Common stock – Class A	950,000,000	32,073,995	32,073,995
Common stock – Class B	300,000,000	289,842,303	289,842,303
Preferred stock	100,000,000	_	_
Total	1,350,000,000	321,916,298	321,916,298

Each share of Class A and Class B common stock entitles the holder to one vote per share. Only holders of Class A common stock have the right to receive dividends. In the event of liquidation, dissolution or winding up of the affairs of the Company, only holders of Class A common stock have the right to receive liquidation proceeds, while the holders of Class B common stock are entitled to only the par value of their shares. Class B common stock can be issued only to members of MarketWise, LLC, their respective successors and permitted transferees. Under the terms of the Third Amended and Restated Limited Liability Company Operating Agreement of MarketWise, LLC (the "MarketWise Operating Agreement"), and subject to certain restrictions set forth therein, the MarketWise Members are entitled to have their LLC Units redeemed or exchanged for shares of our Class A common stock, at our option. If redeemed for cash at the Company's option, such cash would have to be generated through an offering of shares to the market such that there would not be any situation where there would be a net cash obligation to the Company for such redemption. Shares of our Class B common stock held by any such redeeming or exchanging MarketWise Member will be canceled for no additional consideration on a one-for-one basis with the redeemed or exchanged LLC Units whenever such MarketWise Member's LLC Units are so redeemed or exchanged. The MarketWise Members may exercise such redemption rights for as long as their LLC Units remain outstanding. Our Board has discretion to determine the rights, preferences, privileges and restrictions, including voting rights, dividend rights, conversion rights, redemption privileges and liquidation preferences, of each series of preferred stock.

On May 11, 2023, our Board declared a cash dividend to its Class A common stockholders in the amount of \$0.01 per share, totaling \$0.4 million, and a cash distribution to holders of LLC Units in the amount of \$0.01 per unit, totaling \$2.9 million, to stockholders and unitholders of record as of the close of business on June 1, 2023.

Certain of the dividends are attributable to earnout shares and dividend equivalents are attributable to unvested RSUs, and are therefore forfeitable. The Company recorded the full dividend payable as a charge to additional paid-in capital on the declaration date, and forfeitures are reversed as they occur. On July 20, 2023, \$0.3 million of the dividends and the full distribution of \$2.9 million was paid to stockholders and unitholders, respectively.

The Company intends to pay dividends quarterly in the future, subject to Board approval of any such dividends. Dividends payable and distributions payable are included in other current liabilities in the condensed consolidated balance sheet.

During the three and six months ended June 30, 2023, MarketWise Members exchanged an aggregate of 1,250,000 LLC Units together with an equal number of shares of the Company's Class B common stock for 1,250,000 newly-issued shares of the Company's Class A common stock. As a result of the exchanges, the Company recognized adjustments to non-controlling interests. As of June 30, 2023, MarketWise, Inc.'s controlling interest in MarketWise, LLC was 10.0% and the noncontrolling interest was 90.0%.

On November 4, 2021, our Board authorized the repurchase of up to \$35.0 million in aggregate of shares of the Company's Class A common stock, with the authorization to expire on November 3, 2023. We did not repurchase any shares during the three and six months ended June 30, 2023. We repurchased 341,271 shares totaling \$1.6 million in the aggregate, including fees and commissions of \$3, and 2,484,717 shares totaling \$13.1 million in aggregate, including fees and commissions of \$25, for the three and six months ended June 30, 2022, respectively. Since the inception of the program we have repurchased 2,984,987 total shares. The maximum dollar value of shares that may yet to be purchased under the plan was \$18.6 million as of June 30, 2023.

For each share of Class A common stock the Company repurchases under the share repurchase program, MarketWise, LLC, the Company's direct subsidiary, will redeem one LLC Unit held by the Company, decreasing

Notes to the Condensed Consolidated Financial Statements (Unaudited)

(Dollar amounts in thousands, except share, unit, per share, and per unit data)

the percentage ownership of MarketWise, LLC by the Company and relatively increasing the ownership by the other unitholders.

15. Subsequent Events

Subsequent events have been evaluated through August 10, 2023, which is the date that the financial statements were issued.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

The following discussion and analysis of the financial condition and results of operations of MarketWise, Inc., a Delaware corporation ("MarketWise," "we," "us," and "our"), should be read together with our unaudited condensed consolidated financial statements and related notes appearing elsewhere in this Quarterly Report on Form 10-Q and our audited consolidated financial statements and the related notes and the discussion under the heading "Management's Discussion and Analysis of Financial Condition and Results of Operations" in our Annual Report on Form 10-K for the fiscal year ended December 31, 2022 (the "Annual Report"). The following discussion contains forward-looking statements. Our actual results may differ significantly from those projected in the forward-looking statements. Factors that might cause future results to differ materially from those projected in the forward-looking statements include, but are not limited to, those discussed in the section entitled "Cautionary Statement Regarding Forward-Looking Statements" in this report.

Overview

We are a leading multi-brand platform of subscription businesses that provides premium financial research, software, education, and tools for self-directed investors. We offer a comprehensive portfolio of high-quality, independent investment research, as well as several software and analytical tools, on a subscription basis.

MarketWise started in 1999 with the simple idea that, if we could publish intelligent, independent, insightful, and in-depth investment research and treat the subscriber the way we would want to be treated, then subscribers would renew their subscriptions and stay with us. Over the years, we have expanded our business into a comprehensive suite of investment research products and solutions. We now produce a diversified product portfolio from a variety of financial research brands such as Stansberry Research, Palm Beach Research, Chaikin Analytics, InvestorPlace, and Empire Financial Research. Our entire investment research product portfolio is 100% digital and channel agnostic, and we offer all of our research across a variety of platforms, including desktop, laptop, and mobile devices, including tablets and mobile phones.

Today, we benefit from the confluence of a leading editorial team, diverse portfolio of content and brands, and comprehensive suite of investor-centric tools that appeal to a broad subscriber base.

Key Factors Affecting Our Performance

We believe that our growth and future success are dependent upon several factors, including those below and those noted in the "Risk Factors" section in the Annual Report. The key factors below represent significant business opportunities as well as challenges that we must successfully address in order to continue our growth and improve our financial results.

Growing our subscriber base with compelling unit economics. We are highly focused on continuing to acquire new subscribers to support our long-term growth. Our marketing spend is a large driver of new subscriber growth. At the heart of our marketing strategy is our compelling unit economics that combine long-term subscriber relationships, highly scalable content delivery, cost-effective customer acquisition, and high-margin conversions.

Our Paid Subscribers as of December 31, 2022 generated average customer lifetime Billings of approximately \$1,815, resulting in a LTV/CAC (as defined below) ratio of approximately 2.0x. On average, it takes us approximately 0.6 to 1.5 years for a Paid Subscriber's cumulative net revenue to exceed the total cost

of acquiring that subscriber (which includes fixed costs, such as marketing salaries). For more information on our LTV/CAC ratio and the components of this ratio, see "—Definitions of Metrics."

We adjust our marketing spend to drive efficient and profitable customer acquisition. We can adjust our marketing spend in near real-time, and we monitor costs per acquisition relative to the cart value of the initial subscription. We seek 90-day payback periods to cover this variable component of the direct marketing spend.

As of June 30, 2023, our Paid Subscriber base was 750 thousand, down 148 thousand, or 16.4% as compared to 898 thousand at June 30, 2022. Our Paid Subscriber base is comprised of subscribers obtained through both direct-to-paid acquisition and free-to-paid conversions. Since 2019, direct-to-paid acquisition has accounted for approximately two-thirds of our annual Paid Subscriber acquisition, and is largely driven by display ads and targeted email campaigns.

Our free subscription products also serve as a significant source of new Paid Subscribers, accounting for approximately one-third of our annual Paid Subscriber acquisition since 2019. Our free-to-paid and active free-to-paid conversion rate reflects the rate at which Free Subscribers and Active Free Subscribers purchase paid subscription products. Our annual free-to-paid and annual active free-to-paid conversion rates were approximately 1% to 2% and 2% to 4%, respectively, between 2020 and 2022. Over that same three year period, our cumulative free-to-paid and cumulative active free-to-paid conversion rates were 4% and 10%, respectively.

Retaining and expanding relationships with existing subscribers. We believe that we have a significant opportunity to expand our relationships with our large base of Free and Paid Subscribers. Thanks to the quality of our products, we believe our customers will continue their relationship with us and extend and increase their subscriptions over time. As we deepen our engagement with our subscribers, our customers tend to purchase more and higher-value products. Our ARPU (as defined below) as of June 30, 2023 was \$490, which decreased 15.6% from \$580 as of June 30, 2022. For more information on ARPU, see "Key Business Metrics—Average Revenue Per User."

Conversion rates are important to our business because they are an indicator of how engaged and how well we are connecting with our subscribers. The time it takes our customers to move from our free products to our lower-priced paid subscriptions and eventually to high-end products and membership "bundled" offerings impacts our growth in net revenue, Billings, and ARPU.

Our high-value composition rate reflects the rate at which Paid Subscribers that have purchased less than \$600 of our products over their lifetime convert into subscribers that have purchased more than \$600. We believe our high-value composition rate reflects our ability to retain existing subscribers through renewals and our ability to expand our relationship with them when those subscribers purchase higher-value subscriptions. Our ultra high-value composition rate reflects the rate at which Paid Subscribers that have purchased more than \$600 of our products over their lifetime convert into subscribers that have purchased more than \$5,000. We believe our ultra high-value composition rate reflects our ability to successfully build lifetime relationships with our subscribers, often across multiple products and brands. These high-value and ultra high-value subscribers are at the heart of our business model, and are particularly important in periods of time when we are bringing in fewer new subscribers. These high-value and ultra high-value subscribers continue to see the value in our products and are continuing to purchase products at rates similar to historic trends, which is currently driving the majority of our Billings. As of June 30, 2023, our high-value composition rate and ultra high-value composition rate were 48% and 39%, respectively.

Definitions of Metrics

Throughout this discussion and analysis, a number of our financial and operating metrics are referenced which we do not consider to be key business metrics, but which we review to monitor performance, and which we believe may be useful to investors. These are:

Annual free-to-paid and annual active free-to-paid conversion rates: The Company has historically defined free-to-paid conversion rate as the number of Free Subscribers who purchased a subscription during the period divided by the average number of Free Subscribers during the period. In addition, the Company has

historically defined active free-to-paid conversion rate as the number of Active Free Subscribers who purchased a subscription during the period divided by the average number of Active Free Subscribers during the period. There have been no changes to the calculation of free-to-paid or active free-to-paid conversion rates. However, the Company has revised the definition of free-to-paid and active free-to-paid conversion rates to provide clarity in how they are calculated.

We calculate our free-to-paid conversion rate as the sum of (i) the number of Free Subscribers who purchased their first subscription during the period and (ii) the number of Free Subscribers who purchased their first subscription during the period after six months of not having an active paid subscription across all of MarketWise *divided by* the average number of Free Subscribers during the period. We calculate our active free-to-paid conversion rate as the sum of (i) the number of Active Free Subscribers who purchased their first subscription during the period and (ii) the number of Active Free Subscribers who purchased their first subscription during the period and active paid subscription across all of MarketWise *divided by* the average number of Active Free Subscribers during the period. We believe our free-to-paid and active free-to-paid conversion rates are indicators of the type of Free Subscribers that we are signing up and the quality of our content and marketing efforts. Investors should consider free-to-paid and active free-to-paid conversion rates as two of the factors in evaluating our ability to maintain a robust pipeline for new customer acquisition.

Cumulative free-to-paid and cumulative active free-to-paid conversion rates: We calculate our cumulative free-to-paid conversion rate as the sum of (i) the number of Free Subscribers who purchased their first subscription during the trailing three-year period and (ii) the number of Free Subscribers who purchased their first subscription during the trailing three-year period after six months of not having an active paid subscription across all of MarketWise divided by the average number of Free Subscribers during the trailing three-year period. We calculate our cumulative active free-to-paid conversion rate as the sum of (i) the number of Active Free Subscribers who purchased their first subscription during the trailing three-year period and (ii) the number of Active Free Subscribers who purchased their first subscription during the trailing three-year period after six months of not having an active paid subscription across all of MarketWise divided by the average number of Active Free Subscribers during the trailing three-year period.

High-value composition rate: Our high-value composition rate reflects the number of Paid Subscribers who have purchased >\$600 in aggregate over their lifetime as of a particular point in time *divided by* the total number of Paid Subscribers as of that same point in time.

Landing Page Visits: The cumulative number of visits to our standalone web pages created specifically for each marketing campaign. We believe landing page visits are a measure of customer engagement.

LTV/CAC ratio: We calculate LTV/CAC ratio as LTV divided by CAC. We use LTV/CAC ratio because it is a standard metric for subscription-based businesses, and we believe that an LTV/CAC ratio above 3x is considered to be indicative of strong profitability and marketing efficiency. We believe that an increasing LTV per subscriber reflects our existing subscribers recognizing our value proposition, which will expand their relationship with us across our platform over time, either through a combination of additional product purchases or by joining our membership offerings. Investors should consider this metric when evaluating our ability to achieve a return on our marketing investment. Lifetime value ("LTV") represents the average margin on average customer lifetime Billings (that is, the estimated cumulative spend across a customer's lifetime). Customer acquisition cost ("CAC") is defined as direct marketing spend, plus external revenue share expense, plus retention and renewal expenses, plus copywriting and marketing salaries, plus telesales salaries and commissions, plus customer service commissions.

Net revenue retention: Net revenue retention is defined as Billings from all prior period cohorts in the current period, *divided by* all Billings from the prior period. We believe that a high net revenue retention rate is a measure of customer retention and an indicator of the engagement of our subscribers with our products. Investors should consider net revenue retention as an ongoing measure when evaluating our subscribers' interest in continuing to subscribe to our products and spending more with us over time.

Ultra high-value composition rate: Our ultra high-value composition rate reflects the number of Paid Subscribers who have purchased >\$5,000 in aggregate over their lifetime as of a particular point in time *divided by* the number of high-value subscribers as of that same point in time. We believe our ultra high-value composition rate reflects our ability to successfully build lifetime relationships with our subscribers, often across multiple products and brands. Investors should consider ultra high-value composition rate as a factor in evaluating our ability to retain and expand our relationship with our subscribers.

Key Business Metrics

We review the following key business metrics to measure our performance, identify trends, formulate financial projections, and make strategic decisions. We are not aware of any uniform standards for calculating these key metrics, which may hinder comparability with other companies who may calculate similarly titled metrics in a different way.

		and for the ths Ended June 30),			and for the Ended June 30,					
	2023	2022	% change	2023		2022 %		2022 % change		As of and for the Three Months Inded March 31, 2023	% change
Free Subscribers	 15,948,652	14,972,985	6.5 %	15,948,652		14,972,985	6.5%	15,678,205	1.7 %		
Paid Subscribers	750,287	897,941	(16.4)%	750,287		897,941	(16.4)%	777,397	(3.5)%		
ARPU	\$ 490	\$ 580	(15.6)%	\$ 490	\$	580	(15.6)%	\$ 493	(0.6)%		
Billings (in thousands)	\$ 96,170	\$ 117,507	(18.2)%	\$ 193,341	\$	253,502	(23.7)%	\$ 97,171	(1.0)%		

Free Subscribers. Free Subscribers are defined as unique subscribers who have subscribed to one of our free investment publications via a valid email address and continue to remain directly opted in, excluding any Paid Subscribers who also have free subscriptions. Free subscriptions are often daily publications that include some commentary about the stock market, investing ideas, or other specialized topics. Included within our free publications are advertisements and editorial support for our current marketing campaigns. While subscribed to our publications, Free Subscribers learn about our editors and analysts, get to know our products and services, and learn more about ways we can help them be a better investor.

In order to better describe our universe of Free Subscribers, we recognize sub-categories of Free Subscribers – Active and Passive Free Subscribers. Active Free Subscribers are those Free Subscribers with whom we have engaged during the most recent quarter and represent those individuals who have received and/or consumed our content on a regular basis during that same quarter. Our analysis indicates that this population of Active Free Subscribers is more likely to continue to consume content and convert to a Paid Subscriber. Passive Free Subscribers represent those individuals who have not directly received our content during the most recent quarter, however, they remain included in our Free Subscriber population as defined above and may continue to consume content from our platforms. We expect the composition of our Active and Passive Free Subscribers will change over time as we refine our marketing and data analysis techniques aimed at converting Free Subscribers to Paid Subscribers.

Free Subscribers increased by 1.0 million, or 6.5%, to 15.9 million at June 30, 2023 as compared to 15.0 million at June 30, 2022. As of June 30, 2023, Active Free Subscribers decreased by 0.4 million, or 9.5%, to 3.9 million, compared to 4.3 million as of June 30, 2022. The year over year decline in Active Free Subscribers is a

result of decreased engagement with our Free Subscriber community as consumer engagement continues to be soft.

Free Subscribers increased by 0.3 million, or 1.7% to 15.9 million as of June 30, 2023 as compared to 15.7 million as of March 31, 2023. As of June 30, 2023, Active Free Subscribers decreased by 0.1 million, or 2.9% to 3.9 million, compared to 4.0 million as of March 31, 2023.

Paid Subscribers. We define Paid Subscribers as the total number of unique subscribers with at least one paid subscription at the end of the period. We view the number of Paid Subscribers at the end of a given period as a key indicator of the attractiveness of our products and services, as well as the efficacy of our marketing in converting Free Subscribers to Paid Subscribers and generating direct-to-paid acquisitions. We grow our Paid Subscriber base through marketing directly to prospective and existing subscribers across a variety of media, channels, and platforms.

Total Paid Subscribers decreased by 148 thousand, or 16.4%, to 750 thousand as of June 30, 2023 as compared to 898 thousand at June 30, 2022, driven by softening consumer engagement that began in third quarter 2022 as well as a significant decrease in direct marketing spend as we focus on maintenance of profitability.

Total Paid Subscribers decreased by 27 thousand, or 3.5%, to 750 thousand as of June 30, 2023 as compared to 777 thousand as of March 31, 2023. Gross new subscribers increased 11% compared to first quarter 2023. However, this level of acquisition is still below historical levels and was not able to offset a normalized quarter of subscriber churn.

Subscriber count churn has ranged from approximately 1.8% to 2.7% per month between 2020 and 2022. As of June 30, 2023, the average monthly churn over the trailing twelve months was near the higher end of this range. Almost all of the subscribers who churned in second quarter 2023 did so having owned only one entry level publication. This is evidenced by the fact that their ARPU approximately matched the subscription price of our entry level publications. We believe our net revenue retention rate, which has averaged over 80% from 2020 to 2022, is a more meaningful gauge of subscriber satisfaction.

Average Revenue Per User. We calculate ARPU as the trailing four quarters of net Billings *divided by* the average number of quarterly total Paid Subscribers over that period. We believe ARPU is a key indicator of how successful we are in attracting subscribers to higher-value content. We believe that our high ARPU is indicative of the trust we build with our subscribers and of the value they see in our products and services.

ARPU decreased by \$91, or 15.6%, to \$490 as of June 30, 2023 as compared to \$580 as of June 30, 2022. The year-over-year decrease was driven by a 26% decrease in trailing four quarter Billings, while trailing four quarter Paid Subscribers only decreased by 13%. The decrease in trailing four quarter Billings was driven by reduced engagement of prospective and existing subscribers, as further discussed in —*Billings* below.

ARPU decreased by \$4, or 0.8%, to \$490 as of June 30, 2023 as compared to \$493 as of March 31, 2023. The sequential decrease was driven by a 5% decrease in trailing four-quarter Billings, while trailing four-quarter average Paid Subscribers only decreased 4%.

While they have declined recently, our ARPUs remain high relative to other subscription businesses, and we attribute this to the quality of our content and effective sales and marketing efforts regarding higher value content, bundled subscriptions and membership subscriptions. These subscriptions have compelling economics that allow us to recoup our initial marketing spend made to acquire these subscribers. Specifically, our payback period was estimated at 1.5 years, 0.9 years, and 0.6 years for the years ended December 31, 2022, 2021 and 2020, respectively. We have experienced an increase in the 2022 payback period primarily due to a combination of increased customer acquisition costs and the hesitancy of these subscribers to make additional purchases. The payback period reached the low side of the historical range in 2020 as a result of expanded conversion rates and, to a far lesser degree, decreasing costs for media spend as demand dropped as a result of

the pandemic. We have seen the costs for media spend revert back to higher rates in 2021 which continued through second quarter 2023.

Billings. Billings represents amounts invoiced to customers. We measure and monitor our Billings because it provides insight into trends in cash generation from our marketing campaigns. We generally bill our subscribers at the time of sale and receive full cash payment up front, and defer and recognize a portion of the related revenue ratably over time for term and membership subscriptions. For certain subscriptions, we may invoice our Paid Subscribers at the beginning of the term, in annual or monthly installments, and, from time to time, in multi-year installments. Only amounts invoiced to a Paid Subscriber in a given period are included in Billings. While we believe that Billings provides valuable insight into the cash that will be generated from sales of our subscriptions, this metric may vary from period to period for a number of reasons and, therefore, Billings has a number of limitations as a quarter-over-quarter or year-over-year comparative measure. These reasons include, but are not limited to, the following: (i) a variety of contractual terms could result in some periods having a higher proportion of annual or membership subscriptions than other periods; (ii) fluctuations in payment terms may affect the Billings recognized in a particular period; and (iii) the timing of large campaigns may vary significantly from period to period.

Billings decreased by \$21.3 million, or 18.2%, to \$96.2 million for the three months ended June 30, 2023 as compared to \$117.5 million for the three months ended June 30, 2022. We believe the decrease is due in large part to reduced engagement of prospective and existing subscribers. Levels of engagement had plateaued during the second half of 2021 and first half of 2022. The second half of 2022 saw further declines with uncertainty stemming from external factors such as 40-year high inflation, volatility across asset classes, federal reserve tightening, and the war in Ukraine, which we believe further contributed to prospective and existing subscribers delaying their purchases through second quarter 2023.

Approximately 36% of our Billings during the three months ended June 30, 2023 came from membership subscriptions, 63% from term subscriptions, and 1% from other Billings, as compared to 38% from membership subscriptions, 61% from term subscriptions, and 1% from other Billings during the three months ended June 30, 2022.

Billings decreased by \$1.0 million, or 1.0%, to \$96.2 million for the three months ended June 30, 2023 as compared to \$97.2 million for the three months ended March 31, 2023. With overall consumer engagement, as measured by landing pages, slightly up versus first quarter 2023, we attribute the decline in Billings to lower overall conversion rates on our various marketing campaigns within the quarter.

Components of Our Results of Operations

Net Revenue

We generate net revenue primarily from delivering term and membership subscription-based financial research, publications, and SaaS offerings to individual subscribers through our online platforms, advertising arrangements, print products, events, and revenue share agreements.

Net revenue is recognized ratably over the duration of the subscriptions, in an amount that reflects the consideration we expect to be entitled to in exchange for those services. In addition to term subscriptions, we offer membership subscriptions where we receive a large upfront payment when the subscriber enters into the contract, and for which we will receive a lower annual maintenance fee thereafter. Subscribers are typically billed in advance of the subscriptions. Much of our net revenue is generated from subscriptions entered into during previous periods. Consequently, any decreases in new subscriptions or renewals in any one period may not be immediately reflected as a decrease in net revenue for that period, but could negatively affect our net revenue in future quarters. This also makes it difficult for us to rapidly increase our net revenue through the sale of additional subscriptions in any period, as net revenue is recognized over the term of the subscription agreement. We expect subscription net revenue to fluctuate due to the timing and relative mix in recent and prior periods between term subscriptions and related renewals, and membership subscriptions and related maintenance fees.

We also recognize net revenue through revenue share agreements where we earn a commission for successful sales by other parties generated through the use of our customer list. We expect advertising and other net revenue to increase in absolute dollars as our business grows.

Employee Compensation Costs

Employee compensation costs, or payroll and payroll-related costs, include salaries, bonuses, benefits, and stock-based compensation for employees classified within cost of revenue, sales and marketing, and general and administrative, and also includes sales commissions for sales and marketing employees.

We recognized stock-based compensation expenses related to our 2021 Incentive Award Plan and our ESPP of \$3.7 million and \$2.4 million during the three months ended June 30, 2023 and 2022, and \$7.4 million and \$5.0 million during the six months ended June 30, 2023 and 2022, respectively.

The total amount of stock-based compensation expense included within each of the respective line items in the condensed consolidated statement of operations is as follows:

(In thousands)	Three Months Ended June 30,				Six Months Ended June 30,			
	 2023		2022		2023		2022	
Cost of revenue	\$ 683	\$	525	\$	1,691	\$	1,059	
Sales and marketing	716		577		1,808		1,142	
General and administrative	 2,279		1,347		3,882		2,836	
Total stock based-compensation expense	\$ 3,678	\$	2,449	\$	7,381	\$	5,037	

Cost of Revenue

Cost of revenue consists of costs associated with producing and publishing our content, including payroll, payroll-related costs and stock-based compensation expenses related to the 2021 Incentive Award Plan and the ESPP, hosting fees, customer service, credit card processing fees, product costs, and allocated overhead. Cost of revenue is exclusive of depreciation and amortization, which is shown as a separate line item.

We expect cost of revenue to increase as our business grows, including as a result of new acquisitions, joint ventures, and other strategic transactions. However, the level and timing of our variable compensation may not match the pattern of how net revenue is recognized over the subscription term. Therefore, we expect that our cost of revenue will fluctuate as a percentage of net revenue in the future.

Sales and Marketing

Sales and marketing consists of costs associated with marketing and selling our products and services, including payroll, payroll-related costs and stock-based compensation expenses related to the 2021 Incentive Award Plan and the ESPP, amortization of deferred contract acquisition costs, agency costs, advertising campaigns, and branding initiatives. Sales and marketing expenses are exclusive of depreciation and amortization shown as a separate line item.

We expect that our sales and marketing expense will increase in absolute dollars and continue to be our largest operating expense for the foreseeable future as we expand our sales and marketing efforts. However, because we incur sales and marketing expenses up front when we launch campaigns to drive sales, while we recognize net revenue ratably over the underlying subscription term, we expect that our sales and marketing expense will fluctuate as a percentage of our net revenue over the long term. Sales and marketing expenses may fluctuate further as a result of acquisitions, joint ventures, or other strategic transactions we undertake in the future.

Research and Development

Research and development consists of costs associated with technical services, including payroll, payroll-related costs and stock-based compensation expenses related to the 2021 Incentive Award Plan and the ESPP, software

expenses, and hosting expenses. Research and development expenses are exclusive of depreciation and amortization shown as a separate line item.

We expect that our research and development expense will increase in absolute dollars as our business grows, including as a result of new acquisitions, joint ventures, and other strategic transactions, particularly as we incur additional costs related to continued investments in our platform.

General and Administrative

General and administrative consists of costs associated with our finance, legal, information technology, human resources, executive, and administrative personnel, including payroll, payroll-related costs and stock-based compensation expenses related to the 2021 Incentive Award Plan and the ESPP, legal fees, corporate insurance, office expenses, professional fees, and travel and entertainment costs.

We expect to continue to incur general and administrative expenses as a result of operating as a public company, including costs to comply with the rules and regulations applicable to companies listed on a national securities exchange, costs related to compliance and reporting obligations pursuant to the rules and regulations of the SEC, and increased expenses for insurance, investor relations, and professional services. General and administrative expenses may fluctuate further as a result of acquisitions, joint ventures, or other strategic transactions we undertake in the future.

Depreciation and Amortization

Depreciation and amortization expenses consist of amortization of trade names, customer relationship intangibles, and software development costs, as well as depreciation on other property and equipment such as leasehold improvements, furniture and fixtures, and computer equipment. We expect depreciation and amortization expenses to increase on an absolute dollar basis as our business grows, including as a result of new acquisitions, joint ventures, and other strategic transactions, but to remain generally consistent as a percentage of total net revenue.

Related Party Expense

Related party expenses primarily consist of board of director compensation, revenue share expenses, and costs for certain corporate functions performed by a related party.

Other Income, Net

Other income, net primarily consists of the net gains on our embedded derivative instruments.

Interest Income (Expense), Net

Interest income (expense), net primarily consists of interest income from our money market accounts, as well as interest expense related to our 2021 Credit Facility.

Net Income Attributable to Noncontrolling Interests

As of June 30, 2023, MarketWise, Inc.'s controlling interest in MarketWise, LLC was 10.0% and the noncontrolling interest was 90.0%. For the three and six months ended June 30, 2023, net income attributable to controlling interests included a \$0.4 million tax provision and a \$1.4 million tax provision, respectively, both of which are 100% attributable to the controlling interest.

As of June 30, 2022, MarketWise, Inc.'s controlling interest in MarketWise, LLC was 7.2% and the noncontrolling interest was 92.8%. For the three months ended June 30, 2022 net income attributable to controlling interests included a \$12.1 million gain on warrant liabilities and a \$1.0 million tax provision, and for the six months ended June 30, 2022 net income attributable to controlling interests included a \$18.8 million gain on warrant liabilities and a \$2.6 million tax provision, both of which are 100% attributable to the controlling interest.

Results of Operations

The following table sets forth our results of operations for the periods presented:

(In thousands)	Three Months	Ended	Six Months Ended June 30,				
	2023		2022		2023		2022
Net revenue	\$ 103,363	\$	127,466	\$	228,978	\$	264,086
Related party revenue	 281		548		899		726
Total net revenue	 103,644		128,014		229,877		264,812
Operating expenses:							
Cost of revenue ⁽¹⁾	14,635		16,229		29,925		33,846
Sales and marketing ⁽¹⁾	49,033		65,050		97,760		133,287
General and administrative ⁽¹⁾	27,629		20,364		55,662		50,909
Research and development ⁽¹⁾	2,230		2,289		4,693		4,567
Depreciation and amortization	994		613		1,978		1,217
Related party expense	204		97		332		194
Total operating expenses	94,725		104,642		190,350		224,020
Income from operations	 8,919		23,372		39,527		40,792
Other income, net	238		11,923		625		19,219
Interest income (expense), net	 1,013		(218)		1,551		(389)
Income before income taxes	 10,170		35,077		41,703		59,622
Income tax expense	 427		1,040		1,355		2,562
Net income	 9,743		34,037		40,348		57,060
Net income attributable to noncontrolling interests	 9,707		22,156		38,845		39,354
Net income attributable to MarketWise, Inc.	\$ 36	\$	11,881	\$	1,503	\$	17,706

⁽¹⁾ Cost of revenue, sales and marketing, general and administrative, and research and development expenses are exclusive of depreciation and amortization shown as a separate line item.

The following table sets forth our consolidated statements of operations data expressed as a percentage of net revenue for the periods indicated:

	Three Months En	ded June 30,	Six Months Ended June 30,		
	2023	2022	2023	2022	
Net revenue	100.0 %	100.0 %	100.0 %	100.0 %	
Operating expenses:					
Cost of revenue ⁽¹⁾	14.1 %	12.7 %	13.0 %	12.8 %	
Sales and marketing ⁽¹⁾	47.3 %	50.8 %	42.5 %	50.3 %	
General and administrative ⁽¹⁾	26.7 %	15.9 %	24.2 %	19.2 %	
Research and development ⁽¹⁾	2.2 %	1.8 %	2.0 %	1.7 %	
Depreciation and amortization	1.0 %	0.5 %	0.9 %	0.5 %	
Related party expense	0.2 %	0.1 %	0.1 %	0.1 %	
Total operating expenses	91.4 %	81.7 %	82.8 %	84.6 %	
Income from operations	8.6 %	18.3 %	17.2 %	15.4 %	
Other income, net	0.2 %	9.3 %	0.3 %	7.3 %	
Interest income (expense), net	1.0 %	(0.2)%	0.7 %	(0.1)%	
Income before income taxes	9.8 %	27.4 %	18.1 %	22.5 %	
Income tax expense	0.4 %	0.8 %	0.6 %	1.0 %	
Net income	9.4 %	26.6 %	17.6 %	21.5 %	
Net income attributable to noncontrolling interests	9.4 %	17.3 %	16.9 %	14.9 %	
Net income attributable to MarketWise, Inc.	<u> </u>	9.3 %	0.7 %	6.7 %	

⁽¹⁾ Cost of revenue, sales and marketing, general and administrative, and research and development expenses are exclusive of depreciation and amortization shown as a separate line item.

Comparison of Three Months Ended June 30, 2023 and Three Months Ended June 30, 2022

Net Revenue

(In thousands)	Three Months	Ended	d June 30,		
	 2023		2022	\$ Change	% Change
Net revenue	\$ 103,644	\$	128,014	\$ (24,370)	(19.0)%

Net revenue decreased by \$24.4 million, or 19.0%, from \$128.0 million for the three months ended June 30, 2022 to \$103.6 million for the three months ended June 30, 2023. The decrease in net revenue was primarily driven by a \$15.6 million decrease in term subscription revenue and a \$8.4 million decrease in membership subscription revenue.

Term subscription revenue decreased during the three months ended June 30, 2023 primarily due to lower Billings as compared to the 2022 period, which was driven by reduced engagement of prospective and existing subscribers in the 2022 period. Membership subscription revenue, which is initially deferred and recognized over a five-year period, decreased during the three months ended June 30, 2023 as a result of lower volume of membership subscriptions in current and prior years.

Operating Expenses

(In thousands)	Three Months	Ended	June 30,				
	2023 2022				\$ Change	% Change	
Operating expenses:	 						
Cost of revenue	\$ 14,635	\$	16,229	\$	(1,594)	(9.8)%	
Sales and marketing	49,033		65,050		(16,017)	(24.6)%	
General and administrative	27,629		20,364		7,265	35.7 %	
Research and development	2,230		2,289		(59)	(2.6)%	
Depreciation and amortization	994		613		381	62.2 %	
Related party expenses	204		97		107	110.3 %	
Total operating expenses	\$ 94,725	\$	104,642	\$	(9,917)	(9.5)%	

Cost of Revenue

Cost of revenue decreased by \$1.6 million, or 9.8%, from \$16.2 million for the three months ended June 30, 2022 to \$14.6 million for the three months ended June 30, 2023, primarily driven by a \$0.8 million decrease in salaries, taxes and benefits, a \$0.6 million decrease in credit card fees, and a \$0.6 million decrease in outsourced customer service. This was partially offset by a \$0.2 million increase in stock-based compensation expense related to awards under the 2021 Incentive Award Plan, and a \$0.2 million increase in freelance editorial expense.

Sales and Marketing

Sales and marketing expense decreased by \$16.0 million, or 24.6%, from \$65.1 million for the three months ended June 30, 2022 to \$49.0 million for the three months ended June 30, 2023, primarily driven by a \$19.6 million decrease in marketing expense, as we have reduced our marketing spend as part of our cost reduction initiatives, and due to higher per unit subscriber acquisition costs. This was partially offset by a \$3.7 million increase in amortization of deferred contract acquisition costs.

General and Administrative

General and administrative expense increased by \$7.3 million, or 35.7%, from \$20.4 million for the three months ended June 30, 2022 to \$27.6 million for the three months ended June 30, 2023, primarily driven by a \$3.6 million increase related to sales and franchise taxes, a \$3.8 million increase in incentive compensation, a \$1.1 million increase in stock-based compensation expense related to awards under the 2021 Incentive Award Plan and the ESPP, and a \$0.7 million increase in outside labor. This was partially offset by a \$1.1 million decrease in salaries, taxes and benefits, and a \$0.4 million decrease in travel and entertainment expenses.

Comparison of Six Months Ended June 30, 2023 and Six Months Ended June 30, 2022

Net Revenue

(In thousands)	Six Months E	nded June	30,			
	 2023		2022	\$ Change	% Change	
Net revenue	\$ 229,877	\$	264,812	\$ (34,935)	(13.2)%	

Net revenue decreased by \$34.9 million, or 13.2%, from \$264.8 million for the six months ended June 30, 2022 to \$229.9 million for the six months ended June 30, 2023. The decrease in net revenue was primarily driven by a \$23.3 million decrease in term subscription revenue, a \$11.5 million decrease in membership subscription revenue and a \$0.2 million decrease in non-subscription revenue.

Both term subscription revenue and membership subscription revenue decreased during the three and six months ended June 30, 2023 primarily due to lower Billings as compared to the 2022 period which was driven by reduced engagement of prospective and existing subscribers in the 2023 period.

Term subscription revenue decreased during the six months ended June 30, 2023 primarily due to lower Billings as compared to the 2022 period, which was driven by reduced engagement of prospective and existing subscribers in the 2022 period. Membership subscription revenue, which is initially deferred and recognized over a five-year period, decreased during the six months ended June 30, 2023 as a result of lower volume of membership subscriptions in current and prior years.

Operating Expenses

(In thousands)		Six Months E	inded J	une 30,			
		2023		2022	\$ Change	% Change	
Operating expenses:							
Cost of revenue	\$	29,925	\$	33,846	\$ (3,921)	(11.6)%	
Sales and marketing		97,760		133,287	(35,527)	(26.7)%	
General and administrative		55,662		50,909	4,753	9.3 %	
Research and development		4,693		4,567	126	2.8 %	
Depreciation and amortization		1,978		1,217	761	62.5 %	
Related party expenses		332		194	138	71.1 %	
Total operating expenses	\$	190,350	\$	224,020	\$ (33,670)	(15.0)%	

Cost of Revenue

Cost of revenue decreased by \$3.9 million, or 11.6%, from \$33.8 million for the six months ended June 30, 2022 to \$29.9 million for the six months ended June 30, 2023, primarily driven by a \$1.4 million decrease in credit card fees, a \$1.3 million decrease in salaries, taxes and benefits, and a \$1.2 million decrease in outside labor, primarily related to customer service.

Sales and Marketing

Sales and marketing expense decreased by \$35.5 million, or 26.7%, from \$133.3 million for the six months ended June 30, 2022 to \$97.8 million for the six months ended June 30, 2023, primarily driven by a \$44.9 million decrease in marketing expense as we have reduced our marketing spend as part of our cost reduction initiatives and due to higher per unit subscriber acquisition costs. This was partially offset by a \$8.0 million increase in amortization of deferred contract acquisition costs, and a \$1.1 million increase in salaries, taxes and benefits.

General and Administrative

General and administrative expense increased by \$4.8 million, or 9.3%, from \$50.9 million for the six months ended June 30, 2022 to \$55.7 million for the six months ended June 30, 2023, primarily driven by a \$4.9 million increase in incentive compensation, a \$2.8 million increase related to sales tax, and a \$1.0 million increase in stock-based compensation expense related to awards under the 2021 Incentive Award Plan and the ESPP. This was partially offset by a \$2.9 million decrease in salaries, taxes and benefits, and a \$0.6 million decrease in professional fees.

Non-GAAP Financial Measures

In addition to our results determined in accordance with GAAP, we believe that the below non-GAAP financial measures are useful in evaluating our ability to generate cash. We use the below non-GAAP financial measures, collectively, to evaluate our ongoing operations and for internal planning and forecasting purposes. We believe that non-GAAP financial information, when taken collectively, may be helpful to investors because it provides consistency and comparability with past financial performance. This non-GAAP financial information is presented

for supplemental informational purposes only and should not be considered a substitute for financial information presented in accordance with GAAP, and may be different from similarly titled non-GAAP measures used by other companies. A reconciliation is provided below for each non-GAAP financial measure to the most directly comparable financial measure stated in accordance with GAAP. Investors are encouraged to review the related GAAP financial measures and the reconciliations of these non-GAAP financial measures to their most directly comparable GAAP financial measures.

(In thousands)	Three Months	Ended	June 30,		Six Months I		
	2023		2022	% change	2023	2022	% change
Adjusted CFFO	\$ 28,978	\$	26,794	8.2%	\$ 32,846	\$ 27,862	17.9%
Adjusted CFFO Margin	30.1 %		22.8 %		17.0 %	11.0 %	

Adjusted CFFO / Adjusted CFFO Margin

In addition to our results determined in accordance with GAAP, we disclose the non-GAAP financial measures Adjusted CFFO and Adjusted CFFO Margin. We define Adjusted CFFO as cash flow from operations plus or minus any non-recurring items. We define Adjusted CFFO Margin as Adjusted CFFO as a percentage of Billings.

We believe that Adjusted CFFO and Adjusted CFFO Margin are useful indicators that provide information to management and investors about our ability to generate cash (without the effects of non-recurring items), and for internal planning and forecasting purposes.

Adjusted CFFO and Adjusted CFFO Margin have limitations as analytical tools, and should not be considered in isolation or as substitutes for analysis of other GAAP financial measures, such as cash flow from operations or operating cash flow margin. Some of the limitations of using Adjusted CFFO and Adjusted CFFO Margin are that these metrics may be calculated differently by other companies in our industry.

We expect Adjusted CFFO and Adjusted CFFO Margin to fluctuate in future periods as we invest in our business to execute our growth strategy. These activities, along with any non-recurring items as described above, may result in fluctuations in Adjusted CFFO and Adjusted CFFO Margin in future periods.

The following table provides a reconciliation of net cash provided by operating activities, the most directly comparable financial measure calculated in accordance with GAAP, to Adjusted CFFO for each of the periods presented:

(In thousands)	Three Months Ended June 30,					Six Months Ended June 30,				
	 2023		2022 % change		2023		2022		% change	
Net cash provided by operating activities	\$ 28,978	\$	26,794	8.2%	\$	32,846	\$	27,862	17.9%	
Non-recurring expenses				N/M					N/M	
Adjusted CFFO	\$ 28,978	\$	26,794	8.2%	\$	32,846	\$	27,862	17.9%	

N/M: Not meaningful

The following table provides the calculation of net cash provided by operating activities as a percentage of total net revenue, the most directly comparable financial measure in accordance with GAAP, and Adjusted CFFO Margin for each of the periods presented:

(In thousands)	Three Months Ended June 30,					Six Months Ended June 30,				
		2023		2022	% change	2023		2022	% change	
Net cash provided by operating activities	\$	28,978	\$	26,794	8.2%	\$ 32,846	\$	27,862	17.9%	
Total net revenue		103,644		128,014	(19.0)%	229,877		264,812	(13.2)%	
Net cash provided by operating activities margin		28.0 %		20.9 %		 14.3 %		10.5 %		
Adjusted CFFO	\$	28,978	\$	26,794	8.2%	\$ 32,846	\$	27,862	17.9%	
Billings		96,170		117,507	(18.2)%	193,341		253,502	(23.7)%	
Adjusted CFFO Margin		30.1 %		22.8 %		17.0 %		11.0 %		

Net cash provided by operating activities margin increased from 20.9% for the three months ended June 30, 2022 to 28.0% for the three months ended June 30, 2023 primarily due to the increase in net cash provided by operating activities. Cash flow from operations increased by \$2.2 million, or 8.2%, from \$26.8 million for the three months ended June 30, 2022 to \$29.0 million for the three months ended June 30, 2023, primarily due to net income of \$9.7 million, adjusted for net non-cash items which increased cash by \$5.3 million, and net changes in our operating assets and liabilities which increased cash by \$13.9 million.

Net cash provided by operating activities margin increased from 10.5% for the six months ended June 30, 2022 to 14.3% for the six months ended June 30, 2023 primarily due to the increase in net cash provided by operating activities. Cash flow from operations increased by \$5.0 million, or 17.9%, from \$27.9 million for the six months ended June 30, 2022 to \$32.8 million for the six months ended June 30, 2023, primarily due to net income of \$40.3 million, adjusted for net non-cash items which increased cash by \$12.4 million, and net changes in our operating assets and liabilities which reduced cash by \$19.9 million.

Billings represents amounts invoiced to customers. We generally bill our subscribers at the time of sale and receive full cash payment up front, and in accordance with GAAP, we defer and recognize a portion of the related revenue ratably over time for term and membership subscriptions. For certain subscriptions, we may invoice our Paid Subscribers at the beginning of the term, in annual or monthly installments, and, from time to time, in multi-year installments.

Adjusted CFFO margin increased from 22.8% for three months ended June 30, 2022 to 30.1% for the three months ended June 30, 2023 due to the increase in Adjusted CFFO and the decrease in Billings. Adjusted CFFO increased by \$2.2 million, or 8.2%, from \$26.8 million for the three months ended June 30, 2022 to \$29.0 million for the three months ended June 30, 2023, primarily driven by a decrease in operating expenses as a result of cost reduction initiatives.

Adjusted CFFO margin increased from 11.0% for the six months ended June 30, 2022 to 17.0% for the six months ended June 30, 2023 due to the increase in Adjusted CFFO and the decrease in Billings. Adjusted CFFO increased by \$5.0 million, or 18%, from \$27.9 million for the six months ended June 30, 2022 to \$32.8 million for the six months ended June 30, 2023, primarily driven by a decrease in operating expenses as a result of cost reduction initiatives.

Liquidity and Capital Resources

General

As of June 30, 2023, our principal sources of liquidity were cash, cash equivalents, and restricted cash of \$187.0 million. Cash and cash equivalents are comprised of bank deposits, money market funds, and certificates of deposit. Restricted cash is comprised of reserves held with credit card processors for chargebacks and refunds. We have financed our operations primarily through cash received from operations, and our sources of liquidity have enabled us to make continued investments in supporting the growth of our business. Our 2021 Credit Facility (as defined and further discussed below) can be used to finance permitted acquisitions, for working capital and general corporate purposes. We expect that our operating cash flows, in addition to cash on hand, will enable us to continue to make investments in the future, and to pay dividends. We expect our operating cash flows to further improve as we increase our operational efficiency and experience economies of scale.

We believe that our existing cash and cash equivalents and cash flow from operations will be sufficient to support working capital and capital expenditure requirements for at least the next 12 months. While we believe that our existing cash and cash equivalents and cash flow from operations will be sufficient to support working capital and capital expenditure requirements for the long term, our future capital requirements will depend on many factors, including our subscription growth rate, subscription renewal activity, including the timing and the amount of cash received from subscribers, the pace of expansion of sales and marketing activities, the timing and extent of spending to support development efforts, the introduction of new and enhanced products, and the level of costs to operate as a public company. We may, in the future, enter into arrangements to acquire or invest in complementary businesses, products, and technologies.

We may be required to seek additional equity or debt financing. In the event that we require additional financing, we may not be able to raise such financing on terms acceptable to us or at all. If we are unable to raise additional capital or generate cash flows necessary to expand our operations and invest in continued innovation, we may not be able to compete successfully, which would harm our business, operations, and financial condition.

A substantial source of our cash is from our deferred revenue, which is included in the liabilities section of our condensed consolidated balance sheets. Deferred revenue consists of the unearned portion of customer Billings, which is recognized as net revenue in accordance with our revenue recognition policy. As of June 30, 2023, we had deferred revenue of \$628.4 million, of which \$288.4 million was recorded as a current liability and is expected to be recognized as net revenue over the next 12 months, provided all other revenue recognition criteria have been met.

We have incurred and we will continue to incur public company expenses related to our operations, *plus* we expect to incur payment obligations under the Tax Receivable Agreement in the future, which we expect to be significant. MarketWise, Inc. intends to cause MarketWise, LLC to make distributions to MarketWise, Inc. in an amount sufficient to allow MarketWise, Inc. to pay its tax obligations and operating expenses, including distributions to fund any payments due under the Tax Receivable Agreement. If MarketWise, LLC does not have sufficient cash to fund distributions to MarketWise, Inc. in amounts sufficient to cover MarketWise, Inc.'s obligations under the Tax Receivable Agreement, it may have to borrow funds, which could materially adversely affect its liquidity and financial condition and subject it to various restrictions imposed by any such lenders. To the extent that MarketWise, Inc. is unable to make timely payments under the Tax Receivable Agreement for any reason, the unpaid amounts will be deferred and will accrue interest until paid. For additional information regarding the Tax Receivable Agreement, see the section entitled "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity and Capital Resources—Tax Receivable Agreement" in the Annual Report.

Furthermore, to the extent we have taxable income, we will make distributions to the MarketWise Members in amounts sufficient for MarketWise Members to pay taxes due on their share of MarketWise income at prevailing individual income tax rates. Such amounts will be reflected in MarketWise, Inc.'s statement of cash flows as cash used in financing activities, and so will not decrease the amount of cash from operations or net income reflected in MarketWise, Inc.'s financial statements. However, such distributions will decrease the amount of cash available to us for use in our business.

Share Repurchase Program

As previously disclosed in our Annual Report, on November 4, 2021, our Board authorized the repurchase of up to \$35.0 million in aggregate of shares of the Company's Class A common stock, with the authorization to expire on November 3, 2023. We did not repurchase any shares during the three months ended June 30, 2023. During the six months ended June 30, 2022, we repurchased 2,484,717 shares totaling \$13.1 million in the aggregate, including fees and commissions of \$25. Since the inception of the program, we have repurchased 2,984,987 total shares.

For each share of Class A common stock the Company repurchases under the share repurchase program, MarketWise, LLC, the Company's direct subsidiary, will redeem one LLC Unit held by the Company, decreasing the percentage ownership of MarketWise, LLC by the Company and relatively increasing the ownership by the other unitholders.

Credit Facility

In 2021, MarketWise, LLC, entered into a loan and security agreement (as amended the "Loan and Security Agreement") providing for up to \$150 million of commitments under a revolving credit facility (the "2021 Credit Facility"), including a \$5 million letter of credit sublimit, and allows for revolving commitments under the 2021 Credit Facility to be increased or new term commitments to be established by up to \$65 million. The existing lenders under the 2021 Credit Facility are entitled, but not obligated, to provide such incremental commitments. The 2021 Credit Facility has a term of three years, maturing on October 29, 2024. On May 2, 2023, MarketWise, LLC entered into the First Amendment to the Loan and Security Agreement, which provides, among other things, a transition away from LIBOR to SOFR as the basis for the interest rate.

The 2021 Credit Facility is guaranteed by MarketWise, LLC's direct and indirect material U.S. subsidiaries, subject to customary exceptions (the "Guarantors"), pursuant to a guaranty by the Guarantors in favor of HSBC Bank USA, National Association, as agent (the "Guaranty"). Borrowings under the 2021 Credit Facility are secured by a first-priority lien on substantially all of the assets of MarketWise, LLC and the Guarantors, subject to customary exceptions.

Borrowings will bear interest at a floating rate depending on MarketWise, LLC's Net Leverage Ratio (as defined in the Loan and Security Agreement). As of June 30, 2023, there were no outstanding advances under the 2021 Credit Facility.

The Loan and Security Agreement contains customary affirmative and negative covenants for transactions of this type, and contains financial maintenance covenants that require MarketWise, LLC to maintain an Interest Coverage Ratio and Net Leverage Ratio (both as defined in the Loan and Security Agreement), and provides for a number of customary events of default, which could result in the acceleration of obligations and the termination of lending commitments under the Loan and Security Agreement. As of June 30, 2023, we were in compliance with these covenants.

Cash Flows

The following table presents a summary of our consolidated cash flows provided by (used in) operating, investing, and financing activities for the periods indicated:

(In thousands)	Six Months Ended June 30,					
	 2023	2022				
Net cash provided by operating activities	\$ 32,846	5 27,862				
Net cash used in investing activities	(959)	(107)				
Net cash used in financing activities	(3,426)	(16,042)				

Operating Activities

For the six months ended June 30, 2023, net cash provided by operating activities was \$32.8 million, primarily due to net income of \$40.3 million, adjusted for net non-cash items which increased cash by \$12.4 million, and net

changes in our operating assets and liabilities which reduced cash by \$19.9 million, largely due to timing differences in the net receipt of cash. The non-cash items include stock-based compensation expense and depreciation and amortization of \$7.4 million and \$2.0 million, respectively. The changes in operating assets and liabilities were primarily driven by a decrease in deferred revenue, which reduced cash by \$35.1 million due to our overall decrease in sales, and a decrease in accrued expenses, which decreased cash by \$7.6 million, partially offset by a decrease in deferred contract acquisition costs, which increased cash by \$19.9 million.

For the six months ended June 30, 2022, net cash provided by operating activities was \$27.9 million, primarily due to net income of \$57.1 million, adjusted for net non-cash items which reduced cash by \$9.9 million, and net changes in our operating assets and liabilities which reduced cash by \$19.3 million, largely due to timing differences in the net receipt of cash. The non-cash items include a change in fair value of derivative liabilities of \$19.7 million, which was partially offset by stock-based compensation expense of \$5.0 million. The changes in operating assets and liabilities were primarily driven by a decrease in deferred revenue, which reduced cash by \$9.1 million due to our overall decrease in sales, an increase in deferred contract acquisition costs, which reduced cash by \$5.6 million, and a decrease in accounts receivable due to our overall decrease in sales, which increased cash by \$4.5 million.

Investing Activities

For the six months ended June 30, 2023, net cash used in investing activities was \$1.0 million, primarily driven by \$0.9 million of capitalized software development costs.

For the six months ended June 30, 2022, net cash used in investing activities was \$0.1 million.

Financing Activities

On May 11, 2023, the Board declared a cash dividend to the Company's Class A common stockholders in the amount of \$0.01 per share and a cash distribution to its common unit holders in the amount of \$0.01 per unit. The dividend and distribution was paid on July 20, 2023, to stockholders and unitholders of record as of the close of business on June 1, 2023. The total amount of the dividend payment to Class A common stockholders was \$0.4 million and the total amount of the distribution payment to common unit holders was approximately \$2.9 million. The Company intends to pay dividends quarterly in the future, subject to Board approval of any such dividends.

For the six months ended June 30, 2023, net cash used in financing activities was \$3.4 million, primarily due to \$1.1 million in restricted stock units withheld to pay taxes and \$2.7 million in distributions to noncontrolling interests.

For the six months ended June 30, 2022, net cash used in financing activities was \$16.0 million, primarily due to \$13.1 million in share repurchases and \$3.8 million in distributions to noncontrolling interests.

Critical Accounting Estimates

Our condensed consolidated financial statements have been prepared in accordance with GAAP. The preparation of these condensed consolidated financial statements requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenue, costs, and expenses, and related disclosures. On an ongoing basis, management evaluates its estimates and assumptions. Our actual results may differ from these estimates under different assumptions or conditions.

Management believes that, of our significant accounting policies, which are described in Note 2 to our consolidated financial statements for the year ended December 31, 2022 in our Annual Report, the following accounting policies involve a greater degree of judgment and complexity. Accordingly, management believes that "Revenue Recognition" and "Transactions and Valuation of Goodwill and Other Acquired Intangible Assets" are the two policies that are the most critical to aid in fully understanding and evaluating our condensed consolidated financial condition and results of operations. Refer to the 2022 Annual Report on Form 10-K for further discussion of these two policies. During the three months ended June 30, 2023, there were no material changes to these policies.

Item 3. Quantitative and Qualitative Disclosures About Market Risk.

Not applicable.

Item 4. Controls and Procedures.

Evaluation of Disclosure Controls and Procedures

Our management, with the participation of our principal executive officer and principal financial officer, evaluated, as of the end of the period covered by this Quarterly Report on Form 10-Q, the effectiveness of the design and operation of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended (the "Exchange Act")). Based on that evaluation, and as a result of the material weaknesses described below, our principal executive officer and principal financial officer concluded that, as of June 30, 2023, our disclosure controls and procedures were not effective. In light of this fact, our management has performed additional analyses, reconciliations, and other post-closing procedures and has concluded that, notwithstanding the material weaknesses in our internal control over financial reporting, the condensed consolidated financial statements for the periods covered by and included in this Quarterly Report on Form 10-Q fairly state, in all material respects, our financial position, results of operations and cash flows for the periods presented in conformity with U.S. GAAP.

As previously reported, in connection with the audit of our consolidated financial statements for the year ended December 31, 2020, our management determined that material weaknesses existed in our internal control over financial reporting related to: (i) a lack of contemporaneous documentation and account reconciliation, and (ii) the lack of a formal or documented risk assessment process. A material weakness is a deficiency, or combination of deficiencies, in internal control over financial reporting such that there is a reasonable possibility that a material misstatement of our annual or interim consolidated financial statements will not be prevented or detected on a timely basis.

Remediation Efforts to Address the Material Weaknesses

We have taken a number of remediation actions since the identification of the material weakness through June 30, 2023, including:

- hiring personnel with appropriate levels of experience in accounting, technology, risk assessment, and internal controls;
- establishing a comprehensive SOX program that includes the documentation and assessment of the Company's risks and internal controls over financial reporting to comply with the Sarbanes-Oxley Act;
- · implementing an annual risk assessment process over financial reporting;
- · implementing an annual cyber security risk assessment process;
- establishing a Disclosure Task Force with participants across business units to ensure the completeness and accuracy of financial reporting disclosures;
- establishing regular director and executive management meetings to oversee Company risks and performance;
- · implementing a new revenue recognition application and associated internal controls over financial reporting; and
- enhancing account reconciliation processes and internal controls, leveraging the use of a reconciliation application, and training of accounting personnel.

While significant progress has been made to enhance our internal control over financial reporting, we are still in the process of building and enhancing our processes, procedures, and controls. Additional time is required to

complete the remediation of these material weaknesses and the assessment to ensure the sustainability of these remediation actions. We believe the above actions, when complete, will be effective in the remediation of the material weaknesses described above. As such, we have not concluded that the material weaknesses have been fully remediated as of June 30, 2023.

Changes in Internal Control Over Financial Reporting

Other than as described above, there have been no changes in our internal control over financial reporting identified in connection with the evaluation required by Rule 13a-15(d) and 15d-15(d) of the Exchange Act that occurred during the period covered by this Quarterly Report on Form 10-Q that materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

Inherent Limitations on Effectiveness of Controls

Our management, including our Chief Executive Officer and Chief Financial Officer, believes that our disclosure controls and procedures and internal control over financial reporting are designed to provide reasonable assurance of achieving their objectives and are effective at the reasonable assurance level. However, our management does not expect that our disclosure controls and procedures or our internal control over financial reporting will prevent all errors and all fraud. A control system, no matter how well conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the control system are met. Furthermore, the design of a control system must reflect the fact that there are resource constraints, and the benefits of controls must be considered relative to their costs. Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, have been detected. These inherent limitations include the realities that judgments in decision making can be faulty, and that breakdowns can occur because of a simple error or mistake. Additionally, controls can be circumvented by the individual acts of some persons, by collusion of two or more people, or by management override of the controls. The design of any system of controls is also based in part upon certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions; over time, controls may become inadequate because of changes in conditions, or the degree of compliance with policies or procedures may deteriorate. Because of the inherent limitations in a cost-effective control system, misstatements due to error or fraud may occur and not be detected.

PART II—OTHER INFORMATION

Item 1. Legal Proceedings.

Not applicable.

Item 1A. Risk Factors.

Not applicable.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.

Issuer Purchases of Equity Securities

In November 2021, our Board authorized the repurchase of up to \$35.0 million in aggregate of shares of the Company's Class A common stock, with the authorization to expire on November 3, 2023. There were no share repurchases made by or on behalf of the Company of its common stock during the three months ended June 30, 2023. The maximum dollar value of shares that may yet to be purchased under the plan was \$18.6 million as of June 30, 2023.

During the three months ended June 30, 2023, pursuant to the terms of the MarketWise Operating Agreement, members of MarketWise, LLC exchanged an
aggregate of 1,250,000 LLC Units together with an equal number of shares of Class B common stock for 1,250,000 newly-issued shares of the Company's Class
A common stock. These shares of Class A common stock were issued in reliance on an exemption from registration pursuant to Section 4(a)(2) of the Securities
Act of 1933.

Item 3. Defaults Upon Senior Securities.

Not applicable.

Item 4. Mine Safety Disclosures.

Not applicable.

Item 5. Other Information.

Not applicable.

Item 6. Exhibits.

The following exhibits are filed as part of, or incorporated by reference into, this Quarterly Report on Form 10-Q.

Exhibit No.	<u>Description</u>
3.1	Certificate of Incorporation of MarketWise, Inc. (incorporated by reference to Exhibit 3.1 to the Registrant's Current Report on Form 8-K, filed with the SEC on July 28, 2021).
3.2	Bylaws of MarketWise, Inc. (incorporated by reference to Exhibit 3.2 to the Registrant's Current Report on Form 8-K, filed with the SEC on
10.1	<u>July 28, 2021).</u> <u>Voting Agreement, dated as of April 17, 2023, by and between MarketWise, Inc., Monument & Cathedral Holdings, LLC and the other persons and entities listed on Schedule A thereto (incorporated by reference to Exhibit 10.1 to the Registrant's Current Report on Form 8-K, filed with the SEC on April 21, 2023).</u>
10.2†	Settlement Agreement, dated as of April 28, 2023, by and between MarketWise, Inc., F. Porter Stansberry and Stokes Holdings, LLC (incorporated by reference to Exhibit 10.1 to the Registrant's Current Report on Form 8-K, filed with the SEC on May 2, 2023).
10.3†	First Amendment to Loan and Security Agreement, dated as of May 2, 2023, by and among MarketWise, LLC, as borrower, the guarantors party thereto, the lenders from time to time party thereto, HSBC Bank USA, National Association, as administrative agent, collateral agent, joint lead arranger, and joint bookrunner, and BMO Capital Markets Corp., as joint lead arranger and joint bookrunner.
31.1	Certification of Principal Executive Officer Pursuant to Rules 13a-14(a) and 15d-14(a) under the Securities Exchange Act of 1934, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification of Principal Financial Officer Pursuant to Rules 13a-14(a) and 15d-14(a) under the Securities Exchange Act of 1934, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1	Certification of Principal Executive Officer Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
32.2	Certification of Principal Financial Officer Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
101.INS	XBRL Instance Document
101.SCH	XBRL Taxonomy Extension Schema Document
101.CAL	XBRL Taxonomy Extension Definition Linkbase Document
101.DEF	XBRL Taxonomy Extension Label Linkbase Document
101.LAB	XBRL Taxonomy Extension Label Linkbase Document
101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

[†] The annexes, schedules, and certain exhibits to this exhibit have been omitted pursuant to Item 601(a)(5) of Regulation S-K. The Registrant hereby agrees to furnish supplementally a copy of any omitted annex, schedule, or exhibit to the SEC upon request.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

MARKETWISE, INC.

Date: August 10, 2023 By: /s/ Stephen Park

Name: Stephen Park

Title:

Interim Chief Financial Officer

FIRST AMENDMENT TO LOAN AND SECURITY AGREEMENT

This FIRST AMENDMENT TO LOAN AND SECURITY AGREEMENT, dated as of May 2, 2023 (this "<u>Amendment</u>"), is by and between MARKETWISE, LLC, a Delaware limited liability company ("<u>Borrower</u>") and HSBC BANK USA, NATIONAL ASSOCIATION, as administrative agent and collateral agent for the Lenders (in such capacity, "<u>Agent</u>").

WHEREAS, Borrower and Agent, among others, have previously entered into that certain Loan and Security Agreement, dated as of October 29, 2021 (the "Existing Loan Agreement", and as amended by this Amendment and as the same may be further amended, restated, amended and restated, supplemented or otherwise modified from time to time, the "Loan Agreement"), pursuant to which the Lenders have made certain loans and financial accommodations available to Borrower;

WHEREAS, pursuant to <u>Section 3.8(b)</u> of the Existing Credit Agreement, Borrower and Agent have agreed to amend the Existing Credit Agreement to replace USD LIBOR (as defined in the Existing Credit Agreement) in accordance with <u>Section 3.8</u> of the Existing Credit Agreement with one or more SOFR-based rates; and

WHEREAS, this Amendment and the transactions contemplated herein have been posted to the Lenders, and the Required Lenders have not objected to the proposed Amendment within the time period set forth in the definition of "Early Opt-In Effective Date" set forth in the Existing Credit Agreement.

NOW, THEREFORE, in consideration of the premises and agreements, provisions and covenants herein contained, the parties hereto agree as follows:

Section 1. Defined Terms; References.

- (a) Unless otherwise specifically defined herein, each term used herein which is defined in the Loan Agreement has the meaning assigned to such term in the Loan Agreement. The rules of construction and other interpretive provisions specified in Sections 1.1, 1.3 and 1.4 of the Loan Agreement shall apply to this Amendment, including terms defined in the preamble and recitals hereto.
- Section 2. *Amendments to the Loan Agreement.* Subject to the satisfaction in full of the conditions precedent set forth in <u>Section 3</u> hereof, effective as of the First Amendment Effective Date (as defined below):
- (a) the Existing Loan Agreement is hereby amended as set forth in <u>Annex A</u> attached hereto such that all of the newly inserted double underlined text (indicated textually in the same manner as the following example: <u>double-underlined text</u>) and any formatting changes attached hereto shall be deemed to be inserted and all of the stricken text (indicated textually in the same manner in the following example: stricken text) shall be deemed to be deleted therefrom.
 - (b) <u>Exhibit E</u> to the Existing Loan Agreement shall be amended by replacing such Exhibit with <u>Exhibit E</u> attached hereto as <u>Annex B</u>.
 - (c) Exhibit F to the Existing Loan Agreement shall be amended by replacing such Exhibit with Exhibit F attached hereto as Annex C.
- Section 3. *Effectiveness*. This Amendment shall not be effective until each of the following conditions precedent have been fulfilled to the satisfaction of (and in form and substance reasonably satisfactory to) Agent (such date, the "First Amendment Effective Date"):
 - (a) <u>Amendment</u>. This Amendment shall have been duly executed and delivered by Borrower and Agent, and Agent shall have received a fully executed copy hereof.
 - (b) <u>Fees and Expenses</u>. Agent shall have received all fees payable to Agent on or prior to the First Amendment Effective Date pursuant to the Loan Agreement and under the Fee

DB1/ 134130796.8

Letter and all other fees and expenses incurred by Agent on or prior to the First Amendment Effective Date.

- (c) <u>No Objection from Required Lenders</u>. Agent has not received as of 5:00 p.m. (New York City Time) on May 9, 2023 (i.e. the fifth Business Day after the date this Amendment is posted to the Lenders), a written notice of objection to this Amendment from Lenders comprising the Required Lenders.
- Section 4. *Effect of Amendment*. Except as expressly set forth herein or in the Loan Agreement, this Amendment shall not by implication or otherwise limit, impair, constitute a waiver of or otherwise affect the rights and remedies of the Lenders or Agent under the Loan Agreement or under any Loan Document and shall not alter, modify, amend or in any way affect any of the terms, conditions, obligations, covenants or agreements contained in the Loan Agreement or any other provision of the Loan Agreement or any Loan Document, all of which are ratified and affirmed in all respects and shall continue in full force and effect. On and as of the First Amendment Effective Date, each reference in the Loan Agreement to "this Agreement", "hereof", "hereunder", "herein" and "hereby" and each other similar reference, and each reference in any Loan Document to "the Loan Agreement", "thereof", "thereunder", "therein" or "thereby" or any other similar reference to the Loan Agreement shall refer to the Loan Agreement as amended hereby.
- Section 5. *Representations and Warranties.* As of the First Amendment Effective Date, the Borrower hereby represents and warrants to Agent that:
- (a) Borrower has full power, authority and legal right to enter into this Amendment and to perform all its Obligations hereunder and under the Loan Agreement, and the execution, delivery and performance of this Amendment (i) are within Borrower's limited liability company power, have been duly authorized, are not in contravention of applicable law or the terms of Borrower's operating agreement and certificate of formation, or other applicable documents relating to Borrower's formation or to the conduct of Borrower's business or of any agreement or undertaking to which Borrower's is a party or by which Borrower's is bound, and (ii) will neither conflict with nor result in any breach in any of the provisions of or constitute a default under or result in the creation of any Lien except Permitted Encumbrances upon any asset of Borrower under the provisions of any agreement, charter document, by-law, operating agreement, or other instrument to which Borrower or its property is a party or by which it may be bound;
- (b) this Amendment and the Loan Agreement constitute the legal, valid and binding obligation of Borrower, enforceable in accordance with their respective terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws relating to or limiting creditors' rights generally or by equitable principles relating to enforceability;
- (c) after giving effect to this Amendment, the representations and warranties of Borrower in the Loan Agreement and in each other Loan Document to which it is a party are true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations or warranties that already are qualified or modified as to "materiality" or "Material Adverse Effect" in the text thereof, which representations and warranties shall be true and correct in all respects subject to such qualification) on and as of such date as if made on and as of such date except to the extent such representations or warranties are limited by their terms to a specific date (in which case such representation or warranty shall be true and correct on and as of such earlier date in all material respects (except that such materiality qualifier shall not be applicable to any representations or warranties that already are qualified or modified as to "materiality" or "Material Adverse Effect" in the text thereof, which representations and warranties shall be true and correct in all respects subject to such qualification)); and
- (d) no Default or Event of Default has occurred and is continuing on the date hereof or would result from the transactions contemplated by this Amendment.
- Section 6. *Governing Law.* This Amendment shall be governed by and construed in accordance with the laws of the State of New York, without regard to conflicts of law principals.

Section 7. *Counterparts; Telecopied Signatures*. This Amendment may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Amendment, and any separate letter agreements with respect to fees payable to Agent, constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof. Except as provided in <u>Section 3</u> of this Amendment, this Amendment shall become effective when it shall have been executed by Agent and when Agent shall have received counterparts hereof that, when taken together, bear the signatures of each of the other parties hereto. Delivery of an executed counterpart of a signature page of this Amendment or any certificate delivered hereunder, by fax transmission or e-mail transmission (e.g. "pdf" or "tif") shall be effective as delivery of a manually executed counterpart of this Amendment. or such certificate. Without limiting the foregoing, to the extent a manually executed counterpart is not specifically required to be delivered under the terms of this Amendment, upon the request of any party, such fax transmission or e-mail transmission shall be promptly followed by such manually executed counterpart.

Section 8. Eurodollar Rate Loans Outstanding under Existing Loan Agreement. As of the First Amendment Effective Date, there may be one or more outstanding Eurodollar Rate Loans (the "Existing Eurodollar Rate Loans") under the Existing Loan Agreement. Prior to being repaid or prepaid, the Existing Eurodollar Rate Loans shall bear interest, and interest shall be payable by Borrower, in accordance with the terms and conditions of the Existing Loan Agreement applicable thereto. Such terms and conditions shall be applicable solely to Existing Eurodollar Rate Loans, and shall cease to apply or have any further force and effect if there are no Existing Eurodollar Rate Loans outstanding. Notwithstanding anything to the contrary herein, from and after the First Amendment Effective Date, no further Eurodollar Rate Loans will be made under the Loan Agreement and any outstanding Eurodollar Rate Loans will be converted to a SOFR Loan at the end of the applicable Interest Period.

Section 9. *Miscellaneous*. This Amendment constitutes a Loan Document.

[SIGNATURE PAGES FOLLOW]

DB1/ 134130796.8 3

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed by their respective authorized officers as of the day and year first above written.

MARKETWISE, LLC, as Borrower

By: /s/ Amber Mason

Name: Amber Mason Title: Chief Executive Officer

HSBC BANK USA, NATIONAL ASSOCIATION, as Agent

By: <u>/s/ Sebastian Ramirez</u> Name: Sebastian Ramirez Title: Associate Vice President

4 DB1/134130796.8

ANNEX A

[See attached.]

DB1/ 134130796.8 5

LOAN AND SECURITY AGREEMENT

dated as of

October 29, 2021

as amended by that certain First Amendment to Loan and Security Agreement dated as of May 2, 2023

among

MARKETWISE, LLC,

as Borrower,

the Guarantors that are from time to time parties hereto,

the Lenders that are from time to time parties hereto,

HSBC BANK USA, NATIONAL ASSOCIATION,

as Administrative Agent and Collateral Agent,

and

HSBC BANK USA, NATIONAL ASSOCIATION, and BMO CAPITAL MARKETS CORP.,

as Joint Lead Arrangers and Joint Bookrunners



I.	DEFINIT	TIONS. 1
	1.1.	Accounting Terms. 1
	1.2.	General Terms. 1
	1.3.	UCC Terms. 4142
	1.4.	Certain Matters of Construction. 4243
	1.5.	Divisions. 4243
II.	ADVAN	ICES, PAYMENTS. 4344
	2.1.	Revolving Advances and Swingline Loans. 4344
	2.2.	Procedure for Borrowing. 4445
	2.3.	Disbursement of Advance Proceeds. 46
	2.4.	Incremental Loans. 4647
	2.5.	
	2.6.	<u> </u>
	2.7.	
	2.8.	Statement of Account. 5051
		Letters of Credit. 5051
		Issuance of Letters of Credit. 5051
	2.11.	
		Additional Payments. 5253
	2.13.	
	2.14.	
		Use of Proceeds. 5556
	2.16.	-
III.		EST AND FEES. <u>5657</u>
	3.1.	
		Letter of Credit Fees; Cash Collateral. 5657
	3.3.	
		Fee Letter. 5758
	3.5.	r
		Maximum Charges. 5758
	3.7.	Increased Costs. 5859

3.8. Basis for Determining Interest Rate Inadequate or Unfair; Benchmark Replacement Setting. 5859

I.

(continued)

3.9. Capital Adequacy. 60<u>61</u>
3.10. Taxes. 61
IV. COLLATERAL: GENERAL TERMS. 6465
4.1. Security Interest in the Collateral. 6465
4.2. Perfection of Security Interest. 6465
4.3. [Reserved]. 66
4.4. Preservation of Collateral. 66
4.5. Ownership of Collateral. 66
4.6. Defense of Agent's and Lenders' Interests. 6667
4.7. Books and Records. 67
4.8. [Reserved]. 67
4.9. Compliance with Laws. 67
4.10. Inspection of Premises. 6768
4.11. Insurance. 67<u>68</u>
4.12. [Reserved]. 6869
4.13. Payment of Taxes. 68 <u>69</u>
4.14. Payment of Leasehold Obligations. 6869
4.15. [Reserved]. 69
4.16. [Reserved]. 69
4.17. [Reserved]. 69
4.18. Exculpation of Liability. 69
4.19. [Reserved]. 69
4.20. Financing Statements. 69
4.21. [Reserved]. 69 70
4.22. Agent as Collateral Agent. 6970
V. REPRESENTATIONS AND WARRANTIES. 7172
5.1. Authority. 71 <u>72</u>
5.2. Formation and Qualification. 72
5.3. [Reserved]. 72 <u>73</u>
5.4. Tax Returns. 72 73
5.5. Financial Statements. 7273

(continued)

5.6. [Reserv	ved]. 73
5.7. O.S.H.	A. and Environmental Compliance. 73
5.8. Solveno	cy; No Litigation, Violation, Indebtedness or Default. 7374
5.9. Patents	, Trademarks, Copyrights and Licenses. 74 <u>75</u>
5.10. Licens	ses and Permits. 75
5.11. No De	efaults. 75
5.12. No Bu	ardensome Restrictions. 7576
5.13. No La	bor Disputes. 75 <u>76</u>
5.14. Margi	n Regulations. 75 <u>76</u>
5.15. Invest	ment Company Act. 76
5.16. Disclo	osure. 76
5.17. Swaps	s. 76
5.18. Confli	icts. 76 77
5.19. [Rese	rved]. 76 77
5.20. Busin	ess and Property of Loan Parties. 7677
5.21. Mater	ial Contracts. 77
5.22. Sancti	ons. 77
5.23. Anti-0	Corruption and Anti-Bribery Laws. 77
5.24. Federa	al Securities Laws. 7778
5.25. Benef	icial Ownership Certification. 7778
VI. AFFIRMATIVE	E COVENANTS. 7778
6.1. Paymer	nt of Fees. 78
6.2. Conduc	et of Business and Maintenance of Existence and Assets. 78
6.3. Violation	ons. <u>.</u> 78
6.4. Use of	Proceeds 78
6.5. Executi	ion of Supplemental Instruments. <u>.</u> 78
6.6. Paymer	nt of Indebtedness. <u>.</u> 7879
6.7. Standar	rds of Financial Statements 7879
6.8. Financi	al Covenants. 79
6.9. Keepw	ell 79
6.10. Desig	nation of Subsidiaries 7980
6.11. Accou	ints 80

6.12. [Reserved] 80

(continued)

	6.13.	Post-Closing Obligations 80
VII.	NEGA	TIVE COVENANTS. 8081
	7.1.	Merger, Consolidation and Sale of Assets. 8081
	7.2.	Creation of Liens; Negative Pledges. 81
	7.3.	Guarantees. 81
	7.4.	Investments. 8182
	7.5.	[Reserved]. 82
	7.6.	Restricted Payments. 82
	7.7.	Indebtedness. 83
	7.8.	Nature of Business. 83
	7.9.	Transactions with Affiliates. 83
	7.10.	[Reserved]. 83
	7.11.	Subsidiaries 83
	7.12.	Fiscal Year and Accounting Changes. 8384
	7.13.	Compliance with ERISA. 8384
	7.14.	Amendment of Documents and Material Contracts. 84
	7.15.	State of Organization. 8485
	7.16.	Sanctions; Anti-Bribery Laws. 84 85
VIII.	CONI	DITIONS PRECEDENT. 85
	8.1.	Conditions to the Closing Date. 85
	8.2.	Conditions to Each Advance. 87
IX.	INFOR	MATION AS TO LOAN PARTIES. 88
	9.1.	Disclosure of Material Matters. 88
	9.2.	[Reserved]. 8889
	9.3.	[Reserved]. 88 89
	9.4.	Litigation. 8889
	9.5.	Material Occurrences. 8889
	9.6.	[Reserved]. 89
	9.7.	Annual Audited Financial Statements. 89
	9.8.	Quarterly Financial Statements. 89 <u>90</u>
	9.9.	[Reserved]. 90

9.10. Other Reports. 90

(continued)

9.11.	Additional Information. 90
9.12.	Projected Operating Budget. 90
9.13.	[Reserved]. 9091
9.14.	Notice of Suits, Adverse Events. 9091
9.15.	ERISA Notices and Requests. 9091
9.16.	[Reserved]. 91
9.17.	Reconciliations. 91
9.18.	Beneficial Ownership Documentation. 9192
9.19.	[Reserved]. 92
9.20.	Additional Documents. 92
X. EVENTS	OF DEFAULT. 92
XI. LENDE	RS' RIGHTS AND REMEDIES AFTER DEFAULT. 95
11.1.	Rights and Remedies. 95
11.2.	Application of Proceeds. 9596
11.3.	Agent's Discretion. 96
11.4.	Setoff. 96
11.5.	Rights and Remedies Not Exclusive. 96
XII. WAIVE	RS AND JUDICIAL PROCEEDINGS. 9697
12.1.	Waiver of Notice. 9697
12.2.	Delay. 97
12.3.	Jury Waiver. 97
XIII. EFFEC	CTIVE DATE AND TERMINATION. 97
13.1.	Term. 97
13.2.	Termination. 9798
XIV. REGA	RDING AGENT. 98
14.1.	Appointment. 98
14.2.	Nature of Duties. 9899
14.3.	Lack of Reliance on Agent and Resignation. 100
14.4.	Certain Rights of Agent. 101
14.5.	Reliance. 101
14.6.	Notice of Default. 102
14.7.	Indemnification. 102

(continued)

	14.8.	Agent in its Individual Capacity. 102
	14.9.	Delivery of Documents. 102103
	14.10.	Loan Parties' Undertaking to Agent. 102 103
	14.11.	Bankruptcy Proceedings. 103
	14.12.	No Liability for Clean-Up of Hazardous Materials. 103
	14.13.	Certain ERISA Matters. 103 104
	14.14.	Disclaimer of Liability Rates 104
	14.15.	Erroneous Payments. 105
	14.16.	Joint Lead Arrangers and Joint Book Runners 107
XV.	MISCEI	LLANEOUS. 108
	15.1.	Governing Law. 108
	15.2.	Entire Understanding; Amendments. 108
	15.3.	Successors and Assigns; Participations; New Lenders. 110
	15.4.	Application of Payments. 112
	15.5.	Indemnity; Funding Losses. 112
	15.6.	Notice. 113
	15.7.	Survival. 114
	15.8.	Severability. 114
	15.9.	Expenses. 114
	15.10.	Injunctive Relief. 115
	15.11.	Consequential Damages. 115
	15.12.	Captions. 115
	15.13.	Counterparts; Telecopied Signatures. 115
	15.14.	Construction. 116
	15.15.	Confidentiality. 116
	15.16.	Publicity. 117
	15.17.	Electronic Execution of Assignments and Certain Loan Documents 117
	15.18.	Confirmation of Flood Policies and Procedures 117
	15.19.	Patriot Act Notice. 117
	15.20.	Acknowledgement Regarding Any Supported QFCs. 418117
	15.21.	Acknowledgement and Consent to Bail-In of EEA Financial Institutions. 118

15.22. Currency Conversion. 119118

(continued)

15.23. Lender Representations. 119

-vii-

DB1/ 124211234.18134029896.11

List of Exhibits and Schedules

Exhibits

<u>Exhibit A</u>	[Reserved]
Exhibit B	Form of Compliance Certificate
Exhibit C	Form of Revolving Credit Note
Exhibit D	Form of Promissory Note for Swingline Loans
Exhibit E	Form of Notice of Borrowing
Exhibit F	Form of Notice of Conversion
<u>Exhibit G</u>	Form of Term Note
Exhibits H	<u>-1</u> to <u>H-4</u> Form of U.S. Tax Compliance Certificates
Exhibit I	Form of Commitment Transfer Supplement

.

Schedules
Schedule 1.1 Commitments
Schedule 1.2(a) Commercial Tort Claims
Schedule 1.2(b) Subsidiary Guarantors
Schedule 1.3 Unrestricted Subsidiaries
Schedule 4.6 Locations
Schedule 5.2 Subsidiaries and Equityholders
Schedule 5.7 Environmental
Schedule 5.8(b) Litigation
Schedule 5.8(d) Plans
Schedule 5.9 Intellectual Property
Schedule 5.10 Licenses and Permits
Schedule 6.13 Post-Closing Obligations
Schedule 7.2 Existing Liens
Schedule 7.3 Existing Guarantees
Schedule 7.4 Existing Investments
Schedule 7.7 Existing Indebtedness
Schedule 7.9 Existing Affiliate Transactions

LOAN AND SECURITY AGREEMENT

Loan and Security Agreement dated October 29, 2021 among MarketWise, LLC, a limited liability company organized under the laws of the State of Delaware ("Borrower"), the Guarantors (as hereinafter defined) which are now or which hereafter become a party hereto, HSBC BANK USA, NATIONAL ASSOCIATION ("HSBC") and the other financial institutions which are now or which hereafter become a party hereto (each a "Lender" and collectively, the "Lenders"), HSBC, as administrative agent and collateral agent for the Lenders (in such capacity, the "Agent"), and HSBC and BMO CAPITAL MARKETS CORP., as joint lead arrangers (in such capacity, the "Joint Lead Arrangers") and joint bookrunners (in such capacity, the "Joint Bookrunners").

WHEREAS, Borrower has requested that the Lenders provide a revolving credit facility of up to an aggregate principal amount of \$150,000,000 to pay fees and expenses related to the Transactions (as hereinafter defined) and for ongoing working capital and general corporate purposes, and the Lenders have agreed to provide such a revolving credit facility and the Issuer (as hereinafter defined) has indicated its willingness to provide a letter of credit facility (as a sub-facility of such revolving credit facility), in each case, subject to the terms and conditions of this Agreement; and

WHEREAS, the Lenders have indicated their willingness to lend, and the Issuer have indicated their willingness to issue Letters of Credit, on the terms and subject to the conditions set forth herein.

IN CONSIDERATION of the mutual covenants and undertakings herein contained, each Loan Party (as defined below), Lenders and Agent hereby agree as follows:

I. <u>DEFINITIONS</u>.

1.1. Accounting Terms.

As used in this Agreement, the Notes, any Loan Document, or any certificate, report or other document made or delivered pursuant to this Agreement, accounting terms not defined in Section 1.2 or elsewhere in this Agreement and accounting terms partly defined in Section 1.2 to the extent not defined, shall have the respective meanings given to them under GAAP; provided, however, whenever such accounting terms are used for the purposes of determining compliance with financial covenants in this Agreement, such accounting terms shall be defined in accordance with GAAP as applied in preparation of the audited financial statements of Parent for the fiscal year ended December 31, 2021.

1.2. General Terms.

For purposes of this Agreement the following terms shall have the following meanings:

"Accountants" shall have the meaning set forth in Section 9.7.

"Adjusted Term SOFR" means, for purposes of any calculation, the rate per annum equal to (a) Term SOFR for such calculation plus (b) the Term SOFR Adjustment; provided that if Adjusted Term SOFR as so determined shall ever be less than the Floor, then such rate shall be deemed to be equal to the Floor for purposes of this Agreement.

"Adjusted LIBO Rate" shall mean, with respect to any Eurodollar Rate Loan for any Interest Period a rate of interest equal to:

- (a) the offered rate for deposits in Dollars in the London interbank market for the relevant Interest Period which is shown on the Bloomberg Page BBAM1 (or on such other substitute Bloomberg page that displays rates at which Dollar deposits are offered by leading banks in the London interbank market) ("LIBOR") as of 11:00 a.m. (London time) on the day which is two (2) Business Days prior to the first day of such Interest Period for a term comparable to such Interest Period; provided, however, that if, for any reason, such a rate is not published by ICE Benchmark Administration (or any successor thereto) or available on the Bloomberg Page BBAM1 (or on such other substitute Bloomberg page that displays rates at which Dollar deposits are offered by leading banks in the London interbank market), LIBOR shall be equal to a rate per annum equal to the average rate (rounded upwards, if necessary, to the nearest one-eighth of one percent (0.125%)) at which Agent determines that Dollars in an amount comparable to the amount of the applicable Advances are being offered to prime banks at approximately 11:00 a.m. (London time) on the day which is two (2) Business Days prior to the first day of such Interest Period for a term comparable to such Interest Period for settlement in immediately available funds by leading banks in the London interbank market selected by Agent; divided by
- (b) a number equal to 1.0 minus the aggregate (but without duplication) of the rates (expressed as a decimal fraction) of reserve requirements in effect on the day which is two (2) Business Days prior to the beginning of such Interest Period (including, without limitation, basic, supplemental, marginal and emergency reserves under any regulations of the Board of Governors of the Federal Reserve System or other Governmental Body having jurisdiction with respect thereto, as now and from time to time in effect) for eurocurrency funding (currently referred to as "Eurocurrency Liabilities" in Regulation D of such Board) which are required to be maintained by a member bank of the Federal Reserve System (the "Euro-Rate Reserve Percentage"); such rate (if greater than zero) to be rounded upward to the next whole multiple of one-sixteenth of one percent (0.0625%).

Notwithstanding the foregoing, if the Adjusted LIBO Rate determined as provided above would be less than zero, such rate shall be deemed to be zero for purposes of this Agreement.

- "Adjustment Date" shall have the meaning set forth in the definition of "Applicable Margin".
- "Advances" shall mean and include the Revolving Advances, Swingline Loans, any Incremental Term Loan, and Letters of Credit.
- "Affected Financial Institution" means (a) any EEA Financial Institution or (b) any UK Financial Institution.
- "Affiliate" of any Person shall mean (a) any Person which, directly or indirectly, is in control of, is controlled by, or is under common control with such Person, or (b) any Person who is a director, officer, manager, managing member or partner (i) of such Person, (ii) of any Subsidiary of such Person or (iii) of any Person described in clause (a) above. For purposes of this definition, control of a Person shall mean the power, direct or indirect, to direct or cause the direction of the management and policies of such Person whether by ownership of Equity Interests, contract or otherwise.
 - "Agent" shall have the meaning set forth in the preamble to this Agreement and shall include its permitted successors and assigns.
 - "Agent Parties" shall have the meaning set forth in Section 15.23.

"Agreement" shall mean this Loan and Security Agreement, as the same may be amended, restated, modified and/or supplemented from time to time.

"Anticipated Cure Deadline" has the meaning specified therefor in Section 10.15.

"<u>Applicable Law</u>" shall mean all laws, rules and regulations applicable to the Person, conduct, transaction, covenant, Loan Document or contract in question, including all applicable common law and equitable principles; all provisions of all applicable state, federal and foreign constitutions, statutes, rules, regulations and orders of any Governmental Body, and all orders, judgments and decrees of all courts and arbitrators.

"Applicable Margin" shall mean, as of the Closing Date, the applicable percentage specified below:

APPLICABLE MARGIN FOR DOMESTIC RATE LOANS	APPLICABLE MARGIN FOR EURODOLLAR RATESOFR LOANS
0.50%	1.50%

Thereafter, effective as of the first Business Day following receipt by <u>LenderAgent</u> of the financial statements of Borrower and Compliance Certificate for the quarter ending September 30, 2021 required under Section 9.8, and thereafter upon receipt of the financial statements of Borrower required under Sections 9.7 and 9.8, as applicable, for each fiscal quarter or fiscal year ending thereafter (each day of such delivery, an "Adjustment Date"), the Applicable Margin shall be adjusted, if necessary, to the applicable percent per annum set forth in the pricing table set forth below corresponding to the Net Leverage Ratio for the trailing twelve month period ending on the last day of the most recently completed fiscal quarter prior to the applicable Adjustment Date (each such period, a "Calculation Period"):

NET LEVERAGE RATIO	APPLICABLE MARGINS FOR DOMESTIC RATE LOANS	APPLICABLE MARGINS FOR EURODOLLAR RATESOFR LOANS
Less than 0.50:1.00	0.50%	1.50%
Greater than or equal to 0.50:1.00 but less than 1.00:1.00	0.75%	1.75%
Greater than or equal to 1.00:1.00 but less than 1.50:1.00	1.00%	2.00%
Greater than or equal to 1.50:1.00	1.25%	2.25%

If Borrower shall fail to deliver the financial statements, certificates and/or other information required under <u>Sections 9.7</u> and <u>9.8</u>, as applicable, by the dates required pursuant to such sections, each Applicable Margin shall be conclusively presumed to equal the highest Applicable Margin specified in the pricing table set forth above until the date of delivery of such financial statements, certificates and/or other information, at which time the rate will be adjusted based upon the Net Leverage Ratio reflected in such statements.

If, as a result of any restatement of, or other adjustment to, the financial statements of Borrower or for any other reason, Agent determines in its reasonable discretion after consultation with Borrower that (a) the Net Leverage Ratio as previously calculated as of any applicable date was inaccurate, and (b) a proper calculation of the Net Leverage Ratio would have resulted in different pricing for any period, then (i) if the proper calculation of the Net Leverage Ratio would have resulted in higher pricing for such period, Borrower shall automatically and retroactively be required to pay to the Lender, promptly upon demand by the Lender, an amount equal to the excess of the amount of interest that should have been paid for such period over the amount of interest actually paid for such period; and (ii) if the proper calculation of the Net Leverage Ratio would have resulted in lower pricing for such period, Lender shall have no obligation to repay interest to Borrower; provided that, if as a result of any restatement or other event a proper calculation of the Net Leverage Ratio would have resulted in higher pricing for one or more periods and lower pricing for one or more other periods (due to the shifting of income or expenses from one period to another period or any similar reason), then the amount payable by Borrower pursuant to clause (i) above shall be equal to the excess, if any, of the amount of interest that should have been paid for all applicable periods over the amounts of interest actually paid for such periods.

"Approved Fund" shall mean any Fund that is advised or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or Affiliate of an entity that administers or manages a Lender.

"<u>Authority</u>" shall have the meaning set forth in <u>Section 4.19(c)</u>.

"Authorized Officer" shall mean with respect to any Loan Party, the chief executive officer, president, vice president, chief financial officer, senior accountant, treasurer, assistant treasurer, assistant secretary, controller, comptroller or any other Person with an equivalent title or role of such Loan Party or such other officers as Borrower shall designate in writing to Agent, but in any event, with respect to financial matters, the chief financial officer, senior accountant, treasurer, assistant treasurer, controller, comptroller or any other Person with an equivalent title or role of Borrower.

"Available Tenor" shall meanmeans, as of any date of determination and with respect to the then-current Benchmark, as applicable, (x) if the then-currentsuch Benchmark is a term rate, any tenor for such Benchmark (or component thereof) that is or may be used for determining the length of an Interest Period (or any similar or analogous definition) pursuant to this Agreement or (y) otherwise, any payment period for interest calculated with reference to such Benchmark, as applicable, (or component thereof) that is or may be used for determining any frequency of making payments of interest calculated with reference to such Benchmark pursuant to this Agreement, in each case, as of such date and not including, for the avoidance of doubt, any tenor for such Benchmark that is then-removed from the definition of "Interest Period" pursuant to Section 3.8.

"Bail-In Action" means the exercise of any Write-Down and Conversion Powers by the applicable EEA Resolution Authority in respect of any liability of an EEA Financial Institution.

"Bail-In Legislation" means, with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule.

"Bank Product Obligations" shall mean the following services or facilities extended to any Loan Party by Agent, Lender or any of their respective Affiliates: (a) credit cards, (b) credit card processing services, (c) debit cards, (d) purchase cards, (e) ACH transactions, EFT or any other electronic funds transfer, and (f) cash management, including controlled disbursement accounts or services.

"Bankruptcy Code" shall mean Chapter 1 of Title 11 of the United States Code (11 U.S.C §202, et seq.) as amended from time to time.

"Bankruptcy Proceeding" shall have the meaning set forth in Section 14.11.

"Base Rate" shall mean, on any date, a variable rate of interest per annum equal to the highest of (a) the highest of the "prime rate," "reference rate," "base rate" or other similar rate as determined by Agent (or any successor to Agent) announced from time to time by HSBC (or any successor to HSBC) (with the understanding that any such rate may merely be a reference rate and may not necessarily represent the lowest or best rate actually charged to any customer by such bank), (b) the Federal Funds Rate plus ½ of 1%, and (c) the Adjusted LIBO Rate Term SOFR for a one month Interest Period on such day (or if such day is not a Business Day, the immediately preceding Business Day) plus one percent (1.00%) per annum; and each change in any interest rate provided for in this Agreement based upon Base Rate shall take effect at the time of such change in the Base Rate.

"Benchmark" means, initially, USD LIBORthe Term SOFR Reference Rate; provided that,—if a replacement of the—Benchmark Transition Event has occurred pursuant to Section 3.8(b) titled "Benchmark Replacement Setting" with respect to the Term SOFR Reference Rate or the then-current Benchmark, then "Benchmark" means the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate. Any reference to "Benchmark" shall include, as applicable, the published component used in the calculation thereof pursuant to Section 3.8.

"Benchmark Replacement" means, for any Available Tenor:

- (1) For purposes of Section 3.8(b)(i) "Benchmark Replacement" means with respect to any Benchmark Transition Event, the first alternative set forth in the order below that can be determined by Agent for the applicable Benchmark Replacement Date:
- (a) the sum of: (i) TermDaily Simple SOFR and (ii) 0.11448% (11.448 basis points) for an Available Tenor of one-month's duration, 0.26161% (26.161 basis points) for an Available Tenor of three-months' duration, and 0.42826% (42.826 basis points) for an Available Tenor of six-months' duration, or; or
 - (b) the sum of: (i) Daily Simple SOFR and (ii) the spread adjustment selected or recommended by the Relevant Governmental Body for the replacement of the tenor of USD LIBOR with a SOFR-based rate having approximately the same length as the interest payment period specified in Section 3.8(b)(i); and
- (2) For purposes of Section 3.8(b)(ii), the sum of: (ai) the alternate benchmark rate and (b) an adjustment (which may be a positive or negative value or zero), in each case, that has been selected by the Agent and the Borrower as the replacement for such Available Tenor of such Benchmark giving due consideration to (A) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (B) any evolving or then-prevailing market convention; including any applicable recommendations made by the Relevant Governmental Body, for U.S. dollar-denominated for determining a benchmark rate as a replacement to the then-current Benchmark for Dollar-denominated syndicated credit facilities at such time; and (ii) the related Benchmark Replacement Adjustment.

provided that, if If the Benchmark Replacement as determined pursuant to clause (1a) or (2b) above would be less than the Floor, the Benchmark Replacement then such rate will be deemed to be equal to the Floor for the purposes of this Agreement and the other Loan Documents.

"Benchmark Replacement Adjustment" means, with respect to any replacement of LIBORthe then-current Benchmark with an Unadjusted Benchmark Replacement for each applicable Interest Period, the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by Agent and the Borrower giving due consideration to (ia) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of LIBORsuch Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body or (ib) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of LIBORsuch Benchmark with the applicable Unadjusted Benchmark Replacement for Dollar-denominated syndicated credit facilities at such time.

"Benchmark Replacement Conforming Changes" means, with respect to any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Base Rate," the definition of "Business Day," the definition of "Interest Period," timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, the applicability and length of lookback periods, the applicability of breakage provisions, and other technical, administrative or operational matters) that Agent decides may be appropriate to reflect the adoption and implementation of such Benchmark Replacement and to permit the administration thereof by Agent in a manner substantially consistent with market practice (or, if Agent decides that adoption of any portion of such market practice is not administratively feasible or if Agent determines that no market practice for the administration of such Benchmark Replacement exists, in such other manner of administration as Agent decides is reasonably necessary in connection with the administration of this Agreement and the other Loan Documents).

"Benchmark Replacement Date" means the earliest to occur of the following events with respect to the then-current Benchmark:

- (a) in the case of clause (a) or (b) of the definition of "Benchmark Transition Event", the later of (i) the date of the public statement or publication of information referenced therein and (ii) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide all Available Tenors of such Benchmark (or such component thereof); or
- (b) in the case of clause (c) of the definition of "Benchmark Transition Event", the first date on which such Benchmark (or the published component used in the calculation thereof) has been determined and announced the regulatory supervisor for the administrator of such Benchmark (or such component thereof) to be non-representative; provided that such non-representativeness will be determined by reference to the most recent statement or publication referenced in such clause (c) and even if any Available Tenor of such Benchmark (or such component thereof) continues to be provided on such date.

For the avoidance of doubt, the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (a) or (b) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

<u>"Benchmark Transition Event" means the occurrence of one or more of the following events with respect to the then-current Benchmark:</u>

(a) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof);

"Benchmark Transition Event" means, with respect to any then-current Benchmark other than USD LIBOR, the occurrence of (b) a public statement or publication of information by or on behalf of the administrator of the then-current Benchmark, thethe regulatory supervisor for the administrator of such Benchmark, the Board of Governors of the (or the published component used in the calculation thereof), the Federal Reserve SystemBoard, the Federal Reserve Bank of New York, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark, announcing or stating that (a) such (or such component), which states that the administrator of such Benchmark (or such component) has ceased or will cease on a specified date—to provide all Available Tenors of such Benchmark; (or such component thereof) permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof); or

(b)c) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that all Available Tenors of such Benchmark are or will no longer be(or such component thereof) are not, or as of a specified future date will not be, representative of the underlying market and economic reality that such Benchmark is intended to measure and that representativeness will not be restored.

For the avoidance of doubt, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Unavailability Period" means, the period (if any) (a) beginning at the time that a Benchmark Replacement Date has occurred if, at such time, no Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 3.8 and (b) ending at the time that a Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 3.8.

"Beneficial Ownership Certification" means a certification regarding beneficial ownership required by the Beneficial Ownership Regulation.

"Beneficial Ownership Regulation" means 31 C.F.R. Sec. 1010.230.

"Benefit Plan" means any of (a) an "employee benefit plan" (as defined in ERISA) that is subject to Title I of ERISA, (b) a "plan" as defined in and subject to Section 4975 of the Code or (c) any Person

whose assets include (for purposes of ERISA Section 3(42) or otherwise for purposes of Title I of ERISA or Section 4975 of the Code) the assets of any such "employee benefit plan" or "plan".

"BHC Act Affiliate" of a Person means an "affiliate" (as such term is defined under, and interpreted in accordance with, 12 U.S.C. 1841(k)) of such Person.

"Borrower" shall have the meaning set forth in the preamble to this Agreement and shall extend to all of its successors and assigns expressly permitted under this Agreement.

"Borrower Consolidating Financials Delivery Period" means the period during which Borrower is required to deliver Consolidating Annual Financials or Consolidating Quarterly Financials pursuant to Sections 9.7 or 9.8 hereof, as applicable.

"Borrower's Account" shall have the meaning set forth in Section 2.8.

"Business Day." shall mean, with respect to Eurodollar Rate Loans, any day on which commercial banks are open for domestic and international business, including dealings in Dollar deposits, in London, England and New York, New York, and with respect to all other matters, any day other than a day on which commercial banks in New York, New York are authorized or required by law to close.

"Calculation Period" shall have the meaning set forth in the definition of "Applicable Margin".

"Capital Lease" shall mean any lease of any property (whether real, personal or mixed) that, in conformity with GAAP, should be accounted for as a capital lease.

"Capital Lease Obligations" shall have the meaning provided in sub-clause (c) of the definition of "Indebtedness".

"Cash Equivalents" shall mean: (a) marketable direct obligations issued or unconditionally guaranteed by the United States Government or issued by any agency thereof and backed by the full faith and credit of the United States, in each case maturing within one (1) year from the date of acquisition thereof; (b) commercial paper maturing no more than one (1) year from the date issued and, at the time of acquisition, having a rating of at least A-1 from Standard & Poor's Ratings Service or at least P-1 from Moody's Investors Service, Inc.; (c) certificates of deposit or bankers' acceptances maturing within one (1) year from the date of issuance thereof issued by, or overnight reverse repurchase agreements from, any commercial bank organized under the laws of the United States of America or any state thereof or the District of Columbia having combined capital and surplus of not less than \$500,000,000 and whose short-term debt obligations are rated at least P-1 by Moody's Investors Service, Inc. or at least A-1 by Standard & Poor's Ratings Service; (d) up to \$100,000 per institution and up to \$1,000,000 in the aggregate in (i) short-term debt obligations issued by any local commercial bank or trust company located in those areas where Borrower conducts its business, whose deposits are insured by the Federal Deposit Insurance Corporation, or (ii) commercial bank-insured money market funds, or any combination of investments described in clauses (i) and (ii); (e) overnight investments with such financial institutions having a short term deposit rating of at least P-1 by Moody's Investors Service, Inc. or at least A-1 by Standard & Poor's Ratings Service and (f) money market mutual funds that invest solely in the investments described in clauses (a) through (e) above.

"CERCLA" shall mean the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, 42 U.S.C. §§9601 et seq.

"CFC" shall mean a Subsidiary of Borrower that is a "controlled foreign corporation" within the meaning of Section 957 of the Code.

"Change of Control" shall mean any of the following:

- (a) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act, but excluding any employee benefit plan of such person or its Subsidiaries, and any person or entity acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) other than one or more Permitted Holders becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act, except that a "person" or "group" shall be deemed to have "beneficial ownership" of all Equity Interests that such "person" or "group" has the right to acquire, whether such right is exercisable immediately or only after the passage of time (such right, an "option right")), directly or indirectly, of more than thirty-five percent (35%) of the Equity Interests of Parent entitled to vote in the election of members of the board of directors (or equivalent governing body) of Parent and the percentage of such Equity Interest so held is greater than the percentage of the Equity Interests of Parent entitled to vote in the election of members of the board of directors (or equivalent governing body) of Parent held, directly or indirectly, in the aggregate by the Permitted Holders;
- (b) Permitted Holders and Parent shall fail to collectively own, beneficially and of record, the Equity Interests of Borrower representing 100% of the economic interest of Borrower;
 - (c) Parent shall cease to Control all of the business and affairs of Borrower;
- (d) Borrower fails to directly or indirectly own and control one hundred percent (100%) of each class of the outstanding Equity Interests of each other Loan Party, except in connection with a transaction permitted under this Agreement; or
 - (e) any "change in control" (or equivalent concept) shall occur under any Material Indebtedness.

"Closing Date" shall mean the date on which all of the conditions precedent set forth in Section 8.1 shall have been satisfied or waived by Agent and, as applicable, the Lenders or the Required Lenders, which date is October 29, 2021.

"Code" shall mean the Internal Revenue Code of 1986, as the same may be amended or supplemented from time to time, and any successor statute of similar import, and the rules and regulations thereunder, as from time to time in effect.

"Collateral" shall mean and include, with respect to each Loan Party:

- (a) all Receivables;
- (b) all Equipment;
- (c) all General Intangibles (including, without limitation, all Intellectual Property);
- (d) all chattel paper;
- (e) all Inventory;
- (f) all instruments;

- (g) all goods;
- (h) all Investment Property, including, without limitation, all Subsidiary Stock;
- (i) Equity Interests;
- (j) all money, cash and Cash Equivalents;
- (k) all letters of credit, letter-of-credit rights and supporting obligations;
- (l) all deposit accounts and securities accounts with any bank or other financial institution or securities intermediary (including all cash, Cash Equivalents, financial assets, negotiable instruments and other evidence of payment, and other funds on deposit or credited thereto);
 - (m) all commercial tort claims, including those specified on Schedule 1.2(a); and
- (n) all of the Proceeds and products, whether tangible or intangible, of any of the foregoing, including proceeds of insurance or commercial tort claims covering or relating to any or all of the foregoing, and any and all Receivables, books and records, chattel paper, deposit accounts, Equipment, instruments, goods, fixtures, General Intangibles, Inventory, Investment Property, Intellectual Property, securities accounts, Equity Interests, letters of credit, letter-of-credit rights, supporting obligations, money, cash, Cash Equivalents or other tangible or intangible property resulting from the sale, lease, license, exchange, collection, or other disposition of any of the foregoing, the proceeds of any award in condemnation with respect to any of the foregoing, any rebates or refunds, whether for taxes or otherwise, and all proceeds of any such proceeds, or any portion thereof or interest therein, and the proceeds thereof, and all proceeds of any loss of, damage to, or destruction of the above, whether insured or not insured, and, to the extent not otherwise included, any indemnity, warranty, or guaranty payable by reason of loss or damage to, or otherwise with respect to any of the foregoing (the "Proceeds"). Without limiting the generality of the foregoing, the term "Proceeds" includes whatever is receivable or received when Investment Property or proceeds are sold, exchanged, collected, or otherwise disposed of, whether such disposition is voluntary or involuntary, and includes proceeds of any indemnity or guaranty payable to any Loan Party or Agent from time to time with respect to any of the Investment Property.

Notwithstanding any of the other provisions set forth in this Agreement, this Agreement shall not constitute a grant of a security interest in, and "Collateral" shall not include: (A) any property to the extent that such grant of a security interest is prohibited by any Applicable Law or constitutes a breach or default under or results in the termination of or requires any consent not obtained under, any contract, lease, permit, license or license agreement, except to the extent that such Applicable Law or the term in such contract, lease, permit, license or license agreement is ineffective under Section 9-406, 9-407, 9-408 or 9-409 of the UCC (or any successor provision or provisions) of any relevant jurisdiction or any other Applicable Law (including the Bankruptcy Code) or principles of equity; provided, however, that such security interest shall attach immediately at such time as such Applicable Law is not effective or applicable, or such prohibition, breach, default or termination is no longer applicable or is waived, and to the extent severable, shall attach immediately to any portion of the Collateral that does not result in such consequences; and provided, further, that the foregoing shall not be deemed to limit, impair, or otherwise affect Agent's continuing security interests in and Liens upon any rights or interest of any Loan Party in or to (i) monies due or to become due under or in connection with any described contract, lease, permit, license or license agreement (including, without limitation, in respect of any Receivables or proceeds of Inventory), or (ii) any proceeds from the sale, license, lease or other disposition of any such contract, lease, permit, license or license agreement; (B) interests in Real Property; (C) motor vehicles and other assets subject to certificates of title, (D) Subsidiary Stock in excess of 65% of the voting Equity Interests of any FSHCO or CFC to the extent such action would, in the good faith judgment of Borrower, result in

adverse Tax consequences to Borrower or any of its direct or indirect beneficial owners; (E) any assets to the extent that the costs of such perfection shall exceed the benefit of the collateral security provided to Agent, as determined by Agent in its reasonable discretion; (F) any asset subject to capital leases and purchase money financing to the extent such capital leases and purchase money financing are permitted under this Agreement and prohibit the granting of a Lien; (G) any application for registration of a trademark filed with the USPTO on an intent-to-use basis until such time (if any) as a statement of use or amendment to allege use is accepted by the USPTO, at which time such trademark shall automatically become part of the Collateral and subject to the security interest pledged; (H) Excluded Accounts; (I) Equity Interests of any Subsidiary that is a direct or indirect Subsidiary of any FSHCO or CFC; and (J) for the avoidance of doubt, the Equity Interests of Borrower (clauses (A) through (J) above, collectively, the "Excluded Property"); provided that "Excluded Property" shall not include (A) any Proceeds, products, substitutions or replacements would otherwise constitute Excluded Property).

"Collateral Agent" shall mean HSBC, in its capacity as collateral agent for Agent, the Issuer and the Lenders, and its permitted successors and assigns.

"Commitment Fee Percentage" shall mean, as of the most recent Adjustment Date, the applicable percent per annum set forth in the pricing table set forth below corresponding to the Net Leverage Ratio for the Calculation Period:

NET LEVERAGE RATIO	COMMITMENT FEE PERCENTAGE
Less than 0.50:1.00	0.25%
Greater than or equal to 0.50:1.00 but less than 1.00:1.00	0.30%
Greater than or equal to 1.00:1.00 but less than 1.50:1.00	0.30%
Greater than 1.50:1.00	0.35%

<u>provided that</u> for the period from the Closing Date through and including the first Adjustment Date following the Closing Date, the Commitment Fee Percentage shall be 0.25%.

"Commitment Percentage" of any Lender shall mean each percentage set forth below such Lender's under the applicable heading on Schedule 1.1 to this Agreement as same may be adjusted upon any assignment by a Lender pursuant to Section 15.3 or upon the making of an Incremental Term Loan or an Incremental Revolving Commitment pursuant to Section 2.4.

"Commitment Transfer Supplement" shall mean a document in the form of Exhibit I, properly completed and otherwise in form and substance satisfactory to Agent by which the Purchasing Lender purchases and assumes a portion of outstanding Advances and the obligation of Lenders to make Advances under this Agreement.

"Commitments" shall mean, as to any Lender, its obligation to make Advances (including participating in Letters of Credit) in an aggregate amount not to exceed at any one time outstanding the amount set forth below such Lender's name under the applicable heading on Schedule 1.1 to this Agreement under the heading "Commitment Amount", as same may be adjusted upon any assignment by a Lender pursuant to Section 15.3 or upon the making of an Incremental Term Loan or an Incremental Revolving Commitment pursuant to Section 2.4.

"Commodity Exchange Act" shall mean the Commodity Exchange Act (7 U.S.C. §1 et seq.), as amended from time to time, and any successor statute.

"Compliance Certificate" shall mean a certificate of an Authorized Officer of Borrower in the form of Exhibit B hereto.

"Conforming Changes" means, with respect to either the use or administration of Term SOFR or the use, administration, adoption or implementation of any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Base Rate," the definition of "Business Day," the definition of "U.S. Government Securities Business Day," the definition of "Interest Period" or any similar or analogous definition (or the addition of a concept of "interest period"), timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, the applicability and length of lookback periods, the applicability of Section 3.8 and other technical, administrative or operational matters) that Agent decides may be appropriate to reflect the adoption and implementation of any such rate or to permit the use and administration thereof by Agent in a manner substantially consistent with market practice (or, if Agent decides that adoption of any portion of such market practice is not administratively feasible or if Agent decides is reasonably necessary in connection with the administration of this Agreement and the other Loan Documents).

"Connection Income Taxes" shall mean Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.

"Consents" shall mean all filings and all licenses, permits, consents, approvals, authorizations, qualifications and orders of Governmental Bodies and other third parties, domestic or foreign, necessary to carry on any Loan Party's business or necessary (including to avoid a conflict or breach under any agreement, instrument, other document, license, permit or other authorization) for the execution, delivery or performance of this Agreement, the other Loan Documents, including any Consents required under all applicable federal, state or other Applicable Law.

"Consolidated EBITDA" shall mean, for any Person and its Subsidiaries (other than any Unrestricted Subsidiaries), for any fiscal period, an amount equal to the sum of:

- a) Consolidated Net Income for such period, plus
- b) solely to the extent deducted in determining Consolidated Net Income for such period, and without duplication,
 - i. Consolidated Interest Expense,
 - ii. provision for Taxes imposed on Parent and its Subsidiaries based on income or profits or capital as determined on a consolidated basis in accordance with GAAP, including,

without limitation, federal, state, local, foreign, franchise, excise, value added, and similar taxes and foreign withholding taxes paid or accrued during such period including penalties and interest related to such taxes or arising from any tax examinations and any tax distributions related to the foregoing or otherwise permitted under this Agreement, and

- iii. depreciation and amortization (including amortization of deferred financing fees) determined on a consolidated basis in accordance with GAAP, <u>plus</u>
- c) to the extent deducted in determining Consolidated Net Income for such period (if applicable) (other than with respect to <u>clauses (iv)</u> and <u>(vi)</u> of this <u>clause (c)</u> in which case such addbacks shall be permitted at all times subject to the proviso below), and without duplication,
 - i. expenses, losses, charges or write-downs deemed unusual in nature or infrequent in occurrence in accordance with GAAP,
 - ii. non-cash charges, expenses or losses, including, without limitation, any non-cash compensation, non-cash translation (gain) loss and non-cash expense relating to the vesting of warrants (except to the extent such non-cash charge represents an accrual for a future cash charge),
 - iii. cash stock-based compensation associated with Borrower's class B common stock incurred during the fiscal quarter in which the SPAC Transaction closed, but only to the extent that such charges are non-recurring in nature or deemed non-recurring in nature following the SPAC Transaction,
 - iv. restructuring costs, integration costs, business optimization expenses or costs, business line consolidation, non-recurring retention, recruiting, relocation and signing bonuses and expenses, stock option and other equity-based compensation expenses, severance costs and transaction fees and expenses from the transactions contemplated herein,
 - v. reasonably documented expenses related to the transactions contemplated by the Loan Documents and the SPAC Transaction,
 - vi. "run rate" cost savings, operating savings, operating expense reductions and cost synergies reasonably anticipated by Borrower in good faith to be realizable within eighteen (18) months after the Closing Date (calculated on a pro forma basis as though such savings, reductions and synergies have been realized on the first day of such period, net of the aggregate amount of actual savings, reductions and synergy benefits realized) so long as such savings, reductions and synergies are reasonably identifiable, factually supported and set forth in reasonable detail in the applicable compliance certificate for such period; provided that, with respect to this clause (vi), to the extent that such savings, reductions or synergies are not reasonably anticipated by Borrower to be realized within eighteen (18) months after the Closing Date, such savings, reductions and synergies shall not be included in this definition of "Consolidated EBITDA" for any period thereafter,
 - vii. expenses or losses relating to business interruption or, so long as Borrower has made a determination that a reasonable basis exists for indemnification or reimbursement and only to the extent that such amount is in fact indemnified or reimbursed within 365 days of such determination (with a deduction in the applicable future period for any amount so added back to the extent not so indemnified or reimbursed within such 365 days),

charges, losses, lost profit expenses (including litigation expenses, fees and charges) or write-offs to the extent indemnified or incurred by a third party,

- viii. the net amount, if any, by which (i) deferred revenue as of the last day of such period (the "<u>Determination Date</u>") exceeds (ii) deferred revenue as of the date that is 12 months prior to the Determination Date,
- ix. the net amount, if any, by which (i) customer credit liabilities as of the Determination Date exceeds (ii) customer credit liabilities as of the date that is 12 months prior to the Determination Date,
- x. the net amount, if any, by which (i) deferred contract acquisition costs as of the date that is 12 months prior to the Determination Date exceed (ii) deferred contract acquisition costs as of the Determination Date,
- xi. fees payable under Section 3.2, minus

d)

- i) the sum of income and gain items corresponding to those referred to in <u>clauses (c)(i)</u> and <u>(c)(ii)</u> above,
- ii) any extraordinary, one-time or non-recurring gains,
- iii) any distributions made to minority or non-controlling interests attributable to Non-Wholly Owned Subsidiaries (but for the avoidance of doubt, not distributions attributable to Borrower's class B common stock shareholders),
- iv) the net amount, if any, by which clause (c)(viii)(ii) of this definition exceeds clause (c)(viii)(i) of this definition,
- v) the net amount, if any, by which clause (c)(ix)(ii) of this definition exceeds clause (c)(ix)(i) of this definition, and
- vi) the net amount, if any, by which clause (c)(x)(ii) of this definition exceeds clause (c)(x)(i) of this definition;

<u>provided</u> that, notwithstanding the foregoing, (1) the aggregate amount added back to Consolidated Net Income (excluding non-cash amounts) to the extent supported with reasonable documentation made pursuant to <u>clauses (c)(iv)</u> and (<u>c)(vi)</u> shall not in any event exceed 7.5% of Consolidated EBITDA before giving effect to such add-backs and all other add-backs contemplated hereby, and (2) no cap or limitation shall apply with respect to non-cash amounts added back to Consolidated Net Income to the extent supported with documentation reasonably satisfactory to the Agent.

Notwithstanding anything to the contrary contained herein, if during any applicable period any Loan Parties shall have consummated a Permitted Acquisition, or any sale, transfer or other disposition of any Person, business, property or assets, Consolidated EBITDA shall be calculated on a pro forma basis with respect to such Person, business, property or assets so acquired or so disposed of.

"Consolidated Funded Debt" shall mean, as of any date of determination, without duplication, all Indebtedness of any Person and its Subsidiaries (other than any Unrestricted Subsidiaries) of the type

described in <u>clauses (a), (b), (c), (e), (f)</u> (only with respect to unreimbursed amounts thereunder), and (<u>h)</u> (except to the extent relating to Indebtedness of the type described in <u>clause (d)</u> of the definition of Indebtedness) of the definition of Indebtedness and all guarantees by Borrower and its Subsidiaries of the foregoing types of Indebtedness, in each case, measured on a consolidated basis as of such date.

"Consolidated Interest Expense" means, for any fiscal period, Interest Expense of any Person and its Subsidiaries (other than any Unrestricted Subsidiaries) on a consolidated basis with respect to all outstanding Indebtedness of such Person and its Subsidiaries, including all commissions, discounts and other fees and charges owed with respect to letters of credit and net costs under Swap Obligations, but excluding, however, (i) any amount not payable in cash, (ii) any amounts non-cash amounts, that would be reflected in interest expense, payable on or before the Closing Date, including any underwriting fees or original issue discount paid in connection with the consummation of the closing hereof and any agency fees payable to the Agent in connection with the Loan Documents, (iii) any fees and/or expenses paid in connection with any Permitted Acquisition or other investments or in connection with any amendment or waiver with respect to any outstanding Indebtedness or any expenses and upfront fees (including any original issue discount) incurred in connection with any Indebtedness the proceeds of which are applied to fund any Permitted Acquisition or other investment, and (iv) any expenses payable in connection with any Swap Obligations.

"Consolidated Net Income" shall mean, with respect to any Person, for any period, the aggregate of the net income (or loss) of such Person and its Subsidiaries, on a consolidated basis, for such period, excluding to the extent included therein any extraordinary, one-time or non-recurring gains, after deducting all charges which should be deducted before arriving at the net income (or loss) for such period including the Provision for Taxes for such period, all as determined in accordance with GAAP, and before consideration of any net income or loss attributable to (x) minority or non-controlling interests attributable to Non-Wholly Owned Subsidiaries or (y) minority or non-controlling interest attributable to Borrower's class B common stock shareholders; provided that, (a) the net income of any Person that is not a Subsidiary or that is accounted for by the equity method of accounting shall be included only to the extent of the amount of dividends or distributions paid or payable to such Person or a majority-owned Subsidiary of such Person; (b) the effect of any change in accounting principles adopted by (or applicable to) such Person or its Subsidiaries after the date hereof (including any cumulative effects resulting from changes in purchase accounting principles) shall be excluded; (c) the net income (if positive) of any majority-owned Subsidiary to the extent that the declaration or payment of dividends or similar distributions by such majority-owned Subsidiary to such Person or to any other majority-owned Subsidiary of such Person is not at the time permitted by operation of the terms of its charter or any agreement, instrument, judgment, decree, order, statute, rule or governmental regulation applicable to such majority-owned Subsidiary shall be excluded; and (d) the net income (or loss) of any Person accrued prior to the earlier of (i) the date such Person becomes a Subsidiary of Borrower or any of its consolidated Subsidiaries or (ii) the date such Person is merged into or consolidated with Borrower or any of its consolidated Subsidiaries or (iii) the date such Person's assets are acquired by Borrower or any of its consolidated Subsidiaries, in each case pursuant to a Permitted Acquisition, shall be included as if such Permitted Acquisition occurred on the first day of such period. For the purpose of this definition, net income excludes any gain together with any related Provision for Taxes for such gain realized upon the sale or other disposition of any assets other than in the ordinary course of business or of any Equity Interests of such Person or a Subsidiary of such Person.

"Consolidated Total Assets" shall mean, as of any date of determination, the amount that would, in accordance with GAAP, be set forth opposite the caption "total assets" (or any like caption) on a consolidated balance sheet of any Person and its Subsidiaries (other than any Unrestricted Subsidiaries) at such date.

"Consolidating Annual Financials" has the meaning specified therefor in Section 9.7.

- "Consolidating Quarterly Financials" has the meaning specified therefor in Section 9.8.
- "Contract Rate" has the meaning specified therefor in Section 3.1 of this Agreement.
- "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "Controlling" and "Controlled" have meanings correlative thereto.
- "Controlled Group" shall mean, at any time, the Loan Parties and all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control and all other entities which, together with any Loan Party, are treated as a single employer under Section 414 of the Code.

"Covered Entity" means any of the following:

- (a) a "covered entity" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 252.82(b);
- (b) a "covered bank" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 47.3(b); or
- (c) a "covered FSI" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 382.2(b).
- "Covered Party" has the meaning specified therefor in Section 15.20 of this Agreement.
- "Cross-Default" means (i) the failure of Borrower or any of its Restricted Subsidiaries (other than any Immaterial Subsidiary) to pay when due any principal of or interest on or any other amount payable in respect of any Material Indebtedness beyond the grace period, if any, provided therefor; or (ii) breach or default by Borrower or any of its Restricted Subsidiaries (other than any Immaterial Subsidiary) with respect to any other term of (A) any Material Indebtedness or (B) any loan agreement, mortgage, indenture or other agreement relating to such Material Indebtedness, in each case beyond the grace period, if any, provided therefor, if the effect of such breach or default is to cause, or to permit the holder or holders of such Material Indebtedness (or a trustee on behalf of such holder or holders) to cause, such Material Indebtedness to become or be declared due and payable (or subject to a compulsory repurchase or redeemable) prior to its stated maturity or the stated maturity of any underlying obligation, as the case may be.
 - "Cure Amount" has the meaning specified therefor in Section 10.15.
 - "Cure Equity" has the meaning specified therefor in Section 10.15.
 - "Cure Right" has the meaning specified therefor in Section 10.15.
- "<u>Daily Simple SOFR</u>" means, for any day, SOFR, with the conventions for this rate (which will include a lookback) being established by Agent in accordance with the conventions for this rate recommended by the Relevant Governmental Body for determining "Daily Simple SOFR" for syndicated business loans; <u>provided that</u> if Agent decides that any such convention is not administratively feasible for Agent, then Agent may establish another convention in its reasonable discretion.
- "<u>Default</u>" shall mean an event, circumstance or condition which, with the giving of notice or passage of time or both, would constitute an Event of Default.

"Default Rate" shall have the meaning set forth in Section 3.1.

"Default Right" has the meaning assigned to that term in, and shall be interpreted in accordance with, 12 C.F.R. §§ 252.81, 47.2 or 382.1, as applicable.

"Defaulting Lender" shall means, subject to Section 2.16(e), any Lender that (a) has failed to (i) fund all or any portion of any Advance within two Business Days of the date such Advances were required to be funded hereunder unless such Lender notifies Agent and Borrower in writing that such failure is the result of such Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied, or (ii) pay to Agent, the Issuer, any Swingline Lender or any other Lender any other amount required to be paid by it hereunder (including in respect of its participation in Letters of Credit or Swingline Loans) within two Business Days of the date when due, (b) has notified Borrower, Agent or the Issuer or Swingline Lender in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect, (c) has failed, within three Business Days after written request by Agent or Borrower, to confirm in writing to Agent and Borrower that it will comply with its prospective funding obligations hereunder (provided that such Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon receipt of such written confirmation by Agent and Borrower), or (d) has, or has a direct or indirect parent company that has, (i) become the subject of a proceeding under the Bankruptcy Code of the United States of America, or any other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief laws of the United States or other applicable jurisdictions from time to time in effect, (ii) become the subject of a Bail-In Action or (iii) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity; <u>provided</u> that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Body so long as such ownership interest does not result in or provide such Lender with immunity from the jurisdiction of courts within the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Body) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lender. Any determination by Agent that a Lender is a Defaulting Lender under any one or more of clauses (a) through (d) above shall be conclusive and binding absent manifest error, and such Lender shall be deemed to be a Defaulting Lender (subject to Section 2.16(e)) upon delivery of written notice of such determination to Borrower, the Issuer, each Swingline Lender and each Lender.

"Determination Date" shall have the meaning set forth in the definition of "Consolidated EBITDA".

"Disqualified Institutions" shall mean (a) any Person designated by Borrower in writing to the Agent prior to the Closing Date (and any Affiliate thereof clearly identifiable as an Affiliate solely on the basis of the similarity of its name), (b) any Person that is or becomes a competitor designated by Borrower in writing at any time (and any affiliate clearly identifiable as an affiliate solely on the basis of the similarity of its name), (c) additional competitors designated by Borrower in writing at any time (and any Affiliate thereof clearly identifiable as an affiliate thereof solely on the basis of the similarity of its name), and (d) any known Affiliate of a Person described in clauses (a) and (b) above identified in writing by Borrower to the Agent at any time; provided no designation shall have retroactive effect to any prior assignment to any Lender permitted hereunder or under the Loan Documents (but further assignments and participations shall be prohibited); and provided further that any addition to the Disqualified Institutions made in accordance with this paragraph shall not be effective until the 3rd Business Day following the Agent's receipt of written notice of such addition. For the avoidance of doubt, the Agent shall not be

responsible or have any liability for, or have any duty to ascertain, inquire into, monitor or enforce compliance with the provisions under the Loan Documents relating to Disqualified Institutions.

"Dollars" and the sign "\$" shall mean lawful money of the United States of America.

"Domestic Rate Loan" shall mean any Advance that bears interest based upon the Base Rate.

"<u>Domestic Subsidiary</u>" of any Person, shall mean any Subsidiary of such Person that is organized or incorporated in the United States or any State or territory thereof.

<u>"Early Opt-in Effective Date"</u> means, with respect to any Early Opt-in Election, the sixth (6th) Business Day after the date notice of such Early Opt-in Election is provided to the Lenders, so long as Agent has not received, by 5:00 p.m. (New York City time) on the fifth (5th) Business Day after the date notice of such Early Opt-in Election is provided to the Lenders, written notice of objection to such Early Opt-in Election from Lenders comprising the Required Lenders.

"Early Opt-in Election" means the occurrence of:

(i) a notification by Agent to (or the request by Borrower to Agent to notify) each of the other parties hereto that at least five currently outstanding Dollar-denominated syndicated credit facilities at such time contain (as a result of amendment or as originally executed) a SOFR-based rate (including SOFR, a Term SOFR or any other rate based upon SOFR) as a benchmark rate (and such syndicated credit facilities are identified in such notice and are publicly available for review), and

(ii) the joint election by the Agent and Borrower to trigger a fallback from USD LIBOR and the provision by the Agent of written notice of such election to the Lenders.

"EEA Financial Institution" means (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" shall mean any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"EEA Resolution Authority" means any public administrative authority or any person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution.

"Election Period" shall have the meaning set forth in Section 2.4(c).

"Eligible Assignee" shall have the meaning set forth in Section 15.3 (c).

"<u>Eligible Party</u>" means an "eligible contract participant" for purposes of Section 1a(18) of the Commodity Exchange Act, regulations promulgated thereunder and binding guidance thereunder promulgated by the Commodity Futures Trading Commission.

"Environmental Laws" shall mean all federal, state and local environmental, land use, zoning, health, chemical use, safety and sanitation laws, statutes, ordinances and codes relating to the protection

of the environment and/or governing the use, storage, treatment, generation, transportation, processing, handling, production or disposal of Hazardous Substances and the rules, regulations, policies, guidelines, interpretations, decisions, orders and directives of federal, state and local governmental agencies and authorities with respect thereto.

"Equipment" shall mean and include all of each Loan Party's goods (other than Inventory) whether now owned or hereafter acquired and wherever located including, without limitation, all equipment, machinery, apparatus, motor vehicles, fittings, furniture, furnishings, fixtures, parts, accessories and all replacements and substitutions therefor or accessions thereto.

"Equity Interests" of any Person shall mean any and all shares, rights to purchase, options, warrants, general, limited or limited liability partnership interests, member interests, participation or other equivalents of or interest in (regardless of how designated) equity of such Person, whether voting or nonvoting, including common stock, preferred stock, convertible securities or any other "equity security" (as such term is defined in Rule 3a11-1 of the General Rules and Regulations promulgated by the SEC under the Exchange Act).

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time and the rules and regulations promulgated thereunder.

"Erroneous Payment" has the meaning assigned to it in Section 14.15(a).

"Erroneous Payment Deficiency Assignment" has the meaning assigned to it in Section 14.15(d)(i).

"Erroneous Payment Impacted Class" has the meaning assigned to it in Section 14.15(d)(i).

"Erroneous Payment Return Deficiency" has the meaning assigned to it in Section 14.15(d)(i).

"Erroneous Payment Subrogation Rights" has the meaning assigned to it in Section 14.15(e).

"EU Bail-In Legislation Schedule" means the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor person), as in effect from time to time.

"Euro-Rate Reserve Percentage" shall have the meaning set forth in the definition of Adjusted LIBO Rate.

"Eurodollar Rate" shall mean for any Eurodollar Rate Loan for the then current Interest Period relating thereto the rate per annum (such Eurodollar Rate to be adjusted to the next higher 1/100 of one percent (.01%)) equal to the Adjusted LIBO Rate.

"<u>Eurodollar Rate Loan</u>" shall mean any Revolving Advance or Incremental Term Loan that at any time that bears interest based on the Adjusted LIBO Rate.

"Event of Default" shall mean the occurrence of any of the events set forth in Article X.

"Exchange Act" shall mean the Securities Exchange Act of 1934, together with all rules, regulations and interpretations thereunder or related thereto.

"Excluded Accounts" shall mean collectively, payroll accounts, escrow accounts, foreign accounts, fiduciary accounts, trust accounts, zero balance accounts and other accounts held exclusively

for the benefit of an unaffiliated third party, including Tax escrow accounts, and employee benefits accounts maintained in the ordinary course of business.

"Excluded Property" has the meaning assigned to it in the definition of "Collateral".

"Excluded Subsidiary." shall mean any Subsidiary of Borrower that is (a) a Foreign Subsidiary, (b) an Unrestricted Subsidiary, (c) a FSHCO, (d) any Subsidiary acquired by Borrower that, at the time of the relevant acquisition, is an obligor in respect of assumed Indebtedness that is permitted under Section 7.7 and was not incurred or modified in contemplation of such acquisition to the extent (and solely for so long as) the documentation governing such assumed Indebtedness prohibits such Subsidiary from providing a guarantee hereunder, (e) prohibited by applicable law, rule, regulation from guaranteeing the facilities under this Agreement, or which would require governmental (including regulatory) consent, approval, license or authorization to provide a guarantee in each case, unless, such consent, approval, license or authorization has been received (but without obligation to seek the same), (f) prohibited from guaranteeing the Obligations by any contractual obligation in existence (x) on the Closing Date or (y) at the time of the acquisition of such Subsidiary after the Closing Date (to the extent such prohibition was not entered into in contemplation of such acquisition), (g) a not-for-profit Subsidiary, (h) any Special Purpose Securitization Subsidiary, (i) any Domestic Subsidiary that is a direct or indirect Subsidiary of a FSHCO or CFC, (j) any Subsidiary regulated as an insurance company or any other captive insurance Subsidiary, (k) any Immaterial Subsidiary (including, without limitation, any Immaterial Subsidiary that is a Non-Wholly Owned Subsidiary) and (l) any other Subsidiary with respect to which, in the reasonable judgment of the Agent and Borrower, the cost or other consequences (including any adverse Tax consequences) of guaranteeing the Obligations shall be excessive in view of the benefits to be obtained by the Lenders therefrom; provided that, notwithstanding the above, if a Subsidiary executes the Guaranty as a "Guarantor" then it shall not constitute an "Excluded Subsidiary" (unless relea

"Excluded Swap Obligations" means, with respect to any Guarantor, any Swap Obligation if, and to the extent that, all or a portion of the guarantee of such Person of, or the grant by such Person of a security interest to secure, such Swap Obligation (or any guarantee thereof) is or becomes illegal or unenforceable under the Commodity Exchange Act or any rule, regulation or order of the Commodity Futures Trading Commission (or the application or official interpretation of any thereof) by virtue of such Person's failure for any reason not to constitute an Eligible Party.

"Excluded Taxes" shall mean any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient, (a) Taxes imposed on or measured by net income or net profits (however denominated), franchise Taxes, and branch profits Taxes, in each case, (i) imposed as a result of such Recipient being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes, (b) in the case of a Lender, U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in an Advance or Commitment pursuant to a law in effect on the date on which (i) such Lender acquires such interest in an Advance or Commitment (other than pursuant to an assignment request by Borrower hereunder) or (ii) such Lender changes its lending office, except in each case to the extent that, pursuant to Section 3.10, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a party hereto or to such Lender immediately before it changed its lending office, (c) Taxes attributable to such Recipient's failure to comply with Section 3.10(g) and (d) any withholding Taxes imposed under FATCA.

"Extraordinary Receipts" means any cash received by or paid to or for the account of any Person not in the ordinary course of business, including tax refunds, pension plan reversions, proceeds of

insurance (other than proceeds of business interruption insurance to the extent such proceeds constitute compensation for lost earnings), condemnation awards (and payments in lieu thereof), indemnity payments, any purchase price adjustments, and sale-leaseback proceeds.

"FATCA" means Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with) any current or future regulations or official interpretations thereof and any agreement entered into pursuant to Section 1471(b)(1) of the Code (or any amended or successor version described above), and any U.S. or non-U.S. fiscal or regulatory legislation, rules, guidance, notes or practices adopted pursuant to any intergovernmental agreement, treaty or convention among Governmental Body entered into in connection with the implementation of such sections of the Code or analogous provisions of non-U.S. law.

"FCPA" shall have the meaning set forth in Section 5.23.

"Federal Funds Rate" shall mean, for any period, a fluctuating interest rate *per annum* equal to, for each day during such period, the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System, as published for such day (or if such day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or if such rate is not so published for any day which is a Business Day, the average of quotations for such day on such transactions received by Agent from three Federal funds brokers of recognized standing selected by Agent.

"Federal Reserve Bank of New York's Website" means the website of the Federal Reserve Bank of New York at http://www.newyorkfed.org, or any successor source.

"Fee Letter" shall mean the Fee Letter dated as of as of the date hereof between Borrower and HSBC.

"Floor" means the benchmark rate floor, if any, provided in this Agreement initially (as of the execution of this Agreement, the modification, amendment or renewal of this Agreement or otherwise) with respect to USD LIBOR: a rate of interest equal to 0.00%.

"Foreign Lender" means a Lender that is not a U.S. Person within the meaning of Section 7701(a)(30) of the Code.

"Foreign Subsidiary." of any Loan Party, shall mean any Subsidiary of such Loan Party that is not a Domestic Subsidiary.

"Fronting Exposure" means, at any time there is a Defaulting Lender, (a) with respect to the Issuer, such Defaulting Lender's Commitment Percentage of the outstanding Letter of Credit exposure with respect to Letters of Credit issued by such Issuer other than Letter of Credit exposure as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or cash collateralized in accordance with the terms hereof and (b) with respect to any Swingline Lender, such Defaulting Lender's Commitment Percentage of outstanding Swingline Loans made by such Swingline Lender other than Swingline Loans as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders.

"FSHCO" shall mean any Domestic Subsidiary that has no material assets other than Equity Interests, or Equity Interests and Indebtedness in one or more Foreign Subsidiaries that are CFCs.

"Fund" means any Person (other than a natural Person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans, bonds and similar extensions of credit in the ordinary course of its business.

"GAAP" shall mean generally accepted accounting principles in the United States of America in effect from time to time, consistently applied.

"General Intangibles" shall mean and include, as to each Loan Party, all of such Loan Party's general intangibles, whether now owned or hereafter acquired including, without limitation, all payment intangibles, choses in action, commercial tort claims, causes of action, corporate or other business records, inventions, designs, patents, patent applications, equipment formulations, manufacturing procedures, quality control procedures, trademarks, trademark applications, service marks, trade secrets, goodwill, copyrights, design rights, software, computer information, source codes, codes, records and updates, registrations, licenses, franchises, customer lists, tax refunds, tax refund claims, computer programs and computer software, all claims under guaranties, security interests or other security held by or granted to such Loan Party to secure payment of any of the Receivables by a customer, all rights of indemnification and all other intangible property of every kind and nature (other than Receivables).

"Governmental Body" means the government of any nation or any political subdivision thereof, whether at the national, state, territorial, provincial, municipal or any other level, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of, or pertaining to, government (including any supranational bodies such as the European Union or the European Central Bank).

"Guarantor" shall mean (i) on the Closing Date, the Restricted Subsidiaries of Borrower listed on Schedule 1.2(b) and (ii) after the Closing Date, (x) each other Restricted Subsidiary of Borrower that is not an Excluded Subsidiary, and (y) any other Person who may hereafter guarantee payment or performance of the whole or any part of the Obligations, and "Guarantors" means collectively all such Persons. For the avoidance of doubt, no Excluded Subsidiary shall be a Guarantor.

"Guaranty" shall mean any guaranty of the Obligations executed by a Guarantor in favor of Agent for its benefit and for the ratable benefit of Lenders, as the same may be amended, restated, modified and/or supplemented from time to time.

"<u>Hazardous Substance</u>" shall mean, without limitation, any flammable explosives, radon, radioactive materials, asbestos, urea formaldehyde foam insulation, polychlorinated biphenyls, petroleum and petroleum products, methane, hazardous materials, hazardous wastes, hazardous or toxic substances or related materials as defined in CERCLA, the Hazardous Materials Transportation Act, as amended (49 U.S.C. Sections 1801, et seq.), the Resource Conservation and Recovery Act, 42 U.S.C. §§ 6901 et seq., or any other applicable Environmental Law and in the regulations adopted pursuant thereto.

"Hedging Agreement" shall mean any interest rate exchange, collar, cap, swap, adjustable strike cap, adjustable strike corridor or similar agreement entered into by Borrower or any of its Subsidiaries in order to provide protection to, or minimize the impact upon, Borrower or any Subsidiary thereof against fluctuations in interest rates, currency exchange rates or commodity prices in the ordinary course of Borrower's or such Subsidiary's business and not for speculative purposes.

"HSBC" shall have the meaning set forth in the preamble to this Agreement and shall include its successors and assigns.

"Immaterial Subsidiary" shall mean any Restricted Subsidiary of Borrower as to which, as of any relevant date of determination, (a) the consolidated total assets of such Restricted Subsidiary and its Subsidiaries (on a consolidated basis and giving effect to intercompany eliminations) and all other Immaterial Subsidiaries as of such date (and their respective Subsidiaries (on a consolidated basis and giving effect to intercompany eliminations)), do not exceed an amount equal to 5.0% of the Consolidated

Total Assets (giving effect to intercompany eliminations) of Borrower and its Subsidiaries as of the last day of the most recently ended fiscal quarter, (b) the subscription billings of such Restricted Subsidiary and its Subsidiaries (on a consolidated basis and giving effect to intercompany eliminations) for such fiscal quarter and all other Immaterial Subsidiaries as of such date (on a consolidated basis and giving effect to intercompany eliminations) for such fiscal quarter do not exceed an amount equal to 5.0% of the consolidated subscription billings (giving effect to intercompany eliminations) of Borrower and its Subsidiaries for such quarter, unless in either case, Borrower shall have designated that such Restricted Subsidiary shall not be an Immaterial Subsidiary, and (c) does not own any material Intellectual Property at such time; provided, if at any time after the Closing Date, Immaterial Subsidiaries comprise in the aggregate more than 10.0% of the consolidated subscription billings or 10.0% of the Consolidated Total Assets of Parent and its Subsidiaries (or, solely during a Borrower Consolidating Financials Delivery Period, Borrower and its Restricted Subsidiaries), then Borrower shall designate in writing to the Agent that one or more of such Subsidiaries is no longer an Immaterial Subsidiary such that the foregoing condition ceases to be true. As of the Closing Date, Roundtable Research, LLC, KOC Publishing, LLC, Signal to Noise, LLC and Chaikin Investments, LLC are the only Immaterial Subsidiaries.

"Increase Effective Date" shall have the meaning set forth in Section 2.4(d).

"Incremental Revolving Commitment" shall have the meaning set forth in Section 2.4(b).

"Incremental Term Loan" shall have the meaning set forth in Section 2.4(a).

"Indebtedness" of a Person at a particular date shall mean shall mean, without duplication: (a) all obligations of such Person for borrowed money, (b) all obligations of such Person evidenced by bonds, debentures, notes or other similar instruments, (c) all obligations of such Person in respect of the deferred purchase price of property or services or in respect of warrants (other than trade payables incurred in the ordinary course of business, but, subject to clause (a) of the last sentence of this definition, including any obligations in respect of earn-outs, other contingent acquisition consideration, and warrants), (d) all obligations of such Person under any conditional sale or other title retention agreement(s) relating to property acquired by such Person, (e) that portion of obligations with respect to Capital Leases that is properly classified as a liability on a balance sheet in conformity with GAAP ("Capital Lease Obligations") of such Person, (f) all obligations, contingent or otherwise, of such Person in respect of Letters of Credit, acceptances or similar extensions of credit, (g) all guarantees of such Person of the type of Indebtedness described in clauses (a) through (f) above, (h) all Indebtedness of a third party described in clauses (a) through (f) above secured by any Lien on property owned by such Person, whether or not such Indebtedness has been assumed by such Person; provided that the amount of any such Indebtedness under this clause (h) shall be deemed to be the lesser of (A) the total amount of third party Indebtedness secured by such Lien and (B) the fair market value of the property subject to such Lien, (i) all obligations of such Person in respect of disqualified capital stock, and (j) the hedge termination value owed by of such Person under any Hedging Agreements. The Indebtedness of any Person shall include the Indebtedness provide that such Person is not liable therefor.

Notwithstanding the foregoing, Indebtedness shall not include (a) obligations of such Person in respect of earn-outs, other contingent acquisition consideration, and warrants until such obligations become liabilities on the balance sheet of such Person in accordance with GAAP (except to the extent such obligations that are liabilities on the balance sheet of such Person are payable solely in capital stock) and (b) Indebtedness arising as a result of any changes in GAAP which would classify any operating leases so characterized in accordance with GAAP (as GAAP is in effect as of December 15, 2018) as Capital Lease Obligations (or the equivalent) required to be reflected on a consolidated balance sheet of Parent in accordance with GAAP.

"Indemnified Taxes" shall means (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of any Loan Party under this Agreement or any other Loan Document and (b) to the extent not otherwise described in clause (a), Other Taxes.

"Indemnitee" shall have the meaning set forth in Section 15.5(a).

"Intellectual Property" shall mean property constituting under any Applicable Law a patent, patent application, copyright, trademark, service mark, trade name, mask work, or trade secret.

"Intellectual Property Agreements" shall mean any intellectual property security agreement entered into between a Loan Party and Agent pursuant to the terms hereof in form and substance satisfactory to Agent, together with each other intellectual property security agreement and supplement thereto.

"Interest Coverage Ratio" means, with respect to the last day of any fiscal period, the ratio of Consolidated EBITDA for the four consecutive fiscal quarters ending on or immediately prior to such fiscal period to Consolidated Interest Expense for the four consecutive fiscal quarters ending on or immediately prior to such fiscal period.

"Interest Expense" shall mean, for any period, as to any Person, as determined in accordance with GAAP, the total interest expense of such Person (including the interest component of Capital Leases for such period and capitalized interest for such period), whether paid or accrued during such period but without duplication, excluding interest paid in property other than cash.

"Interest Payment Date" means (a) as to any Domestic Rate Loan, the last Business Day of each March, June, September and December and the Termination Date and (b) as to any SOFR Loan, the last day of each Interest Period therefor and, in the case of any Interest Period of more than three months' duration, each day prior to the last day of such Interest Period that occurs at three month intervals after the first day of such Interest Period, and the Termination Date.

"Interest Period" shall mean, with respectmeans, as to any Eurodollar Rate LoanAdvance, the period commencing on the date of such loan is made as, renewed as or converted into a Eurodollar Rate LoanAdvance and ending on the last day of such period as selected by Borrower pursuant to the provisions below and, thereafter, each subsequent period commencing on the last day of the immediately preceding Interest Period and ending on the last day of such period, as selected by Borrower pursuant to the provisions below (in each case, subject to the availability thereof), as specified in the applicable Notice of Borrowing as set forth in Exhibit E or Notice of Conversion as set forth in Exhibit F; provided that (i) if any Interest Period would end on a day other than a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless such next succeeding Business Day would fall in the next calendar month, in which case such Interest Period shall end on the next preceding Business Day, (ii) any Interest Period that commences on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the last calendar month of such Interest Period, (iii) no Interest Period shall extend beyond the Termination Date and (iv) no tenor that has been removed from this definition pursuant to Section 3.8 shall be available for specification in such Notice of Borrowing or Notice of Conversion. For purposes hereof, the date of an Advance initially shall be the date on which such Advance is made and thereafter shall be the effective date of the most recent conversion or continuation of such Advance. The duration of each Interest Period for any Eurodollar Rate SOFR Loan shall be for a number of months selected by Borrower upon notice as set forth in Section 2.2(b); provided that:

- (i) whenever the last day of any Interest Period would otherwise occur on a day other than a Business Day, the last day of such Interest Period shall occur on the next succeeding Business Day; provided that if such extension of time would cause the last day of such Interest Period to occur in the next following calendar month, the last day of such Interest Period shall occur on the last Business Day immediately preceding the last day of such Interest Period; and
- (ii) if Borrower renews any Eurodollar Rate Loan for an additional Interest Period, the first day of the new Interest Period shall be the last day of the preceding Interest Period; however, interest shall only be charged once for such day at the rate applicable to the Eurodollar Rate Loan for the new Interest Period.
- "Interest Rate Hedge" shall mean an interest rate exchange, collar, cap, swap, adjustable strike cap, adjustable strike corridor, option or similar agreements entered into by any Loan Party in order to provide protection to, or minimize the impact upon, such Loan Party of increasing floating rates of interest applicable to Indebtedness.
- "Inventory" shall mean and include all of each Loan Party's now owned or hereafter acquired goods, merchandise and other personal property, wherever located, to be furnished under any consignment arrangement, contract of service or held for sale or lease, all raw materials, work in process, finished goods and materials and supplies of any kind, nature or description which are or might be used or consumed in such Loan Party's business or used in selling or furnishing such goods, merchandise and other personal property, all other inventory of each Loan Party, and all documents of title or other documents representing them.
- "Investment" is any beneficial ownership interest in any Person (including any Equity Interests), and any loan, advance or capital contribution to any Person.
- "Investment Property" as defined in the UCC and shall include all of each Loan Party's now owned or hereafter acquired securities (whether certificated or uncertificated), securities entitlements, securities accounts, commodities contracts, commodities accounts, stocks, mutual fund shares, money market shares and U.S. Government securities.
- "Issuer" shall mean any Person who issues a Letter of Credit and/or accepts a draft pursuant to the terms thereof (it being agreed that so long as HSBC shall be Agent or a Lender, then the Issuer shall be HSBC); provided, however, that in the event that HSBC is neither Agent nor a Lender, the "Issuer" with respect to all subsequently issued Letters of Credit shall be a Lender selected by Borrower and that agrees to be an Issuer hereunder.
 - "Joint Bookrunners" shall have the meaning ascribed to such term in the preamble.
 - "Joint Lead Arrangers" shall have the meaning ascribed to such term in the preamble.
- "Junior Lien Indebtedness" means any Indebtedness of the Loan Parties that is secured by a Lien on the Collateral that is contractually junior to the Liens securing the Obligations to the written satisfaction of Agent and the Required Lenders.
- "Lender" and "Lenders" shall have the meaning ascribed to such term in the preamble to this Agreement and shall include each Person which becomes a transferee, successor or assign of any Lender.
- "<u>Lender-Provided Hedge</u>" shall mean an Interest Rate Hedge or foreign exchange transaction which is provided by any Lender or any direct or indirect Subsidiary or Affiliate of any Lender (or any

Person that was a Lender or a Subsidiary or Affiliate of a Lender at the time of entering into such Interest Rate Hedge), which is entered into for hedging (rather than speculative) purposes.

"Letter of Credit and Guarantee Fees" shall have the meaning set forth in Section 3.2.

"Letter of Credit Application" shall have the meaning set forth in Section 2.10.

"<u>Letter of Credit Fee Percentage</u>" shall mean, as of the most recent Adjustment Date, the applicable percent per annum set forth in the pricing table set forth below corresponding to the Net Leverage Ratio for the Calculation Period:

NET LEVERAGE RATIO	LETTER OF CREDIT FEE PERCENTAGE
Less than 0.50:1.00	1.50%
Greater than or equal to 0.50:1.00 but less than 1.00:1.00	1.75%
Greater than or equal to 1.00:1.00 but less than 1.50:1.00	2.00%
Greater than or equal to 1.50:1.00	2.25%

provided that for the period from the Closing Date through and including the first Adjustment Date following the Closing Date, the Letter of Credit Fee Percentage shall be 1.50%.

"<u>Letter of Credit Reserve</u>" shall mean the sum of (a) the Maximum Undrawn Amount of all outstanding Letters of Credit; <u>plus</u>, (b) all amounts which have been paid or made available by Agent, any Lender or Issuer to the extent not deemed to be a Revolving Advance pursuant to <u>Section 2.9</u> or otherwise reimbursed by Borrower; <u>plus</u> (c) all unpaid interest, fees and expenses related thereto.

"Letters of Credit" shall have the meaning set forth in Section 2.9.

"Leverage Ratio Increase" shall have the meaning set forth in Section 6.8.

"LIBOR" shall have the meaning set forth in clause (a) of the definition of "Adjusted LIBO Rate".

"<u>Lien</u>" shall mean any mortgage, deed of trust, pledge, hypothecation, assignment, security interest, lien (whether statutory or otherwise), charge, claim or encumbrance, or preference, priority or other security agreement or preferential arrangement held or asserted in respect of any asset of any kind or nature whatsoever including, without limitation, any conditional sale or other title retention agreement, any lease having substantially the same economic effect as any of the foregoing, and the filing of, or agreement to give, any financing statement under the UCC or comparable law of any jurisdiction.

- "<u>LLC Agreement</u>" means the Third Amended and Restated Limited Liability Company Agreement, dated as of July 21, 2021, as in effect on the Closing Date and as the same may be amended in accordance with the terms thereof and hereof.
- "Loan Documents" shall mean, collectively, this Agreement, the Notes, the Guaranty, the Perfection Certificate, the Fee Letter, the Intellectual Property Agreements, the Stock Pledge Agreement, and any and all other agreements, instruments and documents, including, without limitation, guaranties, pledges, powers of attorney, consents, and all other writings heretofore, now or hereafter executed by any Loan Party or any Person (as authorized by a Loan Party) or by any Person who becomes a Guarantor of the Obligations and/or delivered to Agent or any Lender in respect of the transactions contemplated by this Agreement.
- "Loan Party" shall mean, individually, Borrower and each Guarantor, and "Loan Parties" shall mean, collectively, Borrower and the Guarantors.
- "<u>Material Acquisition</u>" means the purchase or other acquisition of all or substantially all of the assets of any Person or a business or division of such Person (whether pursuant to a merger or other transaction) or of all or a majority of the Equity Interest of a Person in one or a series of transactions (the "<u>Acquired Entity</u>"), the fair market value of which exceeds 15% of Consolidated Total Assets of Borrower.
- "Material Adverse Effect" shall mean a material adverse effect on (a) the business, assets, operations, results of operations, or condition (financial or otherwise) of Borrower or the Restricted Subsidiaries taken as a whole, (b) any ability of the Loan Parties to pay the Obligations in accordance with the terms hereof, (c) the value of the Collateral, or Agent's Liens on the Collateral or the priority of any such Lien or (d) the practical realization of the benefits of Agent's and each Lender's rights and remedies under this Agreement and the other Loan Documents.
- "<u>Material Contract</u>" shall mean (a) the Tax Receivable Agreement, (b) any contract or other agreement, written or oral, of any Loan Party involving monetary liability of or to any Person in an amount in excess of \$10,000,000 in any fiscal year, other than purchase orders issued by any Loan Party in the ordinary course of its business, and (c) any other contract or other agreement, whether written or oral, to which any Loan Party is a party as to which the breach, nonperformance, cancellation or failure to renew by any party thereto would have a Material Adverse Effect.
- "<u>Material Indebtedness</u>" means Indebtedness of Borrower or any Restricted Subsidiary in an outstanding principal amount of \$10,000,000 or more in the aggregate (or the equivalent thereof in any currency other than Dollars). For purposes of determining the amount of Material Indebtedness at any time, (a) undrawn committed amounts and (b) all amounts owing to all creditors under any syndicated credit arrangement shall be included.
- "<u>Maximum Revolving Advance Amount</u>" shall mean \$150,000,000, as such amount may be changed from time to time pursuant to the terms hereof (including in connection with assignments permitted hereunder and including pursuant to <u>Section 2.4</u>).
 - "Maximum Swingline Loan Amount" shall mean \$5,000,000.
- "Maximum Undrawn Amount" shall mean with respect to any outstanding Letter of Credit, the amount of such Letter of Credit that is or may become available to be drawn, including all automatic increases provided for in such Letter of Credit, whether or not any such automatic increase has become effective.

"Multiemployer Plan" shall mean a Plan that is a "multiemployer plan" as defined in Sections 3(37) and 4001(a)(3) of ERISA.

"<u>Multiple Employer Plan</u>" shall mean a Plan that has two or more contributing sponsors (including any Loan Party or any member of the Controlled Group) at least two of whom are not under common control, as such a plan is described in Section 4064 of ERISA.

"Net Cash Proceeds" means, with respect to the sale or disposition of any asset by a Person, the sum of cash and Cash Equivalent received by such Person in connection with such sale or disposition \underline{less} (i) the reasonable fees, commissions, and expenses related thereto and required to be paid by such Person in connection with such sale or disposition, \underline{less} (ii) all taxes paid or reasonably estimated to be payable in connection with such sale or disposition to the extent, but only to the extent, that the amounts so deducted are, at the time of receipt of such cash, actually paid or payable to a Person that is not an Affiliate of any Loan Party or any of its Subsidiaries, and are properly attributable to such transaction, \underline{less} (iii) any reserve for adjustment in respect of (\underline{x}) the sale price of the property that is the subject of such sale or disposition established in accordance with GAAP and (\underline{y}) any liabilities associated with such property or asset and retained by such Person after such sale or disposition, to the extent that in each case the funds described above in this <u>clause (iii)</u> are (\underline{x}) deposited into escrow with a third party escrow agent and (\underline{y}) paid to Agent as a prepayment of the applicable Obligations in accordance with <u>Section 2.14</u> of this Agreement at such time when such amounts are no longer required to be set aside as such a reserve, <u>less</u> (iv) any amounts used to repay or return any customer deposits required to be repaid or returned as a result of sale or disposition.

"<u>Net Leverage Ratio</u>" means, with respect to the last day of any fiscal period, the ratio of (a) Consolidated Funded Debt as of such date, <u>minus</u> up to \$25,000,000 of unrestricted domestic cash and Cash Equivalents of the Loan Parties as of such date, to (b) Consolidated EBITDA for the four consecutive fiscal quarters ending on or immediately prior to such fiscal period.

"Non-Consenting Lender" means any Lender that does not approve any consent, waiver or amendment that (a) requires the approval of all or all affected Lenders in accordance with the terms of Section 15.2 and (b) has been approved by the Required Lenders.

"Non-Defaulting Lenders" shall have the meaning set forth in Section 2.16(b).

"Non-Guarantor Acquisition" shall have the meaning set forth in the definition of "Permitted Acquisition".

"Non-Wholly Owned Subsidiaries" shall mean any Subsidiaries of Borrower that are not wholly owned by Borrower or one or more Wholly Owned Restricted Subsidiaries of Borrower.

"Note" or "Notes" shall mean, individually or collectively, the Revolving Credit Note and the Swingline Note.

"Notice of Borrowing" shall mean a notice of borrowing provided by Borrower to Agent in substantially the form attached hereto as Exhibit E.

"Notice of Conversion" shall mean a notice of conversion provided by Borrower to Agent in substantially the form attached hereto as Exhibit F.

"Obligations" shall mean and include any and all of each Loan Party's Indebtedness and/or liabilities under this Agreement or the other Loan Documents, any Lender-Provided Hedge or with respect to any Bank Product Obligation, to Agent, Lenders, the Issuer or any Affiliate of Agent, any

Lender or the Issuer, of every kind, nature and description, direct or indirect, secured or unsecured, joint, several, joint and several, absolute or contingent, due or to become due, now existing or hereafter arising, contractual or tortious, liquidated or unliquidated, regardless of how such indebtedness or liabilities arise or by what agreement or instrument they may be evidenced or whether evidenced by any agreement or instrument (including all interest, fees and other amounts accruing after the commencement of any bankruptcy or similar proceeding whether or not enforceable in such proceeding) including all interest, expenses, fees, attorney's fees or other amounts chargeable to any Loan Party pursuant to the terms of this Agreement or under any Loan Document and including all obligations of any Loan Party to Agent, Lenders or the Issuer to perform acts or refrain from taking any action; provided that the Obligations of any Guarantor shall not include any Excluded Swap Obligations.

"Other Connection Taxes" means, with respect to any Recipient, Taxes imposed as a result of a present or former connection between such Recipient and the jurisdiction imposing such Tax (other than connections arising from such Recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced this Agreement or any other Loan Document, or sold or assigned an interest in any Advance or this Agreement or any other Loan Document).

"Other Taxes" shall mean any and all present or future stamp, court or documentary, intangible, recording, filing, mortgage or mortgage recording Taxes or any other excise or property Taxes, charges or similar Taxes arising from any payment made hereunder or from the execution, delivery or enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, this Agreement or any other Loan Document.

"Parent" shall mean MarketWise, Inc., a Delaware corporation, and the direct parent of Borrower.

"Participant" shall mean each financial institution who shall be granted the right by any Lender to participate in any of the Advances and who shall have entered into a participation agreement in form and substance satisfactory to such Lender.

"Payment Office" shall mean initially 452 Fifth Avenue, New York, New York 10018; thereafter, such other office of Agent, if any, which it may designate by notice to the Loan Parties and to each Lender to be the Payment Office.

"Payment Recipient" has the meaning assigned to it in Section 14.11(a).

"PBGC" shall mean the Pension Benefit Guaranty Corporation or any successor.

"Pension Benefit Plan" shall mean at any time any employee pension benefit plan (including a Multiple Employer Plan, but not a Multiemployer Plan) which is covered by Title IV of ERISA or is subject to the minimum funding standards under Section 412 of the Code and either (i) is maintained by any member of the Controlled Group for employees of any member of the Controlled Group; or (ii) has at any time within the preceding five years been maintained by any entity which was at such time a member of the Controlled Group for employees of any entity which was at such time a member of the Controlled Group.

"<u>Perfection Certificate</u>" shall mean the Perfection Certificate and the responses thereto provided by the Loan Parties and delivered to Agent.

"<u>Perfection Exceptions</u>" means that no Loan Party shall be required (and Agent shall not be authorized) to (i) enter into control agreements with respect to, or otherwise perfect any Lien by

"control" (or similar arrangements) over any asset (other than certificated securities and instruments), including commodities accounts, securities accounts, deposit accounts, futures accounts, other bank accounts, cash and Cash Equivalents and accounts related to the clearing, payment processing and similar operations of Borrower and its Restricted Subsidiaries, (ii) perfect the security interest in the following (in each case, other than by the filing of a general UCC financing statement): (1) letter-of-credit rights (as defined in the UCC), (2) commercial tort claims (as defined in the UCC), (3) Fixtures (as defined in the UCC), except to the extent that the same are Equipment (as defined in the UCC) or are related to Real Property covered or intended by the Loan Documents to be covered by a mortgage and (4) all contracts and agreements between any Loan Party and one or more additional parties, (iii) send notices to account debtors or other contractual third-parties unless an Event of Default has not been cured or waived and is continuing and the Agent has exercised its rights pursuant to Article XI of this Agreement, (iv) enter into any security documents to be governed by the law of, or make any filing in, any jurisdiction other than the United States, any state thereof or the District of Columbia, (v) deliver landlord waivers, estoppels or collateral access letters or (vi) deliver any stock certificates or stock powers (or equivalent) with respect to the Equity Interests of any Immaterial Subsidiaries or Non-Wholly Owned Subsidiaries (solely to the extent such Non-Wholly Owned Subsidiary is an Excluded Subsidiary).

"Periodic Term SOFR Determination Day" has the meaning specified in the definition of "Term SOFR".

"<u>Permitted Acquisition</u>" shall mean any acquisition by Borrower or any of its Restricted Subsidiaries of all or substantially all of the assets of any Person or a business or division of such Person (whether pursuant to a merger or other transaction) or of all or a majority of the equity of a Person in one or a series of transactions that occurs; <u>provided</u> that:

- (a) before and after giving effect to such acquisition, no Default or Event of Default has occurred and is continuing or would result therefrom;
- (b) after giving effect to such acquisition, on a pro forma basis, the Net Leverage Ratio does not exceed an amount that is 0.50:1.00 below the Net Leverage Ratio required at such time under <u>Section 6.8(b)</u> (including if a Leverage Ratio Increase has been elected) as of the date of such acquisition and otherwise re-computing such covenants as of the last day of the most recently-ended fiscal quarter for which financial statements shall have been delivered as if such acquisition had occurred, and any Indebtedness incurred in connection therewith was incurred, on the first day of the relevant period for testing compliance;
- (c) the Person or assets being acquired is in the same type of business conducted by Borrower and its Restricted Subsidiaries on Closing Date or any other businesses reasonably related, ancillary or complementary thereto or a reasonable extension thereof;
 - (d) Borrower or the applicable Restricted Subsidiary complies with the requirements of Section 7.11(b); and
- (e) Borrower and its Restricted Subsidiaries shall not make Permitted Acquisitions of Persons that do not become Guarantors or by purchase of assets that are acquired directly by Subsidiaries that are not Guarantors (each a "Non-Guarantor Acquisition") for aggregate consideration, together with any Investments made (in each case determined as of the date of making any such Investment), in excess of \$20,000,000.

"Permitted Dispositions" shall have the meaning set forth in Section 7.1(b).

"Permitted Encumbrances" shall mean:

- (a) Liens granted to or in favor of Agent for the benefit of Agent, Lenders and/or the Issuer, which, in each case, secure Obligations;
- (b) Liens for Taxes, assessments or other governmental charges (1) not delinquent or (2) being contested in good faith and by appropriate proceedings and with respect to which proper

reserves have been taken by the applicable Loan Party; <u>provided</u> that no notice of any such Lien has been filed or recorded under the Code and the Treasury Regulations adopted thereunder or any other Applicable Law;

- (c) deposits or pledges to secure obligations under worker's compensation, social security or similar laws, or under unemployment insurance;
- (d) deposits or pledges to secure bids, tenders, contracts (other than contracts for the payment of money), leases, statutory obligations, surety, appeal bonds, customs bonds and other obligations of like nature arising in the ordinary course of such Loan Party's business;
- (e) judgment Liens that have been stayed or bonded or otherwise would not result in an Event of Default and mechanics', workers', materialmen's or other like Liens arising in the ordinary course of such Loan Party's business with respect to obligations which are not due or which are being contested in good faith by appropriate proceedings;
- (f) Liens for purchase money obligations and Capital Leases, <u>provided</u> that (i) the Indebtedness secured by any such Lien is permitted under <u>Section 7.7</u>, and (ii) such Lien encumbers only the asset so purchased;
- (g) leases or subleases of Real Property granted in the ordinary course of any Loan Party's business (or, if referring to another Person, in the ordinary course of such Person's business), and leases, subleases, non-exclusive licenses or sublicenses of personal property granted in the ordinary course of any Loan Party's business (or, if referring to another Person, in the ordinary course of such Person's business), if the leases, subleases, licenses and sublicenses do not prohibit granting Agent a security interest therein;
- (h) non-exclusive license of Intellectual Property granted to third parties in the ordinary course of business, and licenses of Intellectual Property that could not result in a legal transfer of title of the licensed property that may be exclusive in respects other than territory and that may be exclusive as to territory only as to discrete geographical areas outside of the United States;
- (i) Liens that are replacements of Permitted Encumbrances to the extent that the original Indebtedness is the subject of Permitted Indebtedness and so long as the replacement Liens only encumber those assets that secured the original Indebtedness;
- (j) rights of setoff or bankers' liens upon deposits of funds in favor of banks or other depository institutions, solely to the extent incurred in connection with the maintenance of deposit accounts in the ordinary course of business;
- (k) Liens granted in the ordinary course of business on the unearned portion of insurance premiums securing the financing of insurance premiums to the extent the financing is permitted under the definition of Permitted Indebtedness;

- (l) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods;
 - (m) Liens existing on the Closing Date and disclosed on Schedule 7.2;
- (n) other Liens securing Permitted Indebtedness as to which the aggregate amount of the obligations secured thereby does not exceed the greater of (i) \$5,000,000 and (ii) 2.5% of Consolidated EBITDA as of the most recently-ended test period, in the aggregate outstanding at any time;
- (o) survey exceptions, encumbrances, ground leases, easements or reservations of, or rights of others for, licenses, rights-of-way, servitudes, sewers, electric lines, drains, telegraph and telephone and cable television lines, gas and oil pipelines and other similar purposes, reservations of rights, or zoning, building codes or other restrictions (including, without limitation, minor defects or irregularities in title and similar encumbrances) as to the use of Real Properties or Liens incidental to the conduct of the business of such Person or to the ownership of its properties which do not in the aggregate materially adversely interfere with the ordinary conduct of the business of such Person;
- (p) Liens securing Indebtedness or other obligations of Borrower or a Guarantor owing to Borrower or another Guarantor permitted to be incurred in accordance with Section 7.7;
- (q) Liens arising from, or from Uniform Commercial Code financing statement filings regarding, operating leases or consignments entered into by Borrower or Restricted Subsidiary in the ordinary course of business;
- (r) deposits made or other security provided in the ordinary course of business to secure liability to insurance carriers or under self-insurance arrangements in respect of such obligations;
- (s) Liens (i) of a collection bank arising under Section 4-210 of the Uniform Commercial Code, or any comparable or successor provision, on items in the course of collection; (ii) attaching to pooling, commodity trading accounts or other commodity brokerage accounts incurred in the ordinary course of business; and (iii) in favor of banking or other financial institutions or entities, or electronic payment service providers, arising as a matter of law encumbering deposits (including the right of set-off) and which are within the general parameters customary in the banking or finance industry;
- (t) Liens that are contractual rights of set-off (i) relating to the establishment of depository relations with banks or other Persons not given in connection with the issuance of Indebtedness; (ii) relating to pooled deposit or sweep accounts of Borrower or any Restricted Subsidiary to permit satisfaction of overdraft or similar obligations incurred in the ordinary course of business of Borrower and the Restricted Subsidiaries; or (iii) relating to purchase orders and other agreements entered into with customers of Borrower or any Restricted Subsidiary in the ordinary course of business;
- (u) any encumbrance or restriction (including put and call arrangements) with respect to Equity Interests of any joint venture or similar arrangement pursuant to any joint venture or similar agreement;
 - (v) Liens on vehicles of Borrower or any Restricted Subsidiary granted in the ordinary course of business;
- (w) Liens arising solely by virtue of any statutory or common law provision or customary business provision relating to banker's liens, rights of set off or similar rights;

- (x) Liens on securities (other than Collateral) that are the subject of repurchase agreements constituting Cash Equivalents;
- (y) Liens encumbering reasonable customary initial deposits and margin deposits and similar Liens attaching to commodity trading accounts or other brokerage accounts incurred in the ordinary course of business and not for speculative purposes;
 - (z) restrictive covenants affecting the use to which Real Property may be put; provided that such covenants are complied with;
- (aa) security given to a public utility or any municipality or Governmental Body when required by such utility or authority in connection with the operations of that Person in the ordinary course of business;
- (bb) zoning by-laws and other activity and land use restrictions, including, without limitation, site plan agreements, development agreements, contract zoning agreements and limitations imposed under Environmental Law to secure remedial obligations; and
- (cc) (i) Liens on Securitization Assets and related assets, or created in respect of bank accounts into which only the collections in respect of Securitization Assets have been, sold, conveyed, assigned or otherwise transferred or purported to be so sold, conveyed, assigned or otherwise transferred in connection with a Permitted Securitization Financing and (ii) Liens securing Indebtedness or other obligations of any Special Purpose Securitization Subsidiary;

provided that no Real Property shall be subject to any Permitted Encumbrances except those referenced in <u>clauses (a)</u>, (b), (e), (f), (g), (o), (z), (aa) and (bb) above.

"<u>Permitted Holders</u>" shall mean, as to the Parent, Ascendant Sponsor LP, a Cayman Islands exempted limited partnership, the common equity holders of Borrower as of the Closing Date and such equityholders' Permitted Transferees under the LLC Agreement, and, in each case, their successors and permitted assigns.

"Permitted Indebtedness" shall mean:

- (a) the Obligations;
- (b) Indebtedness existing on the Closing Date and shown on Schedule 7.7;
- (c) unsecured Indebtedness to trade creditors incurred in the ordinary course of business;
- (d) Indebtedness of (i) any Loan Party to any other Loan Party, (ii) any Restricted Subsidiary (which is not a Loan Party) to any other Restricted Subsidiary (which is not a Loan Party), (iii) any Subsidiary (which is not a Loan Party) to any Loan Party to any Subsidiary (which is not a Guarantor); provided that all Indebtedness under this clause (iv) is subordinated to the Obligations of such Loan Party under this Agreement and the other Loan Documents in a manner reasonably satisfactory to Agent; and provided, further, that the aggregate outstanding principal amount of such Indebtedness under this clause (iv) does not exceed \$20,000,000, and (v) any Restricted Subsidiary (which is not a Loan Party) to any unaffiliated Person; provided, that the aggregate outstanding principal amount of such Indebtedness under this clause (v) does not exceed \$5,000,000;

- (e) Indebtedness incurred as a result of endorsing negotiable instruments received in the ordinary course of business;
- (f) Indebtedness secured by purchase money Liens and Capital Lease Obligations (in each case incurred prior to or within 365 days of the acquisition or lease or completion of construction, repair or replacement of, or improvement to or installation of, assets) not exceeding the greater of (i) \$5,000,000 and (ii) 2.5% of Consolidated EBITDA as of the most recently-ended test period, in the aggregate outstanding at any time;
 - (g) [reserved];
 - (h) Indebtedness in connection with Permitted Securitization Financings;
- (i) unsecured Indebtedness not otherwise permitted by <u>Section 7.7</u> not exceeding the greater of (i) \$15,000,000 and (ii) 7.5% of Consolidated EBITDA as of the most recently ended test period, in the aggregate outstanding at any time;
- (j) solely to the extent they constitute Indebtedness, customer deposits and advance payments received in the ordinary course of business from customers for goods or services purchased in the ordinary course of business;
- (k) Indebtedness in respect of appeal, bid, performance or surety or similar bonds, workers' compensation claims, self-insurance obligations and bankers acceptances issued for the account of Borrower or any of its Subsidiaries, each incurred in the ordinary course of business, including guarantees or obligations of any Borrower or any of its Subsidiaries with respect to letters of credit supporting such bid, performance or surety bonds, workers' compensation claims, self-insurance obligations and bankers acceptances (in each case other than for an obligation for borrowed money);
- (l) Indebtedness incurred by Borrower or any of its Restricted Subsidiaries constituting reimbursement obligations with respect to letters of credit or bank guarantees or similar instruments issued in the ordinary course of business, including, without limitation, (i) letters of credit or performance or surety bonds in respect of workers' compensation claims, health, disability or other employee benefits (whether current or former) or property, casualty or liability insurance or self-insurance, or other Indebtedness with respect to reimbursement-type obligations regarding workers' compensation claims, health, disability or other employee benefits (whether current or former) or property, casualty or liability insurance and (ii) guarantees of Indebtedness incurred by customers in connection with the purchase or other acquisition of equipment or supplies in the ordinary course of business;
- (m) unsecured Indebtedness consisting of interest to the extent paid-in-kind (and not in cash or Cash Equivalents) on intercompany Indebtedness;
- (n) Indebtedness incurred in the ordinary course of business in respect of credit cards, credit card processing services, debit cards, stored value cards, commercial cards (including so called "purchase cards", "procurement cards" or "p cards"), or cash management services;
 - (o) guarantees permitted by Section 7.3;
- (p) Indebtedness not otherwise permitted by <u>Section 7.7</u> not exceeding the greater of (i) \$5,000,000 and (ii) 2.5% of Consolidated EBITDA as of the most recently ended test period, in the aggregate outstanding at any time;
- (q) Indebtedness arising from netting services, overdraft protection or the honoring by a bank or other financial institution of a check, draft or similar instrument drawn against insufficient funds in the ordinary course of business;

- (r) Indebtedness of a joint venture to Borrower or a Restricted Subsidiary and to the other holders of Equity Interests or participants of such joint venture, so long as the percentage of the aggregate amount of such Indebtedness of such joint venture owed to such holders of its Equity Interests or participants of such joint venture does not exceed the percentage of the aggregate outstanding amount of the Equity Interests of such joint venture held by such holders or such participant's participation in such joint venture;
- (s) Indebtedness incurred by Borrower or a Restricted Subsidiary in connection with bankers' acceptances, discounted bills of exchange, warehouse receipts or similar facilities or the discounting or factoring of receivables for credit management purposes, in each case incurred or undertaken in the ordinary course of business or consistent with past practice;
- (t) Indebtedness incurred by Borrower or a Restricted Subsidiary as a result of leases entered into by Borrower or such Restricted Subsidiary in the ordinary course of business;
- (u) unfunded pension fund and other employee benefit plan obligations and liabilities to the extent that they are permitted to remain unfunded under applicable law; and
- (w) extensions, refinancings, modifications, amendments and restatements of any items of Permitted Indebtedness (a) through (u) above, provided that the principal amount thereof is not increased or the terms thereof are not modified to impose more burdensome terms upon any Loan Party or its Subsidiary, as the case may be.

"Permitted Investments" shall mean:

- (a) Investments (including, without limitation, Subsidiaries) existing on the Closing Date and shown on Schedule 7.4;
- (b) Investments consisting of cash and Cash Equivalents;
- (c) Investments consisting of the endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of business;
 - (d) Investments consisting of deposit accounts in which Agent has a perfected security interest;
 - (e) Investments accepted in connection with transfers permitted by <u>Section 7.1(b)</u>;
 - (f) [reserved];
- (g) Investments consisting of travel advances and employee relocation loans and other employee loans and advances in the ordinary course of business not to exceed \$1,000,000 in the aggregate in any fiscal year;
- (h) Investments (including debt obligations, Equity Interests or other securities) received in connection with the bankruptcy or reorganization of customers or suppliers and in settlement

of delinquent obligations of, and other disputes with, customers or suppliers arising in the ordinary course of business;

- (i) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions, to customers and suppliers who are not Affiliates, in the ordinary course of business; <u>provided</u> that this <u>paragraph (i)</u> shall not apply to Investments of Borrower in any Subsidiary;
 - (j) Lender-Provided Hedges;
 - (k) Permitted Acquisitions;
- (l) any Investment in (i) any Loan Party by any other Loan Party, (ii) any Restricted Subsidiary (which is not a Loan Party) by any other Restricted Subsidiary (which is not a Loan Party), (iii) any Subsidiary (which is not a Loan Party) by any Loan Party; provided that the aggregate amount of such Investments under this clause (iii) does not exceed \$20,000,000 at any one time outstanding and (iv) any Loan Party by any Restricted Subsidiary (which is not a Loan Party);
 - (m) guarantees permitted by Section 7.3;
- (n) other Investments not otherwise permitted by <u>Section 7.4</u>, so long as (i) no Default or Event of Default shall exist or result from the making of such Investment and (ii) as of the most recently ended fiscal quarter for which financial statements shall have been delivered, calculated on a pro forma basis as if such Investment had been made on the first day of the relevant testing period, the Net Leverage Ratio is not greater than 1.00:1.00;
- (o) Investments consisting of the leasing, licensing, sublicensing of Intellectual Property or pursuant to joint marketing arrangements with other Persons, in each case in the ordinary course of business;
- (p) Investments consisting of purchases or acquisitions of inventory, supplies, materials and equipment or purchases, acquisitions, licenses, sublicenses or leases or subleases of intellectual property, or other rights or assets, in each case in the ordinary course of business;
 - (q) extensions of trade credit in the ordinary course of business by Borrower or any of its Restricted Subsidiaries;
 - (r) Investments consisting of purchases of services in the ordinary course of business;
- (s) Investments in the ordinary course of business consisting of Uniform Commercial Code Article 3 endorsements for collection or deposit and Uniform Commercial Code Article 4 customary trade arrangements with customers;
- (t) intercompany current liabilities owed to Unrestricted Subsidiaries or joint ventures incurred in the ordinary course of business in connection with the cash management operations of Borrower and its Subsidiaries;
 - (u) Investments resulting from pledges and deposits that are Permitted Encumbrances;
 - (v) non-cash Investments made in connection with tax planning and reorganization activities;
- (w) Investments made in the ordinary course of business in connection with obtaining, maintaining or renewing client and customer contracts and loans or advances made to, and guarantees with respect to obligations of, distributors, suppliers, licensors and licensees in the ordinary course of business; and

(x) any Investments in a Special Purpose Securitization Subsidiary in connection with a Permitted Securitization Financing.

"Permitted Securitization Financing" shall mean one or more transactions pursuant to which (i) Securitization Assets or interests therein are sold or transferred to or financed by one or more Special Purpose Securitization Subsidiaries and (ii) such Special Purpose Securitization Subsidiaries finance (or refinance) their acquisition of such Securitization Assets or interests therein, or the financing thereof, by selling or borrowing against Securitization Assets (including conduit and warehouse financings) and any hedging agreements entered into in connection with such Securitization Assets; provided that (x) any Indebtedness incurred in connection with such transaction is non-recourse to Borrower or any Subsidiary thereof (other than the Special Purpose Securitization Subsidiaries) and its respective assets and (y) other than the initial Investment in any such Special Purpose Securitization Subsidiary, none of the Loan Parties or any of their respective Subsidiaries is required to make additional Investments in such Special Purpose Securitization Subsidiary.

"Person" shall mean any individual, sole proprietorship, partnership, corporation, business trust, joint stock company, trust, unincorporated organization, association, limited liability company, limited liability partnership, institution, public benefit corporation, joint venture, entity or Governmental Body (whether Federal, state, county, city, municipal or otherwise, including any instrumentality, division, agency, body or department thereof).

"<u>Plan</u>" shall mean any employee benefit plan within the meaning of Section 3(3) of ERISA, maintained for employees of any Loan Party or any member of the Controlled Group or any such Plan to which any Loan Party or any member of the Controlled Group is required to contribute on behalf of any of its employees.

"Platform" has the meaning specified in Section 6.11.

"Pledged Collateral" shall mean the collective reference to the following: all of such Loan Party's right, title and interest in, to and under (a)(i) the shares of capital stock and other Equity Interests owned by such Loan Party, including those listed opposite the name of such Loan Party in the Perfection Certificate; (ii) any other Equity Interests obtained in the future by such Loan Party and (iii) the certificates and other instruments (if any) representing all such Equity Interests (collectively, the "Pledged Equity Interests"); provided that the Pledged Equity Interests shall not include any Excluded Property, (b)(i) the debt securities owned by such Loan Party, including those listed opposite the name of such Loan Party in the Perfection Certificate, (ii) any debt securities in the future issued to or otherwise acquired by such Loan Party and (iii) the promissory notes and any other instruments evidencing all such debt securities (collectively, the "Pledged Debt Securities"); provided that the Pledged Debt Securities shall not include any Excluded Property; (c) all other property that may be delivered to and held by the Agent pursuant to the terms of Section 4.2; (d) all payments of principal or interest, dividends, cash, instruments and other property from time to time received, receivable or otherwise distributed in respect of, in exchange for or upon the conversion of, and all other Proceeds received in respect of, the Pledged Equity Interests and Pledged Debt Securities; (e) all rights and privileges of such Loan Party with respect to the securities, instruments and other property referred to in Clauses (a), (b), (c) and (d) above; and (f) all Proceeds of any of the foregoing; provided that, in each case, the Pledged Collateral shall not include any Excluded Property.

- "Pledged Debt Securities" has the meaning specified in the definition of "Pledged Collateral".
- "Pledged Equity Interests" has the meaning specified in the definition of "Pledged Collateral".
- "Provision for Taxes" shall mean an amount equal to all Taxes imposed on or measured by net income, whether federal, state, provincial, county or local, and whether foreign or domestic, that are paid or payable by any Person in respect of any period in accordance with GAAP.
- " \underline{PTE} " means a prohibited transaction class exemption issued by the Department of Labor, as any such exemption may be amended from time to time.
 - "Purchasing Lender" shall have the meaning set forth in Section 15.3(c).
- " \overline{QFC} " has the meaning assigned to the term "qualified financial contract" in, and shall be interpreted in accordance with, 12 U.S.C. § 5390(c)(8)(D).
 - "QFC Credit Support" has the meaning specified therefor in Section 15.20 of this Agreement.
- "Qualified ECP Loan Party" means, in respect of any Swap Obligation, (a) each of Borrower and any Guarantor that has total assets exceeding \$10,000,000 at the time such Swap Obligation is incurred, or (b) such other Person as is qualified to give a "letter of credit or keepwell, support, or other agreement" for purposes of Section 1a(18)(A)(v)(II) of the Commodity Exchange Act.
 - "Real Property" shall mean all of each Loan Party's right, title and interest in and to its owned and leased premises.
- "Receivables" shall mean and include as to each Loan Party, all of such Loan Party's accounts (including, without limitation, all health-care insurance receivables), contract rights, instruments (including promissory notes and other instruments evidencing Indebtedness owed to such Loan Party by their Affiliates), documents, chattel paper (whether tangible or electronic), general intangibles relating to accounts, drafts and acceptances, and all other forms of obligations owing to such Loan Party arising out of or in connection with the sale, lease or other disposition of Inventory or the rendition of services, all guarantees and other security therefor, whether secured or unsecured, now existing or hereafter created, and whether or not specifically sold or assigned to Agent hereunder.
 - "Recipient" means (a) Agent, (b) any Lender and (c) the Issuer, as applicable.
- "Reinvestment" shall mean the acquisition of assets or other investment in the relevant Person's business with an amount equal to the Net Cash Proceeds of any sale or other disposition of any other assets (other than Inventory in the ordinary course of business). "Reinveste" and "Reinvested" shall have correlative meanings.
- "Reinvestment Deferred Amount" shall mean with respect to any Reinvestment Event, an amount equal to the aggregate Net Cash Proceeds received by any Loan Party in connection therewith that are not applied to prepay the Advances or other amounts pursuant to Section 2.14(a) as a result of the delivery of a Reinvestment Notice.
- "Reinvestment Event" shall mean any sale or other disposition of assets (other than Inventory in the ordinary course of business) in respect of which Borrower has delivered a Reinvestment Notice to Agent.

"Reinvestment Notice" shall mean a written notice executed by an Authorized Officer of Borrower stating that no Default or Event of Default has occurred and is continuing and that the Loan Parties (directly or indirectly through a Subsidiary) reasonably intend and expect to Reinvest all or a specified portion of an amount equal to the Net Cash Proceeds of a sale or other disposition of assets (other than Inventory in the ordinary course of business) or Extraordinary Receipts, in each case, within 365 days of such sale or disposition (or, if within such 365-day period, the Loan Parties enter into a binding commitment to so Reinvest, and does so Reinvest, such Net Cash Proceeds, within 180 days following such 365-day period during which Borrower so committed to such plan of Reinvestment).

"Reinvestment Prepayment Amount" shall mean, with respect to any Reinvestment Event, the Reinvestment Deferred Amount relating thereto less any amount actually Reinvested prior to the relevant Reinvestment Prepayment Date.

"Reinvestment Prepayment Date" shall mean with respect to any Reinvestment Event, the earlier of (a) 365 days after such sale or disposition (or, if within such 365-day period, the Loan Parties enter into a binding commitment to so Reinvest, and does so Reinvest, such Net Cash Proceeds, within 180 days following such 365-day period during which Borrower so committed to such plan of Reinvestment) and (b) the date on which Borrower shall have determined not to, or shall have otherwise ceased to, make Reinvestments with all or any portion of the relevant Reinvestment Deferred Amount.

"Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, directors, officers, employees, agents, trustees, administrators, managers, advisors and representatives of such Person and of such Person's Affiliates.

"Relevant Governmental Body." means the Board of Governors of the Federal Reserve SystemBoard or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the Board of Governors of the Federal Reserve SystemBoard or the Federal Reserve Bank of New York, or any successor thereto.

"Report" shall have the meaning set forth in Section 9.7.

"Reportable Event" shall mean a reportable event described in Section 4043(c) of ERISA or the regulations promulgated thereunder.

"Required Lenders" shall mean Lenders holding a majority of the Advances (excluding outstanding Swingline Loans as of the most recent Settlement Date, but including any Incremental Term Loans made pursuant to Section 2.4) and, if no Advances are outstanding, shall mean Lenders holding a majority of the Commitment Percentages; provided however, if there are fewer than three (3) Lenders, Required Lenders shall mean all Lenders.

"Resignation Effective Date" shall mean, collectively, the promissory notes referred to in Section 14.3(c).

"Resolution Authority" means an EEA Resolution Authority or, with respect to any UK Financial Institution, a UK Resolution Authority.

"Restricted Payment" shall mean (i) to declare, pay or make any dividend or distribution on any shares of the common stock, preferred stock or other Equity Interests of any Loan Party (other than dividends or distributions payable in its stock or other Equity Interests or split-ups or reclassifications of its stock or other equity interests) or for any Loan Party to apply any of its funds, property or assets to the purchase, redemption or other retirement of any common or preferred stock or other Equity Interests, or of any options to purchase or acquire any such shares of common or preferred stock or other Equity

Interests of any Loan Party, (ii) to repay or prepay any Junior Lien Indebtedness or Subordinated Indebtedness, or repurchase, redeem or retire Junior Lien Indebtedness or Subordinated Indebtedness of any Loan Party, or (iii) any payment by any Loan Party of any management, consulting or similar fees to or any Affiliate, whether pursuant to a management agreement or otherwise.

"Restricted Subsidiary" shall mean each Subsidiary of Borrower other than any Unrestricted Subsidiaries.

"<u>Revolving Advances</u>" shall mean Advances made other than Swingline Loans, Letters of Credit, and the Incremental Term Loan, including, for the avoidance of doubt, any Revolving Advances made pursuant to <u>Section 2.4</u>.

"Revolving Commitment" shall mean, as to any Lender, its obligation to make Revolving Advances in an aggregate amount not to exceed its Revolving Commitment Percentage.

"Revolving Commitment Percentage" of any Lender shall mean the percentage set forth below such Lender's under the applicable heading on Schedule 1.1 to this Agreement under the heading "Revolving Commitment Percentage" as same may be adjusted upon any assignment by a Lender pursuant to Section 15.3.

"Revolving Credit Note" shall mean, collectively, the promissory notes referred to in Section 2.1(a).

"Revolving Interest Rate" shall mean an interest rate per annum equal to (a) the sum of the Base Rate plus the Applicable Margin, with respect to Domestic Rate Loans and (b) the sum of the Eurodollar Rate Adjusted Term SOFR plus the Applicable Margin, with respect to Eurodollar Rate SOFR Loans

"Sanctioned Person" means, at any time, (a) any Person listed in any Sanctions-related list of designated Persons maintained by the Office of Foreign Assets Control of the U.S. Department of the Treasury, the U.S. Department of State or by the United Nations Security Council, the European Union or any European Union member state, Her Majesty's Treasury of the United Kingdom or other relevant sanctions authority, (b) any Person operating, organized or resident in a jurisdiction subject to Sanctions, (c) any Person owned or controlled by any such Person or Persons described in the foregoing clauses (a) or (b), or (d) any Person that is otherwise the subject of any Sanctions.

"Sanctions" shall have the meaning set forth in Section 5.23.

"SEC" shall mean the U.S. Securities and Exchange Commission or any successor thereto.

"Securitization Assets" shall mean any of the following assets (or interests therein) from time to time originated, acquired or otherwise owned by Borrower or any Subsidiary or in which Borrower or any Subsidiary has any rights or interests, in each case, without regard to where such assets or interests are located: (a) Receivables, (b) franchise fees, royalties and other similar payments made related to the use of trade names and other Intellectual Property, business support, training and other services, (c) revenues related to distribution and merchandising of the products of Borrower and its Subsidiaries, (d) rents, real estate taxes and other non-royalty amounts due from franchisees, (e) Intellectual Property rights relating to the generation of any of the types of assets listed in this definition, (f) parcels of or interests in Real Property, together with all appurtenant fixtures and equipment, incidental to the ownership, lease or operation thereof, (g) any Equity Interests of any Special Purpose Securitization Subsidiary or any Subsidiary of a Special Purpose Securitization Subsidiary and any rights under any limited liability company agreement, trust agreement, shareholders agreement, organization or formation documents or other agreement entered into in furtherance of the organization of such entity, (h) any equipment,

contractual rights with unaffiliated third parties, website domains and associated property and rights necessary for a Special Purpose Securitization Subsidiary to operate in accordance with its stated purposes; (i) any rights and obligations associated with gift card or similar programs, and (j) other assets and property (or proceeds of such assets or property) to the extent customarily included in securitization transactions of the relevant type in the applicable jurisdictions (as determined by Borrower in good faith).

"Settlement Date" shall mean the Closing Date and thereafter Tuesday of each week, unless such day is not a Business Day in which case it shall be the next succeeding Business Day, and every other Business Day designated by Agent as a "Settlement Date" by notice from Agent to each Lender. Agent shall be obligated to designate a Settlement Date in the event that Borrower requests Revolving Advances in excess of the Maximum Swingline Loan Amount.

"SOFR" means a rate per annum equal to the secured overnight financing rate for such Business Day published by the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate) on the website of the Federal Reserve Bank of New York, currently at http://www.newyorkfed.org (or any successor source for the secured overnight financing rate identified as such by the administrator of the secured overnight financing rate from time to time).

<u>"SOFR Borrowing" means, as to any borrowing of Revolving Advances, the SOFR Loans comprising such borrowing of Revolving Advances.</u>

"SOFR Loan" means a Revolving Advance that bears interest at a rate based on Adjusted Term SOFR, other than pursuant to clause (c) of the definition of "Base Rate".

"SPAC Transaction" shall mean the merger of Ascendant Digital Acquisition Corp. and Parent, which was effective as of July 21, 2021.

"Special Purpose Securitization Subsidiary" shall mean (i) a direct or indirect Subsidiary of Borrower established in connection with a Permitted Securitization Financing for the acquisition of Securitization Assets or interests therein, and which is organized in a manner (as determined by Borrower in good faith) intended to reduce the likelihood that it would be substantively consolidated with any of Borrower or any of the Subsidiaries (other than Special Purpose Securitization Subsidiaries) in the event Borrower or any such Subsidiary becomes subject to a proceeding under the Title 11 of the Bankruptcy Code (or other insolvency law) and (ii) any subsidiary of a Special Purpose Securitization Subsidiary.

"Standby Letters of Credit" shall mean all Letters of Credit issued in connection with this Agreement as a credit enhancement for certain Indebtedness (other than Indebtedness for borrowed money) of Borrower.

"Stock Pledge Agreement" shall mean a Stock Pledge Agreement, dated as of the Closing Date, pursuant to which each Loan Party pledges to Agent as Collateral for the Obligations the issued and outstanding shares of Subsidiary Stock.

"Subordinated Indebtedness" means (a) with respect to Borrower, any Indebtedness of Borrower which is by its terms expressly subordinated in right of payment to the Obligations, and (b) with respect to any Guarantor, any Indebtedness of such Guarantor which is by its terms expressly subordinated in right of payment to its guarantee of the Obligations, in each case to the written satisfaction of Agent and the Required Lenders.

- "Subsidiary" shall mean, with respect to any Person, a corporation or other entity of whose Equity Interests having ordinary voting power (other than Equity Interests having such power only by reason of the happening of a contingency) to elect a majority of the directors or managers of such corporation or other entity, or other Persons performing similar functions for such entity, are owned, directly or indirectly, by such Person.
- "Subsidiary Stock" shall mean all of the issued and outstanding Equity Interests of any U.S. Wholly Owned Subsidiary owned by any Loan Party.
 - "Supported QFC" has the meaning specified therefor in Section 15.20 of this Agreement.
- "Swap Obligation" means, with respect to any of Borrower or any Guarantor, any obligation to pay or perform under any agreement, contract or transaction that constitutes a "swap" within the meaning of Section 1a(47) of the Commodity Exchange Act.
- "Swingline Lender" means HSBC, or if HSBC shall resign as Swingline Lender, another Lender selected by Agent or the successor Agent and reasonably acceptable to Required Lenders and Borrower.
 - "Swingline Loan" means each Advance made by Swingline Lender pursuant to Section 2.1(c).
 - "Swingline Note" shall mean the promissory note referred to in Section 2.1(c).
- "<u>Tax Receivable Agreement</u>" shall mean the Tax Receivable Agreement, dated as of July 21, 2021, by and among Parent, Borrower and the other parties thereto, as the same may be amended, restated, modified and/or supplemented from time to time in accordance with the terms thereof and hereof.
- "Taxes" means any and all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Body, including any interest, additions to tax or penalties applicable thereto.
 - "Term" shall mean the period commencing on the Closing Date and ending on the Termination Date.

"Term SOFR" means:

- (a) for any calculation with respect to a SOFR Loan, the Term SOFR Reference Rate for a tenor comparable to the applicable Interest Period on the day (such day, the "Periodic Term SOFR Determination Day") that is two (2) U.S. Government Securities Business Days prior to the first day of such Interest Period, as such rate is published by the Term SOFR Administrator; provided, however, that if as of 5:00 p.m. (New York City time) on any Periodic Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as published by the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such Periodic Term SOFR Determination Day, and
- (b) for any calculation with respect to a Domestic Rate Loan on any day, the Term SOFR Reference Rate for a tenor of one month on the day (such day, the "Base Rate Term SOFR

Determination Day") that is two (2) U.S. Government Securities Business Days prior to such day, as such rate is published by the Term SOFR Administrator; provided, however, that if as of 5:00 p.m. (New York City time) on any Base Rate Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as published by the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such Base Rate Term SOFR Determination Day;

<u>"Term SOFR Adjustment" means, for any calculation with respect to a SOFR Loan, a percentage per annum as set forth below for the applicable Interest Period therefor:</u>

Interest Period	<u>Percentage</u>
One month	<u>0.11448%</u>
Three months	<u>0.26161%</u>
Six months	<u>0.42826%</u>

<u>"Term SOFR Administrator" means CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the Term SOFR Reference Rate selected by the Agent in its reasonable discretion).</u>

"<u>Term SOFR</u>" means, for the applicable corresponding tenor, <u>Reference Rate</u>" means the forward-looking term rate based on SOFR that has been selected or recommended by the Relevant Governmental Body.

"Termination Date" shall have the meaning set forth in Section 13.1.

"Termination Event" shall mean (i) a Reportable Event with respect to any Plan or Multiemployer Plan; (ii) the withdrawal of any Loan Party or any member of the Controlled Group from a Plan or Multiemployer Plan during a plan year in which such entity was a "substantial employer" as defined in Section 4001(a)(2) of ERISA; (iii) the providing of notice of intent to terminate a Plan in a distress termination described in Section 4041(c) of ERISA; (iv) the institution by the PBGC of proceedings to terminate a Plan or Multiemployer Plan; (v) any event or condition (a) which might reasonably be expected to constitute grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Plan or Multiemployer Plan, or (b) that may reasonably be expected to result in termination of a Multiemployer Plan pursuant to Section 4041A of ERISA; or (vi) the partial or complete withdrawal within the meaning of Sections 4203 and 4205 of ERISA, of any Loan Party or any member of the Controlled Group from a Multiemployer Plan.

"Transactions" means the transactions contemplated by this Agreement.

"U.S. Government Securities Business Day" means any day except for (a) a Saturday, (b) a Sunday or (c) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities.

"<u>U.S. Person</u>" means any Person that is a "United States Person" as defined in Section 7701(a)(30) of the Code.

- "U.S. Special Resolution Regimes" has the meaning specified therefor in Section 15.20 of this Agreement.
- "UCC" shall have the meaning set forth in Section 1.3.
- "UK Bribery Act" shall have the meaning set forth in Section 5.23.
- "<u>UK Financial Institution</u>" means any BRRD Undertaking (as such term is defined under the PRA Rulebook (as amended <u>formfrom</u> time to time) promulgated by the United Kingdom Prudential Regulation Authority) or any person falling within IFPRU 11.6 of the FCA Handbook (as amended from time to time) promulgated by the United Kingdom Financial Conduct Authority, which includes certain credit institutions and investment firms, and certain affiliates of such credit institutions or investment firms.
- "<u>UK Resolution Authority</u>" means the Bank of England or any other public administrative authority having responsibility for the resolution of any UK Financial Institution.
- "<u>Unadjusted Benchmark Replacement</u>" means the <u>applicable</u> Benchmark Replacement excluding the <u>related</u> Benchmark Replacement Adjustment.
 - "United States" or "U.S." means the United States of America.
- "<u>Unrestricted Subsidiary</u>" shall mean (i) on the Closing Date, each Subsidiary of Borrower listed on <u>Schedule 1.3</u> and (ii) any other Subsidiary of Borrower designated by the board of directors of Borrower as an Unrestricted Subsidiary pursuant to <u>Section 6.10</u> subsequent to the Closing Date, in each case, except to the extent redesignated as a Restricted Subsidiary in accordance with such <u>Section 6.10</u>.
- "<u>USA PATRIOT Act</u>" shall mean the Uniting and Strengthening America by Providing Appropriate Tools Required To Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56, as the same has been, or shall hereafter be renewed, extended, amended or replaced from time to time.
 - "USCO" shall mean the U.S. Copyright Office.
 - "USD LIBOR" means the London interbank offered rate for Dollars.
 - "USPTO" shall mean the U.S. Patent and Trademark Office.
 - "Wholly Owned Restricted Subsidiary" means any Wholly Owned Subsidiary that is a Restricted Subsidiary.
- "Wholly Owned Subsidiary" of any Person means a direct or indirect Subsidiary of such Person 100% of the outstanding Equity Interests or other ownership interests of which (other than directors' qualifying shares or shares or interests required to be held by foreign nationals or other third parties to the extent required by applicable law) shall at the time be owned by such Person or by one or more Wholly Owned Subsidiaries of such Person.
 - "Withholding Agent" means any Loan Party and Agent.
- "<u>Write-Down and Conversion Powers</u>" means, (a) with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule, and (b) with respect to the United Kingdom, any

powers of the applicable Resolution Authority under the Bail-In Legislation to cancel, reduce, modify or change the form of a liability of any UK Financial Institution or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability or any of the powers under that Bail-In Legislation that are related to or ancillary to any of those powers.

1.3. UCC Terms.

All terms used herein and defined in the Uniform Commercial Code as adopted in the State of New York from time to time (the "<u>UCC</u>") shall have the meaning given therein unless otherwise defined herein. Without limiting the foregoing, the terms "accounts", "chattel paper", "commercial tort claims", "instruments", "general intangibles", "goods", "payment intangibles", "supporting obligations", "securities", "documents", "deposit accounts", "securities accounts", "proceeds", "software", "letter of credit", "letter-of-credit rights", "inventory", "equipment" and "fixtures", as and when used in the description of Collateral shall have the meanings given to such terms in Articles 8 or 9 of the UCC. To the extent the definition of any category or type of collateral is expanded by any amendment, modification or revision to the UCC, such expanded definition will apply automatically as of the date of such amendment, modification or revision.

1.4. <u>Certain Matters of Construction</u>.

The terms "herein", "hereof" and "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular section, paragraph or subdivision. Each reference to a Section, an Exhibit or a Schedule shall be deemed to refer to a Section, an Exhibit or a Schedule, as applicable, of this Agreement, as modified or supplemented with the consent of Agent in its sole discretion unless otherwise specified. Any pronoun used shall be deemed to cover all genders. Wherever appropriate in the context, terms used herein in the singular also include the plural and vice versa. All references to statutes (including the UCC) and related regulations shall include any amendments of same and any successor statutes and regulations. Unless otherwise provided, all references to any instruments or agreements to which Agent is a party, including, without limitation, references to any of the Loan Documents, shall include any and all modifications or amendments thereto and any and all extensions or renewals thereof. For purposes of <u>Sections 3.1, 3.5, 3.6, 3.7, 3.8, 3.10(a)</u>, and <u>Articles V, VIII, XI, XII, XIII, XIV</u>, and <u>XV</u>, the term "<u>Lender</u>" shall include each Lender and Swingline Lender. All references herein to the time of day shall mean the time in New York, New York. A Default or Event of Default shall be deemed to exist at all times during the period commencing on the date that such Default or Event of Default occurs to the date on which such Default or Event of Default is waived in writing pursuant to this Agreement or, in the case of a Default, is cured within any period of cure expressly provided for in this Agreement; and an Event of Default shall "continue" or be "continuing" until such Event of Default has been waived in writing by the Required Lenders. Any Lien referred to in this Agreement or any of the Loan Documents as having been created in favor of Agent, any agreement entered into by Agent pursuant to this Agreement or any of the Loan Documents, any payment made by or to or funds received by Agent pursuant to or as contemplated by this Agreement or any of the Loan Documents, or any act taken or omitted to be taken by Agent, shall, unless otherwise expressly provided, be created, entered into, made or received, or taken or omitted, for the benefit or account of Agent and Lenders. Wherever the phrase "to the best of any Loan Party's knowledge" or words of similar import relating to the knowledge or the awareness of any Loan Party are used in this Agreement or Loan Documents, such phrase shall mean and refer to (i) the actual knowledge of a senior officer of such Loan Party or (ii) the knowledge that a senior officer would have obtained if he had engaged in good faith and diligent performance of his duties, including the making of such reasonably specific inquiries as may be necessary of the employees or agents of such Loan Party and a good faith attempt to ascertain the existence or accuracy of the matter to which such phrase relates. All covenants hereunder shall be given independent

effect so that if a particular action or condition is not permitted by any of such covenants, the fact that it would be permitted by an exception to, or otherwise within the limitations of, another covenant shall not avoid the occurrence of a Default if such action is taken or condition exists. In addition, all representations and warranties hereunder shall be given independent effect so that if a particular representation or warranty proves to be incorrect or is breached, the fact that another representation or warranty concerning the same or similar subject matter is correct or is not breached will not affect the incorrectness of a breach of a representation or warranty hereunder.

1.5. Divisions.

For all purposes under this Agreement and the other Loan Documents, in connection with any division or plan of division under Delaware law (or any comparable event under a different jurisdiction's laws): (a) if any asset, right, obligation or liability of any Person becomes the asset, right, obligation or liability of a different Person, then it shall be deemed to have been transferred from the original Person to the subsequent Person, and (b) if any new Person comes into existence, such new Person shall be deemed to have been organized on the first date of its existence by the holders of its Equity Interests at such time.

II. ADVANCES, PAYMENTS.

2.1. Revolving Advances and Swingline Loans.

(a) <u>Revolving Advances</u>. Subject to the terms and conditions set forth in this Agreement (including, without limitation, <u>Section 2.1(b)</u> and <u>Article VIII)</u>, each Lender, severally and not jointly, will make Revolving Advances to Borrower during the Term in aggregate amounts outstanding at any time not to exceed such Lender's Commitment Percentage of the Maximum Revolving Advance Amount <u>minus</u>, in either case, the sum of (1) the outstanding amount of the Revolving Advances and Swingline Loans <u>plus</u> (2) the Letter of Credit Reserve.

The Revolving Advances may be evidenced by one or more secured promissory notes (each, a "Revolving Credit Note") executed by Borrower in favor of each Lender in the amount of such Lender's Commitment Percentage of the Revolving Advances and shall be in substantially in the form attached hereto as Exhibit C.

(b) [Reserved].

(c) Swingline Loans. (i) Subject to the requirements of this clause (c), so long as the aggregate amount of outstanding Swingline Loans plus the amount of the requested Swingline Loan does not exceed the Maximum Swingline Loan Amount, Borrower may request that Swingline Lender make available to Borrower by transfer of immediately available funds a Swingline Loan. The Swingline Loans shall be advanced by Agent (subject to the requirements of Section 8.2 hereof) as Domestic Rate Loans and shall not exceed in the aggregate at any time outstanding the Maximum Swingline Loan Amount. In the event that on any Business Day, Borrower desires that all or any portion of the outstanding Swingline Loans should be reduced in whole or in part, Borrower shall promptly notify Agent to that effect and indicate the portion of the Swingline Loans to be reduced. Borrower hereby agrees that it shall notify Agent to reduce the outstanding Swingline Loans to \$0 at least once every week and, in any event, at any time that the aggregate outstanding principal amount of Swingline Loan equals the Maximum Swingline Loan Amount. Swingline Lender shall notify Agent to reduce the outstanding Swingline Loans to \$0 by conversion of such Swingline Loans to Revolving Advances as described in sub-clause (ii) of this Section 2.1(c), at least once each week if Borrower fails to do so. Agent agrees to promptly transmit to Lenders the information contained in each notice received by Agent from Borrower or Swingline Lender and shall concurrently notify Lenders of each Lender's Commitment Percentage of

the obligation to make a Revolving Advance to repay the Swingline Loan (or portion thereof). In no event shall the aggregate outstanding Advances exceed the Maximum Revolving Advance Amount.

(i)—(ii)—Each of the Lenders hereby unconditionally and irrevocably agrees to fund to Agent for the benefit of Swingline Lender, in lawful money of the United States and in same day funds, not later than 1:00 p.m. (New York time) on the Settlement Date, such Lender's Commitment Percentage of a Revolving Advance (which Revolving Advance shall be a Domestic Rate Loan and shall be deemed to be requested by Borrower) in the principal amount of such portion of the Swingline Loans which is required to be paid to Swingline Lender under this Section 2.1(c) (regardless of whether the conditions precedent thereto set forth in Article VIII are then satisfied and whether or not Borrower has provided a Notice of Borrowing under Section 2.2 and whether or not any Default or Event of Default exists or all or any of the Advances have been accelerated, but subject to the other provisions of this Section 2.1(c)). The proceeds of any such Revolving Advance shall be immediately paid over to Agent for the benefit of Swingline Lender for application to the Swingline Loan.

2.2. Procedure for Borrowing.

- (a) Borrower may notify by delivery of a Notice of Borrowing to Agent prior to 11:00 a.m. (New York time) on a Business Day of Borrower's request to incur, on that day, or in the alternative, on the immediately succeeding Business Day thereafter (as specified in the Notice of Borrowing), a Revolving Advance hereunder. Any amount required to be paid as interest hereunder, or as fees or other charges under this Agreement or any other agreement with Agent, any Lender and/or the Issuer, or with respect to any other Obligation, which shall become due, if not otherwise paid when due, shall be deemed a request for a Revolving Advance to be maintained as a Domestic Rate Loan as of the date such payment is due, in the amount required to pay in full such interest, fee, charge or Obligation under this Agreement, or any other agreement with Agent, any Lender and/or the Issuer and such request shall be irrevocable.
- (b) Notwithstanding the provisions of <u>subsection (a)</u> above, in the event Borrower desires to obtain a <u>Eurodollar RateSOFR</u> Loan, Borrower shall deliver a Notice of Borrowing to Agent by no later than 11:00 a.m. (New York time) on the day which is three (3) <u>U.S. Government Securities</u> Business Days prior to the date such <u>Eurodollar RateSOFR</u> Loan is to be borrowed, specifying (i) the date of the proposed borrowing (which shall be a Business Day), (ii) the type of borrowing and the amount on the date of such Revolving Advance to be borrowed, which amount shall be in a minimum

amount of \$5,000,000 and in integral multiples of \$500,000 in excess thereof, and (iii) the duration of the first Interest Period therefor. Interest Periods for Eurodollar RateSOFR Loans consisting of Revolving Advances shall be for one, three or six months. No Eurodollar RateSOFR Loan shall be made available to Borrower during the continuance of a Default or an Event of Default. After giving effect to each such borrowing, there shall not be outstanding more than seven (7) Eurodollar RateSOFR Loans consisting of Revolving Advances, in the aggregate at any time. Agent shall provide Borrower with a quote of the actual interest rate available for the Eurodollar RateSOFR Loan requested by Borrower, which quote shall be given on the day after such Eurodollar RateSOFR Loan is requested and such quote shall be effective from the day provided by Agent until two (2) Business Days thereafter.

- (c) Subject to the definition of "Interest Period", each Interest Period of a Eurodollar RateSOFR Loan shall commence on the date such Eurodollar RateSOFR Loan is made and shall end on such date as Borrower may elect as set forth in subsection (b)(iii) aboves provided that the exact length of each Interest Period shall be determined in accordance with the practice of the interbank market for offshore Dollar deposits and no Interest Period shall end after the Termination Date.
- (d) Borrower shall elect the initial Interest Period applicable to a **Eurodollar RateSOFR** Loan by its Notice of Borrowing given to Agent pursuant to Section 2.2(b) or by its Notice of Conversion given to Agent pursuant to Section 2.2(e) as the case may be. Borrower shall elect the duration of each succeeding Interest Period by giving irrevocable written notice to Agent of such duration not less than three (3) **U.S. Government Securities** Business Days prior to the last day of the then current Interest Period applicable to such **Eurodollar RateSOFR** Loan. If Agent does not receive timely notice of the Interest Period elected by Borrower, Borrower shall be deemed to have elected to convert to a Domestic Rate Loan subject to Section 2.2(e).
- (e) Provided that no Event of Default shall have occurred and be continuing, Borrower may, on the last Business Day of the then current Interest Period applicable to any outstanding Eurodollar RateSOFR Loan, or on any Business Day with respect to Domestic Rate Loans, convert any such loan into a loan of another type in the same aggregate principal amount; provided that any conversion of a Eurodollar RateSOFR Loan shall be made only on the last Business Day of the then current Interest Period applicable to such Eurodollar RateSOFR Loan. If Borrower desires to convert a loan, Borrower shall give Agent a Notice of Conversion by no later than 11:00 a.m. (New York time) (i) on the day which is three (3) U.S. Government Securities Business Days²-prior to the date on which such conversion is to occur with respect to a conversion from a Domestic Rate Loan to a Eurodollar RateSOFR Loan, or (ii) on the day which is one (1) U.S. Government Securities Business Day prior to the date on which such conversion is to occur with respect to a conversion from a Eurodollar RateSOFR Loan to a Domestic Rate Loan, specifying, in each case, the date of such conversion, the loans to be converted and if the conversion is from a Domestic Rate Loan to any other type of loan, the duration of the first Interest Period therefor. After giving effect to each request for a Eurodollar RateSOFR Loan, there shall not be outstanding more than six (6) Eurodollar RateSOFR Loans consisting of Revolving Advances, in the aggregate.
- (f) At its option and upon written notice given prior to 11:00 a.m. (New York time) three (3) <u>U.S. Government Securities</u> Business Days²-prior to the date of such prepayment, Borrower may prepay the <u>Eurodollar RateSOFR</u> Loans in whole at any time or in part from time to time, without premium or penalty (except amounts which may be owed pursuant to <u>Section 15.5(b)</u>), but with accrued interest on the principal being prepaid to the date of such repayment. Borrower shall specify the date of prepayment of Revolving Advances which are <u>Eurodollar RateSOFR</u> Loans and the amount of such prepayment. In the event that any prepayment of a <u>Eurodollar RateSOFR</u> Loan is required or permitted on a date other than the last Business Day of the then current Interest Period with respect thereto, Borrower shall pay any amounts which may be owed pursuant to <u>Section 15.5(b)</u>.

Notwithstanding any other provision hereof, if any Applicable Law, treaty, regulation or directive, or any change therein or in the interpretation or application thereof, shall make it unlawful for any Lender (for purposes of this Section 2.2(g), the term "Lender" shall include any Lender and the office or branch where any Lender or any corporation or bank controlling such Lender makes or maintains any Eurodollar RateSOFR Loans) to make or maintain its Eurodollar RateSOFR Loans, the obligation of Lenders to make Eurodollar RateSOFR Loans hereunder shall forthwith be cancelled and Borrower shall, if any affected Eurodollar RateSOFR Loans are then outstanding, promptly upon request from Agent, either pay all such affected Eurodollar RateSOFR Loans or convert such affected Eurodollar RateSOFR Loans into Domestic Rate Loans. In the event that (i) any payment of a Eurodollar RateSOFR Loan is required, made or permitted on a date other than the last day of the then current Interest Period applicable thereto (including upon demand by Agent or Lenders), (ii) the conversion of any Eurodollar RateSOFR Loan other than on the last day of the Interest Period applicable thereto, or (iii) the failure to convert, continue, borrow or prepay any Eurodollar RateSOFR Loan on the date specified in any notice delivered pursuant hereto, then, in any such event, Borrower shall compensate the Lenders for the loss, cost and expense attributable to such event in an amount computed as follows: the excess, if any, of (x) the amount of interest that would have accrued on the principal amount of such Eurodollar Rate Loan had such event not occurred, at the interest rate that would have been applicable to such Eurodollar Rate Loan, for the period from the date of such event to the last day of the then current Interest Period therefor (or, in the case of a failure to borrow, convert or continue, for the period that would have been the Interest Period for such Eurodollar Rate Loan), over (y) the amount of interest that would accrue on such principal amount for such period at the interest rate with such Lender (or an affiliate of such Lender) would bid were it to bid, at the commencement of the interest period, for U.S. Dollar deposits of a comparable amount and period from other banks in the London interbank eurodollar market, including any loss, cost or expense arising from the liquidation or redeployment of funds. A certificate of the Lenders delivered to Borrower and setting forth any amount or amounts that the Lenders are entitled to receive pursuant to this paragraph shall be conclusive absent manifest error. Borrower shall pay the Lenders the amount shown as due on any such certificate upon demand.

2.3. Disbursement of Advance Proceeds.

All Advances shall be disbursed from whichever office or other place Agent may designate from time to time and, together with any and all other Obligations of Borrower to Agent or Lenders, shall be charged to Borrower's Account on Agent's books. During the Term, Borrower may use the Revolving Advances by borrowing, prepaying and reborrowing, all in accordance with the terms and conditions hereof. The proceeds of each Revolving Advance requested by Borrower or deemed to have been requested by Borrower under Section 2.2(a) shall, with respect to requested Revolving Advances to the extent Lenders make such Revolving Advances, be made available to Borrower by the close of business on the day so requested by Borrower by way of credit to Borrower's operating account maintained with Agent or such other bank as Borrower may designate following notification to Agent, in immediately available federal funds or other immediately available funds or, with respect to Revolving Advances deemed to have been requested by Borrower, be disbursed to Agent to be applied to the outstanding Obligations giving rise to such deemed request.

2.4. Incremental Loans.

(a) <u>Incremental Term Loans</u>. At any time commencing on the Closing Date until the Termination Date, provided no Default or Event of Default has occurred and is continuing and subject to the conditions set forth in <u>clause (e)</u> below, upon notice to the Agent, Borrower may, from time to time, request the funding of a term loan (but, together with increases in respect of Incremental Revolving Commitments, not more than three (3) increases in the aggregate) from one or more existing Lenders or from other Eligible Assignees reasonably acceptable to the Agent and Borrower (each, an "<u>Incremental</u>

<u>Term Loan</u>"), in an aggregate amount for all such Incremental Term Loans and any Incremental Revolving Commitments, not to exceed \$65,000,000. Any Incremental Term Loan shall be in the amount of at least \$10,000,000 (or such lower amount that represents all remaining availability pursuant to this <u>Section 2.4(a)</u>) and integral multiples of \$5,000,000 in excess thereof (or such lower amount that represents all remaining availability pursuant to this <u>Section 2.4(a)</u>).

- (b) Revolving Loans. At any time during the Term, provided no Default or Event of Default has occurred and is continuing and subject to the conditions set forth in <u>clause (e)</u> below, upon notice to the Agent, Borrower may, from time to time, request one or more increases (but, together with the requests for Incremental Term Loans, not more than three (3) increases in the aggregate) to the Commitments (the "Incremental Revolving Commitment"); provided that, in no event shall the aggregate amount of Incremental Revolving Commitment plus the Incremental Term Loans exceed \$65,000,000. Any Incremental Revolving Commitment shall be in the amount of at least \$5,000,000 (or such lower amount that represents all remaining availability pursuant to this <u>Section 2.4(b)</u>) and integral multiples of \$5,000,000 in excess thereof (or such lower amount that represents all remaining availability pursuant to this <u>Section 2.4(b)</u>).
- (c) Lender Election to Increase; Prospective Lenders. At the time of sending such request, Borrower (in consultation with the Agent) shall specify the time period (such period, the "Election Period") within which each Lender is requested to respond (which Election Period shall in no event be less than ten (10) Business Days from the date of delivery of such request to the Lenders), and the Agent shall promptly thereafter notify each Lender of Borrower's request for such Incremental Term Loan and/or such Incremental Revolving Commitment and the Election Period during which each Lender is requested to respond to such Borrower request; provided that, if such notice indicates that it is conditioned upon the occurrence of a specified event, such request may be revoked if such event does not occur prior to the requested funding date. No Lender shall be obligated to participate in any Incremental Term Loan or any Incremental Revolving Commitment, and each such Lender's determination to participate shall be in such Lender's sole and absolute discretion. Any Lender not responding by the end of such Election Period shall be deemed to have declined to increase its respective Commitment. To the extent Lenders (or their Affiliates) do not agree to provide an Incremental Term Loan or Incremental Revolving Commitment, as applicable, on terms acceptable to Borrower, Borrower may invite any prospective lender that satisfies the criteria of being an "Eligible Assignee" and is reasonably satisfactory to the Agent to become a Lender pursuant to a joinder agreement in form and substance satisfactory to the Agent in connection with the proposed Incremental Term Loan or Incremental Revolving Commitment, as applicable, shall not require the consent of any other Lender (including any other "Lender" that is joining this Agreement to provide all or part of such Incremental Term Loan or Incremental Revolving Commitment, as applicable)).
- (d) <u>Effective Date and Allocations</u>. If the Commitments are increased in accordance with this <u>Section 2.4</u>, the Agent and Borrower shall determine the effective date (the "<u>Increase Effective Date</u>") and the final allocation of such Incremental Revolving Commitment or Incremental Term Loan, as applicable. The Agent shall promptly notify Borrower and the Lenders of the final allocation of such Incremental Revolving Commitment or Incremental Term Loan, as applicable, and the Increase Effective Date.
- (e) Each of the following shall be the only conditions precedent to the making of an Incremental Term Loan or Incremental Revolving Commitment:
- (i) Borrower shall deliver to the Agent a certificate of each Loan Party dated as of the Increase Effective Date (in sufficient copies for each Lender) signed by an Authorized Officer of

each such Loan Party certifying and attaching the resolutions adopted by such Loan Party approving or consenting to such Incremental Revolving Commitment or Incremental Term Loan, as applicable;

- (ii) Each of the conditions precedent set forth in Section 8.2 shall be satisfied;
- (iii) The then applicable financial covenants set forth in <u>Section 6.8</u> hereof shall have been met, immediately after giving effect to the making of the Incremental Term Loan or Incremental Revolving Commitment on a pro forma basis (treating any Incremental Revolving Commitment as fully funded); <u>provided that</u> the Net Leverage Ratio on a pro forma basis (treating any Incremental Revolving Commitment as fully funded) shall be 0.50:1.00 lower than the Net Leverage Ratio then required under <u>Section 6.8(b)</u> (<u>provided</u>, the cash proceeds of any Incremental Term Loan or Incremental Revolving Loan then being incurred shall not be netted from the numerator in the Net Leverage Ratio for purposes of calculating the Net Leverage Ratio);
- (iv) Borrower shall have delivered to the Agent a certificate certifying as to compliance with the requirements of <u>clauses</u> (<u>ii)</u> and (<u>iii)</u> above, together with all reasonably detailed calculations evidencing compliance with <u>clause</u> (<u>iii)</u> above;
- (v) Borrower shall (x) deliver to any Lender providing an increase in the Commitments hereunder (or any new Lender providing such Commitment) any Notes requested by such Lender in connection with the making of such increased or new Commitment, and (y) have executed any amendments to this Agreement and the other Loan Documents as may be required by the Agent to effectuate the provisions of this Section 2.4, including, if applicable, any amendment that may be necessary to ensure and demonstrate that the Liens and security interests granted by the Loan Documents are perfected under the UCC or other applicable law to secure the Obligations in respect of the Incremental Term Loans and Incremental Revolving Commitment;
- (vi) With respect to any Incremental Revolving Commitment, Borrower shall prepay any Revolving Advances outstanding on the Increase Effective Date (and pay any additional amounts required pursuant to <u>Section 2.2(f)</u>) to the extent necessary to keep the outstanding Revolving Advances ratable with any revised Commitment Percentages resulting from any non-ratable increase in the Commitments undertaken pursuant to this <u>Section 2.4</u>.
- (f) <u>Distribution of Revised Commitments Schedule</u>. The Agent shall promptly distribute to the parties an amended <u>Schedule 1.1</u> (which shall be deemed incorporated into this Agreement), to reflect any such changes in the Commitments of the existing Lenders, or the addition of any new Lenders and their Commitment amounts, and the respective Commitment Percentages resulting therefrom.
 - (g) <u>Conflicting Provisions</u>. This Section shall supersede any provisions in <u>Sections 2.13</u> or <u>15.2</u> to the contrary.

(h) Treatment.

(i) The Incremental Revolving Commitments and any additional Revolving Advances made available pursuant to any such Incremental Revolving Commitment shall be treated on the same terms (including with respect to pricing and maturity date, but excluding any commitment, arrangement, upfront or similar fees, which shall be determined by Borrower and the lenders providing such Incremental Revolving Commitments) as, and made pursuant to the same documentation as is applicable to, the original Revolving Commitment and the original revolving facility, and shall be entitled to all the benefits afforded by, this Agreement and the other Loan Documents, and shall, without limiting

the foregoing, benefit equally and ratably from any guarantees and the security interests created by the Loan Documents and will be secured on a pari passu basis by the same Collateral. Borrower and the other Loan Parties shall take any actions reasonably required by the Agent to ensure and demonstrate that the Liens and security interests granted by the Loan Documents continue to be perfected under the UCC or otherwise after giving effect to the establishment of any such new Incremental Revolving Commitment and additional Revolving Advances.

- (ii) The Incremental Term Loans shall be treated as agreed by the Lenders providing the Incremental Term Loans hereunder (or any new Lender providing an Incremental Term Loan) and the other Lenders in accordance with Section 15.2, and shall be entitled to all the benefits afforded by this Agreement and the other Loan Documents, and shall, without limiting the foregoing, benefit equally and ratably from any guarantees and the security interests created by the Loan Documents and will be secured on a pari passu or junior basis by the same Collateral; provided that, in any case, no Incremental Term Loan shall have a final maturity date sooner than the Termination Date. Borrower and the other Loan Parties shall take any actions reasonably required by the Agent to ensure and demonstrate that the Liens and security interests granted by the Loan Documents continue to be perfected under the UCC or otherwise after giving effect to the establishment of any such new Incremental Term Loans.
- (i) <u>Effect of Increase</u>. Upon the increase in the Maximum Revolving Advance Amount under this <u>Section 2.4</u>, all references in this Agreement and in any other Loan Document (i) to the Maximum Revolving Advance Amount of any Lender shall be deemed to include any increase in such Lender's Revolving Commitment pursuant to this <u>Section 2.4</u>, and (ii) to the Maximum Revolving Advance Amount shall be deemed to include the increase in the Maximum Revolving Advance Amount made pursuant to this <u>Section 2.4</u>.

2.5. Maximum Advances and Letters of Credit.

The aggregate balance of Revolving Advances outstanding <u>plus</u> the Letter of Credit Reserve <u>plus</u> the aggregate principal amount of Swingline Loans outstanding at any time shall not exceed the Maximum Revolving Advance Amount.

2.6. Repayment of Advances.

(a) The Revolving Advances shall be due and payable in full in cash on the Termination Date subject to earlier prepayment as herein provided. The Swingline Loans shall be due and payable in accordance with <u>Section 2.1(c)</u>, subject to earlier prepayment as herein provided.

(b) [Reserved].

(c) Borrower recognizes that the amounts evidenced by checks, notes, drafts or any other items of payment relating to and/or proceeds of Collateral may not be collectible by Agent on the date received. In consideration of Agent's agreement to conditionally credit Borrower's Account as of the Business Day on which Agent receives those items of payment in immediately available funds, Agent agrees that, in computing the charges under this Agreement, all items of payment shall be deemed applied by Agent on account of the Obligations on the Business Day (i) Agent receives such payments via wire transfer or electronic depository check or (ii) in the case of payments received by Agent in any other form, such payment constitutes good funds in Agent's account. Agent is not, however, required to credit Borrower's Account for the amount of any item of payment which is unsatisfactory to Agent and Agent may charge Borrower's Account for the amount of any item of payment which is returned to Lender unpaid. All repayments of Revolving Advances shall be applied first, to any outstanding Swingline Loans, second, to any Revolving Advances maintained as Domestic Rate Loans, and third, to any

Eurodollar Rate SOFR Loans (subject to Section 2.2(f) hereof).

- (d) All payments of principal, interest and other amounts payable hereunder, or under any of the Loan Documents shall be made to Agent at the Payment Office not later than 1:00 p.m. (New York time) on the due date therefor in lawful money of the United States of America in federal funds or other funds immediately available to Agent. Agent shall have the right to effectuate payment on any and all Obligations due and owing hereunder by charging Borrower's Account or by making Revolving Advances maintained as a Domestic Rate Loan as provided in Section 2.2 in the amount of all such Obligations due and owing. In the event Agent charges Borrower's Account or makes any such Revolving Advances, the statement of account required to be delivered to Borrower under Section 2.8 shall reflect all such charges or Revolving Advances which occurred in the prior month.
- (e) Borrower shall pay principal, interest, and all other amounts payable hereunder, or under any related agreement, without any deduction whatsoever, including, but not limited to, any deduction for any setoff or counterclaim.

2.7. Mandatory Prepayments.

- (a) The aggregate balance of Revolving Advances outstanding at any time in excess of the maximum amount of Revolving Advances permitted hereunder shall be immediately due and payable without the necessity of any demand, at the Payment Office, whether or not a Default or Event of Default has occurred, together with any amounts which may be due under <u>Section 15.5(b)</u> if a <u>Eurodollar RateSOFR</u> Loan is required to be prepaid as a consequence of this section.
- (b) Within one Business Day of the date of receipt by Borrower of the proceeds of any Cure Amount pursuant to <u>Section 10.15</u>, Borrower shall prepay the outstanding Advances in an amount equal to 100% of such proceeds.

2.8. Statement of Account.

Agent shall maintain, in accordance with its customary procedures, a loan account (the "Borrower's Account") in the name of Borrower in which shall be recorded the date and amount of each Advance made by Lenders and the date and amount of each payment in respect thereof; provided, however, the failure by Agent to record the date and amount of any Advance shall not adversely affect Agent or any Lender. Each month, Agent shall send to Borrower a statement showing the accounting for the Advances made, payments made or credited in respect thereof, and other transactions between Lenders and Borrower during such month. The monthly statements shall be deemed correct and binding upon Borrower in the absence of manifest error and shall constitute an account stated between Lenders and Borrower unless Agent receives a written statement of Borrower's specific exceptions thereto within thirty (30) days after such statement is received by Borrower. The records of Agent with respect to Borrower's Account shall be conclusive evidence absent manifest error of the amounts of Advances and other charges thereto and of payments applicable thereto.

2.9. Letters of Credit.

Subject to the terms and conditions hereof, Agent shall issue or cause the issuance of Standby Letters of Credit (collectively, "<u>Letters of Credit</u>") by Issuer on behalf of Borrower <u>provided</u>, <u>however</u>, that Agent will not be required to issue or cause to be issued any Letters of Credit to the extent that the Maximum Undrawn Amount of such Letters of Credit would then cause the sum of (i) the outstanding Revolving Advances and Swingline Loans <u>plus</u> (ii) the Letter of Credit Reserve (after giving effect to such issuance or creation) to exceed the Maximum Revolving Advance Amount; and <u>provided further</u>,

that, notwithstanding anything in this Agreement to the contrary, Issuer shall have the right not to issue a Letter of Credit if the issuance of such Letter of Credit would violate one or more policies of Issuer applicable to letters of credit generally. The Maximum Undrawn Amount of outstanding Letters of Credit shall not exceed \$5,000,000 in the aggregate at any time. All disbursements or payments related to Letters of Credit shall be deemed to be Domestic Rate Loans consisting of Revolving Advances and shall bear interest at the Revolving Interest Rate for Domestic Rate Loans; Letters of Credit that have not been drawn upon shall not bear interest.

2.10. <u>Issuance of Letters of Credit</u>.

- (a) Borrower may request Agent to issue or cause the issuance of a Letter of Credit by delivering to Agent at the Payment Office, Issuer's standard form of letter of credit and security agreement and standard form of letter of credit application (collectively, the "Letter of Credit Application") and any draft if applicable, completed to the satisfaction of Agent; and such other certificates, documents and other papers and information as Agent or Issuer may reasonably request. Agent shall use its best efforts to cause all Letters of Credit requested by Borrower and approved by Agent in accordance with the terms of this Agreement to be issued no later than three (3) Business Days after the day so requested by Borrower.
- Each Letter of Credit shall, among other things, (i) provide for the payment of sight drafts or acceptances of issuance drafts when presented for honor thereunder in accordance with the terms thereof and when accompanied by the documents described therein and (ii) have an expiry date not later than twelve (12) months after such Letter of Credit's date of issuance. With respect to clause (ii) above, in no event shall any Letters of Credit issued hereunder have an expiry date later than the Termination Date unless Borrower provides cash collateral equal to not less than one hundred five percent (105%) of the face amount thereof to be held by Agent pursuant to a cash collateral agreement in form and substance satisfactory to Agent. All Letters of Credit shall be subject to the laws or rules designated in such Letter of Credit, or if no laws or rules are designated, the International Standby Practices (ISP98 – International Chamber of Commerce Publication Number 590) (the "ISP98 Rules") and, as to matters not governed by the ISP98 Rules, the laws of the State of New York and applicable United States Federal law. If, at Borrower's request, the Letter of Credit expressly chooses a letter of credit governing rule that is not the ÎSP98 Rules or a state or country law other than New York state law and United States Federal law or is silent with respect to the choice of the ISP98 Rules or a governing law, neither Agent nor Issuer shall be liable for any payment, cost, expense or loss resulting from any action or inaction taken by Agent or Issuer if such action or inaction is or would be justified under the ISP98 Rules, New York law, applicable United States Federal law or the law governing the Letter of Credit. Borrower agrees that for matters not addressed by the ISP98 Rules, the Letter of Credit shall be subject to and governed by the laws of the State of New York and applicable United States Federal laws. If, at Borrower's request, the Letter of Credit expressly chooses a state or country law other than New York state law and United States Federal law or is silent with respect to the choice of the ISP98 Rules or a governing law, Issuer shall not be liable for any payment, cost, expense or loss resulting from any action or inaction taken by Issuer if such action or inaction is or would be justified under the ISP98 Rules, New York law, applicable United States Federal law or the law governing the Letter of Credit.
 - (c) Agent shall use its reasonable efforts to notify Lenders of the request by Borrower for a Letter of Credit hereunder.

2.11. Requirements for Issuance of Letters of Credit.

(a) In connection with the issuance of any Letter of Credit, Borrower shall indemnify, save and hold Agent, each Lender and the Issuer harmless from any loss, cost, expense or

liability, including, without limitation, payments made by Agent, any Lender or the Issuer and expenses and, subject to Section 15.5(c), reasonable attorneys' fees incurred by Agent, any Lender or the Issuer arising out of, or in connection with, any Letter of Credit to be issued or created for Borrower. Borrower shall be bound by Agent's or Issuer's regulations and good faith interpretations of any Letter of Credit issued or created for Borrower's Account, although this interpretation may be different from its own; and, neither Agent, nor any Lender, nor the Issuer nor any of their correspondents shall be liable for any error, negligence, or mistakes, whether of omission or commission, in following Borrower's instructions or those contained in any Letter of Credit or of any modifications, amendments or supplements thereto or in issuing or paying any Letter of Credit except for Agent's, any Lender's, the Issuer's or such correspondents' willful misconduct or gross negligence (as determined by a court of competent jurisdiction in a final and non-appealable judgment).

- (b) Borrower shall authorize and direct the Issuer of a Letter of Credit to deliver to Agent all related payment/acceptance advices, to deliver to Agent all instruments, documents, and other writings and property received by Issuer pursuant to the Letter of Credit and to accept and rely upon Agent's instructions and agreements with respect to all matters arising in connection with the Letter of Credit or the application therefor. With respect to documents presented which appear on their face to be in substantial compliance with the terms of a Letter of Credit, Issuer may, in its sole discretion, either accept and make payment upon such documents without responsibility for further investigation, regardless of any notice or information to the contrary, or refuse to accept and make payment upon such documents if such documents are not in strict compliance with the terms of such Letter of Credit.
- (c) In connection with all Letters of Credit issued or caused to be issued by Agent under this Agreement, Borrower hereby appoints Agent, or its designee, as its attorney, with full power and authority: (i) to sign and/or endorse Borrower's name upon any warehouse or other receipts, letter of credit applications and acceptances; (ii) to sign Borrower's name on bills of lading; and (iii) to complete in Borrower's name or Agent's, or in the name of Agent's designee, any order, sale or transaction, obtain the necessary documents in connection therewith, and collect the proceeds thereof. None of Agent, Issuer nor any of their respective attorneys will be liable for any acts or omissions nor for any error of judgment or mistakes of fact or law, except for Agent's, Issuer's or their respective attorney's willful misconduct or gross negligence (as determined by a court of competent jurisdiction in a final and non-appealable judgment). This power, being coupled with an interest, is irrevocable as long as any Letters of Credit remain outstanding.
- (d) Each Lender shall to the extent of the amount equal to the product of such Lender's Commitment Percentage times the aggregate amount of all unreimbursed reimbursement obligations arising from disbursements made or obligations incurred with respect to the Letters of Credit be deemed to have irrevocably purchased an undivided participation in (i) each such unreimbursed reimbursement obligation, and (ii) each Revolving Advance made as a consequence of the issuance of a Letter of Credit and all disbursements thereunder, in each case in an amount equal to such Lender's applicable Commitment Percentage times the outstanding amount of the Letters of Credit and disbursements thereunder. In the event that at the time a disbursement is made the unpaid balance of Revolving Advances exceeds or would exceed, with the making of such disbursement, the amount permitted under Section 2.1(a), and such disbursement is not reimbursed by Borrower within one (1) Business Day, Agent shall promptly notify each Lender and upon Agent's demand each Lender shall pay to Agent such Lender's proportionate share of such unreimbursed costs and expenses relating to such unreimbursed disbursement. Upon receipt by Agent of a repayment from Borrower of any amount disbursed by Agent for which Agent had already been reimbursed by Lenders, Agent shall deliver to each Lender that Lender's pro rata share of such repayment. Each Lender's participation commitment shall continue until the last to occur of any of the following events: (A) Issuer ceases to be obligated to issue or cause to be issued Letters of Credit hereunder; (B) no Letters of Credit issued hereunder remain outstanding and

uncancelled or (C) all Persons (other than any Loan Party) have been fully reimbursed for all payments made under or relating to Letters of Credit.

2.12. Additional Payments.

Any sums expended by Agent or any Lender due to any Loan Party's failure to perform or comply with its obligations under this Agreement or any other Loan Document including, without limitation, any Loan Party's obligations under <u>Sections 4.2</u>, <u>4.4</u>, <u>4.12</u>, <u>4.13</u>, <u>4.14</u> and <u>6.1</u>, may be charged to Borrower's Account as a Revolving Advance maintained as a Domestic Rate Loan and added to the Obligations.

2.13. Manner of Borrowing and Payment.

- (a) Each borrowing of Revolving Advances shall be advanced according to the applicable Revolving Commitment Percentages of Lenders.
- (b) Each payment (including each prepayment) by Borrower on account of the principal of the Revolving Advances, shall be applied to the Revolving Advances <u>pro rata</u> according to the applicable Revolving Commitment Percentages of Lenders. Except as expressly provided herein, all payments (including prepayments) to be made by Borrower on account of principal, interest and fees shall be made without set-off or counterclaim and shall be made to Agent on behalf of the Lenders to the Payment Office, in each case on or prior to 1:00 p.m. (New York time), in Dollars and in immediately available funds.
- (c) (f) Notwithstanding anything to the contrary contained in Sections 2.13(a) and 2.13(b), commencing with the first Business Day following the Closing Date, each borrowing of Revolving Advances shall be advanced by Agent and each payment by Borrower on account of Revolving Advances shall be applied first to those Revolving Advances advanced by Agent. On or before 1:00 p.m. (New York time) on each Settlement Date commencing with the first Settlement Date following the Closing Date, Agent and Lenders shall make certain payments as follows: (I) if the aggregate amount of new Revolving Advances made by Agent during the preceding week (if any) exceeds the aggregate amount of repayments applied to outstanding Revolving Advances during such preceding week, then each Lender shall provide Agent with funds in an amount equal to its applicable Commitment Percentage of the difference between (w) such Revolving Advances and (x) such repayments and (II) if the aggregate amount of repayments applied to outstanding Revolving Advances during such week exceeds the aggregate amount of new Revolving Advances made during such week, then Agent shall provide each Lender with funds in an amount equal to its applicable Commitment Percentage of the difference between (y) such repayments and (z) such Revolving Advances.
- (i) (ii) Each Lender shall be entitled to earn interest at the applicable Contract Rate on outstanding Advances which it has funded.
- (ii) ___(iii) Promptly following each Settlement Date, Agent shall submit to each Lender a certificate with respect to payments received and Revolving Advances made during the week immediately preceding such Settlement Date. Such certificate of Agent shall be conclusive in the absence of manifest error.
- (d) If any Lender or Participant (a "benefited Lender") shall at any time receive any payment of all or part of its Advances, or interest thereon, or receive any Collateral in respect thereof (whether voluntarily or involuntarily or by set-off) in a greater proportion than any such payment to and Collateral received by any other Lender, if any, in respect of such other Lender's Advances, or interest thereon, and such greater proportionate payment or receipt of Collateral is not expressly permitted

hereunder, such benefited Lender shall purchase for cash from the other Lenders a participation in such portion of each such other Lender's Advances, or shall provide such other Lender with the benefits of any such Collateral, or the proceeds thereof, as shall be necessary to cause such benefited Lender to share the excess payment or benefits of such Collateral or proceeds ratably with each of Lenders; provided, however, that if all or any portion of such excess payment or benefits is thereafter recovered from such benefited Lender, such purchase shall be rescinded, and the purchase price and benefits returned, to the extent of such recovery, but without interest. Each Lender so purchasing a portion of another Lender's Advances may exercise all rights of payment (including, without limitation, rights of set-off) with respect to such portion as fully as if such Lender were the direct holder of such portion.

(e) Unless Agent shall have been notified by telephone, confirmed in writing, by any Lender that such Lender will not make the amount which would constitute its applicable Commitment Percentage of the Advances available to Agent, Agent may (but shall not be obligated to) assume that such Lender shall make such amount available to Agent on the next Settlement Date and, in reliance upon such assumption, make available to Borrower a corresponding amount. Agent will promptly notify Borrower of its receipt of any such notice from a Lender. If such amount is made available to Agent on a date after such next Settlement Date, such Lender shall pay to Agent on demand an amount equal to the product of (i) the daily average Federal Funds Rate (computed on the basis of a year of 360 days) during such period as quoted by Agent, times (ii) such amount, times (iii) the number of days from and including such Settlement Date to the date on which such amount becomes immediately available to Agent. A certificate of Agent submitted to any Lender with respect to any amounts owing under this <u>paragraph (e)</u> shall be conclusive, in the absence of manifest error. If such amount is not in fact made available to Agent by such Lender within three (3) Business Days after such Settlement Date, Agent shall be entitled to recover such an amount, with interest thereon at the rate per annum then applicable to such Revolving Advances hereunder, on demand from Borrower; <u>provided</u>, <u>however</u>, that Agent's right to such recovery shall not prejudice or otherwise adversely affect Borrower's rights (if any) against such Lender.

2.14. Mandatory Prepayments.

- (a) When any Loan Party sells or otherwise disposes of any Collateral, other than Inventory in the ordinary course of business, then, unless a Reinvestment Notice shall be delivered in respect thereof within ten (10) days of such Loan Party's receipt of proceeds of such sale or other disposition, Borrower shall repay the Advances made to Borrower in an amount equal to the Net Cash Proceeds, such repayments to be made promptly but in no event more than one (1) Business Day following receipt of such Net Cash Proceeds, and until the date of payment, such proceeds shall be held in trust for Agent; provided that notwithstanding the foregoing, on each Reinvestment Prepayment Date, an amount equal to the Reinvestment Prepayment Amount with respect to the relevant Reinvestment Event shall be applied toward the prepayment of the Advances and other amounts as set forth in this Section 2.14(a); and provided further, that so long as no Default or Event of Default shall exist or would result therefrom, and such Net Cash Proceeds of such a sale or such a disposition do not exceed (i) \$5,000,000 in any single transaction or series of related sales or dispositions or (ii) \$10,000,000 in the aggregate for all such sales or dispositions in any fiscal year, then no such prepayment shall be required. The foregoing shall not be deemed to be implied consent to any such sale otherwise prohibited by the terms and conditions hereof. Such repayments shall be applied to the Advances in such order as Agent may determine, subject to Borrower's ability to reborrow Revolving Advances in accordance with the terms hereof.
- (b) Subject to the provisions of <u>Section 4.11</u>, Agent shall apply the proceeds of any insurance settlements from casualty losses which are received by Agent to the Advances in such order as Agent may determine, subject to Borrower's ability to reborrow Revolving Advances in accordance with the terms hereof.

(c) Unless a Reinvestment Notice shall be delivered in respect thereof within ten (10) days of any Loan Party's receipt of proceeds of any Extraordinary Receipts, Borrower shall repay the Advances made to Borrower in an amount equal to the Net Cash Proceeds of such Extraordinary Receipts, such repayments to be made promptly but in no event more than one (1) Business Day following receipt of such Net Cash Proceeds, and until the date of payment, such proceeds shall be held in trust for Agent; provided that notwithstanding the foregoing, on each Reinvestment Prepayment Date, an amount equal to the Reinvestment Prepayment Amount with respect to the relevant Reinvestment Event shall be applied toward the prepayment of the Advances and other amounts as set forth in this Section 2.14(c). Such repayments shall be applied to the Advances in such order as Agent may determine subject to Borrower's ability to reborrow Revolving Advances in accordance with the terms hereof.

(d) [Reserved].

(e) If any Loan Party receives any proceeds from the issuance or incurrence of any Indebtedness (other than Permitted Indebtedness), Borrower shall repay the Advances made to Borrower in an amount equal to the net proceeds of such issuance or incurrence (i.e., gross proceeds less the reasonable costs of such issuance or incurrence), such repayments to be made promptly but in no event more than one (1) Business Day following receipt of such net proceeds, and until the date of payment, such proceeds shall be held in trust for Agent. The foregoing shall not be deemed to be implied consent to any such sale or issuance otherwise prohibited by the terms and conditions hereof. Such repayments shall be applied to the other Advances in such order as Agent may determine, subject to Borrower's ability to reborrow Revolving Advances in accordance with the terms hereof.

2.15. Use of Proceeds.

Borrower shall apply the proceeds of Advances to (i) finance Permitted Acquisitions, (ii) pay fees and expenses relating to the Transactions, (iii) provide for its working capital needs and reimburse drawings under Letters of Credit and (iv) for other general corporate purposes.

2.16. <u>Defaulting Lender</u>.

- (a) Notwithstanding anything to the contrary contained herein, in the event any Lender is a Defaulting Lender, all rights and obligations hereunder of such Lender and of the other parties hereto shall be modified to the extent of the express provisions of this <u>Section 2.16</u> while such Lender is a Defaulting Lender.
- (b) Advances shall be incurred <u>pro rata</u> from Lenders (the "<u>Non-Defaulting Lenders</u>") which are not Defaulting Lenders based on their respective Commitment Percentages, and no Commitment Percentage of any Lender or any <u>pro rata</u> share of any Advances required to be advanced by any Lender shall be increased as a result of such Lender being a Defaulting Lender. Amounts received in respect of principal of any type of Advances shall be applied to reduce the applicable Advances of each Lender (other than any Defaulting Lender) <u>pro rata</u> based on the aggregate of the outstanding Advances of that type of all Lenders at the time of such application; <u>provided</u> that Agent shall not be obligated to transfer to a Defaulting Lender any payments received by Agent for the Defaulting Lender's benefit, nor shall a Defaulting Lender be entitled to the sharing of any payments hereunder (including any principal, interest or fees). Amounts payable to a Defaulting Lender shall instead be paid to or retained by Agent. Agent may hold and, in its discretion, re-lend to Borrower the amount of such payments received or retained by it for the account of such Defaulting Lender.
- (c) A Defaulting Lender shall not be entitled to give instructions to Agent or to approve, disapprove, consent to or vote on any matters relating to this Agreement and the other Loan

Documents other than (i) any such amendment, waiver or consent to the extent that it relates to any matter that disproportionately affects any Defaulting Lender and (ii) any of the matters governed by Section 15.2(a)(i) through (iii) that affect such Lender. All amendments, waivers and other modifications of this Agreement and the other Loan Documents may be made without regard to a Defaulting Lender and, for purposes of the definition of "Required Lenders", a Defaulting Lender shall be deemed not to be a Lender and not to have either Advances outstanding or a Commitment Percentage.

- (d) Other than as expressly set forth in this <u>Section 2.16</u>, the rights and obligations of a Defaulting Lender (including the obligation to indemnify Agent) and the other parties hereto shall remain unchanged. Nothing in this <u>Section 2.16</u> shall be deemed to release any Defaulting Lender from its obligations under this Agreement and the other Loan Documents, shall alter such obligations, shall operate as a waiver of any default by such Defaulting Lender hereunder, or shall prejudice any rights which Borrower, Agent or any Lender may have against any Defaulting Lender as a result of any default by such Defaulting Lender hereunder.
- (e) If Borrower, Agent, the Swingline Lender and the Issuer agree in writing that a Lender is no longer a Defaulting Lender, Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein (which may include arrangements with respect to any cash collateral), that Lender will, to the extent applicable, purchase at par that portion of outstanding Advances of the other Lenders or take such other actions as Agent may determine to be necessary to cause the Advances and funded and unfunded participations in Letters of Credit and Swingline Loans to be held <u>pro rata</u> by the Lenders in accordance with the Commitments under the applicable facility, whereupon such Lender will cease to be a Defaulting Lender; <u>provided</u> that no adjustment will be made retroactively with respect to fees accrued or payments made by or on behalf of Borrower while that Lender was a Defaulting Lender; and <u>provided</u>, <u>further</u>, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.
- (f) So long as any Lender is a Defaulting Lender, (i) the Swingline Lender shall not be required to fund any Swingline Loans unless it is satisfied that it will have no Fronting Exposure after giving effect to such Swingline Loan and (ii) no Issuer shall be required to issue, extend, renew or increase any Letter of Credit unless it is satisfied that it will have no Fronting Exposure after giving effect thereto.

III. INTEREST AND FEES.

3.1. <u>Interest</u>.

Interest on Advances shall be payable to Agent for the benefit of Lenders in arrears on the last day of each calendar quarter with respect to Domestic Rate Loans and, with respect to Eurodollar Rate Loans, at the end of each Interest Period or, for Eurodollar Rate Loans with an Interest Period in excess of three months, on each date which is a three month anniversary date of the commencement of such Eurodollar Rate Loan and at the end of each Interest Periodapplicable Interest Payment Date. Interest charges shall be computed on the actual principal amount of Advances outstanding during the calendar month at a rate per annum equal to (i) with respect to Revolving Advances, the applicable Revolving Interest Rate and (ii) with respect to Swingline Loans, the Revolving Interest Rate applicable to Revolving Advances maintained as Domestic Rate Loans (as applicable, the "Contract Rate"). Whenever, subsequent to the date of this Agreement, the Base Rate is increased or decreased, the applicable Contract Rate for Domestic Rate Loans shall be similarly changed without notice or demand of any kind by an amount equal to the amount of such change in the Base Rate during the time such change or changes remain in effect. Upon and after the occurrence of any (x) Default under

<u>Section 10.1(a)</u> or Event of Default under <u>Section 10.1(a)</u>, and during the continuation thereof, all overdue Obligations shall bear interest at the applicable Contract Rate <u>plus</u> two percent (2.0%) per annum (the "<u>Default Rate</u>"), and (y) any other Default or Event of Default, all Obligations shall bear interest at the Default Rate.

3.2. Letter of Credit Fees; Cash Collateral.

- (a) Borrower shall pay (w) to Agent, for the ratable benefit of Lenders, fees for each Letter of Credit for the period from and excluding the date of issuance of same to and including the date of expiration or termination, equal to the average daily face amount of each outstanding Letter of Credit multiplied by the Letter of Credit Fee Percentage, the fees under this Section 3.2(a)(w) to be calculated on the basis of a 360-day year for the actual number of days elapsed and to be payable quarterly in arrears on the first day of each quarter and on the last day of the Term, (x) to Agent for the benefit of Issuer, any and all customary fees and expenses in connection with any Letter of Credit, including, without limitation, in connection with the issuance, amendment or renewal of any such Letter of Credit, and (y) to Agent for the ratable benefit of Lenders, a fee equal to the greater of (i) one-quarter of one percent (0.25%) of the amount of each draft negotiated with respect to any Letter of Credit upon the payment thereof and (ii) \$100 (all of the foregoing fees, the "Letter of Credit and Guarantee Fees"). All such charges shall be deemed earned in full on the date when the same are due and payable hereunder and shall not be subject to rebate or pro-ration upon the termination of this Agreement for any reason. Any such charge in effect at the time of a particular transaction shall be the charge for that transaction, notwithstanding any subsequent change in Issuer's prevailing charges for that type of transaction. Upon and after the occurrence of an Event of Default, and during the continuation thereof, Agent may, and at the direction of the Required Lenders shall, increase the Letter of Credit and Guarantee Fees by two percent (2.0%) per annum. All Letter of Credit and Guarantee Fees payable hereunder shall be deemed earned in full on the date when the same are due and payable hereunder and shall not be subject to rebate or pro-ration upon the termination of this Agreement for any reason.
- (b) On demand from Agent or the Required Lenders at any time following (x) the occurrence of an Event of Default (whether or not such Event of Default is continuing) or (y) any termination of the Lenders' commitment to make Revolving Advances, Borrower will cause cash to be deposited and maintained in an account with Agent, as cash collateral for the Obligations, in an amount equal to one hundred and five percent (105%) of the Letter of Credit Reserve, and Borrower hereby irrevocably authorizes Agent, in its discretion, on Borrower's behalf and in Borrower's name, to open such an account and to make and maintain deposits therein, or in an account opened by Borrower, in the amounts required to be made by Borrower, out of the proceeds of Receivables or other Collateral or out of any other funds of Borrower coming into any Lender's possession at any time. Agent will invest such cash collateral (less applicable reserves) in such short-term money-market items as to which Agent and Borrower mutually agree and the net return on such investments shall be credited to such account and constitute additional cash collateral. Borrower may not withdraw amounts credited to any such account except upon payment and performance in full in cash of all Obligations (other than (A) contingent indemnification obligations as to which no claim has been asserted and (B) obligations and liabilities under Lender-Provided Hedges and Bank Product Obligations) and termination of this Agreement.

3.3. <u>Unused Commitment Fee</u>.

Borrower shall pay to Agent, for the ratable benefit of the Lenders, a fee in an amount equal to (x) the Maximum Revolving Advance Amount $\underline{\text{minus}}$ the sum of (1) the average daily balance of the Revolving Advances during the immediately preceding quarter, plus, (2) the average daily face amount of the Letter of Credit Reserve during the immediately preceding quarter, multiplied by the applicable Commitment Fee Percentage. Such fee to be calculated on the basis of a three hundred sixty (360) day year for the actual number of days elapsed and to be payable quarterly in arrears on the first day of each

quarter following the Closing Date. For the avoidance of doubt, outstanding Swingline Loans shall not be counted towards or considered usage of Revolving Advances for purposes of determining the unused commitment fee.

3.4. <u>Fee Letter</u>. Borrower shall pay the amounts required to be paid in the Fee Letter in the manner and at the times required by the Fee Letter.

3.5. Computation of Interest and Fees.

Interest and fees hereunder shall be computed on the basis of a year of 360 days and for the actual number of days elapsed except that, with respect to Advances the rate of interest on which is calculated on the basis of the Base Rate, the interest thereon shall be calculated on the basis of a 365- (or 366-, as the case may be) day year for the actual days elapsed. If any payment to be made hereunder becomes due and payable on a day other than a Business Day, the due date thereof shall be extended to the next succeeding Business Day and interest thereon shall be payable at the applicable Contract Rate during such extension; provided that with respect to Eurodollar RateSOFR Loans, if extending such payment would cause the last day of the applicable Interest Period to be extended into the next calendar month, then the due date for such payment shall be the immediately preceding Business Day.

3.6. Maximum Charges.

In no event whatsoever shall interest and other charges charged hereunder exceed the highest rate permissible under any Applicable Law. In the event interest and other charges as computed hereunder would otherwise exceed the highest rate permitted under Applicable Law, such excess amount shall be first applied to any unpaid principal balance owed by Borrower, and if the then remaining excess amount is greater than the previously unpaid principal balance, Lenders shall promptly refund such excess amount to Borrower and the provisions hereof shall be deemed amended to provide for such permissible rate.

3.7. <u>Increased Costs</u>.

In the event that any Applicable Law, treaty or governmental regulation coming into effect after the Closing Date, or any change therein or any change in the interpretation or application thereof after the Closing Date, or compliance by any Lender (for purposes of this Section 3.7, the term "Lender" shall include Agent or any Lender and any corporation or bank controlling Agent or any Lender) and the office or branch where Agent or any Lender (as so defined) makes or maintains any Eurodollar RateSOFR Loans with any request or directive (whether or not having the force of law) from any central bank or other Governmental Body arising after the Closing Date, shall:

- (a) subject Agent or any Lender to any Tax (other than (i) Indemnified Taxes, (ii) Taxes described in clauses (b) through (d) of the definition of Excluded Taxes and (iii) Connection Income Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto;
- (b) impose, modify or hold applicable any reserve, special deposit, assessment or similar requirement against assets held by, or deposits in or for the account of, advances or loans by, or other credit extended by, any office of Agent or any Lender, including (without limitation) pursuant to Regulation D of the Board of Governors of the Federal Reserve System; or
- (c) impose on Agent or any Lender or the London interbank Eurodollar market any other condition with respect to this Agreement or any other Loan Document;

and the result of any of the foregoing is to increase the cost to Agent or any Lender of making, renewing or maintaining its Advances hereunder by an amount that Agent or such Lender reasonably deems to be material or to reduce the amount of any payment (whether of principal, interest or otherwise) in respect of any of the Advances by an amount that Agent or such Lender reasonably deems to be material, then, in any case Borrower shall promptly pay Agent or such Lender, upon its demand, such additional amount as will compensate Agent or such Lender for such additional cost or such reduction, as the case may be, <u>provided</u> that the foregoing shall not apply to increased costs which are reflected in the Adjusted LIBO Rate Term SOFR. Agent or such Lender shall certify the amount of such additional cost or reduced amount to Borrower, and such certification shall be conclusive absent manifest error.

3.8. Basis for Determining Interest Rate Inadequate or Unfair; Benchmark Replacement Setting.

(a) Basis for Determining Interest Rate Inadequate or Unfair. Subject to the provisions set forth in Section 3.8(b) below, in the event that Agent or any Lender shall have determined that; (i) reasonable means do not exist for ascertaining the Adjusted LIBO Rate applicable pursuant to Section 2.2 for any Interest Period; or (ii) Dollar deposits in the relevant amount and for the relevant maturity are not available in the London interbank Eurodollar market, with respect to an outstanding Eurodollar Rate Loan, a proposed Eurodollar Rate Loan, or a proposed conversion of a Domestic Rate Loan into a Eurodollar Rate Loan, then Agent shall give Borrower prompt written, telephonic or telegraphic notice of such determination. If such notice is given, (i) any such requested Eurodollar Rate Loan shall be made as a Domestic Rate Loan, unless Borrower shall notify Agent no later than 10:00 a.m. (New York time) two (2) Business Days prior to the date of such proposed borrowing, that its request for such borrowing shall be cancelled or made as an unaffected type of Eurodollar Rate Loan, (ii) any Domestic Rate Loan or Eurodollar Rate Loan which was to have been converted to an affected type of Eurodollar Rate Loan shall be continued as or converted into a Domestic Rate Loan, or, if Borrower shall notify Agent, no later than 10:00 a.m. (New York time) two (2) Business Days prior to the proposed conversion, shall be maintained as an unaffected type of Eurodollar Rate Loan, and (iii) any outstanding affected Eurodollar Rate Loans shall be converted into a Domestic Rate Loan, or, if Borrower shall notify Agent, no later than 10:00 a.m. (New York time) two (2) Business Days prior to the last Business Day of the then current Interest Period applicable to such affected Eurodollar Rate Loan, shall be converted into an unaffected type of Eurodollar Rate Loan, on the last Business Day of the then current Interest Period for such affected Eurodollar Rate Loans. Until such notice has been withdrawn, Lenders shall have no obligation to make an affected type of Eurodollar Rate Loan or maintain outstanding affected Eurodollar Rate Loans and Borrower shall have no right to convert a Domestic Rate Loan or an unaffected type of Eurodollar Rate Loan into an affected type of Eurodollar Rate Loan.

(a) Benchmark Replacement.

(b) Benchmark Replacement Setting. Notwithstanding anything to the contrary herein or in any other Loan Document:

(i) Replacing USD LIBOR. On March 5, 2021 the Financial Conduct Authority ("FCA"), the regulatory supervisor of USD LIBOR's administrator ("IBA"), announced in a public statement the future cessation or loss of representativeness of overnight/Spot Next, 1-month, 3-month, 6-month and 12-month USD LIBOR tenor settings. On the earlier of (A) the date that all Available Tenors of USD LIBOR have either permanently or indefinitely ceased to be provided by IBA or have been announced by the FCA pursuant to public statement or publication of information to be no longer representative and (B) the Early Opt-in Effective Date, if, if a Benchmark Transition Event and its related Benchmark Replacement Date have occurred prior

any setting of the then-current Benchmark is USD LIBOR, the Benchmark, then (x) if a Benchmark Replacement is determined in accordance with clause (a) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any other Loan Document in respect of any setting of such Benchmark on such day and all setting and subsequent Benchmark settings without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document. If the and (y) if a Benchmark Replacement is Daily Simple SOFR, all interest payments will be payable on a monthly basis.

(ii) determined in accordance with clause (b) of Replacing Future Benchmarks. Upon the occurrence of a Benchmark Transition Event, the the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace the then currentsuch Benchmark for all purposes hereunder and under any other Loan Document in respect of any Benchmark setting at or after 5:00 p.m. (New York City time) on the fifth (5th) Business Day after the date notice of such Benchmark Replacement is provided to the Lenders without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document so long as the Agent has not received, by such time, written notice of objection to such Benchmark Replacement from Lenders comprising the Required Lenders. At any time that the administrator of the then-current Benchmark has permanently or indefinitely ceased to provide such Benchmark or such Benchmark has been announced by the regulatory supervisor for the administrator of such Benchmark pursuant to public statement or publication of information to be no longer representative of the underlying market and economic reality that such Benchmark is intended to measure and that representativeness will not be restored, Borrower may revoke any request for a borrowing of, conversion to or continuation of Advances to be made, converted or continued that would bear interest by reference to such Benchmark until Borrower will be deemed to have converted any such request into a request for a borrowing of conversion to Domestic Rate Loans. During the period referenced in the foregoing sentence, the component of the Base Rate based upon the Benchmark will not be used in any determination of the Base Rate. If the Benchmark Replacement is Daily Simple SOFR, all interest payments will be payable on a monthly basis.

(i) No Hedging Agreement shall be deemed to be a "Loan Document" for purposes of this Section 3.8.

(b) (iii) Benchmark Replacement Conforming Changes. In connection with the implementation and use, administration, adoption or implementation of a Benchmark Replacement; and in consultation with the Borrower, Agent will have the right to make Benchmark Replacement Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Benchmark Replacement Conforming Changes will become effective without any further action or consent of any other party to this Agreement.

(c) (iv) Notices; Standards for Decisions and Determinations. The Agent will promptly notify the Borrower and the Lenders of (Ai) the implementation of any Benchmark Replacement and (Bii) the effectiveness of any Benchmark Replacement Conforming Changes in connection with the use, administration, adoption or implementation of a Benchmark Replacement. Agent will notify the Borrower of (x) the removal or reinstatement of any tenor of a Benchmark pursuant to Section 3.8(d) and (y) the commencement of any Benchmark Unavailability Period. Any determination, decision or election that may be made by the Agent or, if applicable, any Lender (or group of Lenders) pursuant to this Section 3.8(b), including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made in its or their sole discretion and without consent from any other party heretoto

this Agreement or any other Loan Document, except, in each case, as expressly required pursuant to this Section 3.8(b).

(d) (v) Unavailability of Tenor of Benchmark. At Notwithstanding anything to the contrary herein or in any other Loan Document, at any time (including in connection with the implementation of a Benchmark Replacement), (i) if the then-current Benchmark is a term rate (including the Term SOFR or USD LIBOR), then the Agent may remove any tenor Reference Rate) and either (A) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by Agent in its reasonable discretion or (B) the regulatory supervisor for the administrator of such Benchmark that is has provided a public statement or publication of information announcing that any tenor for such Benchmark is not or will not be representative, then Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for any Benchmark settings at or after such time to remove such unavailable or non-representative for Benchmark (including Benchmark Replacement) settings and (ii) the Agent may tenor and (ii) if a tenor that was removed pursuant to clause (i) above either (A) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (B) is not, or is no longer, subject to an announcement that it is not or will not be representative for a Benchmark (including a Benchmark Replacement), then Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for all Benchmark settings at or after such time to reinstate any—such previously removed tenor—for Benchmark (including Benchmark Replacement) settings.

(e) Benchmark Unavailability Period. Upon the Borrower's receipt of notice of the commencement of a Benchmark Unavailability Period, the Borrower may revoke any pending request for a SOFR Borrowing of, conversion to or continuation of SOFR Loans to be made, converted or continued during any Benchmark Unavailability Period and, failing that, the Borrower will be deemed to have converted any such request into a request for an Advance or conversion to Domestic Rate Loans. During a Benchmark Unavailability Period or at any time that a tenor for the then-current Benchmark is not an Available Tenor, the component of Base Rate based upon the then-current Benchmark or such tenor for such Benchmark, as applicable, will not be used in any determination of Base Rate.

3.9. Capital Adequacy.

(a) In the event that Agent or any Lender shall have determined that any Applicable Law, rule, regulation or guideline regarding capital adequacy or liquidity, or any change therein, or any change in the interpretation, application or administration thereof by any Governmental Body, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by Agent or any Lender (for purposes of this Section 3.9, the term "Lender" shall include Agent or any Lender and any corporation or bank controlling Agent or any Lender) and the office or branch where Agent or any Lender (as so defined) makes or maintains any Eurodollar RateSOFR Loans with any request or directive regarding capital adequacy or liquidity (whether or not having the force of law) of any such authority, central bank or comparable agency, has or would have the effect of reducing the rate of return on Agent or any Lender's capital as a consequence of its obligations hereunder to a level below that which Agent or such Lender would have achieved but for such adoption, change or compliance (taking into consideration Agent's and each Lender's policies with respect to capital adequacy or liquidity) by an amount reasonably deemed by Agent or any Lender to be material, then, from time to time, Borrower shall pay upon demand to Agent or such Lender such additional amount or amounts as will compensate Agent or such Lender for such reduction. In determining such amount or amounts, Agent or such Lender may use any reasonable averaging or attribution methods. The protections of this Section 3.9 shall be available to Agent and each Lender regardless of any possible contention of invalidity or inapplicability with respect to the Applicable Law, regulation or condition. Notwithstanding anything herein to the

contrary, for all purposes under this Agreement, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, regulations, guidelines or directives thereunder or issued in connection therewith and (y) all requests, rules, guidelines or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall, in either case, be deemed to have gone into effect after the Closing Date, regardless of the date enacted, adopted or issued.

(b) A certificate of Agent or such Lender setting forth such amount or amounts as shall be necessary to compensate Agent or such Lender with respect to <u>Section 3.9(a)</u> when delivered to Borrower shall be conclusive absent manifest error.

3.10. <u>Taxes</u>.

- (a) For purposes of this Section 3.10, the term "Lender" includes the Issuer and the term "Applicable Law" includes FATCA.
- (b) Any and all payments by or on account of any obligation of any Loan Party under this Agreement or any other Loan Document shall be made without deduction or withholding for any Taxes, except as required by Applicable Law. If any Applicable Law (as determined in the good faith discretion of an applicable Withholding Agent) requires the deduction or withholding of any Tax from any such payment by a Withholding Agent, then the applicable Withholding Agent shall be entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Body in accordance with Applicable Law and, if such Tax is an Indemnified Tax, then the sum payable by the applicable Loan Party shall be increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this Section) the applicable Recipient receives an amount equal to the sum it would have received had no such deduction or withholding been made.
- (c) The Loan Parties shall timely pay to the relevant Governmental Body in accordance with Applicable Law, or at the option of Agent timely reimburse it for the payment of, any Other Taxes.
- (d) The Loan Parties shall jointly and severally indemnify each Recipient, within 10 days after demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section) payable or paid by such Recipient or required to be withheld or deducted from a payment to such Recipient and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Body. Promptly upon having knowledge that any such Indemnified Taxes have been levied, imposed or assessed, and promptly upon notice by the Agent or any Lender, the Loan Parties shall pay such Indemnified Taxes directly to the relevant Governmental Body; provided that neither Agent nor any Lender shall be under any obligation to provide any such notice to the Loan Parties. A certificate as to the amount of such payment or liability delivered to Borrower by a Lender (with a copy to Agent), or by Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error. The Loan Parties hereby indemnify Agent, and shall make payment in respect thereof within ten days after demand therefor, for any amount which a Lender for any reason fails to pay to Agent as required by the Agreement.
- (e) Each Lender shall severally indemnify Agent, within 10 days after demand therefor, for (i) any Indemnified Taxes attributable to such Lender (but only to the extent that any Loan Party has not already indemnified Agent for such Indemnified Taxes and without limiting the obligation

of the Loan Parties to do so), (ii) any Taxes attributable to such Lender's failure to comply with the provisions of Section 15.3 relating to the maintenance of the Participant Register and (iii) any Excluded Taxes attributable to such Lender, in each case, that are payable or paid by Agent in connection with this Agreement or any other Loan Document, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Body. A certificate as to the amount of such payment or liability delivered to any Lender by Agent shall be conclusive absent manifest error. Each Lender hereby authorizes Agent to set off and apply any and all amounts at any time owing to such Lender under this Agreement or any other Loan Document or otherwise payable by Agent to the Lender from any other source against any amount due to Agent under this paragraph (e).

- (f) As soon as practicable after any payment of Taxes by any Loan Party to a Governmental Body pursuant to this <u>Section 3.10</u>, such Loan Party shall deliver to Agent the original or a certified copy of a receipt issued by such Governmental Body evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to Agent.
- (g) (i) Any Lender that is entitled to an exemption from or reduction of withholding Tax with respect to payments made under this Agreement or any other Loan Document shall deliver to Borrower and Agent, at the time or times reasonably requested by Borrower or Agent, such properly completed and executed documentation reasonably requested by Borrower or Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if reasonably requested by Borrower or Agent, shall deliver such other documentation prescribed by Applicable Law or reasonably requested by Borrower or Agent as will enable Borrower or Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in Section 3.10(g)(ii)(A), (ii)(B) and (ii)(D) below) shall not be required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.
 - (i)—(ii) Without limiting the generality of the foregoing, in the event that Borrower is a Borrower that is a U.S. Person,
- (A) any Lender that is a U.S. Person shall deliver to Borrower and Agent on or prior to the date on which such Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of Borrower or Agent), executed copies of IRS Form W-9 certifying that such Lender is exempt from U.S. federal backup withholding tax;
- (B) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to Borrower and Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of Borrower or Agent), whichever of the following is applicable:
 - (1) in the case of a Foreign Lender claiming the benefits of an income tax treaty to which the United States is a party (x) with respect to payments of interest under this Agreement or any other Loan Document, properly completed and duly executed copies of IRS Form W-8BEN establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "interest" article of such tax treaty and (y) with respect to any other applicable payments under this Agreement or any other Loan Document, IRS Form W-8BEN establishing an exemption from, or reduction of, U.S.

federal withholding Tax pursuant to the "business profits" or "other income" article of such tax treaty;

- (2) properly completed and duly executed copies of IRS Form W-8ECI;
- (3) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under Section 881(c) of the Code, (x) a certificate substantially in the form of Exhibit H-1 to the effect that such Foreign Lender is not a "bank" within the meaning of Section 881(c)(3)(A) of the Code, a "10 percent shareholder" of Borrower within the meaning of Section 881(c)(3)(B) of the Code, or a "controlled foreign corporation" described in Section 881(c)(3)(C) of the Code (a "U.S. Tax Compliance Certificate") and (y) properly completed and duly executed copies of IRS Form W-8BEN; or
- (4) to the extent a Foreign Lender is not the beneficial owner, properly completed and duly executed copies of IRS Form W-8IMY, accompanied by IRS Form W-8ECI, IRS Form W-8BEN, a U.S. Tax Compliance Certificate substantially in the form of Exhibit H-2 or Exhibit H-3, IRS Form W-9, and/or other certification documents from each beneficial owner, as applicable; provided that if the Foreign Lender is a partnership and one or more direct or indirect partners of such Foreign Lender are claiming the portfolio interest exemption, such Foreign Lender may provide a U.S. Tax Compliance Certificate substantially in the form of Exhibit H-4 on behalf of each such direct and indirect partner;
- (C) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to Borrower and Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of Borrower or Agent), executed copies of any other form prescribed by Applicable Law as a basis for claiming exemption from or a reduction in U.S. federal withholding Tax, duly completed, together with such supplementary documentation as may be prescribed by Applicable Law to permit Borrower or Agent to determine the withholding or deduction required to be made; and
- (D) if a payment made to a Lender under this Agreement or any other Loan Document would be subject to U.S. federal withholding Tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable), such Lender shall deliver to Borrower and Agent at the time or times prescribed by law and at such time or times reasonably requested by Borrower or Agent such documentation prescribed by Applicable Law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and such additional documentation reasonably requested by Borrower or Agent as may be necessary for Borrower and Agent to comply with their obligations under FATCA and to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this clause (D), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

(ii) (iii) On or before the date the Agent (or any successor Agent) becomes the Agent hereunder, it shall deliver to Borrower two duly executed copies of either (A) IRS Form W-9 (or any successor forms) certifying that it is exempt from U.S. federal backup withholding Tax or (B) a U.S. branch withholding certificate on IRS Form W-8IMY (or any successor forms) evidencing its agreement with Borrower to be treated as a U.S. Person (with respect to amounts received on account of

any Lender party to this Agreement) and IRS Form W-8ECI (or any successor forms) (with respect to amounts received on its own account), with the effect that, in either case, Borrower will be entitled to make payments hereunder to the Agent without withholding or deduction on account of U.S. federal withholding Tax. The Agent agrees that if any form or certification it previously delivered becomes expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification.

Each Lender agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify Borrower and Agent in writing of its legal inability to do so.

- (h) If any party determines, in its sole discretion exercised in good faith, that it has received a refund of any Taxes as to which it has been indemnified pursuant to this Section 3.10 (including by the payment of additional amounts pursuant to this Section 3.10), it shall pay to the indemnifying party an amount equal to such refund (but only to the extent of indemnity payments made under this Section with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses (including Taxes) of such indemnified party and without interest (other than any interest paid by the relevant Governmental Body with respect to such refund). Such indemnifying party, upon the request of such indemnified party, shall repay to such indemnified party the amount paid over pursuant to this paragraph (h) (plus any penalties, interest or other charges imposed by the relevant Governmental Body) in the event that such indemnified party is required to repay such refund to such Governmental Body. Notwithstanding anything to the contrary in this paragraph (h), in no event will the indemnified party be required to pay any amount to an indemnifying party pursuant to this paragraph (h) the payment of which would place the indemnified party in a less favorable net after-Tax position than the indemnified party would have been in if the Tax subject to indemnification and giving rise to such refund had not been deducted, withheld or otherwise imposed and the indemnification payments or additional amounts with respect to such Tax had never been paid. This paragraph shall not be construed to require any indemnified party to make available its Tax returns (or any other information relating to its Taxes that it deems confidential) to the indemnifying party or any other Person.
- (i) Notwithstanding anything to the contrary in this <u>Section 3.10</u>, Agent shall be entitled to withhold United States federal income taxes at the full 30% withholding rate if in its reasonable judgment it is required to do so under the due diligence requirements imposed upon a withholding agent under Treasury Regulations Section 1.1441-7(b). Further, Agent is indemnified under Treasury Regulations Section 1.1461-1(e) against any claims and demands of any Lender or Participant for the amount of any tax it deducts and withholds in accordance with regulations under Section 1441 of the Code.
- (j) Each party's obligations under this <u>Section 3.10</u> shall survive the resignation or replacement of Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the Commitments and the repayment, satisfaction or discharge of all obligations under this Agreement or any other Loan Document.

IV. COLLATERAL: GENERAL TERMS.

4.1. Security Interest in the Collateral.

To secure the prompt payment and performance to Agent, the Issuer and each Lender of the Obligations, each Loan Party hereby assigns, pledges and grants to the Collateral Agent for the ratable benefit of Agent, the Issuer and each Lender a continuing security interest in and to all of the Collateral, whether now owned or existing or hereafter acquired or arising and wheresoever located.

4.2. <u>Perfection of Security Interest</u>.

- (a) Each Loan Party shall take all action that may be necessary or desirable that Agent may reasonably request, so as at all times to maintain the validity, perfection, enforceability and priority of Agent's security interest in the Collateral or to enable Agent to protect, exercise or enforce its rights hereunder and in the Collateral, including, but not limited to, (i) promptly discharging all Liens other than Permitted Encumbrances, (ii) delivering to Agent, endorsed or accompanied by such instruments of assignment as Agent may reasonably specify, any and all chattel paper, instruments, letters of credits and advices thereof and documents evidencing or forming a part of the Collateral, in each case, to the extent such Collateral comprises part of the Pledged Collateral required to be delivered to the Collateral Agent under Section 4.2 and (iii) executing and/or delivering financing statements or other instruments of pledge, notices and assignments, in each case in form and substance satisfactory to Agent, relating to the creation, validity, perfection, maintenance or continuation of Agent's security interest under the UCC or other Applicable Law.
- (b) Agent may at any time and from time to time file, without the signature of any Loan Party in accordance with Section 9-509 of the UCC, financing statements, continuation statements and amendments thereto that describe the Collateral as "all assets" (or equivalent language) of such Loan Party and which contain any other information required by the UCC for the sufficiency or filing office acceptance of any financing statements, continuation statements or amendments. Each Loan Party agrees to furnish any such information to Agent promptly upon request.
- (c) Each Loan Party shall, at any time and from time to time, subject to the Perfection Exceptions, take such steps as Agent may reasonably request to insure the continued perfection and priority of Agent's security interest in any of the Collateral for the benefit of the Lenders and of its rights therein. If any Loan Party shall at any time, acquire a "commercial tort claim" (as such term is defined in the UCC) with a value in excess of \$5,000,000, such Loan Party shall promptly notify Agent thereof in writing, therein providing a reasonable description and summary thereof, and upon delivery thereof to Agent, such Loan Party shall be deemed to thereby grant to Agent for the benefit of the Lenders (and each Loan Party hereby grants to Agent, for the benefit of each Lender) a security interest and lien in and to such commercial tort claim and all proceeds thereof, all upon the terms of and governed by this Agreement.
- (d) Each Loan Party agrees to deliver or cause to be delivered to the Agent any and all (i) Pledged Equity Interests of Borrower and the Guarantors on the date hereof (subject to Section 6.13), (ii) Pledged Debt Securities in a principal amount in excess of \$10,000,000 on the date hereof (subject to Section 6.13) and (iii) all other Pledged Equity Interests or Pledged Debt Securities required to be delivered pursuant to the terms of this Agreement, as promptly as practicable after the Closing Date (subject to Section 6.13), in each case, in the case of any such Pledged Equity Interests or Pledged Debt Securities owned by such Loan Party on the date hereof, and within 30 days of receipt by such Loan Party (or such later date as the Agent may agree in its sole discretion), in each case, in the case of any such Pledged Equity Interests or Pledged Debt Securities s acquired by such Loan Party after the date hereof. Each Loan Party acknowledges and agrees that (i) to the extent any interest in any limited liability company or limited partnership controlled now or in the future by such Loan Party (or by such Loan Party and one or more other Loan Parties) and pledged hereunder is a "security" within the meaning of Article 8 of the UCC and is governed by Article 8 of the UCC, such interest shall be certificated or subject to a control agreement for such uncertificated interest; and such certificate shall be delivered to the Agent in accordance with this Section 4.2(d) and (ii) each such interest shall at all times hereafter continue to be such a security and represented by such certificate or subject to a control agreement. Each Loan Party further acknowledges and agrees that with respect to any interest in any limited liability company or limited partnership controlled now or in the future by such Loan Party (or by such Loan Party and one or more other Loan Parties) and pledged hereunder that is not a "security" within the meaning of Article 8 of

the UCC, the terms of such interest shall at no time provide that such interest is a "security" within the meaning of Article 8 of the UCC, nor shall such interest be represented by a certificate, unless such Loan Party provides prior written notification to the Agent that the terms of such interest so provide that such interest is a "security" within the meaning of Article 8 of the UCC and such interest is thereafter represented by a certificate or subject to a control agreement; and such certificate shall be delivered to the Agent in accordance with this <u>Section 4.2(d)</u>.

(e) All charges, expenses and fees Agent may incur in doing any of the foregoing shall be paid by the Loan Parties in accordance with Section 15.9 of this Agreement.

4.3. [<u>Reserved</u>].

4.4. <u>Preservation of Collateral</u>.

In addition to the rights and remedies set forth in Section 11.1, Agent may following the occurrence and during the continuation of an Event of Default: (a) at any time take such steps as Agent deems necessary in its reasonable credit judgment to protect Agent's interest in and to preserve the Collateral, including the hiring of such security guards or the placing of other security protection measures as Agent may deem appropriate; (b) employ and maintain at any Loan Party's premises a custodian who shall have full authority to do all acts necessary to protect Agent's interests in the Collateral; (c) lease warehouse facilities to which Agent may move all or part of the Collateral; and (d) use any Loan Party's owned or leased lifts, hoists, trucks and other facilities or equipment for handling or removing the Collateral. In addition, following the occurrence and during the continuation of an Event of Default or in connection with any inspections or field examinations performed by or on behalf of Agent, Agent shall have, and is hereby granted, a right of ingress and egress to the places where the Collateral is located, and may proceed over and through any Loan Party's owned or leased property. Each Loan Party shall cooperate fully with all of Agent's efforts to preserve the Collateral and will take such actions to preserve the Collateral as Agent may direct. All of Agent's expenses of preserving the Collateral, including any expenses relating to the bonding of a custodian, shall be paid by the Loan Parties in accordance with Section 15.9 of this Agreement.

4.5. Ownership of Collateral.

(a) With respect to the Collateral, at the time the Collateral becomes subject to Agent's security interest: (a) each Loan Party shall be the sole owner of and fully authorized and able to sell, transfer, pledge and/or grant a first priority security interest in each and every item of its respective Collateral to Agent; and, except for Permitted Encumbrances, the Collateral shall be free and clear of all Liens and encumbrances whatsoever; (b) each document and agreement with respect to the Collateral executed by any Loan Party or delivered to Agent or any Lender in connection with this Agreement shall be true and correct in all material respects; and (c) all signatures and endorsements of each Loan Party that appear on such documents and agreements shall be genuine and each Loan Party shall have full capacity to execute same.

4.6. <u>Defense of Agent's and Lenders' Interests</u>.

Until (a) payment and performance in full in cash of all of the Obligations (other than (A) contingent indemnification obligations as to which no claim has been asserted and (B) obligations and liabilities under Lender-Provided Hedges and Bank Product Obligations) and (b) termination of this Agreement, Agent's security interests in the Collateral shall continue in full force and effect. During such period no Loan Party shall, without Agent's prior written consent, pledge, sell (except Permitted Dispositions), assign, transfer, create or suffer to exist a Lien upon or encumber or allow or suffer to be encumbered in any way except for Permitted Encumbrances, any part of the Collateral. Each Loan Party

shall defend Agent's interests in the Collateral against any and all Persons whatsoever. Subsequent to the occurrence and during the continuation of an Event of Default, Agent shall have the right to take possession of the indicia of the Collateral and the Collateral in whatever physical form contained, including without limitation, labels, stationery, documents, instruments and advertising materials. If Agent exercises this right to take possession of the Collateral, each Loan Party shall, upon demand, assemble and make it available to Agent at one of such Loan Party's locations set forth in Schedule 4.6 hereto. In addition, with respect to all Collateral, Agent and Lenders shall be entitled to all of the rights and remedies set forth herein and further provided by the UCC or other Applicable Law. Upon the occurrence and during the continuation of an Event of Default, each Loan Party shall, and Agent may, at its option, instruct all suppliers, carriers, forwarders, warehouses or others receiving or holding cash, checks, Inventory, documents or instruments in which Agent holds a security interest to deliver same to Agent and/or subject to Agent's order and if they shall come into any Loan Party's possession, they shall be held by such Loan Party in trust as Agent's trustee, and such Loan Party will immediately deliver them to Agent in their original form together with any necessary endorsement.

4.7. Books and Records.

Each Loan Party shall in all material respects, (a) keep proper books of record and account in which full, true and correct entries will be made of all dealings or transactions of or in relation to its business and affairs; (b) set up on its books accruals with respect to all taxes, assessments, charges, levies and claims; and (c) on a reasonably current basis set up on its books, from its earnings, allowances against doubtful Receivables, advances and investments and all other proper accruals (including without limitation by reason of enumeration, accruals for premiums, if any, due on required payments and accruals for depreciation, obsolescence, or amortization of properties), which should be set aside from such earnings in connection with its business.

4.8. [<u>Reserved</u>].

4.9. Compliance with Laws.

Each Loan Party shall comply with all acts, rules, regulations and orders of any Governmental Body applicable to the Collateral or any part thereof or to the operation of such Loan Party's business the non-compliance with which would reasonably be expected to have a Material Adverse Effect. The Collateral at all times shall be maintained in all material respects in accordance with the requirements of all insurance carriers which provide insurance with respect to the Collateral so that such insurance shall remain in full force and effect.

4.10. <u>Inspection of Premises</u>.

At (x) reasonable times during normal business hours (to be not more than once per fiscal year) with reasonable prior notice to Borrower (it being understood that at least two (2) Business Days' notice will be considered reasonable prior notice) or (y) at any time following the occurrence and during the continuance of an Event of Default (provided that no prior notice shall be required following the occurrence and during the continuance of an Event of Default), Agent shall have full access to and the right to audit, check, inspect and make abstracts and copies from each Loan Party's books, records, audits, correspondence and all other papers, in each case, relating to the Collateral and the operation of such Loan Party's business. At (x) reasonable times during business hours (to be not more than once per fiscal year) with reasonable prior notice to Borrower (it being understood that at least two (2) Business Days' notice will be considered reasonable prior notice) or (y) at any time following the occurrence and during the continuance of an Event of Default (provided that no prior notice shall be required following the occurrence and during the continuance of an Event of Default), Agent and its agents may enter upon

any of any Loan Party's premises for the sole purpose of inspecting the Collateral and any and all records pertaining thereto and the operation of such Loan Party's business.

4.11. Insurance.

Each Loan Party shall bear the full risk of any loss of any nature whatsoever with respect to the Collateral. At each Loan Parties' own cost and expense in amounts and with carriers acceptable to Agent, Loan Parties shall: (a) keep all its insurable properties and properties in which any Loan Party has an interest insured against the hazards of fire, flood, sprinkler leakage, those hazards covered by extended coverage insurance and such other hazards, and for such amounts, as is customary in the case of companies engaged in businesses similar to the Loan Parties including, without limitation, business interruption insurance; (b) maintain a bond in such amount as is customary in the case of companies in the same industry and of comparable size as the Loan Parties against larceny, embezzlement or other criminal misappropriation of insured's officers and employees who may either singly or jointly with others at any time have access to the assets or funds of the Loan Parties either directly or through authority to draw upon such funds or to direct generally the disposition of such assets; (c) [reserved]; (d) maintain public and product liability insurance against claims for personal injury, death or property damage suffered by others; (e) maintain all such worker's compensation or similar insurance as may be required under the laws of any state or jurisdiction in which the Loan Parties are engaged in business; (f) [reserved]; (g) [reserved]; and (h) furnish Agent with (i) copies of all policies and evidence of the maintenance of such policies by the renewal thereof at least thirty (30) days before any expiration date, and (ii) appropriate endorsements in form and substance reasonably satisfactory to Agent, naming Agent as a co-insured and lender loss payee as its interests may appear with respect to all insurance coverage referred to in clauses (a), (c), (d), (e) and (<u>f</u>) above, as applicable, and providing (A) that all proceeds thereunder shall be payable to Agent (excluding insurance coverage referred to in <u>clauses</u> (<u>d</u>) and (<u>e</u>) above), (B) no such insurance shall be affected by any act or neglect of the insured or owner of the property described in such policy other than nonpayment of premiums, and (C) that such policy and lender loss payable clauses may not be cancelled, amended or terminated unless at least thirty (30) days' prior written notice is given to Agent. In the event of any loss thereunder (excluding insurance coverage referred to in <u>clauses</u> (d) and (e) above), the carriers named therein hereby are directed by Agent and the Loan Parties to make payment for such loss to Agent and not to the applicable Loan Party and Agent jointly. If any insurance losses are paid by check, draft or other instrument payable to a Loan Party and Agent jointly, Agent may endorse such Loan Party's name thereon and do such other things as Agent may deem advisable to reduce the same to cash. Following the occurrence and during the continuation of an Event of Default, Agent is hereby authorized to adjust and compromise claims under insurance coverage referred to <u>clause (a)</u> above. All loss recoveries received by Agent upon any such insurance may be applied to the Obligations, in such order as Agent in its sole discretion shall determine. Any surplus shall be paid by Agent to Borrower within five (5) Business Days of receipt thereof or applied as may be otherwise required by law. Any deficiency thereon shall be paid by the Loan Parties to Agent, on demand.

4.12. [<u>Reserved</u>].

4.13. Payment of Taxes.

Each Loan Party will pay, before delinquency or before the expiration of any extension period, all Taxes, assessments and other charges lawfully levied or assessed upon such Loan Party or any of the Collateral by any Governmental Body, except (i) where such Taxes, assessments, fees or other charges are being contested in good faith by appropriate proceedings diligently conducted and which proper reserves have been taken by such Loan Party or (ii) where the failure to make such payments would not, in the aggregate, reasonably be expected to have a Material Adverse Effect.

4.14. Payment of Leasehold Obligations.

Each Loan Party shall at all times pay, when and as due, its rental obligations under all leases under which it is a tenant except where the failure to make such payments would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, and shall otherwise comply, in all material respects, with all other terms of such leases and keep them in full force and effect.

- 4.15. [Reserved].
- 4.16. [Reserved].
- 4.17. [Reserved].
- 4.18. Exculpation of Liability.

Nothing herein contained shall be construed to constitute Agent or any Lender as Loan Parties' agent for any purpose whatsoever, nor shall Agent or any Lender be responsible or liable for any shortage, discrepancy, damage, loss or destruction of any part of the Collateral wherever the same may be located and regardless of the cause thereof. Neither Agent nor any Lender, whether by anything herein or in any assignment or otherwise, assume any of any Loan Party's obligations under any contract or agreement assigned to Agent or such Lender, and neither Agent nor any Lender shall be responsible in any way for the performance by any Loan Party of any of the terms and conditions thereof.

- 4.19. [Reserved].
- 4.20. <u>Financing Statements</u>.

Other than the financing statements filed by Agent with respect to the transactions contemplated under this Agreement and the financing statements evidencing Permitted Encumbrances, including (but not limited to) those described on <u>Schedule 7.2</u>, no Loan Party has authorized the filing of any financing statement covering any of the Collateral.

- 4.21. [<u>Reserved</u>].
- 4.22. Agent as Collateral Agent.

(a) Each Lender hereby authorizes Agent to (i) execute, deliver and perform as a collateral agent under this Agreement and each other Loan Document to which Agent is or is intended to be a party, (ii) exercise and enforce any and all rights, powers and remedies provided to Agent or any Lender by this Agreement and each other Loan Document to which Agent is or is intended to be a party, any Applicable Law, or any other document, instrument, or agreement, and (iii) take any other action under this Agreement and each other Loan Document to which Agent is or is intended to be a party which Agent in its sole discretion shall deem advisable and in the best interests of the Lenders. Notwithstanding the foregoing, Agent shall not commence an enforcement action except at the direction of the Required Lenders; provided that if Agent is prohibited by any court order or applicable law from commencing any enforcement action, Agent shall not be obligated to commence such enforcement action until such authority is obtained. All decisions with respect to the type of enforcement action which is to be commenced shall be made by, and all actions with respect to prosecution and settlement of such enforcement action shall require the direction of the Required Lenders, and Agent shall not be required to take any enforcement action in the absence of any such direction. Agent will use its commercially reasonable efforts to pursue diligently the prosecution of any enforcement action, which Agent is so authorized or directed to initiate pursuant to this Agreement. Agent shall make available to the Lenders

copies of all notices it receives in connection with the Collateral or any enforcement action promptly upon receipt.

- (b) Agent may, but shall not be obligated, to take such action as it deems necessary to perfect or continue the perfection of the Liens on the Collateral held for the benefit of the Lenders. Agent shall not release any of the Collateral held for the benefit of the Lenders, or any Liens on the Collateral held for the benefit of the Lenders, except (i) upon the written direction of the Required Lenders (or of all Lenders if required under Section 15.2(b)), (ii) upon payment in full in cash of the Obligations (other than (A) contingent indemnification obligations as to which no claim has been asserted and (B) obligations and liabilities under Lender-Provided Hedges and Bank Product Obligations), the termination of all Commitments under this Agreement and the cash collateralization of all Letters of Credit in accordance with this Agreement (or as otherwise acceptable to the Issuer in its sole discretion), (iii) for Collateral consisting of a debt instrument if the indebtedness evidenced thereby has been paid in full, (iv) where such release is expressly permitted under the Loan Documents to which it is a party, or (v) with respect to any Securitization Asset that is sold or pledged pursuant to a Permitted Securitization Financing.
- (c) Subject to the terms of this Agreement, Agent agrees to administer and enforce this Agreement and the other Loan Documents to which it is a party and to foreclose upon, collect and dispose of the Collateral and to apply the proceeds therefrom, for the benefit of Agent, the Issuer and each Lender, as provided in this Agreement, and otherwise to perform its duties and obligations as the collateral agent hereunder in accordance with the terms hereof; provided, however, that Agent shall have no duties or responsibilities except those expressly set forth in the Loan Documents to which it is a party as Collateral Agent, and no implied covenants or obligations shall be read into any such Loan Documents against Agent. Agent will use its commercially reasonable efforts to pursue diligently the enforcement of this Agreement and the other Loan Documents, which Agent is so authorized or directed to initiate pursuant to this Agreement.
- (d) Notwithstanding anything contained herein to the contrary, Agent shall not be required to exercise any discretion or take any action but shall only be required to act or refrain from acting (and shall be fully protected in so acting or refraining from acting) upon the written instructions of the Required Lenders, in each case, as specified therein, and such instructions shall be binding upon Agent, the Issuer and each Lender; provided, however, that the written instructions of Agent, the Issuer and each Lender shall be required where expressly provided for herein; and provided, further, that Agent shall not be required to take any action which is contrary to any provision herein or Applicable Law.
- (e) Agent may at any time request instructions from the Required Lenders as to a course of action to be taken by it hereunder and any of the other Loan Documents or in connection herewith and therewith or any other matters relating hereto and thereto.
- (f) Unless otherwise consented to in writing by Agent (acting at the direction of the Required Lenders), no Lender or Issuer, individually or together with any other Lenders or the Issuer, shall have the right, nor shall it, exercise or enforce any of the rights, powers or remedies which Agent is authorized to exercise or enforce under this Agreement or any of the other Loan Documents.
- (g) Notwithstanding any other provision herein, in no event shall Agent be required to foreclose on, or take possession of, the Collateral, if, in the judgment of Agent, such action would be in violation of any Applicable Law, rule or regulation pertaining thereto, or if Agent reasonably believes that such action would result in the incurrence of liability by Agent for which it is not fully indemnified by Agent, the Issuer and each Lender.

- (h) Neither Agent nor any of its directors, officers, employees or agents shall be liable or responsible for any action taken or omitted to be taken by it or them hereunder or in connection herewith, except for its or their own gross negligence or willful misconduct (as determined by a court of competent jurisdiction in a final and non-appealable judgment).
- (i) Agent shall not be responsible to the Issuer and each Lender for (i) any recital statements, representations or warranties by the Borrower or any of the Issuer and Lenders contained in this Agreement or the Loan Documents, or any certificate or other document delivered by the Borrower or the Issuer and each Lender thereunder, (ii) the value, validity, effectiveness, genuineness, enforceability (other than as to Agent with respect to such documents to which Agent is a party) or sufficiency of this Agreement or any other document referred to or provided for herein or therein or of the Collateral held by Agent hereunder, (iii) the performance or observance by the Borrower or any of the Issuer and Lenders of any of their respective agreements contained herein or therein, nor shall Agent be liable because of the invalidity or unenforceability of any provisions of this Agreement (other than as to itself) or (iv) the validity, perfection, priority or enforceability of the Liens in any of the Collateral, whether impaired by operation of law or by reason of any action or omission to act on its part hereunder (except to the extent such action or omission constitutes gross negligence or willful misconduct, as determined by a court of competent jurisdiction in a final and non-appealable judgment, on the part of Agent), the validity of the title of the Borrower to the Collateral, insuring the Collateral or the payment of taxes, charges, assessments or Liens upon the Collateral or otherwise as to the maintenance of the Collateral.
- (j) The powers conferred on Agent under this Agreement and any of the other Loan Documents are solely to protect its interest in the Collateral and shall not impose any duty upon it to exercise any such powers. Except for the safe custody and preservation of the Collateral in its possession and the accounting for monies actually received by it, Agent shall have no other duty as to the Collateral, whether or not Agent or any of the other Lenders or the Issuer has or is deemed to have knowledge of any matters, or as to the taking of any necessary steps to preserve rights against any parties or any other rights pertaining to the Collateral. Agent hereby agrees to exercise reasonable care in respect of the custody and preservation of the Collateral. Agent shall be deemed to have exercised reasonable care in the custody and preservation of the Collateral in its possession if such Collateral is accorded treatment substantially equal to that which Agent accords its own property.
- (k) Upon the payment in full in cash of the Obligations (other than (A) contingent indemnification obligations as to which no claim has been asserted and (B) obligations and liabilities under Lender-Provided Hedges and Bank Product Obligations), the termination of all Commitments under this Agreement, the cash collateralization of all Letters of Credit in accordance with this Agreement (or as otherwise acceptable to the Issuer in its sole discretion), and the termination of this Agreement or as may be otherwise directed by Required Lenders (or of all Lenders if required under Section 15.2(b)) in accordance with the applicable provisions of this Agreement, all rights to the Collateral as shall not have been sold or otherwise applied, in each case, pursuant to the terms hereof, shall revert to the applicable Loan Parties, its successors or assigns, or otherwise as a court of competent jurisdiction may direct. Upon any such termination, Agent shall, at Borrower's expense, execute and deliver to Borrower such documents as Borrower shall reasonably request to evidence such termination and release.

V. REPRESENTATIONS AND WARRANTIES.

Each Loan Party represents and warrants as of the date hereof and on every other date thereafter on which an Advance is made (solely to the extent required to be true and correct for such Advance pursuant to Section 8.2), to Agent, Collateral Agent and the Lenders that:

5.1. Authority.

Each Loan Party has full power, authority and legal right to enter into this Agreement and the other Loan Documents and to perform all its Obligations hereunder and thereunder. The execution, delivery and performance of this Agreement and of the other Loan Documents (a) are within such Loan Party's corporate or limited liability company power, as applicable, have been duly authorized, are not in contravention of Applicable Law or the terms of such Loan Party's by-laws, operating agreement, certificate of incorporation, certificate of formation, as applicable, or other applicable documents relating to such Loan Party's organization or formation or to the conduct of such Loan Party's business or of any agreement or undertaking to which such Loan Party is a party or by which such Loan Party is bound, and (b) will not conflict with nor result in any breach in any of the provisions of or constitute a default under or result in the creation of any Lien except Permitted Encumbrances upon any asset of such Loan Party under the provisions of the charter document or by-laws or operating agreement, as applicable, of such Loan Party. This Agreement and the other Loan Documents, as applicable, constitute the legal, valid and binding obligation of such Loan Party, enforceable in accordance with their respective terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws relating to or limiting creditors' rights generally or by equitable principles relating to enforceability.

5.2. Formation and Qualification.

- (a) Each Loan Party (i) is duly formed or incorporated, (ii) in good standing (to the extent such concept is applicable in the relevant jurisdiction) under the laws of the jurisdiction of its incorporation or organization and (iii) is qualified and is authorized to do business and in good standing (to the extent such concept is applicable in the relevant jurisdiction) under the laws of each jurisdiction where its ownership, lease or operation of properties or the conduct of its business requires such qualification except, in the case of this clause (iii), where the failure to so qualify would not reasonably be expected to have a Material Adverse Effect. Each Loan Party has delivered to Agent true and complete copies of its certificate of incorporation, certificate of formation, by-laws, operating agreement and stockholder's agreement (or any other agreements among the equity holders of such Loan Party), as applicable, and will promptly notify Lender of any amendment or changes thereto.
- (b) The only Subsidiaries of and owners of each Loan Party (including owners of Equity Interests issued by each Loan Party) are listed on <u>Schedule 5.2</u>.
 - 5.3. [<u>Reserved</u>].
 - 5.4. Tax Returns.

Each Loan Party has filed or caused to be filed all federal, Tax returns and other material state and other Tax returns that are required by law to be filed by it and has paid, or caused to be paid, all federal, state and other Taxes that are due and payable by it any assessments made against it or any of its property by any Governmental Body, other than any such Taxes which are being contested in good faith and by appropriate proceedings and with respect to which proper reserves have been taken by such Loan Party. The Provision for Taxes for such Loan Party on the books of each Loan Party are adequate for all years not closed by applicable statutes, and for its current fiscal year, and no Loan Party has any knowledge of any deficiency or additional assessment in connection therewith not provided for on its books.

5.5. Financial Statements.

(a) [Reserved].

(b) The quarterly cash flow projections of Borrower for the period January 1, 2021 through and including December 31, 2023, (the "<u>Projections</u>") were prepared by Authorized Officer of Borrower, are based on underlying assumptions which provide a reasonable basis for the projections contained therein and reflect Borrower's judgment based on present circumstances of the most likely set of conditions and course of action for the projected period.

(c) [Reserved].

- (d) The audited consolidated balance sheets of Borrower as of December 31, 2018, December 31, 2019 and December 31, 2020, and the related consolidated statements of income, changes in stockholder's equity, and changes in cash flow for the period ended on such date, all accompanied by reports thereon containing opinions without qualification by the Accountants (other than any such exception, qualification or explanatory paragraph that is with respect to, or resulting from, (i) an upcoming maturity date of any other Indebtedness incurred in compliance with this Agreement, (ii) any actual or potential inability to satisfy a financial maintenance covenant, including the financial covenants set forth in Section 6.8 hereof, on a future date or in a future period, or (iii) the activities, operations, financial results, assets or liabilities of any Unrestricted Subsidiary), copies of which have been delivered to Agent, have been prepared in accordance with GAAP, consistently applied (except for changes in application in which such accountants concur) and present fairly the financial position of Borrower at such date and the results of its operations for such period.
- (e) Since December 31, 2020, no event, condition or state of facts has occurred which has had, or would reasonably be expected to have, a Material Adverse Effect individually or in the aggregate.

5.6. [<u>Reserved</u>].

5.7. O.S.H.A. and Environmental Compliance.

- (a) Except as would not reasonably be expected to have a Material Adverse Effect, or except as set forth on <u>Schedule 5.7</u>, each Loan Party has duly complied with applicable Environmental Laws, hold all permits, licenses and approvals required under applicable Environmental Laws in relation to its business and properties, and is not subject to any outstanding citations, notices or orders of non-compliance issued to such Loan Party under any Environmental Laws.
- (b) Except as would not reasonably be expected to have a Material Adverse Effect, or as set forth on <u>Schedule 5.7</u>, to Borrower's knowledge, (i) there are no visible signs of releases, spills, discharges, leaks or disposal of Hazardous Substances at, upon, under or within any Real Property; and (ii) no Hazardous Substances are present on any Real Property, excepting such quantities as are handled in accordance with all applicable manufacturer's instructions and governmental regulations and in proper storage containers and as are necessary or appropriate for the operation of the commercial business of any Loan Party or of its tenants.

5.8. Solvency; No Litigation, Violation, Indebtedness or Default.

(a) After giving effect to the Transactions, Borrower and its Subsidiaries, on a consolidated basis, will be solvent, able to pay its debts as they mature, have capital sufficient to carry on its business and all businesses in which it is about to engage, and (i) as of the Closing Date, the fair present saleable value of each Loan Party's assets, is in excess of the amount of its liabilities and (ii) immediately subsequent to the Closing Date, the fair saleable value of each Loan Party's assets (calculated on a going concern basis) will be in excess of the amount of its liabilities. For purposes of this

<u>Section 5.8(a)</u>, the amount of any contingent liability at any time shall be computed as the amount that, in light of all of the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability (irrespective of whether such contingent liabilities meet the criteria for accrual under Statement of Financial Accounting Standard No. 5).

- (b) Except as disclosed in <u>Schedule 5.8(b)</u>, no Loan Party has any pending or threatened litigation, arbitration, actions or proceedings which would reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect,.
- (c) No Loan Party is in violation of any applicable statute, regulation or ordinance in any material respect, nor is any Loan Party in violation of any order of any court, Governmental Body or arbitration board or tribunal.
- No Loan Party nor any member of the Controlled Group maintains or contributes to any Pension Benefit Plan other than those listed on Schedule 5.8(d). No Loan Party nor any member of the Controlled Group has any liability to, or has ever contributed to or had an obligation to contribute to a Multiemployer Plan. No Loan Party nor any member of the Controlled Group has or ever has maintained a defined benefit pension plan subject to the minimum funding requirements of Code Section 412 or ERISA Section 302. Except as would not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect, (i) no Pension Benefit Plan has incurred any "accumulated funding deficiency," as defined in Section 302(a)(2) of ERISA and Section 412(a) of the Code, whether or not waived, and Loan Parties and each member of the Controlled Group have met all applicable minimum funding requirements under Section 302 of ERISA in respect of each Pension Benefit Plan, (ii) each Plan which is intended to be a qualified plan under Section 401(a) of the Code has received a favorable determination letter, or a favorable opinion letter as to its qualification under Section 401(a) of the Code and no event or circumstance has occurred which would reasonably be expected to result in the revocation of such qualified status of the form of the Plan document under the Code, (iii) no Loan Party nor any member of the Controlled Group has incurred any liability to the PBGC other than for the payment of premiums, and there are no premium payments which have become due which are unpaid, (iv) no Pension Benefit Plan has been terminated within the last five years by the plan administrator thereof nor by the PBGC, and there is no occurrence which would cause the PBGC to institute proceedings under Title IV of ERISA to terminate any Pension Benefit Plan, (v) at this time, the current value of the assets of each Pension Benefit Plan or funded Plan exceeds the present value of the accrued benefits and other liabilities of such Plan and no Loan Party nor any member of the Controlled Group knows of any facts or circumstances which would materially change the value of such assets and accrued benefits and other liabilities, (vi) each Loan Party and any member of the Controlled Group has breached any of the responsibilities, obligations or duties imposed on it by ERISA with respect to any Plan, (vii) no Loan Party nor any member of a Controlled Group has incurred any material liability for any excise tax arising under Section 4972 or 4980B of the Code, and no fact exists which would give rise to any such material liability, (viii) no Loan Party nor any member of the Controlled Group nor any fiduciary of, nor any trustee to, any Plan, has engaged in a "prohibited transaction" described in Section 406 of ERISA or Section 4975 of the Code nor taken any action which would constitute or result in a Termination Event with respect to any such Plan which is subject to ERISA, (ix) each Loan Party and each member of the Controlled Group has made all contributions due and payable with respect to each Plan, (x) there exists no event described in Section 4043(b) of ERISA for which the thirty (30) day notice period contained in 29 CFR §2615.3 has not been waived, (xi) no Loan Party nor any member of the Controlled Group has any fiduciary responsibility for investments with respect to any plan existing for the benefit of Persons other than employees or former employees of Loan Parties and any member of the Controlled Group, and (xii) no Loan Party nor any member of the Controlled Group has withdrawn, completely or partially, from any Multiemployer Plan so as to incur liability under the Multiemployer Pension Plan Amendments Act of 1980.

5.9. Patents, Trademarks, Copyrights and Licenses.

All material patents, patent applications, trademarks, trademark applications, service marks, service mark applications, copyright registrations, copyright applications, design rights, tradenames, assumed names, trade secrets and licenses owned or utilized by any Loan Party are set forth on Schedule 5.9, and are valid and have been duly registered or filed the USPTO or USCO, as applicable, and constitute all of the intellectual property rights which are necessary for the operation of its business; there is no objection to or pending challenge to the validity of any such material patent, trademark, copyright, design right, tradename, trade secret or license and no Loan Party is aware of any grounds for any challenge, except as set forth in Schedule 5.9 or otherwise disclosed in writing to Agent after the Closing Date. Each patent, patent application, patent license, trademark, trademark application, trademark license, service mark, service mark application, service mark license, copyright, copyright application and copyright license owned or held by any Loan Party and all trade secrets used by any Loan Party in each case consist of original material or property developed by such Loan Party or was lawfully acquired by such Loan Party from the proper and lawful owner thereof. Each of such items has been maintained by the Loan Parties so as to preserve the value thereof from the date of creation or acquisition thereof to the extent that the applicable Loan Party, in its reasonable business judgment, has determined that it would be commercially reasonable to do so.

5.10. Licenses and Permits.

Except as set forth in <u>Schedule 5.10</u>, each Loan Party (a) is in compliance with and (b) has procured and is now in possession of, all material licenses or permits required by any applicable federal, state or local law or regulation for the operation of its business in each jurisdiction wherein it is now conducting or proposes to conduct business except, in each case, where the failure to be in such compliance, or to procure such licenses or permits would not reasonably be expected to have a Material Adverse Effect.

5.11. No Defaults.

- (a) No Loan Party is in default in the payment of the principal of or interest on any Material Indebtedness or under any instrument or agreement under or subject to which any Material Indebtedness has been issued and no event has occurred under the provisions of any such instrument or agreement which with the lapse of any grace period or failure to give of notice, or both, constitutes or would constitute an event of default thereunder.
 - (b) No Loan Party is in default in the payment or performance of any Material Contract.

5.12. No Burdensome Restrictions.

No Loan Party has agreed or consented to cause or permit in the future (upon the happening of a contingency or otherwise) any of its property, whether now owned or hereafter acquired, to be subject to a Lien which is not a Permitted Encumbrance.

5.13. No Labor Disputes.

No Loan Party is involved in any labor dispute; there are no strikes or walkouts or to its knowledge any union organization campaigns of such Loan Party's employees in existence or to its knowledge threatened and no collective bargaining, labor contract is scheduled by its terms to expire during the Term.

5.14. Margin Regulations.

No Loan Party is engaged, nor will it engage, principally or as one of its important activities, in the business of extending credit for the purpose of "purchasing" or "carrying" any "margin stock" within the meaning of the quoted term under Regulation U of the Board of Governors of the Federal Reserve

System as now and from time to time hereafter in effect. No part of the proceeds of any Advance will be used, directly or indirectly, for "purchasing" or "carrying" "margin stock", or to extend credit to others to "purchase" or "carry" any "margin stock", as defined in Regulation U of such Board of Governors or for any other purpose that entails a violation of any Regulations of such Board of Governors, including Regulation T, U or X.

5.15. <u>Investment Company Act</u>.

No Loan Party is an "investment company" registered or required to be registered under the Investment Company Act of 1940, as amended, nor is it controlled by such a company.

5.16. <u>Disclosure</u>.

No representation or warranty made by any Loan Party in any financial statement, report, certificate or any other document furnished by or on behalf of any Loan Party (other than projected financial information, pro forma financial information and information of a general economic or industry nature) to the Agent or any Lender in connection with the transactions contemplated hereby (as modified or supplemented by other information so furnished), when taken as a whole, contains any material misstatement of fact or omits to state any material fact necessary to make the statements therein (when taken as a whole), in the light of the circumstances under which they were made, not materially misleading; provided that, with respect to projected and pro forma financial information, the Loan Parties represent only that such information was prepared in good faith based upon assumptions believed to be reasonable at the time of preparation and delivery; it being understood that actual results may vary from such forecasts and that such variances may be material.

5.17. Swaps.

Except for Lender-Provided Hedges, no Loan Party is a party to, nor will it be a party to, any Interest Rate Hedge or foreign exchange transaction whereby such Loan Party has agreed or will agree to swap interest rates or currencies, other than those Interest Rate Hedges or foreign exchange transactions, as applicable, which have been entered into for non-speculative purposes.

5.18. Conflicts.

To the Loan Parties' knowledge, no provision of any mortgage, indenture, contract, agreement, judgment, decree or order binding on any Loan Party or affecting the Collateral or any provision of Applicable Law of any Governmental Body conflicts with, or requires any Consent which (1) has not already been obtained, (2) would in any way prevent the execution, delivery or performance of, the terms of this Agreement or the Loan Documents, or (3) the failure to obtain or make would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

5.19. [<u>Reserved</u>].

5.20. Business and Property of Loan Parties.

Upon and after the Closing Date, the Loan Parties do not propose to engage in any business other than financial research, software, education and tools for investors and business activities reasonably related, complementary or ancillary thereto or that are reasonable extensions thereof. On the Closing Date, each Loan Party will own all the property and possess all of the rights and Consents necessary for the conduct of its business.

5.21. <u>Material Contracts</u>. Each Material Contract is in full force and effect and no material defaults enforceable against any Loan Party exist thereunder. No Loan Party has received written notice from any party to any such contract stating that it intends to terminate or amend such contract.

5.22. Sanctions.

- (a) None of Borrower, any of its Subsidiaries, any director or officer, or any employee, agent, or Affiliate, of Borrower or any of its Subsidiaries is a Person that is, or is owned or controlled by Persons that are, (i) the target of any sanctions administered or enforced by the US Department of the Treasury's Office of Foreign Assets Control, the US Department of State, the United Nations Security Council, the European Union, Her Majesty's Treasury or other relevant sanctions authority (collectively, "Sanctions"), or (ii) located, organized or resident in a country or territory that is the target of Sanctions, including currently, the Crimea region, Cuba, Iran, North Korea and Syria.
- (b) No Collateral is or will become property (i) in which a Sanctioned Person holds an interest; (ii) beneficially owned, directly or indirectly, by a Sanctioned Person; (iii) that is due to or from a Sanctioned Person; (iv) that is located in a jurisdiction subject to Sanctions; or (v) that would otherwise cause any violation by a Lender of any applicable anti-terrorism law if such Lender were to obtain an Lien on such property or provide services in consideration of such property.

5.23. <u>Anti-Corruption and Anti-Bribery Laws</u>.

None of Borrower or any of its Subsidiaries nor to the knowledge of Borrower, any director, officer, agent, employee, Affiliate or other Person acting on behalf of Borrower or any of its Subsidiaries is aware of or has taken any action, directly or indirectly, that would result in a violation by such Persons of any applicable anti-bribery law, including but not limited to, the United Kingdom Bribery Act 2010 (the "<u>UK Bribery Act</u>") and the U.S. Foreign Corrupt Practices Act of 1977 (the "<u>FCPA</u>"). Furthermore, Borrower and, to the knowledge of Borrower, its Affiliates have conducted their businesses in compliance with the UK Bribery Act, the FCPA and similar laws, rules or regulations and have instituted and maintain policies and procedures designed to ensure, and which are reasonably expected to continue to ensure, continued compliance therewith.

5.24. Federal Securities Laws.

Neither Borrower nor any of its Subsidiaries (i) is required to file periodic reports under the Exchange Act, (ii) has any securities registered under the Exchange Act or (iii) has filed a registration statement that has not yet become effective under the Securities Act of 1933, as amended.

5.25. Beneficial Ownership Certification.

As of the Closing Date, the information included in the Beneficial Ownership Certification, if applicable, is true and correct in all respects.

VI. AFFIRMATIVE COVENANTS.

Each Loan Party shall, until payment in full in cash of the Obligations (other than (\underline{A}) contingent indemnification obligations as to which no claim has been asserted and (\underline{B}) obligations and liabilities under Lender-Provided Hedges and Bank Product Obligations), the termination of all Commitments under this Agreement, the cash collateralization of all Letters of Credit in accordance with this Agreement (or as otherwise acceptable to the Issuer in its sole discretion), and the termination of this Agreement:

6.1. Payment of Fees.

Pay to Agent on demand all usual and customary fees and expenses which Agent incurs in connection with the forwarding of Advance proceeds and the other Transactions. Agent may, without making demand, charge Borrower's Account for all such fees and expenses.

- 6.2. Conduct of Business and Maintenance of Existence and Assets. (i) Conduct continuously and operate actively its business according to good business practices consistent with past practices and maintain all of its properties useful or necessary in its business in good working order and condition (reasonable wear and tear excepted and except as may be disposed of in accordance with the terms of this Agreement), including, without limitation, all material licenses, patents, copyrights, design rights, tradenames, trade secrets and trademarks and take all actions necessary to enforce and protect the validity of any intellectual property right or other right included in the Collateral; (ii) keep in full force and effect its existence (except to the extent not otherwise prohibited by this Agreement) and comply in all material respects with the laws and regulations governing the conduct of its business, in each case, where the failure to do so would reasonably be expected to have a Material Adverse Effect; and (iii) make all such reports and pay all such franchise and other Taxes and license fees and do all such other acts and things as may be lawfully required to maintain its rights, licenses, leases, powers and franchises under the laws of the United States or any political subdivision thereof where the failure to do so would reasonably be expected to have a Material Adverse Effect.
- 6.3. <u>Violations</u>. Promptly notify Agent in writing of any violation of any law, statute, regulation or ordinance of any Governmental Body, or of any agency thereof, applicable to (and legally binding on) any Loan Party which would reasonably be expected to have a Material Adverse Effect.
 - 6.4. Use of Proceeds. Use the proceeds of each Advance only for the purposes specified in Section 2.15.
- 6.5. <u>Execution of Supplemental Instruments</u>. Execute and deliver to Agent from time to time, promptly following demand, such reasonable supplemental agreements, statements, assignments and transfers, or instructions or documents relating to the Collateral, and such other instruments as Agent may reasonably request, and take such further action as the Agent reasonably deems necessary to perfect, protect, ensure the priority of or continue the Agent's Lien on the Collateral in order that the full intent of this Agreement may be carried into effect.
- 6.6. <u>Payment of Indebtedness</u>. Subject at all times to any applicable subordination arrangement in favor of Agent or Lenders, pay, discharge or otherwise satisfy at or before maturity (subject, where applicable, to specified grace periods and, in the case of the trade payables, to normal payment practices) all its obligations and liabilities of whatever nature, except when the failure to do so

would not reasonably be expected to have a Material Adverse Effect or when the amount or validity thereof is currently being contested in good faith by appropriate proceedings and each Loan Party shall have provided for such reserves as are proper and necessary in accordance with GAAP.

6.7. <u>Standards of Financial Statements</u>. Cause all financial statements, projections and budgets referred to in <u>Sections 9.7</u>, <u>9.8</u>, and <u>9.12</u> as to which GAAP is applicable to be complete and correct in all material respects (subject, in the case of interim financial statements, to normal year-end audit adjustments) and to be prepared in reasonable detail and in accordance with GAAP applied consistently throughout the periods reflected therein.

6.8. Financial Covenants.

- (a) <u>Interest Coverage Ratio</u>. The Interest Coverage Ratio of the Parent (or, solely during a Borrower Consolidating Financials Delivery Period, Borrower), as of the end of each fiscal quarter with respect to the four fiscal quarters then ended, shall not be less than 3.00 to 1.00.
- (b) Net Leverage Ratio. The Net Leverage Ratio of the Parent (or, solely during a Borrower Consolidating Financials Delivery Period, Borrower), as of the end of each fiscal quarter with respect to the four fiscal quarters then ended, shall not be greater than 2.00 to 1.00; provided that, so long as no Default or Event of Default has occurred and is continuing, Borrower may elect, upon written notice to the Agent, to increase the maximum Net Leverage Ratio required under this clause (b) to 2.50:1.00 (each, a "Leverage Ratio Increase") in connection with a Permitted Acquisition for which the aggregate cash consideration paid by the Borrower and its Restricted Subsidiaries shall exceed \$50,000,000 for the fiscal quarter in which such Permitted Acquisition is consummated and for the next succeeding three fiscal quarters (such four fiscal quarter period during with a Leverage Ratio Increase is in effect, the "Leverage Ratio Increase Period"); and provided, further, that there shall be at least one full fiscal quarter following the cessation of each such Leverage Ratio Increase Period during which no Leverage Ratio Increase shall then be in effect.
- 6.9. <u>Keepwell</u>. Each Qualified ECP Loan Party, jointly and severally, hereby absolutely, unconditionally and irrevocably undertakes to provide such funds or other support as may be needed from time to time by any of Borrower or any Guarantor hereunder to honor all of such Person's obligations under this Agreement in respect of Swap Obligations (<u>provided</u>, <u>however</u>, that each Qualified ECP Loan Party shall only be liable under this <u>Section 6.9</u> for the maximum amount of such liability that can be hereby incurred without rendering its obligations under this <u>Section 6.9</u>, or otherwise under this Agreement, voidable under Applicable Law, including Applicable Law relating to fraudulent conveyance or fraudulent transfer, and not for any greater amount). The obligations of each Qualified ECP Loan Party under this <u>Section 6.9</u> shall remain in full force and effect until all of the Obligations and all other amounts payable under this Agreement shall have been paid in full in cash (other than (A) contingent indemnification obligations as to which no claim has been asserted and (B) obligations and liabilities under Lender-Provided Hedges and Bank Product Obligations), all Letters of Credit shall have expired or been terminated and the Commitments shall have expired or been terminated. Each Qualified ECP Loan Party intends that this <u>Section 6.9</u> constitute, and this <u>Section 6.9</u> shall be deemed to constitute, a "keepwell, support, or other agreement" for the benefit of each of Borrower and each Guarantor for all purposes of Section 1a(18)(A)(v)(II) of the Commodity Exchange Act.
- 6.10. <u>Designation of Subsidiaries</u>. Borrower may at any time as of or after the Closing Date designate any Restricted Subsidiary of (or acquired by) Borrower as an Unrestricted Subsidiary (unless such Restricted Subsidiary or any of its Subsidiaries owns any Equity Interests or Indebtedness of, or owns or holds any Lien on, any property of, Borrower or any Subsidiary (other than solely any Subsidiary of the Subsidiary to be so designated)) or any Unrestricted Subsidiary as a Restricted Subsidiary by

written notice to the Agent; provided that (i) immediately before and after such designation, no Event of Default shall exist (including after giving effect to the reclassification of Investments in, Indebtedness of and Liens on the assets of, the applicable Restricted Subsidiary or Unrestricted Subsidiary), (ii) the financial covenants set forth in Section 6.8 shall have been met, as determined on a pro forma basis as of the last day of the most recently ended Calculation Period for which financial statements were required to have been delivered in accordance with Sections 9.7 and 9.8 after giving effect to such designation (and determined on the basis of the financial statements for the most recently-ended test period at or prior to such time), (iii) in the case of the designation of any Subsidiary as an Unrestricted Subsidiary, such designation shall constitute an Investment in such Unrestricted Subsidiary (calculated as an amount equal to the sum of (x) the fair market value of the Subsidiary designated immediately prior to such designation (such fair market value to be calculated without regard to any Obligations of such Subsidiary under the Guaranty) and (y) the aggregate principal amount of any Indebtedness owed by the Subsidiary to Borrower or any of its Subsidiaries immediately prior to such designation, all calculated, except as set forth in the parenthetical to clause (x) above, on a consolidated basis in accordance with GAAP), and such designation shall only be permitted to the extent such Investment is permitted under Section 7.4, (iv) immediately after giving effect to the designation of an Unrestricted Subsidiary as a Restricted Subsidiary, Borrower shall comply with the provisions of Section 7.11 with respect to such designated Restricted Subsidiary, (v) no Restricted Subsidiary may be a Subsidiary of an Unrestricted Subsidiary, (vi) each of (A) the Subsidiary to be so designated and (B) its Subsidiaries has not, at the time of designation, and does not thereafter, create, incur, issue, assume, guarantee or otherwise become directly or indirectly liable with respect to any Indebtedness pursuant to which the applicable lender has recourse to any of the assets of Borrower or any Restricted Subsidiary (other than Equity Interests in an Unrestricted Subsidiary); and (vii) Borrower shall have delivered to the Agent and each Lender a certificate executed by its chief financial officer or treasurer, certifying to the compliance with the requirements of the preceding <u>clauses (i)</u> through (v), inclusive, and containing the calculations (in reasonable detail) required by the preceding <u>clause (ii)</u>. The designation of any Unrestricted Subsidiary as a Restricted Subsidiary shall constitute (i) the incurrence at the time of designation of any Investment, Indebtedness or Liens of such Subsidiary existing at such time and (ii) a return on any Investment by Borrower in Unrestricted Subsidiaries pursuant to the preceding sentence in an amount equal to the fair market value at the date of such designation of Borrower's Investment in such Subsidiary. In no event shall an Unrestricted Subsidiary be the legal owner or exclusive licensee of material Intellectual Property that is used in the business of Borrower and its Restricted Subsidiaries.

- 6.11. Accounts. Each Loan Party shall maintain its primary depository and operating accounts (other than accounts the aggregate average daily balance of which does not exceed \$5,000,000 for all such accounts) with HSBC or with any Lender(s).
 - 6.12. [<u>Reserved</u>].
- 6.13. <u>Post-Closing Obligations</u>. Borrower shall complete the actions specified in <u>Schedule 6.13</u> within the time periods specific therein, or such longer period of time that Agent may agree to in writing in its sole discretion.

VII. NEGATIVE COVENANTS.

No Loan Party shall, until payment in full in cash of the Obligations (other than (A) contingent indemnification obligations as to which no claim has been asserted and (B) obligations and liabilities under Lender-Provided Hedges and Bank Product Obligations), the termination of all Commitments under the Agreement, the cash collateralization of all Letters of Credit in accordance with this Agreement (or as otherwise acceptable to the Issuer in its sole discretion), and the termination of this Agreement:

7.1. Merger, Consolidation and Sale of Assets.

- (a) Enter into any merger, consolidation or other reorganization with or into any other Person or liquidate, wind up or dissolve itself (or suffer any liquidation or dissolution) or permit any other Person to consolidate with or merge with it; provided that (i) Restricted Subsidiaries of Borrower may merge into Borrower (so long as Borrower is the surviving entity of such merger), (ii) Loan Parties (other than Borrower) may merge into each other, and (iii) Foreign Subsidiaries may merge into each other.
- (b) Sell, lease, transfer or otherwise dispose of any of its properties or assets (including by an allocation of assets among newly divided limited liability companies pursuant to a "plan of division"), except (a) sales or other dispositions of assets in the ordinary course of business and (b) sales and other dispositions of assets outside the ordinary course of business having a fair market value, in the case of this clause (b), not to exceed (i) \$15,000,000 in the aggregate in any fiscal year and (ii) \$25,000,000 in the aggregate during the Term, so long as, in each case, the Net Cash Proceeds of any such disposition are remitted to Agent in accordance with, and to the extent required by, Section 2.14 (collectively, "Permitted Dispositions"); provided that, in the case of any disposition of Intellectual Property material and necessary for the operation of the assets of the Loan Parties which constitute Collateral, such Intellectual Property shall be subject to a non-exclusive royalty-free worldwide license in favor of Agent solely for the purpose of Agent's exercise of rights and remedies under this Agreement and the other Loan Documents in connection with the Collateral.
- (c) License material Intellectual Property (other than licenses entered into by a Loan Party in, or incidental to, the ordinary course of business), or amend or permit the amendment of any of the licenses in a manner that materially and adversely affects the right to receive payments thereunder, or in any manner that could reasonably be expected to materially impair, in the good faith judgment of the Agent, the value of the Intellectual Property or the Lien on and security interest in the Intellectual Property created therein hereby, without the consent of the Agent;

7.2. Creation of Liens; Negative Pledges.

(a) Create or suffer to exist any Lien or transfer upon or against any property or assets of any Loan Party or any Restricted Subsidiary now owned or hereafter acquired, except Permitted Encumbrances, or (b) enter into any agreement prohibiting the creation or assumption of any Lien upon its properties or assets now owned or hereafter acquired.

7.3. Guarantees.

Become liable upon the obligations of any Person by assumption, endorsement or guaranty thereof or otherwise except (i) to Agent, Issuer and Lenders, (ii) the endorsement of checks in the ordinary course of business, (iii) guarantees existing on the Closing Date and set forth on Schedule 7.3, (iv) guarantees by a Loan Party of Indebtedness of another Loan Party permitted by Section 7.7, (v) unsecured guarantees incurred in the ordinary course of business with respect to Permitted Indebtedness described in clause (k) of the definition thereof, (vi) unsecured guarantees arising with respect to customary indemnification obligations to purchasers in connection with Permitted Dispositions; (vii) unsecured guarantees with respect to Indebtedness of Borrower or one of its Subsidiaries, to the extent that the Person that is obligated under such guaranty could have incurred such underlying Indebtedness hereunder and (viii) to the extent such guarantees constitute Indebtedness, any other guarantees constituting Permitted Indebtedness.

7.4. Investments.

Directly or indirectly make any Investment, other than Permitted Investments; <u>provided</u> that to the extent any Investment includes dispositions of Intellectual Property material and necessary for the operation of the assets of the Loan Parties which constitute Collateral, such Intellectual Property shall be subject to a non-exclusive royalty-free worldwide license in favor of Agent solely for the purpose of Agent's exercise of rights and remedies under this Agreement and the other Loan Documents in connection with the Collateral.

7.5. [<u>Reserved</u>].

7.6. Restricted Payments.

Make any Restricted Payment; except that: (a) any Restricted Subsidiary of Borrower may pay dividends to Borrower or another Restricted Subsidiary of a Borrower; (b) Borrower may make distributions to Parent to the extent necessary to enable Parent to (i) pay "Cash Settlements" (as defined in the LLC Agreement) pursuant to and in accordance with the LLC Agreement, (ii) pay "Early Termination Payments" (as defined in the Tax Receivable Agreement) pursuant to and in accordance with the Tax Receivable Agreement and (iii) reimburse Parent for its expenses incurred on behalf of Borrower to the extent required by the LLC Agreement; provided, that, (Å) after giving effect to any such payment referenced in <u>clauses (i)</u>, <u>(ii)</u>, and <u>(iii)</u> (and any incurrence of Indebtedness in connection therewith), the then applicable financial covenant set forth in <u>Section 6.8(b)</u> hereof shall have been met, both as of the end of the most recently ended fiscal quarter prior to the making of such payment and immediately after giving effect to the making of such payment on a pro forma basis, and (B) no Event of Default has occurred and is continuing at the time of such payment or would result therefrom, and (C) Borrower shall have delivered to Agent a certificate certifying as to compliance with the requirements of clauses (A) and (B) above, together with reasonably detailed calculations evidencing compliance therewith; (c) Borrower may make payments to Parent for customary public company expenses; (d) Borrower may make, without duplication, (i) "Tax Distributions" (as defined in the LLC Agreement) in accordance with Section 4.01(b) of the LLC Agreement, (ii) for any taxable period for which Borrower is either wholly owned by Parent or a member of a consolidated, combined, unitary or similar tax group for U.S. federal and/or applicable state or local tax purposes, payments to Parent or the relevant member of such tax group to discharge the tax liabilities of Parent or the consolidated, combined, unitary or similar Tax liabilities of such tax group when and as due, in each case to the extent such liabilities are attributable to the income of Borrower and/or any Subsidiary of Borrower, (iii) payments to Parent in amounts sufficient to enable Parent to pay any franchise Taxes or other fees, Taxes and expenses required to maintain the corporate existence of Parent, and (iv) make "Tax Benefit Payments" (as defined in the Tax Receivable Agreement) pursuant to and in accordance with <u>Article III</u> of the Tax Receivable Agreement; <u>provided, however,</u> that payments described in this <u>clause (iv)</u> shall be made only to the extent amounts paid to Parent pursuant to this <u>Section 7.6(b)(ii)</u> in any fiscal year of Parent are less than or equal to the sum of Parent's actual income Tax liability (as determined in the reasonable discretion of Borrower and Parent) at any point within such fiscal year of Parent (including required quarterly estimated tax payments) and the amount of Tax Benefit Payments due and payable in such fiscal year of Parent; (e) any Non-Wholly Owned Subsidiary that is a Restricted Subsidiary of Borrower may make payments consisting of profit distributions to Persons holding non-controlling interests in such Non-Wholly Owned Subsidiary pursuant to the charter documents of such Non-Wholly Owned Subsidiary on a ratable basis; (f) Borrower may make distributions to former employees, officers, or directors of Parent (or any spouses, ex-spouses, or estates of any of the foregoing) on account of redemptions or repurchases of Equity Interests of Parent; and (g) that the Loan Parties may make other Restricted Payments so long as (i) no Default or Event of Default shall exist or would result from the making of such Restricted Payment and (ii) the Net Leverage Ratio, calculated on a pro forma basis as if such Restricted Payment had been made on the first day of the relevant testing period, is not greater than 1.00:1.00 as of the most recently ended fiscal quarter for which

financial statements shall have been delivered; <u>provided</u>, <u>further</u>, that from and after the period from consummation of a Material Acquisition until the earlier to occur of (x) the addition of the applicable Acquired Entity to the Loan Documents as a Loan Party (whether by merger with an existing Loan Party, joinder to the Loan Documents as a new Loan Party or otherwise) or (y) the delivery of Consolidating Annual Financials or Consolidating Quarterly Financials or for the most recent fiscal quarter or fiscal year (as the case may be) ended prior to the fiscal quarter in which the restricted payment is made, no voluntary or discretionary Restricted Payment shall be permitted other than Restricted Payments made pursuant to <u>clauses (b)</u>, (c), and (d).

7.7. Indebtedness.

Create, incur, assume or suffer to exist any Indebtedness except Permitted Indebtedness.

7.8. Nature of Business.

- (a) Substantially change the nature of the business in which it is presently engaged (it being understood that any other businesses reasonably related, ancillary or complementary thereto or a reasonable extension thereof shall be permitted); nor
- (b) except as permitted under this Agreement, purchase or invest, directly or indirectly, in any assets or property other than (i) purchases or investments in the ordinary course of business or (ii) purchases of, or investments in, assets or property which are useful in, necessary for, or reasonably related, ancillary or complementary to, or a reasonable extension of the business.

7.9. Transactions with Affiliates.

Directly or indirectly, purchase, acquire or lease any property from, or sell, transfer or lease any property to, any Affiliate, or enter into or permit to exist any transaction with any Affiliate except (a) transactions in the ordinary course of business or on an arm's-length basis on terms no less favorable than terms which would have been obtainable from a Person other than an Affiliate; (b) transactions between or among Borrower and/or any of its Restricted Subsidiaries (or an entity that becomes a Restricted Subsidiary as a result of such transaction), (c) transactions expressly permitted pursuant to Section 7.6, and (d) as expressly contemplated under (or by) the LLC Agreement, Tax Receivable Agreement and the other agreements and transactions described on Schedule 7.9.

7.10. [<u>Reserved</u>].

7.11. <u>Subsidiaries</u>. Form any Restricted Subsidiary unless (a) such Subsidiary is an Excluded Subsidiary or (b) to the extent such Subsidiary is not an Excluded Subsidiary, within 30 days of the date of formation of such Restricted Subsidiary, (x) such Restricted Subsidiary enters into a guaranty of the Obligations on terms acceptable to Agent and the Lenders, (y) such Restricted Subsidiary grants Agent for its benefit and for the ratable benefit of Lenders a perfected and first priority security interest in its assets in accordance with the terms of this Agreement and the other Loan Documents, and (z) Agent shall have received all documents, including customary legal opinions in form and substance reasonably

satisfactory to Agent which shall cover such customary matters incident to the transactions contemplated by this <u>Section 7.11(b)</u> and the other Loan Documents as Agent may reasonably require.

7.12. Fiscal Year and Accounting Changes.

Change its fiscal year from December 31 of each calendar year or make any change (i) in accounting treatment and reporting practices except as permitted by GAAP or (ii) in tax reporting treatment except as permitted by law.

7.13. Compliance with ERISA.

- (a) (x) Maintain, or permit any member of the Controlled Group to maintain, or (y) become obligated to contribute, or permit any member of the Controlled Group to become obligated to contribute, to any Plan, other than an employee welfare plan as defined in §3(1) of ERISA or those Plans disclosed on Schedule 5.8(d).
- (b) Engage, or permit any member of the Controlled Group to engage, in any non-exempt "prohibited transaction", as that term is defined in section 406 of ERISA and Section 4975 of the Code.
- (c) Incur, or permit any member of the Controlled Group to incur, any "accumulated funding deficiency", as that term is defined in Section 302 of ERISA or Section 412 of the Code.
- (d) Terminate, or permit any member of the Controlled Group to terminate, any Plan (other than an employee welfare plan as defined in § 3(1) of ERISA) where such event could result in any liability of any Loan Party or any member of the Controlled Group or the imposition of a lien on the property of any Loan Party or any member of the Controlled Group pursuant to Section 4068 of ERISA.
- (e) Assume, or permit any member of the Controlled Group to assume, any obligation to contribute to any Multiemployer Plan not disclosed on $\underline{\text{Schedule 5.8(d)}}$.
 - (f) Incur, or permit any member of the Controlled Group to incur, any withdrawal liability to any Multiemployer Plan.
 - (g) Fail promptly to notify Agent of the occurrence of any Termination Event.
- (h) Fail to comply, or permit a member of the Controlled Group to fail to comply, with the requirements of ERISA or the Code or other Applicable Laws in respect of any Plan.
- (i) Fail to meet, or permit any member of the Controlled Group to fail to meet, all minimum funding requirements under ERISA or the Code.
- (j) Postpone or delay or allow any member of the Controlled Group to postpone or delay any funding requirement with respect of any Plan.

7.14. Amendment of Documents and Material Contracts.

Amend, modify or waive any term or provision of its certificate of incorporation (or certificate of formation) or by-laws (or operating agreement, including, without limitation, the LLC Agreement) or equivalent document, any shareholders' agreement, the Tax Receivable Agreement or Material Contract in a manner materially adverse to Agent and the Lenders taken as a whole.

7.15. State of Organization.

Change the State in which it is incorporated or otherwise organized, unless it has given Agent not less than thirty (30) days prior written notice thereof.

7.16. Sanctions; Anti-Bribery Laws.

- (a) Directly or indirectly, use the proceeds of any Advance, or lend, contribute or otherwise make available such proceeds to any Subsidiary, joint venture partner or other Person, (i) to fund any activities or business of or with any Person, or in any country or territory, that, at the time of such funding, is the target of Sanctions, or (ii) in any other manner that would result in a violation of Sanctions by any Person (including any Person participating in the Advances, whether as administrative agent, collateral agent, issuing bank, arranger, bookrunner, underwriter, advisor, investor, lender or otherwise).
- (b) Directly or indirectly, use any part of the proceeds of any Advance for any payments that could constitute a violation of any applicable anti-bribery law.

VIII. CONDITIONS PRECEDENT.

8.1. Conditions to the Closing Date.

The effectiveness of this Agreement and the occurrence of the Closing Date are subject to the satisfaction, or waiver by all Lenders of the following conditions precedent:

- (a) <u>Loan Documents</u>. Agent shall have received executed counterparts of (i) this Agreement from each Loan Party, (ii) the Guaranty from each Guarantor, (iii) the Stock Pledge Agreement from each Loan Party that is a party thereto, in each case, in form and substance reasonably satisfactory to Agent and Lenders together with the share certificates referred to in the Stock Pledge Agreement (if any) and stock powers relating thereto (if applicable), and (iv) the other Loan Documents;
- (b) <u>Notes</u>. Agent shall have received for the account of each Lender requesting a Note, a Note duly executed and delivered by an Authorized Officer of Borrower;
- (c) <u>Filings, Registrations, Recordings and Searches</u>. Each document (including, without limitation, any UCC financing statement, termination statement or release) required by this Agreement or any other Loan Document or reasonably requested by Agent to be filed, registered or recorded in order to create, in favor of Agent, a perfected first priority security interest in or lien upon the Collateral shall have been properly filed, registered or recorded (or duly prepared for filing, registration or recording) in each jurisdiction in which the filing, registration or recordation thereof is so required or requested. Agent shall also have received customary UCC, intellectual property, tax, ERISA, litigation, bankruptcy and judgment lien searches (or the foreign equivalent thereof, if any) with respect to the Loan Parties in such jurisdictions as Agent shall reasonably require, and the results of such searches shall be reasonably satisfactory to Agent;
- (d) <u>Proceedings of the Loan Parties</u>. Agent shall have received a copy of the resolutions, in form and substance reasonably satisfactory to Agent, of the Board of Directors (or equivalent authority) of each Loan Party authorizing (i) the execution, delivery and performance of this Agreement and the other Loan Documents, the Notes (for Borrower only) and any related agreements by each Loan Party and (ii) the granting by each Loan Party of the security interests in and liens upon the Collateral, in each case certified by an Authorized Officer of such Loan Party as of the Closing Date; and,

such certificate shall state that the resolutions thereby certified have not been amended, modified, revoked or rescinded as of the date of such certificate;

- (e) <u>Incumbency Certificates of Loan Parties</u>. Agent shall have received a certificate of an Authorized Officer of each Loan Party, dated the Closing Date, as to the incumbency and signature of the officers of such Loan Party executing this Agreement, any certificate or other documents to be delivered by it pursuant hereto, together with evidence of the incumbency of such Authorized Officer;
 - (f) [Reserved];
 - (g) [Reserved];
- (h) <u>Certificates</u>. Agent shall have received a copy of the certificate of incorporation, certificate of organization or equivalent document of each Loan Party, and all amendments thereto, certified by the Secretary of State or other appropriate official of its jurisdiction of organization together with copies of the bylaws or operating agreement or other constitutive documents, as applicable, of each Loan Party certified as accurate and complete by an Authorized Officer of such Loan Party;
- (i) <u>Good Standing Certificates</u>. To the extent applicable in such Loan Party's jurisdiction of organization or formation, Agent shall have received good standing certificates or certificates of existences (or equivalent thereof, if any) for each Loan Party dated as of a recent date prior to the Closing Date, issued by the Secretary of State or other appropriate official of such Loan Party's jurisdiction of organization or formation:
- (j) <u>Legal Opinion</u>. Agent shall have received a customary, executed legal opinion of each of (i) Latham & Watkins LLP, (ii) Foley & Lardner LLP, special Florida counsel to the Loan Parties, and (iii) Ballard Spahr LLP, special Maryland counsel to the Loan Parties, in each case in form and substance reasonably satisfactory to Agent which shall cover such customary matters incident to the transactions contemplated by this Agreement and the other Loan Documents as Agent may reasonably require and each Loan Party hereby authorizes and directs such counsel to deliver such opinions to Agent and Lenders;
 - (k) [Reserved];
- (l) <u>Solvency Certificate</u>. Agent shall have received a solvency certificate executed by the chief financial officer of Borrower certifying that the representations and warranties set forth in <u>Section 5.8(a)</u> are true and correct as of the Closing Date and otherwise in form and substance satisfactory to Agent;
- (m) <u>Fees and Expenses</u>. Agent shall have received all fees payable to Agent and Lenders on or prior to the Closing Date pursuant to <u>Article III</u> and under the Fee Letter and all other fees and expenses incurred by Agent on or prior to the Closing Date;
- (n) <u>Financial Statements</u>. Agent shall have received (i) a copy of the Projections, (ii) the financial statements referred to in <u>Section 5.5(c)</u> and (<u>d</u>), and (iii) unaudited balance sheets, statements of income and statements of cash flow for Borrower for each fiscal month ended after the date of the applicable financial statements delivered pursuant to <u>clause (ii)</u> above and at least 15 days before the Closing Date, which shall, in each case, be reasonably satisfactory in all respects to Lenders;
- (o) <u>Indebtedness</u>. All Indebtedness of the Loan Parties not expressly permitted hereunder shall have been terminated or contemporaneously paid in full and any Liens on the assets of the Loan Parties securing such Indebtedness shall have been terminated;

- (p) <u>Insurance</u>. Subject to <u>Section 6.13</u>, Agent shall have received in form and substance satisfactory to Agent, lender loss payee endorsements on Agent's standard form of lender loss payee endorsement naming Agent as lender loss payee, and certified copies of the Loan Parties' liability insurance policies, together with endorsements naming Agent as a co-insured;
- (q) <u>Payment Instructions</u>. Agent shall have received written instructions from Borrower directing the application of proceeds of the initial Advances made pursuant to this Agreement;

(r) [Reserved];

(s) <u>No Material Adverse Effect</u>. Since December 31, 2020, there shall not have occurred any change in the business, assets, operations, results or condition (financial or otherwise) of Borrower and its Restricted Subsidiaries which has had, or would reasonably be expected to have, a Material Adverse Effect;

(t) [Reserved];

- (u) <u>Closing Certificate</u>. Agent shall have received a closing certificate signed by an Authorized Officer of Borrower dated as of the date hereof, stating that (i) all representations and warranties set forth in this Agreement and the other Loan Documents are true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations or warranties that already are qualified or modified as to "materiality" or "Material Adverse Effect" in the text thereof, which representations and warranties shall be true and correct in all respects subject to such qualification) on and as of such date as if made on and as of such date except to the extent such representations or warranties are limited by their terms to a specific date (in which case such representation or warranty shall be true and correct on and as of such earlier date in all material respects (except that such materiality qualifier shall not be applicable to any representations or warranties that already are qualified or modified as to "materiality" or "Material Adverse Effect" in the text thereof, which representations and warranties shall be true and correct in all respects subject to such qualification) on and as of such earlier date), (ii) the Loan Parties are on such date in compliance with all the terms and provisions set forth in this Agreement and the other Loan Documents, (iii) on such date no Default or Event of Default has occurred or is continuing or would result from the consummation of the Transactions, and (iv) confirming that the condition precedent set forth in <u>clause (s)</u> of this Section 8.1 has been satisfied; and
- (v) <u>Beneficial Ownership Certification</u>. To the extent Borrower qualifies as a "legal entity customer" under the Beneficial Ownership Regulation, at least three (3) Business Days prior to the Closing Date, if any Lender has requested, in a written notice to Borrower at least ten (10) Business Days prior to the Closing Date, a Beneficial Ownership Certification in relation to Borrower, then such Lender shall have received such Beneficial Ownership Certification.

8.2. Conditions to Each Advance.

The agreement of Lenders to make any Advance requested to be made on any date (including, without limitation, the initial Advance, but excluding Advances, the proceeds of which are to reimburse Agent for amounts drawn under a Letter of Credit), is subject to the satisfaction of the following conditions precedent as of the date such Advance is made:

(a) <u>Representations and Warranties</u>. Each of the representations and warranties made by each Loan Party in or pursuant to this Agreement or any other Loan Document shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations or warranties that already are qualified or modified as to "materiality" or "Material

Adverse Effect" in the text thereof, which representations and warranties shall be true and correct in all respects subject to such qualification) on and as of such date as if made on and as of such date except to the extent such representations or warranties are limited by their terms to a specific date (in which case such representation or warranty shall be true and correct on and as of such earlier date in all material respects (except that such materiality qualifier shall not be applicable to any representations or warranties that already are qualified or modified as to "materiality" or "Material Adverse Effect" in the text thereof, which representations and warranties shall be true and correct in all respects subject to such qualification) on and as of such earlier date);

- (b) No Default. No Event of Default or Default shall have occurred and be continuing on such date, or would exist after giving effect to the Advances requested to be made, on such date; <u>provided</u>, <u>however</u> that with the approval of Required Lenders (or of all Lenders if required under <u>Section 15.2(b)</u>), Lenders, in their sole discretion, may continue to make Advances notwithstanding the existence of an Event of Default or Default and that any Advances so made shall not be deemed a waiver of any such Event of Default or Default;
- (c) <u>Maximum Revolving Advances and Swingline Loans</u>. In the case of any Revolving Advances requested to be made, after giving effect thereto, the aggregate Revolving Advances shall not exceed the maximum amount of Revolving Advances permitted under <u>Sections 2.1</u> and <u>2.5</u>, and in the case of any Swingline Loans requested to be made, after giving effect thereto, the aggregate outstanding principal amount of Swingline Loans shall not exceed the Maximum Swingline Loan Amount;
- (d) <u>Maximum Letters of Credit</u>. In the case of any Letters of Credit requested to be made, after giving effect thereto, the aggregate face amount and reimbursement obligations outstanding in respect of Letters of Credit shall not exceed the maximum amount permitted under Section 2.9; and
- (e) <u>Notice of Borrowing</u>. Agent shall have received a completed Notice of Borrowing executed by Borrower and otherwise complying with the requirements of <u>Section 2.2</u>.

Each request for an Advance by Borrower hereunder shall constitute a representation and warranty by Borrower as of the date of such Advance that the conditions contained in this subsection shall have been satisfied.

IX. INFORMATION AS TO LOAN PARTIES.

The Loan Parties shall, until satisfaction in full in cash of the Obligations (other than (\underline{A}) contingent indemnification obligations as to which no claim has been asserted and (\underline{B}) obligations and liabilities under Lender-Provided Hedges and Bank Product Obligations) and the termination of this Agreement:

9.1. Disclosure of Material Matters.

Immediately upon learning thereof, report to Agent all matters materially affecting the value, enforceability or collectability of any portion of the Collateral including, without limitation, any Loan Party's reclamation or repossession of, or the return to any Loan Party of, a material amount of goods or claims or disputes asserted by any customer or other obligor.

- 9.2. [Reserved].
- 9.3. [Reserved].
- 9.4. Litigation.

Promptly notify Agent in writing of any litigation, suit or administrative proceeding affecting any Loan Party, whether or not the claim is covered by insurance, and of any suit or administrative proceeding, which in any such case would reasonably be expected to have a Material Adverse Effect.

9.5. Material Occurrences.

Promptly notify Agent in writing upon the occurrence of (a) any Event of Default or Default; (b) any event, development or circumstance whereby any financial statements or other reports furnished to Agent fail in any material respect to present fairly (and, solely with respect to the financial statements delivered pursuant to Section 9.7 and 9.8, in accordance with GAAP consistently applied) the financial condition or operating results of Loan Parties as of the date of such statements; and (c) any other development in the business or affairs of any Loan Party which would reasonably be expected to have a Material Adverse Effect; in each case describing the nature thereof and the action such Loan Party proposes to take with respect thereto.

9.6. [<u>Reserved</u>].

9.7. Annual Audited Financial Statements.

Furnish Agent within one hundred and twenty (120) days after the end of each fiscal year of Parent, consolidated financial statements of Parent including, but not limited to, consolidated statements of income and stockholders' equity and cash flow reflecting results of operations from the beginning of the current fiscal year to the end of such fiscal year, and the balance sheet as at the end of such fiscal year, all prepared in accordance with GAAP applied on a basis consistent with prior practices, and in reasonable detail and reported upon without qualification (other than any such exception, qualification or explanatory paragraph that is with respect to, or resulting from, (i) an upcoming maturity date of any Advances or other Indebtedness incurred in compliance with this Agreement or (ii) the activities, operations, financial results, assets or liabilities of any Unrestricted Subsidiary) (a "Report") by an independent certified public accounting firm selected by Parent and satisfactory to Agent (the "Accountants"); provided that, if Parent consummates a Material Acquisition (x) during the first 2 fiscal quarters of any fiscal year and the applicable Acquired Entity is not added to the Loan Documents as a Loan Party (whether by merger with an existing Loan Party, joinder to the Loan Documents as a new Loan Party or otherwise) by December 31st of such fiscal year, Borrower shall also deliver consolidated financial statements referenced above for such fiscal year or (y) during the 3rd or 4th fiscal quarter of any fiscal year, (1) Borrower shall also deliver consolidating management-prepared financial statements of the Parent concurrently with the consolidated financial statements referenced above for the immediately succeeding fiscal year if the Acquired Entity is not added to the Loan Documents as a Loan Party (whether by merger with an existing Loan Party, joinder to the Loan Documents as a new Loan Party or otherwise) by December 31st of the fiscal year if the Acquired Entity is not added to the Loan Docume

9.8. Quarterly Financial Statements.

Furnish Agent within forty-five (45) days after the end of each of the first three fiscal quarters of Parent, the unaudited consolidated balance sheet of Parent and unaudited consolidated statements of income and stockholders' equity and cash flow of Parent reflecting results of operations from the beginning of the fiscal year to the end of such fiscal quarter and for such fiscal quarter prepared on a basis consistent with prior practices and complete and correct in all material respects, subject to normal and recurring year-end adjustments that individually and in the aggregate are not material to the business of Parent; provided that, if Parent consummates a Material Acquisition (x) during the first 30 days of any fiscal quarter, Borrower shall also deliver consolidating management-prepared financial statements of the Parent with respect to such fiscal quarter (collectively, "Consolidating Quarterly Financials") concurrently with the consolidated financial statements referenced above for such fiscal quarter or (y) after the 30th day of any fiscal quarter, Borrower shall also deliver Consolidating Quarterly Financials of the Parent concurrently with the consolidated financial statements referenced above for the immediately succeeding fiscal quarter. Each such balance sheet, statement of income and stockholders' equity and statement of cash flow shall set forth a comparison of the figures for (x) the current fiscal period and (y) the current year-to-date with the figures for the same fiscal period and year-to-date period of the immediately preceding fiscal year. The financial statements shall be accompanied by a Compliance Certificate.

9.9. [<u>Reserved</u>].

9.10. Other Reports.

Furnish Agent as soon as available, but in any event within ten (10) days after the issuance thereof, copies of all annual, regular, periodic and special reports and registration statements which Parent may file or be required to file, copies of any report, filing or communication with the SEC under Section 13 or 15(d) of the Exchange Act, or with any Governmental Body that may be substituted therefor, or with any national securities exchange, and in any case not otherwise required to be delivered to the Agent pursuant hereto.

9.11. <u>Additional Information</u>.

Promptly upon request, furnish Agent with such additional information regarding the business, financial, legal or corporate affairs of any Loan Party or any Subsidiary thereof, or compliance with the terms of the Loan Documents, as the Agent or any Lender may from time to time reasonably request.

9.12. Projected Operating Budget.

Furnish Agent, within one hundred and twenty (120) days after the end of each fiscal year of Parent, quarter-by-quarter consolidated and, if applicable, consolidating projected operating budgets of Parent for such fiscal year (including a consolidated and, if applicable, consolidating income statement and cash flow statement for each quarter and a balance sheet as at the end of the last quarter and proposed business plan for such fiscal year), such projections to be accompanied by a certificate signed by an Authorized Officer of Borrower to the effect that such projections have been prepared on the basis of sound financial planning practice consistent with past budgets and financial statements and that such officer has no reason to question the reasonableness of any material assumptions on which such projections were prepared. In addition, Borrower will provide Agent with such projections and business plans as and when requested by Agent in connection with any request by Borrower to increase or permanently reduce the amount of Advances available under this Agreement.

9.13. [Reserved].

9.14. Notice of Suits, Adverse Events.

Furnish Agent with prompt written notice of (i) any lapse or other termination of any Consent issued to any Loan Party by any Governmental Body or any other Person that is material to the operation of such Loan Party's business, (ii) any refusal by any Governmental Body or any other Person to renew or extend any such Consent; and (iii) copies of any periodic or special reports filed by any Loan Party with any Governmental Body or Person, if such reports indicate any material change in the business, operations, affairs or condition of any Loan Party, or if copies thereof are requested by Agent or any Lender, (iv) copies of any material notices and other communications from any Governmental Body or Person which specifically relate to any Loan Party.

9.15. ERISA Notices and Requests.

Furnish Agent with immediate written notice in the event that (i) any Loan Party or any member of the Controlled Group knows or has reason to know that a Termination Event has occurred, together with a written statement describing such Termination Event and the action, if any, which any Loan Party or member of the Controlled Group has taken, is taking, or proposes to take with respect thereto and, when known, any action taken or threatened by the Internal Revenue Service, Department of Labor or PBGC with respect thereto, (ii) any Loan Party or any member of the Controlled Group knows or has reason to know that a prohibited transaction (as defined in Sections 406 of ERISA and 4975 of the Code) has occurred together with a written statement describing such transaction and the action which such Loan Party or any member of the Controlled Group has taken, is taking or proposes to take with respect thereto, (iii) a funding waiver request has been filed with respect to any Plan together with all communications received by any Loan Party or any member of the Controlled Group with respect to such request, (iv) any increase in the benefits of any existing Plan or the establishment of any new Plan or the commencement of contributions to any Plan to which any Loan Party or any member of the Controlled Group was not previously contributing shall occur, (v) any Loan Party or any member of the Controlled Group shall receive from the PBGC a notice of intention to terminate a Plan or to have a trustee appointed to administer a Plan, together with copies of each such notice, (vi) any Loan Party or any member of the Controlled Group shall receive any favorable or unfavorable determination letter from the Internal Revenue Service regarding the qualification of a Plan under Section 401(a) of the Code, together with copies of each such letter; (vii) any Loan Party or any member of the Controlled Group shall receive a notice regarding the imposition of withdrawal liability, together with copies of each such notice; (viii) any Loan Party or any member of the Controlled Group shall fail to make a required installment or any other required payment under Section 412 of the Code on or before the due date for such installment or payment; (ix) any Loan Party or any member of the Controlled Group knows that (a) a Multiemployer Plan has been terminated, (b) the administrator or plan sponsor of a Multiemployer Plan intends to terminate a Multiemployer Plan, or (c) the PBGC has instituted or will institute proceedings under Section 4042 of ERISA to terminate a Multiemployer Plan.

9.16. [<u>Reserved</u>].

9.17. Reconciliations.

(a) Furnish Agent, concurrently with the delivery of any financial statements pursuant to <u>Section 9.8</u>, (i) quarterly tax payment and distribution reconciliations and (ii) quarterly reconciliations of cash balances held by Parent, on the one hand, and Borrower and its Restricted Subsidiaries, on the other hand; <u>provided</u> that such reconciliations shall not be required if Borrower is required to deliver Consolidating Quarterly Financials in accordance with <u>Section 9.8</u>.

(b) Furnish Agent, concurrently with the delivery of any financial statements pursuant to <u>Section 9.7</u>, (i) annual tax payment and distribution reconciliations and (ii) annual reconciliations of cash balances held by Parent, on the one hand, and Borrower and its Restricted Subsidiaries, on the other hand; <u>provided</u> that such reconciliations shall not be required if Borrower is required to deliver Consolidating Annual Financials in accordance with <u>Section 9.7</u>.

9.18. Beneficial Ownership Documentation.

Promptly following any request therefor, provide to Agent or any Lender information and documentation reasonably requested by Agent or any Lender for purpose of compliance with applicable "know your customer" and anti-money-laundering rules and regulations, including, without limitation, the USA PATRIOT Act and the Beneficial Ownership Regulation.

9.19. [<u>Reserved</u>].

9.20. <u>Additional Documents</u>. Execute and deliver to Agent, upon request, such documents and agreements as Agent may, from time to time, reasonably request to carry out the purposes, terms or conditions of this Agreement.

X. EVENTS OF DEFAULT.

The occurrence of any one or more of the following events shall constitute an "Event of Default":

- 10.1. (a) Failure by Borrower to pay any principal of any Advance when due in accordance with the terms hereof; (b) failure by Borrower to pay any amount payable to Issuer when due in reimbursement of any drawing under a Letter of Credit; or (c) failure by Borrower to pay any interest or premium on any Advance, or any other amount payable hereunder or under any other Loan Document other than those specified in <u>clauses (a)</u> and <u>(b)</u> above, within three (3) Business Days after any such interest or other amount becomes due in accordance with the terms hereof or thereof;
- 10.2. (i) Failure by any Loan Party to perform, keep or observe any provision of Sections 4.2(a), 4.10, 4.11, 6.2(ii) (only with respect to Borrower), 6.4, 6.8, 6.13, Article VII, 9.5(a), 9.7, 9.8, 9.10 or 9.17 (ii) failure by any Loan Party to perform, keep or observe any provision of any covenant or agreement (other than those specified in Sections 10.1 or 10.2(2)(i) above) contained in any Loan Document on its part to be performed or observed and such failure continues for 30 days after the earlier of (x) the date on which such failure shall first become known to any officer of Borrower or (y) the date on which written notice thereof is given to Borrower by Agent or (iii) any representation or warranty made by any Loan Party in this Agreement or any other Loan Document or in any certificate, document or financial or other statement furnished at any time in connection herewith or therewith shall prove to have been misleading in any material respect (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) on the date when made or deemed to have been made;

10.3. [reserved];

- 10.4. Issuance of a notice of Lien, levy, assessment, injunction or attachment against a material portion of any Loan Party's property which is not stayed or lifted within sixty (60) days;
- 10.5. Failure or neglect of any Loan Party to perform, keep or observe any term, provision, condition, covenant contained in this Agreement, or contained in any other Loan Document, now or hereafter entered into between such Loan Party, Agent and/or any Lender (to the extent such breach is not

otherwise embodied in any other provision of this $\underline{\text{Article } X}$ for which a different grace or cure period is specified or which constitute an immediate Event of Default under this Agreement or the Loan Documents), and such failure continues for thirty (30) days after the earlier of (x) the date on which such failure shall first become known to any officer of Borrower or (y) the date on which written notice thereof is given to Borrower by Agent;

- 10.6. Any judgment or judgments are rendered or judgment liens filed against any Loan Party for an aggregate amount in excess of \$5,000,000 (except to the extent fully covered (other than to the extent of customary deductibles) by insurance pursuant to which the insurer has not denied coverage) which within thirty (30) days of such rendering or filing is not either satisfied, stayed or discharged of record;
- 10.7. Any Loan Party or any Restricted Subsidiary shall (i) apply for, consent to or suffer the appointment of, or the taking of possession by, a receiver, custodian, trustee, liquidator or similar fiduciary of itself or of all or a substantial part of its property, (ii) make a general assignment for the benefit of creditors, (iii) commence a voluntary case under any state or federal bankruptcy laws (as now or hereafter in effect), (iv) be adjudicated a bankrupt or insolvent, (v) file a petition seeking to take advantage of any other law providing for the relief of debtors, (vi) acquiesce to, or fail to have dismissed, within sixty (60) days, any petition filed against it in any involuntary case under such bankruptcy laws, or (vii) take any action for the purpose of effecting any of the foregoing;
- 10.8. Any Loan Party or any Subsidiary of a Loan Party shall admit in writing its inability, or be generally unable, to pay its debts as they become due or cease operations of its present business;
- 10.9. The validity or enforceability of any Loan Document shall at any time for any reason (other than solely as the result of an action or failure to act on the part of Agent) be declared to be null and void, or a proceeding shall be commenced by a Loan Party or its Subsidiaries, or by any Governmental Body having jurisdiction over a Loan Party or its Subsidiaries, seeking to establish the invalidity or unenforceability thereof, or a Loan Party or its Subsidiaries shall deny that such Loan Party or its Subsidiaries has any liability or obligation purported to be created under any Loan Document;
 - 10.10. Any Cross-Default shall occur;
- 10.11. Termination or breach of any Guaranty or similar agreement executed and delivered to Agent in connection with the Obligations of Borrower, or if any Guarantor of the Obligations attempts to terminate, challenges the validity of, or its liability under, any such Guaranty or similar agreement;
 - 10.12. Any Change of Control shall occur;
- 10.13. The Guaranty or any other Loan Document that purports to create a Lien, shall, for any reason, fail or cease to create a valid and perfected and, (except to the extent of Permitted Encumbrances which are non-consensual Permitted Encumbrances, permitted purchase money Liens or the interests of lessors under Capital Leases) first priority Lien on the Collateral covered thereby, except (a) as a result of a disposition of the applicable Collateral in a transaction permitted under this Agreement or (b) as the result of an action or failure to act on the part of Agent;
- 10.14. An event or condition specified in <u>Section 7.15</u> or <u>Section 9.15</u> shall occur or exist with respect to any Plan and, as a result of such event or condition, together with all other such events or conditions, any Loan Party or any member of the Controlled Group shall incur, or in the opinion of Agent be reasonably likely to incur, a liability to a Plan or the PBGC (or both) which, in the reasonable judgment of Agent, would have a Material Adverse Effect.

10.15. Equity Cure.

- (a) Notwithstanding anything to the contrary contained in Article X or XI, in the event that Parent or Borrower, as applicable, fails to comply with the requirements of the financial covenants set forth in Section 6.8, then from the end of the most recently ended fiscal quarter of Borrower until the expiration of the tenth Business Day subsequent to the date the relevant financial statements and related Compliance Certificate are required to be delivered pursuant to Section 9.7 or 9.8 (the last day of such period being the "Anticipated Cure Deadline"), Parent and Borrower, as applicable, shall have the right (the "Cure Right") to issue common Equity Interests (or another form of Equity Interests reasonably acceptable to Agent) to, in the case of an exercise by the Borrower of the Cure Right, the Permitted Holders, for cash, and contribute the proceeds therefrom in immediately available funds to Borrower, in the case of an exercise by the Parent of the Cure Right ("Cure Equity"), and upon the receipt by Borrower of such cash (the "Cure Amount") pursuant to the exercise of such Cure Right, the calculation of Consolidated EBITDA as used in the financial covenants set forth in Section 6.8 shall be recalculated giving effect to the following pro forma adjustments:
- (i) Consolidated EBITDA for such fiscal quarter (and for any subsequent period that includes such fiscal quarter) shall be increased, solely for the purpose of measuring the financial covenants set forth in Section 6.8 and not for any other purpose under this Agreement (including but not limited to determining pricing or the availability or amount of any covenant baskets or carve-outs); provided that (1) the receipt by Borrower of the Cure Amount pursuant to the Cure Right shall be deemed to have no other effect on a consolidated basis under this Agreement (including but not limited to determining pricing or the availability or amount of any covenant baskets or carve-outs); and (2) no Cure Amount shall reduce Indebtedness on a pro forma basis for the fiscal quarter for which the Cure Right was exercised for purposes of calculating the financial covenants set forth in Section 6.8 (whether as a result of a prepayment of the Advances or via netting of such Cure Amount);
- (ii) if, after giving effect to the foregoing recalculations, Parent or Borrower, as applicable, shall then be in compliance with the requirements of the financial covenants set forth in Section 6.8, Borrower shall be deemed to have satisfied the requirements of the financial covenants set forth in Section 6.8 as of the relevant date of determination with the same effect as though there had been no failure to comply therewith at such date, and the applicable breach or default of the financial covenants set forth in Section 6.8 that had occurred (and any other Default as a result thereof, including the failure to meet any condition requiring no Default or Event of Default based solely on the basis of any actual or purported Event of Default under the financial covenants set forth in Section 6.8) shall be deemed cured for the purposes of this Agreement; and
- (iii) upon receipt by Agent of notice from Borrower, on or prior to the Anticipated Cure Deadline, that Parent or Borrower intends to exercise the Cure Right in respect of a fiscal quarter, neither Agent nor the Lenders (i) shall not be permitted to accelerate Advances held by them, to terminate the Revolving Commitments held by them or to exercise remedies against the Collateral on the basis of a failure to comply with the requirements of the financial covenants set forth in Section 6.8, unless such failure is not cured pursuant to the exercise of the Cure Right on or prior to the Anticipated Cure Deadline and (ii) shall not be obligated to make any Advances until such Cure Amount has been received by Borrower.
- (b) Borrower shall promptly notify Agent of its receipt of the Cure Amount and shall immediately apply the full amount of the Cure Amount to the payment of outstanding Advances in the manner specified in <u>Section 2.7(b)</u>.

(c) Notwithstanding anything to the contrary contained herein, (i) the Cure Right may not be exercised not more than 3 times during the term of this Agreement, (ii) the Cure Right may not be exercised (x) more than 2 times in any twelve fiscal month period or (y) in consecutive fiscal quarters and (iii) the amount of any Cure Amount will be no greater than the amount required to cause Parent or Borrower, as applicable, to be in compliance with all of the financial covenants set forth in Section 6.8 as at such date (any amount contributed in excess of the amount required to cause Parent or Borrower, as applicable, to be in compliance with all of the financial covenants set forth in Section 6.8 as at such date shall not constitute Cure Equity and shall be used to prepay the outstanding Advances in accordance with Section 2.7(b)).

XI. <u>LENDERS' RIGHTS AND REMEDIES AFTER DEFAULT</u>.

11.1. Rights and Remedies.

Upon the occurrence and during the continuance of (i) an Event of Default pursuant to Section 10.7 or 10.8, all Obligations shall be immediately due and payable and the obligation of Lenders to make Advances shall be deemed terminated, or (ii) the other Events of Default specified in Article X, and at any time thereafter (such Event of Default not having previously been waived in accordance with this Agreement), Agent may (but shall not be obligated to) (and at the direction of the Required Lenders shall) upon written notice to the Loan Parties, declare all of the Obligations immediately due and payable and terminate this Agreement and terminate the obligation of Lenders to make Advances and (iii) a filing of a petition against any Loan Party in any involuntary case under any state or federal bankruptcy laws, the obligation of Lenders to make Advances hereunder shall be terminated other than as may be required by an appropriate order of the bankruptcy court having jurisdiction over the Loan Parties. Upon the occurrence and continuation of any Event of Default, Agent shall have the right to exercise any and all other rights and remedies provided for herein, under the UCC and at law or equity generally, including, without limitation, the right to foreclose the security interests granted herein and to realize upon any Collateral by any available judicial procedure and/or to take possession of and sell any or all of the Collateral with or without judicial process. In furtherance of such exercise Agent may enter any Loan Party's premises or other premises without legal process and without incurring liability to such Loan Party therefor, and Agent may thereupon, or at any time thereafter, in its discretion without notice or demand, take the Collateral and remove the same to such place as Agent may deem advisable and Agent may require such Loan Party to make the Collateral available to Agent at a convenient place. With or without having the Collateral at the time or place of sale, Agent may sell the Collateral, or any part thereof, at public or private sale, at any time or place, in one or more sales, at such price or prices, and upon such terms, either for cash, credit or future delivery, as Agent may elect. Except as to that part of the Collateral which is perishable or threatens to decline speedily in value or is of a type customarily sold on a recognized market, Agent shall give the Loan Parties reasonable notification of such sale or sales, it being agreed that in all events written notice mailed to the Loan Parties at least ten (10) days prior to such sale or sales is reasonable notification. At any public sale Agent or any Lender may bid for and become the purchaser, and Agent, any Lender or any other purchaser at any such sale thereafter shall hold the Collateral sold absolutely free from any claim or right of whatsoever kind, including any equity of redemption and such right and equity are hereby expressly waived and released by each Loan Party. Agent may specifically disclaim any warranties of title or the like at any sale of Collateral. In connection with the exercise of the foregoing remedies, Agent is granted permission to use without charge all of each Loan Party's trademarks, trade styles, trade names, patents, patent applications, licenses, franchises and other proprietary rights which are used in connection with the foregoing; provided, that with respect to any Event of Default resulting solely from noncompliance with a financial covenant set forth in Section 6.8, neither Agent nor the Required Lenders may exercise the foregoing remedies in this Section 11.1 until the date that is the earlier of (i) the Anticipated Cure Deadline, and (ii) the date that Agent receives notice that the Parent will not exercise the Cure Right for the applicable fiscal quarter.

11.2. <u>Application of Proceeds</u>.

The proceeds realized from the sale of any Collateral shall be applied as follows: <u>first</u>, to the reasonable costs, expenses and attorneys' fees and expenses incurred by Agent for collection and for acquisition, completion, protection, removal, storage, sale and delivery of the Collateral; <u>second</u>, to interest due upon any of the Swingline Loans; <u>third</u>, to the principal of the Swingline Loans; <u>fourth</u>, to fees payable in connection with this Agreement; <u>fifth</u>, to interest due upon any of the Obligations; <u>sixth</u>, ratably, to (i)furnish to Agent cash collateral in an amount not less than 105% of the Letter of Credit Reserve, such cash collateral arrangements to be in form and substance reasonably satisfactory to Agent and (ii) ratably, to the principal of the Obligations, including Lender-Provided Hedges and Bank Product Obligations; and <u>seventh</u>, ratably, to the other Obligations. All such payments hereunder shall be shared ratably among the Lenders holding such Obligations based upon their respective Commitment Percentages. If any deficiency shall arise, each Loan Party shall remain liable to Agent and Lenders therefor. If it is determined by an authority of competent jurisdiction that a disposition by Agent did not occur in a commercially reasonable manner, Agent may obtain a deficiency judgment for the difference between the amount of the Obligation and the amount that a commercially reasonable sale would have yielded. Agent will not be considered to have offered to retain the Collateral in satisfaction of the Obligations unless Agent, subject to <u>Section 15.2(b)</u> hereof, has entered into a written agreement with Loan Parties to that effect.

11.3. Agent's Discretion.

Agent (acting with the consent of the Required Lenders or all Lenders to the extent required by Section 15.2(b)) shall have the right in its sole discretion to determine which rights, Liens, security interests or remedies Agent may at any time pursue, relinquish, subordinate, or modify or to take any other action with respect thereto and such determination will not in any way modify or affect any of Agent's or Lenders' rights hereunder; provided, however, that, in the absence of such direction, Agent may (but shall not be obligated or have any duty to) take such action, or refrain from taking such action, with respect to any Event of Default as it shall deem advisable and in the best interests of the Lenders and the Issuer and solely to the extent permitted hereunder or pursuant to the other Loan Documents; provided, further, that Agent shall not be obligated to follow any direction by Required Lenders if Agent reasonably determines that such direction is in conflict with any provisions hereunder or under any Applicable Law, and Agent shall not, under any circumstances, be liable to any Lenders, the Issuer, Borrower, the Guarantors or any other Person for following the direction of Required Lenders. At all times, if Agent acting at the direction of the Required Lenders advises the Lenders that it wishes to proceed in good faith with respect to any of its rights and remedies, each of the Lenders will cooperate in good faith with respect to such rights and remedies and will not unreasonably delay the enforcement of the security interests provided for herein.

11.4. <u>Setoff</u>.

In addition to any other rights which Agent, any Lender or the Issuer may have under Applicable Law, upon the occurrence and during the continuation of an Event of Default hereunder, Agent, such Lender and such Issuer and their Affiliates shall have a right to apply any Loan Party's property held by Agent, such Lender or such Issuer and their Affiliates to reduce the Obligations. If any party (or its Affiliate) exercises the right of setoff provided for hereunder, such party shall be obligated to share any such setoff in accordance with Section 2.13(d).

11.5. <u>Rights and Remedies Not Exclusive</u>.

The authority to enforce rights and remedies under this Agreement and other Loan Documents against Borrower or any Guarantor shall be vested in, and all actions and proceedings at law in connection with such enforcement shall be instituted and maintained by, Agent for the benefit of all the Lenders and

the Issuer. The enumeration of the foregoing rights and remedies is not intended to be exhaustive and the exercise of any right or remedy shall not preclude the exercise of any other right or remedies provided for herein or otherwise provided by Applicable Law, all of which shall be cumulative and not alternative.

XII. WAIVERS AND JUDICIAL PROCEEDINGS.

12.1. Waiver of Notice.

Each Loan Party hereby waives notice of non-payment of any of the Receivables, demand, presentment, protest and notice thereof with respect to any and all instruments, notice of acceptance hereof, notice of loans or advances made, credit extended, Collateral received or delivered, or any other action taken in reliance hereon, and all other demands and notices of any description, except such as are expressly provided for herein to the extent not prohibited by law.

12.2. <u>Delay</u>.

No delay or omission on Agent's or any Lender's part in exercising any right, remedy or option shall operate as a waiver of such or any other right, remedy or option or of any default.

12.3. Jury Waiver.

EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION (A) ARISING UNDER THIS AGREEMENT OR ANY OTHER INSTRUMENT, DOCUMENT OR AGREEMENT EXECUTED OR DELIVERED IN CONNECTION HEREWITH, OR (B) IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO THIS AGREEMENT OR ANY OTHER INSTRUMENT, DOCUMENT OR AGREEMENT EXECUTED OR DELIVERED IN CONNECTION HEREWITH, OR THE TRANSACTIONS RELATED HERETO OR THERETO IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT OR TORT OR OTHERWISE AND EACH PARTY HEREBY CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENTS OF THE PARTIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

XIII. EFFECTIVE DATE AND TERMINATION.

13.1. Term.

This Agreement, which shall inure to the benefit of and shall be binding upon the respective successors and permitted assigns of each Loan Party, Agent and each Lender, shall become effective on the date hereof and shall continue in full force and effect until the earliest of (x) October 29, 2024, (y) the acceleration of all Obligations pursuant to the terms of this Agreement or (z) the date on which this Agreement shall be terminated in accordance with the provisions hereof or by operation of law (the "Termination Date"). Borrower may permanently reduce the Maximum Revolving Advance Amount, at any time upon ten (10) days' prior written notice to Agent, upon payment in full in cash of any Obligations in excess of the maximum amount of Revolving Advances then permitted under Section 2.1 and payment of all amounts which may be owed pursuant to Section 15.5(b), after giving effect to the reduced Maximum Revolving Advance Amount; provided that the Maximum Revolving Advance

Amount may not be reduced to an amount less than \$25,000,000 except in connection with the termination of all Commitments and the payment in full in cash of all Obligations.

13.2. <u>Termination</u>.

The termination of the Agreement shall not affect any Loan Party's, Agent's or any Lender's rights, or any of the Obligations having their inception prior to the effective date of such termination, and the provisions hereof shall continue to be fully operative until all transactions entered into, rights or interests created or Obligations have been fully disposed of, concluded or liquidated. The security interests, Liens and rights granted to Agent and Lenders hereunder and the financing statements filed hereunder shall continue in full force and effect, notwithstanding the termination of this Agreement or the fact that Borrower's Account may from time to time be temporarily in a zero or credit position, until all of the Obligations of the Loan Parties have been paid or performed in full after the termination of this Agreement or the Loan Parties have furnished Agent and Lenders with an indemnification satisfactory to Agent and Lenders with respect thereto. Accordingly, each Loan Party waives any rights which it may have under Section 9-513(c) of the UCC to demand the filing of termination statements with respect to the Collateral, and Agent shall not be required to send such termination statements to any Loan Party, or to file them with any filing office, unless and until this Agreement shall have been terminated in accordance with its terms and all Obligations (other than Obligations exclusively with respect to the Letter of Credit Reserve if Borrower shall have furnished Agent with cash collateral in an amount not less than 105% of the Letter of Credit Reserve) paid in full in cash in immediately available funds. All representations, warranties, covenants, waivers and agreements contained herein or in any Loan Document shall survive termination hereof until all Obligations are paid or performed in full in cash, as such representations and warranties may change from time to time as contemplated in Section 8.2.

XIV. REGARDING AGENT.

14.1. Appointment.

Each Lender hereby irrevocably appoints and designates HSBC to act as Agent for such Lender under this Agreement and the other Loan Documents. Each Lender hereby irrevocably authorizes Agent to take such action on its behalf under the provisions of this Agreement and the other Loan Documents and to exercise such powers and to perform such duties hereunder and thereunder as are specifically delegated to or required of Agent by the terms hereof and such other powers as are reasonably incidental thereto and Agent shall hold all Collateral, payments of principal and interest, fees (except the fees set forth in Section 3.4 and the Fee Letter), charges and collections (without giving effect to any collection days) received pursuant to this Agreement, for the ratable benefit of Lenders. Agent may perform any of its duties hereunder by or through its agents or employees. As to any matters not expressly provided for by this Agreement (including without limitation, collection of the Notes), Agent shall not be required to exercise any discretion or take any action, but shall be required to act or to refrain from acting (and shall be fully protected in so acting or refraining from acting) upon the written instructions of the Required Lenders or all Lenders to the extent required under Section 15.2(b), and such instructions shall be binding; provided, however, that Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose Agent to liability or that is contrary to any Loan Document or Applicable Law, including, for the avoidance of doubt, any action that may be in violation of the automatic stay or that may effect a forfeiture, modification or termination of a property interest in violation of any applicable bankruptcy/insolvency laws and Agent shall in all cases be fully justified in failing or refusing to act under this Agreement or any other Loan Document unless it first receives further assurances of its indemnification from the Lenders that Agent reasonably believes it

Unless otherwise consented to in writing by Agent (acting at the direction of the Required Lenders), no Lender or Issuer, individually or together with any other Lenders or the Issuer, shall have the right, nor shall it, exercise or enforce any of the rights, powers or remedies which Agent is authorized to exercise or enforce under this Agreement or any of the other Loan Documents.

14.2. Nature of Duties.

Agent shall have no duties, obligations or responsibilities except those expressly set forth in this Agreement and the other Loan Documents, including, without limitation, any security agreements, documents or instruments executed in connection therewith, and its duties thereunder shall be administrative in nature only, whether or not a Default or Event of Default has occurred and is continuing. None of Agent, any Lender, or the Issuer nor any of their respective officers, directors, employees or agents shall be (i) liable for any action taken or omitted by them as such hereunder or in connection herewith, unless caused by their gross negligence or willful misconduct (as determined by a court of competent jurisdiction in a final and non-appealable judgment), or (ii) responsible in any manner for (A) any recitals, statements, representations or warranties made by any Loan Party or any officer thereof contained in this Agreement or in any of the Loan Documents, or in any certificate, report, statement or other document referred to or provided for in, or received by Agent under or in connection with, this Agreement or any of the Loan Documents, or (B) the value, validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any of the Loan Documents, or of the Collateral held by Agent hereunder, (C) any failure of any Loan Party to perform their respective obligations hereunder, (D) the validity, perfection, priority or enforceability of the Liens in any of the Collateral, whether impaired by operation of law or by reason of any action or omission to act on Agent's part hereunder, or (E) the validity of the title of Borrower to the Collateral, insuring the Collateral or the payment of taxes, charges, assessments or Liens upon the Collateral or otherwise as to the maintenance of the Collateral. Agent shall not be responsible for, have any duty or be under any obligation to any Lender to ascertain or to inquire as to (i) any statement, warranty or representation made in or in connection with the Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the observance or performance of any of the covenants or agreements contained in, or terms or conditions of, this Agreement or any of the Loan Documents, including the satisfaction of any conditions precedent set forth in this Agreement, other than to confirm receipt of items expressly required to be delivered to Agent, or the occurrence of any Default or Event of Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document, or (v) to inspect the properties, books or records of any Loan Party. Agent shall not have any duty to disclose, and shall not be liable for the failure to disclose, except as expressly set forth in this Agreement and in the other Loan Documents, any information relating to Borrower or any of its Affiliates that is communicated to or obtained by Agent or any of its Affiliates in any capacity. Agent shall not be liable for any damages arising for the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated thereby. In no event shall Agent be required to expend or risk any of its own funds or otherwise incur any liability, financial or otherwise, in the performance of its duties under the Loan Documents or in the exercise of any of its rights or powers under this Agreement or the other Loan Documents. The duties of Agent as with respect to the Advances shall be mechanical and administrative in nature; Agent shall not have by reason of this Agreement a fiduciary relationship in respect of any Lender; and nothing in this Agreement, expressed or implied, is intended to or shall be so construed as to impose upon Agent any obligations in respect of this Agreement except as expressly set forth herein.

Use of the term "agent" in the Agreement or in any other Loan Documents (or any other similar term) with reference to Agent does not connote (and is not intended to connote), any fiduciary or other implied (or express) obligation arising under agency doctrine of any applicable law. Instead such term is used as a matter of market custom, and is intended to create or reflect only an administrative relationship between the contracting parties.

Agent shall not incur any liability for not performing any act or fulfilling any duty, obligation or responsibility hereunder by reason of any occurrence beyond the control of Agent (including but not limited to any act or provision of any present or future law or regulation or Governmental Body, any act

of God or war, civil unrest, local or national disturbance or disaster, any act of terrorism, or the unavailability of the Federal Reserve Bank wire or facsimile or other wire or communication facility).

14.3. Lack of Reliance on Agent and Resignation.

- (a) Independently and without reliance upon Agent, the Issuer or any other Lender, each Lender has made and shall continue to make (i) its own independent investigation of the financial condition and affairs of each Loan Party in connection with the making and the continuance of the Advances hereunder and the taking or not taking of any action in connection herewith, and (ii) its own appraisal of the creditworthiness of each Loan Party. Except as set forth in Section 14.9, Agent shall have no duty or responsibility, either initially or on a continuing basis, to provide any Lender with any credit or other information with respect thereto, whether coming into its possession before making of the Advances or at any time or times thereafter. Agent shall not be responsible to any Lender for any recitals, statements, information, representations or warranties herein or in any agreement, document, certificate or a statement delivered in connection with or for the execution, effectiveness, genuineness, validity, enforceability, collectability or sufficiency of this Agreement or any other Loan Document, or of the financial condition of any Loan Party, or be required to make any inquiry concerning either the performance or observance of any of the terms, provisions or conditions of this Agreement, the Notes, the Loan Documents or the financial condition of any Loan Party, or the existence of any Event of Default or any Default.
- (b) Any entity into which Agent in its individual capacity may be merged or converted or with which it may be consolidated, or any corporation resulting from any merger, conversion or consolidations which Agent in its individual capacity may be party, or any corporation to which substantially all of the corporate trust or agency business of Agent in its individual capacity may be transferred, shall be Agent under this Agreement without further action.
- (c) Agent may resign on thirty (30) days' written notice to each of Lenders, the Issuer and Borrower (the "Resignation Effective Date") and upon such resignation, the Required Lenders will promptly designate a successor Agent reasonably satisfactory to Borrower. If no such successor Agent is appointed at the end of such thirty (30) day period, Agent may (but shall not be obligated to) designate one of the Lenders as a successor Agent, subject to the acceptance of such Lender; provided that, if no Lender accepts appointment as a successor Agent, then Agent shall be entitled to apply to a court of competent jurisdiction for the appointment of a successor Agent or for other appropriate relief. The costs and expenses (including its' attorneys' fees and expenses) incurred by Agent in connection with such proceeding shall be paid by the Lenders.
- (d) With effect from the Resignation Effective Date, (i) the retiring Agent shall be discharged from its duties and obligations under this Agreement and other Loan Documents and (ii) except for any indemnity payments owed to the retiring or removed Agent, all payments, communications and determinations provided to be made by, to or through Agent shall instead be made by or to each Lender and the Issuer directly, until such time, if any, a successor Agent has been appointed as provided for above. Upon the acceptance of a successor's appointment as Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring or removed Agent (other than any rights to indemnity payments owed to the retiring Agent). Until a successor Agent is appointed, the retiring Agent shall continue to hold the Liens on the Collateral on

behalf of the Lenders, Issuer and Swingline Lender, but otherwise shall be discharged from all of its other duties and obligations under this Agreement and other Loan Documents as of the Resignation Effective Date. Notwithstanding the Resignation Effective Date, all rights of the retiring Agent (and any sub-agent) to indemnification by Borrower and Lenders shall continue in effect for the benefit of such retiring Agent, its sub-agents and their Indemnitees in respect of any action taken or omitted to be taken by any of them while the retiring Agent was acting as Agent. After Agent's resignation as Agent, the provisions of this <u>Article XIV</u> shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Agent under this Agreement.

14.4. Certain Rights of Agent.

If Agent shall request instructions from Lenders with respect to any act or action (including failure to act) in connection with this Agreement or any other Loan Document, Agent shall be entitled to refrain from such act or taking such action unless and until Agent shall have received written instructions from the Required Lenders; and Agent shall not incur liability to any Person by reason of so refraining. Without limiting the foregoing, Lenders shall not have any right of action whatsoever against Agent as a result of its acting or refraining from acting hereunder in accordance with the instructions of the Required Lenders or all Lenders as required by Section 15.2(b).

Agent shall not have any duty to take any discretionary action or exercise any discretionary power, except discretionary rights and powers expressly contemplated by the Agreement that Agent is required to exercise and only so long as so directed in writing to take such discretionary action by the Required Lenders *provided*, *however*, that Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose Agent to liability or that is contrary to any Loan Document or Applicable Law, including, for the avoidance of doubt, any action that may be in violation of the automatic stay or that may effect a forfeiture, modification or termination of a property interest in violation of any applicable bankruptcy/insolvency laws and Agent shall in all cases be fully justified in failing or refusing to act under the Agreement or any other Loan Document unless it first receives further assurances of its indemnification from the Lenders that Agent reasonably believes it may require, including prepayment of any related expenses and any other protection it requires against any and all costs, expenses and liabilities it may incur in taking or continuing to take any such discretionary action at the direction of the Required Lenders.

14.5. Reliance.

Agent shall be entitled to conclusively rely, and shall be fully protected in and shall not incur any liability for relying, upon any note, writing, resolution, notice, request, consent, statement, certificate, instrument, teletype, cablegram, order, document or other writing (including any electronic or telecopier message, internet or intranet website posting or other distribution) in good faith believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or entity, and, with respect to all legal matters pertaining to this Agreement and the other Loan Documents and its duties hereunder, upon advice of counsel selected by it. Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition to the making of any Advance, Agent may presume that such condition is satisfactory to such Lender or Issuer unless Agent receives notice to the contrary from such Lender or Issuer prior to the making of such Advance.

Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents or attorneys-in-fact appointed by Agent. Agent and any such sub-agent or attorney-in-fact may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of the Agreement shall apply to any such sub-agent or attorney-in-fact. Agent shall not be responsible for the negligence or misconduct of any sub-agent or attorney-in-fact except to the extent that a court of

competent jurisdiction determines in a final and non-appealable judgment that Agent acted with gross negligence or willful misconduct in the selection of such sub-agent or attorney-in-fact.

Agent shall be entitled to take any action or refuse to take any action which Agent regards as necessary for Agent to comply with any Applicable Law, regulation or court order. Agent may consult with legal counsel, independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

14.6. Notice of Default.

Except for actual knowledge of non-payment of the Obligations, Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default hereunder or under the Loan Documents, except with respect to defaults in the payment of principal, interest and fees required to be paid to Agent for the account of the Lenders, unless Agent has received written notice from a Lender or any Loan Party referring to this Agreement or the other Loan Documents, describing such Default or Event of Default and stating that such notice is a "notice of default". In the event that Agent receives such a notice, Agent shall give prompt notice thereof to Lenders. Agent shall take such action with respect to such Default or Event of Default (including, without limitation, the institution of the Default Rate pursuant to Section 3.1 hereof) as shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default (including, without limitation, the institution of the Default Rate pursuant to Section 3.1 hereof) as it shall deem advisable in the best interests of Lenders.

14.7. <u>Indemnification</u>.

To the extent Agent is not reimbursed and indemnified by the Loan Parties for any amount required under <u>Section 15.5(a)</u> to be paid by the Loan Parties to Agent, each Lender will severally reimburse Agent in proportion to its respective portion of the Advances (or, if no Advances are outstanding, according to its Commitment Percentage), for such unpaid amount (including any such unpaid amount in respect of a claim asserted by such Lender); <u>provided</u> that Lenders shall not be liable for any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements resulting from the indemnified party's gross negligence or willful misconduct (as determined by a court of competent jurisdiction in a final and non-appealable judgment); and <u>provided</u> <u>further</u>, that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against Agent in its capacity as such.

14.8. Agent in its Individual Capacity.

With respect to the obligation of Agent to lend under this Agreement, the Advances made by it shall have the same rights and powers hereunder as any other Lender and as if it were not performing the duties as Agent specified herein; and the term "Lender" or any similar term shall, unless the context clearly otherwise indicates, include Agent in its individual capacity as a Lender. Agent and its Affiliates may accept deposits from, lend money to, own securities of, act as the financial advisor or in any other advisory capacity for, and generally engage in any kind of business with any Loan Party as if it were not performing the duties specified herein, and may accept fees and other consideration from the Loan Parties for services in connection with this Agreement or otherwise without having to account for the same to Lenders. Each Loan Party agrees to pay to Agent, for each of their own account, the fees payable in the amounts and at the times set forth in the Fee Letter.

14.9. Delivery of Documents.

To the extent Agent receives documents and information from the Loan Parties pursuant to <u>Sections</u> <u>9.7</u>, <u>9.8</u>, and <u>9.12</u>, Agent will promptly furnish such documents and information to Lenders.

14.10. Loan Parties' Undertaking to Agent.

Without prejudice to its obligations to Lenders under the other provisions of this Agreement, each Loan Party hereby undertakes with Agent to pay to Agent from time to time on demand all amounts from time to time due and payable by it for the account of Agent or Lenders or any of them pursuant to this Agreement to the extent not already paid. Any payment made pursuant to any such demand shall <u>protanto</u> satisfy such Loan Party's obligations to make payments for the account of Lenders or the relevant one or more of them pursuant to this Agreement.

14.11. Bankruptcy Proceedings.

In case of any bankruptcy or other insolvency proceeding involving Borrower (a "Bankruptcy Proceeding"), Agent shall be entitled, but not obligated to, to intervene in such Bankruptcy Proceeding to (a) file and prove a claim for the whole amount of principal, interest and unpaid fees in respect of the Revolving Advances, issued Letters of Credit and all other Obligations that are owing and unpaid under the terms of this Agreement and other Loan Documents and to file such documents as may be necessary or advisable in order to have the claims of the Lenders, the Issuer and Agent (including any claim for reasonable compensation, expenses, disbursements and advances of any of the foregoing entities and their respective agents, counsel and other advisors) allowed in such Bankruptcy Proceedings; and (b) to collect and receive any monies or other property payable or deliverable on account of any such claims and to distribute the same to the Lenders and the Issuer under the terms of this Agreement. Further, any custodian, receiver, assignee, trustee, liquidator or similar official in any such Bankruptcy Proceeding is (i) authorized to make payments or distributions in a Bankruptcy Proceeding directly to Agent on behalf of all of the Lenders or the Issuer to whom any amounts are owed under this Agreement and other Loan Documents, unless Agent expressly consents in writing to the making of such payments or distributions directly to such Lenders and the Issuer; and (ii) required to pay to Agent any amount due for the reasonable compensation, expenses, disbursements and advances of Agent and its agents and counsel, and any other amounts due to Agent under this Agreement and other Loan Documents.

14.12. No Liability for Clean-Up of Hazardous Materials.

In the event that Agent is required to acquire title to an asset for any reason, or take any managerial action of any kind in regard thereto, in order to carry out any obligation for the benefit of another, which in Agent's sole discretion may cause Agent to be considered an "owner or operator" under the provisions of CERCLA, or otherwise cause Agent to incur liability under CERCLA or any other federal, state or local law, Agent reserves the right, instead of taking such action, to either resign as Agent or arrange for the transfer of the title or control of the asset to a court-appointed receiver. Except for such claims or actions arising directly from the gross negligence, or willful misconduct of Agent, Agent shall not be liable to any Person or entity for any environmental claims or contribution actions under any federal, state or local law, rule or regulation by reason of Agent's actions and conduct as authorized, empowered and directed hereunder, including in or relating to the discharge, release or threatened release of Hazardous Substances into the environment. If at any time after any foreclosure on the Collateral (or a transfer in lieu of foreclosure) upon the exercise of remedies in accordance with the terms of this Agreement it is necessary or advisable to take possession, own, operate or manage any portion of the Collateral by any Person or entity other than Borrower, Agent shall appoint an appropriately qualified Person to possess, own, operate or manage such Collateral.

14.13. Certain ERISA Matters.

- (a) Each Lender (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto to the date such Person ceases being a Lender party hereto, for the benefit of, Agent and not, for the avoidance of doubt, to or for the benefit of Borrower or any other Loan Party, that at least one of the following is and will be true:
- (i) such Lender is not using "plan assets" (within the meaning of Section 3(42) of ERISA or otherwise) of one or more Benefit Plans with respect to such Lender's entrance into, participation in, administration of and performance of the Advances, the Commitments or this Agreement,
- (ii) the transaction exemption set forth in one or more PTEs, such as PTE 84-14 (a class exemption for certain transactions determined by independent qualified professional asset managers), PTE 95-60 (a class exemption for certain transactions involving insurance company general accounts), PTE 90-1 (a class exemption for certain transactions involving insurance company pooled separate accounts), PTE 91-38 (a class exemption for certain transactions involving bank collective investment funds) or PTE 96-23 (a class exemption for certain transactions determined by in-house asset managers), is applicable with respect to such Lender's entrance into, participation in, administration of and performance of the Advances, the Commitments and this Agreement,
- (iii) (A) such Lender is an investment fund managed by a "Qualified Professional Asset Manager" (within the meaning of Part VI of PTE 84-14), (B) such Qualified Professional Asset Manager made the investment decision on behalf of such Lender to enter into, participate in, administer and perform the Advances, the Commitments and this Agreement, (C) the entrance into, participation in, administration of and performance of the Advances, the Commitments and this Agreement satisfies the requirements of sub-sections (b) through (g) of Part I of PTE 84-14 and (D) to the best knowledge of such Lender, the requirements of subsection (a) of Part I of PTE 84-14 are satisfied with respect to such Lender's entrance into, participation in, administration of and performance of the Advances, the Commitments and this Agreement, or
- (iv) such other representation, warranty and covenant as may be agreed in writing between Agent, in its sole discretion, and such Lender.
- (b) In addition, unless either (1) <u>sub-clause (i)</u> in the immediately preceding <u>clause (a)</u> is true with respect to a Lender or (2) a Lender has provided another representation, warranty and covenant in accordance with <u>sub-clause (iv)</u> in the immediately preceding <u>clause (a)</u>, such Lender further (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto, for the benefit of, Agent and not, for the avoidance of doubt, to or for the benefit of Borrower or any other Loan Party, that Agent is not a fiduciary with respect to the assets of such Lender involved in such Lender's entrance into, participation in, administration of and performance of the Advances, the Commitments and this Agreement (including in connection with the reservation or exercise of any rights by Agent under this Agreement, any other Loan Document or any documents related hereto or thereto).
- 14.14. Rates. The interest rate on an Advance denominated in Dollars may be derived from an interest rate benchmark that may be discontinued or is, or may in the future become, the subject of regulatory reform. Upon the occurrence of a Benchmark Transition Event, Section 3.8 provides a mechanism for determining an alternative rate of interest. Agent does not warrant or accept responsibility for, and shall not have any liability with respect to (a) the continuation of, administration of, submission of, calculation of or any other matter related to the Base Rate, the

Term SOFR Reference Rate, Adjusted Term SOFR or Term SOFR, or any component definition thereof or rates referred to in the definition thereof, or any alternative, successor or replacement rate thereto (including any Benchmark Replacement), including whether the composition or characteristics of any such alternative, successor or replacement rate (including any Benchmark Replacement) will be similar to, or produce the same value or economic equivalence of, or have the same volume or liquidity as, the Base Rate, the Term SOFR Reference Rate, Adjusted Term SOFR, Term SOFR or any other Benchmark prior to its discontinuance or unavailability, or (b) the effect, implementation or composition of any Conforming Changes. Agent and its affiliates or other related entities may engage in transactions that affect the calculation of the Base Rate, the Term SOFR Reference Rate, Term SOFR, Adjusted Term SOFR, any alternative, successor or replacement rate (including any Benchmark Replacement) or any relevant adjustments thereto, in each case, in a manner adverse to the Borrower. Agent may select information sources or services in its reasonable discretion to ascertain the Base Rate, the Term SOFR Reference Rate, Term SOFR, Adjusted Term SOFR or any other Benchmark including any component thereof, or rates referenced in the definition thereof, in each case pursuant to the terms of this Agreement, and shall have no liability to the Borrower, any Lender or any other person or entity for damages of any kind, including direct or indirect, special, punitive, incidental or consequential damages, costs, losses or expenses (whether in tort, contract or otherwise and whether at law or in equity), for any error or calculation of any such rate (or component thereof) provided by any such information source or service.

14.14. Disclaimer of Liability.

(a) The Agent does not warrant or accept any responsibility for, and shall not have any liability in respect of (a) the monitoring, determination or verification of the unavailability or cessation of LIBOR (or other applicable Benchmark), (b) the administration of, submission of or any other matter related to the Adjusted LIBO Rate, Revolving Interest Rate, any component definition thereof or rates referenced in the definition thereof or any alternative, comparable or successor rate or adjustment thereto (including any then-current Benchmark, any Benchmark Replacement or any Benchmark Replacement Adjustment), including whether the composition or characteristics of any such alternative, comparable or successor rate or adjustment (including any Benchmark Replacement or any Benchmark Replacement Adjustment) will be similar to, or produce the same value of economic equivalence of, Adjusted LIBO Rate, or any other Benchmark or any Benchmark Replacement Adjustment, or (c) the effect, implementation or composition of any Benchmark Replacement Conforming Changes.

(b) The Agent shall not be liable for any inability, failure or delay on its part to perform any of its duties set forth in this Agreement as a result of the unavailability of LIBOR (or other applicable Benchmark) and absence of a designated replacement Benchmark, including as a result of any inability, delay, error or inaccuracy on the part of any other transaction party, including without limitation the Required Lenders, in providing any direction, instruction, notice or information required or contemplated by the terms of this Agreement and reasonably required for the performance of such duties. In the event the Agent on any interest determination date is required, but is unable, to determine USD LIBOR or other applicable or replacement Benchmark in accordance with the procedures set out in this Agreement, USD LIBOR or the applicable or replacement Benchmark will be USD LIBOR as determined on the previous interest determination date.

14.15. Erroneous Payments.

(a) If Agent (x) notifies a Lender, Issuer or secured party, or any Person who has received funds on behalf of a Lender, Issuer or secured party (any such Lender, Issuer, secured party or

other recipient (and each of their respective successors and assigns), a "Payment Recipient") that Agent has determined in its sole discretion (whether or not after receipt of any notice under immediately succeeding clause (b)) that any funds (as set forth in such notice from Agent) received by such Payment Recipient from Agent or any of its Affiliates were erroneously or mistakenly transmitted to, or otherwise erroneously or mistakenly received by, such Payment Recipient (whether or not known to such Lender, Issuer, secured party or other Payment Recipient on its behalf) (any such funds, whether transmitted or received as a payment, prepayment or repayment of principal, interest, fees, distribution or otherwise, individually and collectively, an "Erroneous Payment") and (y) demands in writing the return of such Erroneous Payment (or a portion thereof), such Erroneous Payment shall at all times remain the property of Agent pending its return or repayment as contemplated below in this Section 14.15 and held in trust for the benefit of Agent, and such Lender, Issuer or secured party shall (or, with respect to any Payment Recipient who received such funds on its behalf, shall cause such Payment Recipient to) promptly, but in no event later than two Business Days thereafter (or such later date as Agent may, in its sole discretion, specify in writing), return to Agent the amount of any such Erroneous Payment (or portion thereof) as to which such a demand was made, in same day funds (in the currency so received), together with interest thereon (except to the extent waived in writing by Agent) in respect of each day from and including the date such Erroneous Payment (or portion thereof) was received by such Payment Recipient to the date such amount is repaid to Agent in same day funds at the greater of the Federal Funds Rate and a rate determined by Agent in accordance with banking industry rules on interbank compensation from time to time in effect. A notice of Agent to any Payment Recipient under this clause (a) sha

- (b) Without limiting immediately preceding <u>clause (a)</u>, each Lender, Issuer, secured party or any Person who has received funds on behalf of a Lender, Issuer or secured party (and each of their respective successors and assigns), agrees that if it receives a payment, prepayment or repayment (whether received as a payment, prepayment or repayment of principal, interest, fees, distribution or otherwise) from Agent (or any of its Affiliates) (x) that is in a different amount than, or on a different date from, that specified in this Agreement or in a notice of payment, prepayment or repayment sent by Agent (or any of its Affiliates) with respect to such payment, prepayment or repayment, (y) that was not preceded or accompanied by a notice of payment, prepayment sent by Agent (or any of its Affiliates), or (z) that such Lender, Issuer or secured party, or other such recipient, otherwise becomes aware was transmitted, or received, in error or by mistake (in whole or in part), then in each such case:
- (i) it acknowledges and agrees that (A) in the case of immediately preceding <u>clauses (x)</u> or <u>(y)</u>, an error and mistake shall be presumed to have been made (absent written confirmation from Agent to the contrary) or (B) an error and mistake has been made (in the case of immediately preceding <u>clause (z)</u>), in each case, with respect to such payment, prepayment or repayment; and
- (ii) such Lender, Issuer or secured party shall (and shall cause any other recipient that receives funds on its respective behalf to) promptly (and, in all events, within one Business Day of its knowledge of the occurrence of any of the circumstances described in immediately preceding <u>clauses (x)</u>, $\underline{(y)}$ and $\underline{(z)}$) notify Agent of its receipt of such payment, prepayment or repayment, the details thereof (in reasonable detail) and that it is so notifying Agent pursuant to this <u>Section 14.15(b)</u>.
- (c) Each Lender, Issuer or secured party hereby authorizes Agent to set off, net and apply any and all amounts at any time owing to such Lender, Issuer or secured party under any Loan Document, or otherwise payable or distributable by Agent to such Lender, Issuer or secured party under any Loan Document with respect to any payment of principal, interest, fees or other amounts, against any amount that Agent has demanded to be returned under immediately preceding <u>clause (a)</u>.

For the avoidance of doubt, the failure to deliver a notice to Agent pursuant to this <u>Section 14.15(b)</u> shall not have any effect on a Payment Recipient's obligations pursuant to <u>Section 14.15(a)</u> or on whether or not an Erroneous Payment has been made.

(i) In the event that an Erroneous Payment (or portion thereof) is not recovered by Agent for any reason, after demand therefor in accordance with immediately preceding clause (a), from any Lender that has received such Erroneous Payment (or portion thereof) (and/or from any Payment Recipient who received such Erroneous Payment (or portion thereof) on its respective behalf) (such unrecovered amount, an "Erroneous Payment Return Deficiency"), upon Agent's notice to such Lender at any time, then effective immediately (with the consideration therefor being acknowledged by the parties hereto), (A) such Lender shall be deemed to have assigned its Advances (but not its Commitments) with respect to which such Erroneous Payment was made (the "Erroneous Payment Impacted Class") in an amount equal to the Erroneous Payment Return Deficiency (or such lesser amount as Agent may specify) (such assignment of the Advances (but not Commitments) of the Erroneous Payment Impacted Class, the "Erroneous Payment Deficiency Assignment") (on a cashless basis and such amount calculated at par plus any accrued and unpaid interest (with the assignment fee to be waived by Agent in such instance)), and is hereby (together with Borrower) deemed to execute and deliver a Commitment Transfer Supplement (or, to the extent applicable, an agreement incorporating a Commitment Transfer Supplement by reference pursuant to an applicable electronic platform as to which Agent and such parties are participants) with respect to such Erroneous Payment Deficiency Assignment, and such Lender shall deliver any Notes evidencing such Advances to Borrower or Agent (but the failure of such Person to deliver any such Notes shall not affect the effectiveness of the foregoing assignment), (B) Agent as the assignee Lender shall be deemed to have acquired the Erroneous Payment Deficiency Assignment, (C) upon such deemed acquisition, Agent as the assignee Lender shall become a Lender, as applicable, hereunder with respect to such Erroneous Payment Deficiency Assignment and the assigning Lender shall cease to be a Lender, as applicable, hereunder with respect to such Erroneous Payment Deficiency Assignment, excluding, for the avoidance of doubt, its obligations under the indemnification provisions of this Agreement and its applicable Commitments which shall survive as to such assigning Lender, (D) Agent and Borrower shall each be deemed to have waived any consents required under this Agreement to any such Erroneous Payment Deficiency Assignment, and (E) Agent will reflect in the Register its ownership interest in the Advances subject to the Erroneous Payment Deficiency Assignment. For the avoidance of doubt, no Erroneous Payment Deficiency Assignment will reduce the Commitments of any Lender and such Commitments shall remain available in accordance with the terms of this Agreement.

(i) _(ii) Subject to Section 15.3 (but excluding, in all events, any assignment consent or approval requirements (whether from Borrower or otherwise)), Agent may, in its discretion, sell any Advances acquired pursuant to an Erroneous Payment Deficiency Assignment and upon receipt of the proceeds of such sale, the Erroneous Payment Return Deficiency owing by the applicable Lender shall be reduced by the net proceeds of the sale of such Advance (or portion thereof), and Agent shall retain all other rights, remedies and claims against such Lender (and/or against any recipient that receives funds on its respective behalf). In addition, an Erroneous Payment Return Deficiency owing by the applicable Lender (x) shall be reduced by the proceeds of prepayments or repayments of principal and interest, or other distribution in respect of principal and interest, received by Agent on or with respect to any such Advances acquired from such Lender pursuant to an Erroneous Payment Deficiency Assignment (to the extent that any such Advances are then owned by Agent) and (y) may, in the sole discretion of Agent, be reduced by any amount specified by Agent in writing to the applicable Lender from time to time.

(e) The parties hereto agree that (x) irrespective of whether Agent may be equitably subrogated, in the event that an Erroneous Payment (or portion thereof) is not recovered from any Payment Recipient that has received such Erroneous Payment (or portion thereof) for any reason, Agent shall be subrogated to all the rights and interests of such Payment Recipient (and, in the case of any

Payment Recipient who has received funds on behalf of a Lender, Issuer or secured party, to the rights and interests of such Lender, Issuer or secured party, as the case may be) under the Loan Documents with respect to such amount (the "Erroneous Payment Subrogation Rights") (provided that the Loan Parties' Obligations under the Loan Documents in respect of the Erroneous Payment Subrogation Rights shall not be duplicative of such Obligations in respect of Advances that have been assigned to Agent under an Erroneous Payment Deficiency Assignment) and (y) an Erroneous Payment shall not pay, prepay, repay, discharge or otherwise satisfy any Obligations owed by Borrower or any other Loan Party; provided that this Section 14.15 shall not be interpreted to increase (or accelerate the due date for), or have the effect of increasing (or accelerating the due date for), the Obligations of Borrower relative to the amount (and/or timing for payment) of the Obligations that would have been payable had such Erroneous Payment not been made by Agent; provided, further, that for the avoidance of doubt, immediately preceding clauses (x) and (y) shall not apply to the extent any such Erroneous Payment is, and solely with respect to the amount of such Erroneous Payment that is, comprised of funds received by Agent from Borrower for the purpose of making such Erroneous Payment.

- (f) To the extent permitted by applicable law, no Payment Recipient shall assert any right or claim to an Erroneous Payment, and hereby waives, and is deemed to waive, any claim, counterclaim, defense or right of set-off or recoupment with respect to any demand, claim or counterclaim by Agent for the return of any Erroneous Payment received, including, without limitation, or waiver of, any defense based on "discharge for value" or any similar doctrine.
- (g) Each party's obligations, agreements and waivers under this <u>Section 14.15</u> shall survive the resignation or replacement of Agent, any transfer of rights or obligations by, or the replacement of, a Lender or Issuer, the termination of the Commitments and/or the repayment, satisfaction or discharge of all Obligations (or any portion thereof) under any other Loan Document.
- 14.16. <u>Joint Lead Arrangers and Joint Book Runners</u>. Each of the Joint Lead Arrangers and Joint Book Runners, in such capacities, shall not have any right, power, obligation, liability, responsibility, or duty under this Agreement other than those applicable to it in its capacity as a Lender, as Agent, as Swingline Lender, or as Issuer. Without limiting the foregoing, each of the Joint Lead Arrangers and Joint Book Runners, in such capacities, shall not have or be deemed to have any fiduciary relationship with any Lender or any Loan Party. Each Lender, Agent, Swingline Lender, Issuer, and each Loan Party acknowledges that it has not relied, and will not rely, on the Joint Lead Arrangers, Joint Book Runners, in deciding to enter into this Agreement or in taking or not taking action hereunder. Each of the Joint Lead Arrangers and Joint Book Runners, in such capacities, shall be entitled to resign at any time by giving notice to Agent and Borrower.

XV. MISCELLANEOUS.

15.1. Governing Law.

This Agreement shall be governed by and construed in accordance with the laws of the State of New York applied to contracts to be performed wholly within the State of New York, without regard to conflicts of law principals except Title 14 of Article 5 of the New York General Obligations law. Any judicial proceeding brought by or against any Loan Party with respect to any of the Obligations, this Agreement or any related agreement may be brought in any court of competent jurisdiction in the County of New York, State of New York, United States of America, and, by execution and delivery of this Agreement, each Loan Party accepts for itself and in connection with its properties, generally and unconditionally, the non-exclusive jurisdiction of the aforesaid courts, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this Agreement. Each Loan Party hereby waives right to personal service of any and all process upon it and consents that all such service of process may

be made by registered mail (return receipt requested) directed to such Loan Party at its address set forth in <u>Section 15.6</u> and service so made shall be deemed completed five (5) days after the same shall have been so deposited in the mails of the United States of America. Nothing herein shall affect the right to serve process in any manner permitted by law or shall limit the right of Agent or any Lender to bring proceedings against any Loan Party in the courts of any other jurisdiction. Each Loan Party waives any objection to jurisdiction and venue of any action instituted hereunder and shall not assert any defense based on lack of jurisdiction or venue or based upon <u>forum non conveniens</u>. Any judicial proceeding by any Loan Party against Agent or any Lender involving, directly or indirectly, any matter or claim in any way arising out of, related to or connected with this Agreement or any related agreement, shall be brought only in a federal or state court located in the County of New York, State of New York.

15.2. Entire Understanding; Amendments.

- (a) This Agreement and the documents executed concurrently herewith contain the entire understanding between the Loan Parties, Agent and each Lender and supersedes all prior agreements and understandings, if any, relating to the subject matter hereof. Any promises, representations, warranties or guarantees not herein contained and hereinafter made shall have no force and effect unless in writing, signed by each Loan Party's, Agent's and each Lender's respective officers. Neither this Agreement nor any portion or provisions hereof may be changed, modified, amended, waived, supplemented, discharged, cancelled or terminated orally or by any course of dealing, or in any manner other than by an agreement in writing, signed by the party to be charged. Each Loan Party acknowledges that it has been advised by counsel in connection with the execution of this Agreement and Loan Documents and is not relying upon oral representations or statements inconsistent with the terms and provisions of this Agreement.
- (b) The Required Lenders, Agent with the consent in writing of the Required Lenders, and the Loan Parties may, subject to the provisions of this <u>Section 15.2(b)</u>, from time to time enter into written supplemental agreements to this Agreement or the Loan Documents executed by the Loan Parties, for the purpose of adding or deleting any provisions or otherwise changing, varying or waiving in any manner the rights of Lenders, Agent or any Loan Party thereunder or the conditions, provisions or terms thereof or waiving any Event of Default thereunder, but only to the extent specified in such written agreements; <u>provided</u>, <u>however</u>, that no such supplemental agreement shall:
 - (A) without the written consent of each Lender affected thereby:
 - (i) increase the Commitment Percentage or the Commitment of any Lender;
 - (ii) increase the Maximum Revolving Advance Amount (except, for the avoidance of doubt, for any increase to the Maximum Revolving Advance Amount resulting from, and in accordance with, the provisions of Section 2.4);
 - (iii) extend the maturity of any Note or the due date for any amount payable hereunder, or decrease the rate of interest or reduce any scheduled principal payment, fee or any other amount payable by any Loan Party to Lenders pursuant to this Agreement (it being understood that a waiver of any Default, Event of Default or mandatory prepayment shall not constitute a reduction of the rate of interest or reduction of any principal payment, fee or other amount payable by any Loan Party);
 - (iv) alter the definition of the term Required Lenders or alter, amend or modify or waive any provision of this Section 15.2(b), including, without

limitation, eliminating or reducing the voting rights of the Lenders under this Section 15.2(b);

- (v) other than in connection with a merger, liquidation, dissolution or sale of such Person expressly permitted by the terms hereof or the other Loan Documents, release any Borrower from any obligation for the payment of money or consent to the assignment or transfer by any Borrower or any Guarantor of any of its rights or duties under this Agreement or the other Loan Documents;
- (vi) release all or substantially all of the Collateral or release all or substantially all of the Guarantors from their obligations under the applicable Guaranty executed by such Guarantor;
- (vii) amend, modify or waive the pro rata requirements of <u>Section 2.13</u> or <u>Section 11.2</u> or any other provision of the Loan Documents requiring pro rata treatment of the Lenders; or
 - (viii) [reserved]; or
 - (B) without the consent of Agent, alter, amend, modify or waive the rights or duties of Agent;
 - (C) without the consent of the Issuer, alter, amend, modify or waive the rights or duties of such Issuer; and
- (D) without the consent of Swingline Lender, alter, amend, modify or waive the rights or duties of the Swingline Lender or increase the Maximum Swingline Loan Amount.

Any such supplemental agreement shall apply equally to each Lender and shall be binding upon each Loan Party, Lenders and Agent and all future holders of the Obligations. In the case of any waiver, the Loan Parties, Agent and Lenders shall be restored to their former positions and rights, and any Event of Default waived shall be deemed to be cured and not continuing, but no waiver of a specific Event of Default shall extend to any subsequent Event of Default (whether or not the subsequent Event of Default is the same as the Event of Default which was waived), or impair any right consequent thereon.

(c) In the event a Lender is a Non-Consenting Lender, then Borrower may, at its option, require such Lender to assign its Commitment Percentage of the Advances to HSBC (with HSBC's consent) or to another Lender (with such Lender's consent) or to any other Person that shall assume such obligations (the "Designated Lender"), for a price equal to the then outstanding principal amount thereof plus accrued and unpaid interest and fees and all other Obligations due such Lender under this Agreement and the other Loan Documents, which interest and fees shall be paid in full at the time of such assignment. In the event Borrower elects to require any Lender to assign its interest to HSBC or to the Designated Lender, Borrower will so notify such Lender in writing within forty five (45) days following such Lender's denial, and such Lender will assign its interest to HSBC or the Designated Lender no later than five (5) days following receipt of such notice pursuant to a Commitment Transfer Supplement executed by such Lender, HSBC or the Designated Lender, as appropriate, and Agent.

15.3. Successors and Assigns; Participations; New Lenders.

(a) This Agreement shall be binding upon and inure to the benefit of each Loan Party, Agent, each Lender, all future holders of the Obligations and their respective successors and

permitted assigns, except that no Loan Party may assign or transfer any of its rights or obligations under this Agreement without the prior written consent of Agent and each Lender.

- (b) Each Loan Party acknowledges that in the regular course of commercial banking business one or more Lenders may at any time and from time to time sell participating interests in the Advances to Participants (other than Disqualified Institutions). Each Participant may exercise all rights of payment (including without limitation rights of set-off) with respect to the portion of such Advances held by it or other Obligations payable hereunder as fully as if such Participant were a Lender and the direct holder of such Advance provided that Borrower shall not be required to pay to any Participant more than the amount which it would have been required to pay to Lender which granted an interest in its Advances or other Obligations payable hereunder to such Participant had such Lender retained such interest in the Advances hereunder or other Obligations payable hereunder and in no event shall Borrower be required to pay any such amount arising from the same circumstances and with respect to the same Advances or other Obligations payable hereunder to both such Lender and such Participant. For the avoidance of doubt, each Participant shall not be entitled to the benefits of Section 3.10 unless such Participant complies with Section 3.10(g) as if it were a Lender. Each Loan Party hereby grant to any Participant a continuing security interest in any deposits, moneys or other property actually or constructively held by such Participant as security for the Participant's interest in the Advances. Each Lender shall retain the sole right to approve, without the consent of the Participant, any amendment, modification or waiver of any provision of this Agreement and the other Loan Documents other than any amendment, modification or waiver of the type specified in clause (i), (iv) or (vi) of Section 15.2(b) as it relates to Participant's interest in the Obligations. Each Lender that sells participations to a Participant, acting solely for this purpose as a non-fiduciary agent of Borrower, shall maintain a register on which it enters the name and address of each Participant and the principal amount of and interest owing with respect to the participation sold to each such Participant (the "Participant Register"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any participant or any information relating to a participant's interest in any commitments, loans, letters of credit or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan, letter of credit or other obligation is in registered form under Section 5f.103-1(c)of the U.S. Treasury Regulations. The entries in the Participant Register shall be conclusive (absent manifest error), and Borrower and the Lenders shall treat each Person whose name is recorded in such Participant Register pursuant to the terms hereof as a participant for all purposes of this Agreement, notwithstanding notice to the contrary. For the avoidance of doubt, the Agent (in its capacity as such) shall have no responsibility for maintaining a Participant Register.
- (c) Any Lender may, with the consent of Agent and Borrower, which consent shall not be unreasonably withheld or delayed, sell, assign or transfer all or any part of its rights under this Agreement and the other Loan Documents to another Lender or to one or more additional banks or financial institutions (other than Disqualified Institutions) and one or more additional banks or financial institutions may commit to make Advances hereunder (each a "Purchasing Lender"), in minimum amounts of not less than \$5,000,000 and integral multiples of \$1,000,000 in excess thereof, pursuant to a Commitment Transfer Supplement, executed by a Purchasing Lender, the transferor Lender, and Agent and delivered to Agent for recording; provided that no assignment shall be made to (i) a Loan Party or any of a Loan Party's Affiliates or Subsidiaries or (ii) any Defaulting Lender or any of its Subsidiaries, or any Person who, upon becoming a Lender hereunder, would constitute any of the foregoing Persons described in this clause (ii) (provided that Borrower shall be deemed to have consented to any such assignment unless it shall have objected thereto by written notice to the Agent within 10 Business Days after having received notice thereof). The consent of Agent and Borrower shall not be required in the case of an assignment by a Lender to another Lender or to an Affiliate of a Lender or an Approved Fund (each, an "Eligible Assignee"), and the consent of Borrower shall not be required at any time that an Event of Default has occurred and is continuing hereunder. Upon such execution, delivery, acceptance and recording, from and after the transfer effective date determined pursuant to such Commitment

Transfer Supplement, (i) Purchasing Lender thereunder shall be a party hereto and, to the extent provided in such Commitment Transfer Supplement, have the rights and obligations of a Lender thereunder with a Commitment Percentage as set forth therein (including, for the avoidance of doubt, such obligations under Section 3.10(g)), and (ii) the transferor Lender thereunder shall, to the extent provided in such Commitment Transfer Supplement, be released from its obligations under this Agreement, the Commitment Transfer Supplement creating a novation for that purpose. Such Commitment Transfer Supplement shall be deemed to amend this Agreement to the extent, and only to the extent, necessary to reflect the addition of such Purchasing Lender and the resulting adjustment of the Commitment Percentages arising from the purchase by such Purchasing Lender of all or a portion of such Purchasing Lender and the resulting adjustment of the Commitment Percentages arising from the purchase by such Purchasing Lender of all or a portion of the rights and obligations of such transferor Lender under this Agreement and the other Loan Documents. Each Loan Party shall execute and deliver such further documents and do such further acts and things in order to effectuate the foregoing.

- (d) Agent shall maintain at its address an electronic or written record (the "Register") in which it will record (i) the names and addresses of the Lenders (including Purchasing Lenders) and (ii) the amount of each Advance and amounts owing to each Lender from time to time and (iii) all amounts received by Agent hereunder from Borrower and each Lender's share thereof. Agent will also maintain at its address a copy of each Commitment Transfer Supplement delivered to it, and no Commitment Transfer Supplement will be effective until it is recorded in the Register. The entries in the Register shall be conclusive, in the absence of manifest error, and Borrower, Agent and Lenders may treat each Person whose name is recorded in the Register as the owner of the Advance recorded therein for the purposes of this Agreement. The Register shall be available for inspection by Borrower or any Lender at any reasonable time and from time to time upon reasonable prior notice. Agent shall be available for inspection of \$3,500 (which may be waived or reduced at the sole discretion of the Agent) payable by the applicable Purchasing Lender upon the effective date of each transfer or assignment to such Purchasing Lender except in connection with any assignments by a Lender to an Affiliate of such Lender or to another Lender. If either the assigning Lender or the Purchasing Lender shall have failed to make any payment required to be made by it pursuant to this Agreement, Agent shall have no obligation to accept such assignment and record the information therein in the Register unless and until such payment shall have been made in full, together with all accrued interest thereon. No assignment shall be effective for purposes of this Agreement unless it has been recorded in the Register as provided in this paragraph.
- (e) Each Loan Party authorizes each Lender to disclose to any Participant or Purchasing Lender and any prospective Participant or Purchasing Lender any and all financial information in such Lender's possession concerning such Loan Party which has been delivered to such Lender by or on behalf of such Loan Party pursuant to this Agreement or in connection with such Lender's credit evaluation of such Loan Party so long as such Participant, Purchasing Lender or prospective Participant or Purchasing Lender agrees to abide by the provisions of Section 15.15.
- (f) Any other provision in this Agreement notwithstanding, any Lender may at any time create a security interest in, or pledge, all or any portion of its rights under and interest in this Agreement to secure obligations of such Lender, including any pledge in favor of any Federal Reserve Bank in accordance with Regulation A of the Federal Reserve Bank or U.S. Treasury Regulation 31 CFR §203.24, and such Federal Reserve Bank may enforce such pledge or security interest in any manner permitted under applicable law; provided, that no such pledge shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.

15.4. Application of Payments.

Agent shall have the continuing and exclusive right to apply or reverse and re-apply any payment and any and all proceeds of Collateral to any portion of the Obligations. To the extent that any Loan Party makes a payment or Agent or any Lender receives any payment or proceeds of the Collateral for such Loan Party's benefit, which are subsequently invalidated, declared to be fraudulent or preferential, set aside or required to be repaid to a trustee, debtor in possession, receiver, custodian or any other party under any bankruptcy law, common law or equitable cause, then, to such extent, the Obligations or part thereof intended to be satisfied shall be revived (as to each Lender based upon its Commitment Percentage of any such Obligations) and continue as if such payment or proceeds had not been received by Agent or such Lender.

15.5. <u>Indemnity; Funding Losses</u>.

- (a) Each Loan Party shall indemnify Agent, the Issuer, each Lender and each of their respective officers, directors, Affiliates, advisors, employees, partners, trustees, administrators, managers, counsel, representatives, advisors and agents (each such Person being called an "Indemnitee") from and against, and hold each Indemnitee harmless from, any and all liabilities, obligations, claims, losses, damages, penalties, actions, judgments, suits, costs, expenses and disbursements of any kind or nature whatsoever (including, without limitation, reasonable fees and disbursements of counsel) which may be imposed on, incurred by, or asserted against Agent, such Issuer or any Lender by any Person (including Borrower or any Guarantor) in connection with, or as a result of (i) the execution or delivery of this Agreement or any other Loan Documents contemplated thereby, (ii) the performance by an Indemnitee of its respective obligations under this Agreement and other Loan Documents; (iii) consummation of the transactions contemplated thereby; (iv) environmental claims and liabilities; or (v) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or by Borrower or any Guarantor, and regardless of whether any Indemnitee is a party thereto, subject to the condition that claims or losses on which indemnification is sought by Indemnitee not arise by such Indemnitee's own willful misconduct, bad faith, or gross negligence as determined by final, non-appealable judgment of a court of competent jurisdiction.
- (b) In the event that (i) any payment of a Eurodollar RateSOFR Loan is required, made or permitted on a date other than the last day of the then current Interest Period applicable thereto (including upon demand by any Lender or Agent), (ii) the conversion of any Eurodollar RateSOFR Loan other than on the last day of the Interest Period applicable thereto, or (iii) the failure to convert, continue, borrow or prepay any Eurodollar RateSOFR Loan on the date specified in any notice delivered pursuant hereto, then, in any such event, Borrower shall compensate each Lender for theany loss, cost and expense attributable to such event in an amount computed as follows: the excess, if any, of (x) the amount of interest that would have accrued on the principal amount of such Eurodollar Rate Loan had such event not occurred, at the interest rate that would have been applicable to such Eurodollar Rate Loan, for the period from the date of such event to the last day of the then current Interest Period therefor (or, in the case of a failure to borrow, convert or continue, for the period that would have been the Interest Period for such Eurodollar Rate Loan), over (y) the amount of interest that would accrue on such principal amount for such period at the interest rate with such Lender (or an affiliate of such Lender) would bid were it to bid, at the commencement of the interest period, for Dollar deposits of a comparable amount and period from other banks in the London interbank eurodollar market, including any loss, cost or expense arising from the liquidation or redeployment of funds. A certificate of a Lender delivered to Borrower and setting forth any amount or amounts that such Lender is entitled to receive pursuant to this clause (b) shall be conclusive absent manifest error. Borrower shall pay such Lender the amount shown as due on any such certificate upon demand.

(c) In connection with the issuance of any Letter of Credit, each Loan Party shall indemnify, save and hold Agent, each Lender and the Issuer harmless from any loss, cost, expense or liability, including, without limitation, any claims, damages, costs and expenses, and other payments made by Agent, any Lender or the Issuer and expenses and reasonable attorneys' fees (limited to one outside counsel for the Agent, Issuer and Lenders and, if reasonably necessary, a single local counsel in each jurisdiction for which local counsel is reasonably deemed necessary) incurred by Agent, any Lender or the Issuer arising out of, or in connection with, any Letter of Credit to be issued or created for Borrower.

15.6. Notice.

Any notice or request hereunder may be given to any Loan Party or to Agent or any Lender at their respective addresses set forth below or at such other address as may hereafter be specified in a notice designated as a notice of change of address under this Section 15.6. Any notice or request hereunder shall be given by (a) hand delivery, (b) overnight courier, (c) registered or certified mail, return receipt requested, or (d) telecopy to the number set out below (or such other number as may hereafter be specified in a notice designated as a notice of change of address) with electronic confirmation of its receipt. Any notice or other communication required or permitted pursuant to this Agreement shall be deemed given (a) when personally delivered to any officer of the party to whom it is addressed, (b) on the earlier of actual receipt thereof or three (3) days following posting thereof by certified or registered mail, postage prepaid, or (c) upon actual receipt thereof when sent by a recognized overnight delivery service or (d) upon actual receipt thereof when sent by telecopier to the number set forth below with electronic confirmation of its receipt, in each case addressed to each party at its address set forth below or at such other address as has been furnished in writing by a party to the other by like notice:

(A) If to Agent or HSBC: HSBC Bank USA, N.A., as Administrative Agent and Collateral Agent 452 Fifth Avenue New York, NY, 10018 Attention: Loan Agency Email: [*]

> with copies to: Morgan, Lewis & Bockius LLP 101 Park Avenue New York, NY 10178

Attention: Rick Denhup Phone: [*]
Email: [*]

- (B) If to any other Lender, as specified on the signature pages hereof.
- (C) If to any Loan Party: MarketWise, LLC

1125 N. Charles Street Baltimore, MD 21201 Attention: Dale Lynch, CFO

Phone: [*] Email: [*]

with copies to: MarketWise, LLC

1125 N. Charles Street Baltimore, MD 21201

Attention: Gary Anderson, General Counsel

Phone: [*] Email: [*]

and

Latham & Watkins LLP 555 Eleventh Street, NW Suite 1000 Washington, D.C. 20004-1304 Attention: Benjamin Berman Phone: [*] Email: [*]

15.7. Survival.

The obligations of each Loan Party under <u>Sections 2.11</u>, <u>3.6</u>, <u>3.8</u> and <u>15.5</u> and of Lenders under <u>Section 14.7</u> shall survive termination of this Agreement and the other Loan Documents, the expiration or termination of the Letters of Credit and all Commitments, the resignation and removal of Agent, and payment in full in cash of the Obligations (other than (A) contingent indemnification obligations as to which no claim has been asserted and (B) obligations and liabilities under Lender-Provided Hedges and Bank Product Obligations).

15.8. Severability.

If any part of this Agreement is contrary to, prohibited by, or deemed invalid under Applicable Laws or regulations, such provision shall be inapplicable and deemed omitted to the extent so contrary, prohibited or invalid, but the remainder hereof shall not be invalidated thereby and shall be given effect so far as possible.

15.9. Expenses.

All out-of-pocket costs and expenses of Agent and Lenders, including, without limitation:

(a) reasonable attorneys' fees, charges and disbursements (limited to one outside counsel for the Agent and Lenders and, if reasonably necessary, a single local counsel in each jurisdiction for which local counsel is reasonably deemed necessary) incurred by Agent and, with respect to <u>clause (iv)</u> below, the applicable Lenders solely to the extent such Lenders are represented by the same counsel, and with respect to <u>clause (vii)</u> below, the Issuer, (i) in all efforts made to enforce payment of any Obligation or effect collection of any Collateral, or (ii) in connection with the entering into, modification, amendment and administration of this Agreement or any consents or waivers hereunder and all related agreements, documents and instruments (whether or not the transactions contemplated hereby or thereby are consummated), or (iii) in instituting, maintaining, preserving, enforcing and foreclosing on Agent's security interest in or Lien on any of the Collateral, whether through judicial proceedings or otherwise, or (iv) in defending or prosecuting any actions or proceedings arising out of or relating to Agent's or any Lender's transactions with the Loan Parties, or (v) in connection with any advice given to Agent with respect to its rights and obligations under this Agreement and all related agreements, (vi) in connection with the enforcement of this Agreement or any consent or waivers hereunder and all related agreements, documents and instruments or (vii) in connection with the issuance, amendment or extension of any Letter of Credit or any demand for payment thereunder, in each case, whether before or after a Default or Event of Default has occurred under any of the Loan Documents, relating to a workout, restructuring, or other negotiations with the Loan Parties in respect of such rights, obligations and transactions arising under the Loan Documents; and

(b) reasonable fees and disbursements incurred by Agent or Agent on behalf of Lenders in connection with any appraisals of Inventory or other Collateral, field examinations, collateral analysis or monitoring or other business analysis conducted by outside Persons in connection with this Agreement and all related agreements;

may be charged to Borrower's Account and shall be part of the Obligations.

15.10. <u>Injunctive Relief</u>.

Each Loan Party recognizes that, in the event such Loan Party fails to perform, observe or discharge any of its obligations or liabilities under this Agreement, any remedy at law may prove to be inadequate relief to Lenders; therefore, Agent, if Agent so requests, shall be entitled to temporary and permanent injunctive relief in any such case without the necessity of proving that actual damages are not an adequate remedy.

15.11. Consequential Damages.

TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, NONE OF AGENT, THE ISSUER, ANY LENDER, NOR ANY AGENT OR ATTORNEY FOR ANY OF THEM, SHALL BE LIABLE TO ANY LOAN PARTY FOR INDIRECT, PUNITIVE, EXEMPLARY, INCIDENTAL, SPECIAL OR CONSEQUENTIAL DAMAGES (AS OPPOSED TO DIRECT OR ACTUAL DAMAGES) ARISING OUT OF, IN CONNECTION WITH, OR AS A RESULT OF, THIS AGREEMENT, ANY OTHER LOAN DOCUMENT OR THE USE OF PROCEEDS THEREOF AND FROM ANY BREACH OF CONTRACT, TORT OR OTHER WRONG RELATING TO THE ESTABLISHMENT, ADMINISTRATION OR COLLECTION OF THE OBLIGATIONS.

15.12. Captions.

The captions at various places in this Agreement are intended for convenience only and do not constitute and shall not be interpreted as part of this Agreement.

15.13. <u>Counterparts; Telecopied Signatures</u>.

This Agreement and each of the Loan Documents may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Agreement, the Loan Documents, and any separate letter agreements with respect to fees payable to Agent or the Issuer, constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof. Except as provided in Section 8.1, this Agreement shall become effective when it shall have been executed by Agent and when Agent shall have received counterparts hereof that, when taken together, bear the signatures of each of the other parties hereto. Delivery of an executed counterpart of a signature page of this Agreement or any other Loan Document, or any certificate delivered thereunder, by fax transmission or e-mail transmission (e.g. "pdf" or "tif") shall be effective as delivery of a manually executed counterpart of this Agreement. or such Loan Document or certificate. Without limiting the foregoing, to the extent a manually executed counterpart is not specifically required to be delivered under the terms of this Agreement or any other Loan Document, upon the request of any party, such fax transmission or e-mail transmission shall be promptly followed by such manually executed counterpart.

15.14. Construction.

The parties acknowledge that each party and its counsel have participated in the preparation of this Agreement and that the normal rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Agreement or any amendments, schedules or exhibits thereto.

15.15. Confidentiality.

Each of Agent, the Lenders and the Issuer agree to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates and to its Related Parties (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential and such disclosure is on a need-to-know basis); (b) to the extent required or requested by any regulatory authority purporting to have jurisdiction over such Person or its Related Parties (including any self-regulatory authority, such as the National Association of Insurance Commissioners); (c) to the extent required by Applicable Laws or regulations or by any subpoena or similar legal process; (d) to any other party hereto; (e) in connection with the exercise of any remedies hereunder or under any Loan Document or any action or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder; (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights and obligations under this Agreement, or (ii) any actual or prospective party (or its Related Parties) to any swap, derivative or other transaction under which payments are to be made by reference to Borrower and its obligations, this Agreement or payments hereunder; (g) on a confidential basis to (i) any rating agency in connection with rating Borrower or its Subsidiaries or the credit facilities hereunder or (ii) the CUSIP Service Bureau or any similar agency in connection with the issuance and monitoring of CUSIP numbers with respect to the credit facilities hereunder; (h) with the consent of Borrower; or (i) to the extent such Information (x) becomes publicly available other than as a result of a breach of this Section, or (y) becomes available to Agent, any Lender, the Issuer or any of their respective Affiliates on a non-confidential basis from a source other than Borrower. In addition, Agent and the Lenders may disclose the existence of this Agreement and information about this Agreement to market data collectors, similar service providers to the lending industry and service providers to Agent and the Lenders in connection with the administration of this Agreement, the Loan Documents, and the Commitments.

For purposes of this Section, "<u>Information</u>" means all information received from Borrower or any of its Subsidiaries relating to Borrower or any of its Subsidiaries or any of their respective businesses, other than any such information that is available to Agent, any Lender or the Issuer on a nonconfidential basis prior to disclosure by Borrower or any of its Subsidiaries; <u>provided</u> that, in the case of information received from Borrower or any of its Subsidiaries after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

15.16. Publicity.

Each Loan Party hereby authorizes Agent, upon prior notice, to make appropriate announcements of the financial arrangement entered into among the Loan Parties, Agent and Lenders, including, without limitation, announcements which are commonly known as tombstones, in such publications and to such selected parties as Agent shall in its sole and absolute discretion deem appropriate. In addition, each Loan Party upon prior notice authorizes Agent to include such Loan Party's name and logo in select

transaction profiles and client testimonials prepared by Agent for use in publications, company brochures and other marketing materials of Agent.

- 15.17. <u>Electronic Execution of Assignments and Certain Loan Documents</u>. The words "delivery," "execute," "execution," "signed," "signature," and words of like import in this Agreement or any other Loan Document or any other document executed in connection herewith shall be deemed to include electronic signatures, the electronic matching of assignment terms and contract formations on electronic platforms approved by Agent, or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature, physical delivery thereof or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any Applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act; provided that notwithstanding anything contained herein to the contrary neither Agent or the Issuer nor any Lender is under any obligation to agree to accept electronic signatures in any form or in any format unless expressly agreed to by Agent, such Issuer or such Lender pursuant to procedures approved by it and provided further without limiting the foregoing, upon the request of any party, any electronic signature shall be promptly followed by such manually executed counterpart.
- 15.18. <u>Confirmation of Flood Policies and Procedures</u>. HSBC has adopted internal policies and procedures that address requirements placed on federally regulated lenders under the National Flood Insurance Reform Act of 1994 and related legislation (the "<u>Flood Laws</u>"). HSBC, as administrative agent or collateral agent on a syndicated facility, will post on the applicable electronic platform (or otherwise distribute to each lender in the syndicate) documents that it receives in connection with the Flood Laws. However, HSBC reminds each Lender, Participant and Purchasing Lender in connection with the facility that, pursuant to the Flood Laws, each federally regulated Lender (whether acting as a Lender or Purchasing Lender in the facility) is responsible for assuring its own compliance with the flood insurance requirements.

15.19. Patriot Act Notice.

Each Lender and Agent (for itself and not on behalf of any Lender) hereby notifies each Loan Party that pursuant to the requirements of the USA PATRIOT Act, it is required to obtain, verify and record, and Borrower shall provide, information that identifies each Loan Party, which information includes the names and addresses of each Loan Party and other information that will allow such Lender or Agent, as applicable, to identify each Loan Party in accordance with the USA PATRIOT Act and other applicable "know your customer" and anti-money laundering rules and regulations under Applicable Law.

15.20. Acknowledgement Regarding Any Supported QFCs.

To the extent that any of this Agreement or the Loan Documents provide support, through a guarantee or otherwise, for Hedging Agreements or any other agreement or instrument that is a QFC (such support, "QFC Credit Support" and each such QFC a "Supported QFC"), the parties acknowledge and agree as follows with respect to the resolution power of the Federal Deposit Insurance Corporation under the Federal Deposit Insurance Act and Title II of the Dodd-Frank Wall Street Reform and Consumer Protection Act (together with the regulations promulgated thereunder, the "U.S. Special Resolution Regimes") in respect of such Supported QFC and QFC Credit Support (with the provisions below applicable notwithstanding that this Agreement, the Loan Documents and any Supported QFC may in fact be stated to be governed by the laws of the State of New York and/or of the United States or any other state of the United States): In the event a Covered Entity that is party to a Supported QFC (each, a "Covered Party") becomes subject to a proceeding under a U.S. Special Resolution Regime, the transfer of such Supported QFC and the benefit of such QFC Credit Support (and any interest and obligation in or under such Supported QFC and such QFC Credit Support, and any rights in property securing such

Supported QFC or such QFC Credit Support) from such Covered Party will be effective to the same extent as the transfer would be effective under the U.S. Special Resolution Regime if the Supported QFC and such QFC Credit Support (and any such interest, obligation and rights in property) were governed by the laws of the United States or a state of the United States. In the event a Covered Party or a BHC Act Affiliate of a Covered Party becomes subject to a proceeding under a U.S. Special Resolution Regime, Default Rights under this Agreement and the other Loan Documents that might otherwise apply to such Supported QFC or any QFC Credit Support that may be exercised against such Covered Party are permitted to be exercised to no greater extent than such Default Rights could be exercised under the U.S. Special Resolution Regime if the Supported QFC and this Agreement and the other Loan Documents were governed by the laws of the United States or a state of the United States. Without limitation of the foregoing, it is understood and agreed that rights and remedies of the parties with respect to a Defaulting Lender shall in no event affect the rights of any Covered Party with respect to a Supported QFC or any QFC Credit Support.

- 15.21. Acknowledgement and Consent to Bail-In of EEA Financial Institutions. Notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any Affected Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the write-down and conversion powers of the applicable Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:
- (a) the application of any Write-Down and Conversion Powers by the applicable Resolution Authority to any such liabilities arising hereunder which may be payable to it by any party hereto that is an Affected Financial Institution; and
 - (b) the effects of any Bail-in Action on any such liability, including, if applicable:
 - (i) a reduction in full or in part or cancellation of any such liability;
- (ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such Affected Financial Institution, its parent undertaking, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or
- (iii) the variation of the terms of such liability in connection with the exercise of the write-down and conversion powers of the applicable Resolution Authority.

15.22. <u>Currency Conversion</u>.

If any payment by Borrower or the proceeds of any Collateral shall be made in a currency other than Dollars, such amount shall be converted into Dollars at the rate determined by Agent or Issuer, as applicable, as the rate quoted by it in accordance with methods customarily used by such Person for such or similar purposes as the spot rate for the purchase by such Person of Dollars with the currency of actual payment through its principal foreign exchange trading office (including, in the case of Agent, any Affiliate), provided that Agent or Issuer, as applicable, may obtain such spot rate from another financial institution actively engaged in foreign currency exchange if Agent or Issuer, as applicable, does not then have a spot rate for Dollars.

15.23. <u>Lender Representations</u>.

Each Lender and Issuer represents that it is engaged in making, acquiring or holding commercial loans in the ordinary course of its business and that it has, independently and without reliance upon Agent or any sub-agents, or any of their officers, directors, employees, agents, advisors, attorneys-in-fact or affiliates thereof (collectively the "Agent Parties"), and based on such documents and information (which may contain material, non-public information within the meaning of the United States securities laws concerning Borrower and its Affiliates) as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender and Issuer also acknowledges that it will, independently and without reliance upon Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed by their respective Authorized Officers as of the day and year first above written.

MARKETWISE, LLC, as Borrower

By: /s/ Mark Arnold

Name: Mark Arnold

Title: Chief Executive Officer

MARKETWISE SOLUTIONS, LLC _, as a Guarantor

By: /s/ Mark Arnold

Name: Mark Arnold

Title: Chief Executive Officer

BONNER AND PARTNERS, LLC, as a Guarantor

By: /s/ Mark Arnold Name: Mark Arnold
Title: Manager

CASEY RESEARCH, LLC, as a Guarantor

By: /s/ Mark Arnold

Name: Mark Arnold Title: Manager

CHAIKIN ANALYTICS, LLC, as a Guarantor

By: /s/ Mark Arnold

Name: Mark Arnold Title: Manager

CHAIKIN RESOURCES, LLC, as a Guarantor

By: /s/ Mark Arnold
Name: Mark Arnold
Title: Manager

COMMON SENSE PUBLISHING, LLC, as a Guarantor

By: /s/ Mark Arnold Name: Mark Arnold

Signature page to Loan and Security Agreement

Title: Manager **INVESTORPLACE MEDIA, LLC, as a Guarantor**

By: /s/ Mark Arnold
Name: Mark Arnold
Title: Manager

INVESTORPLACE MEDIA HOLDINGS, LLC, as a Guarantor

By: /s/ Mark Arnold
Name: Mark Arnold

Title: Manager
LEGACY RESEARCH GROUP, LLC, as a Guarantor

By: /s/ Mark Arnold

Name: Mark Arnold
Title: Manager
OMNIA RESEARCH, LLC, as a Guarantor By: /s/ Mark Arnold

Name: Mark Arnold

Title: Manager
STANSBERRY & ASSOCIATES INVESTMENT RESEARCH, LLC, as a Guarantor

By: /s/ Mark Arnold_

Name: Mark Arnold
Title: Manager

TRADESMITH, LLC, as a Guarantor

By: /s/ Mark Arnold
Name: Mark Arnold
Title: Manager

Signature page to Loan and Security Agreement

HSBC BANK USA, NATIONAL ASSOCIATION, as Agent

By: /s/ Nimish Pandey Name: Nimish Pandey Title: Vice President

HSBC BANK USA, NATIONAL ASSOCIATION,, Issuer and as a Lender

By: /s/ Jan Luehrs

Name: Jan Luehrs

Title: Senior Vice President

WELLS FARGO BANK, N.A., as a Lender By: /s/ Lynn S. Manthy

Name: Lynn S. Manthy Title: Senior Vice President
SILICON VALLEY BANK, as a Lender

By: /s/ Will Deevy
Name: Will Deevy
Title: Managing Director

BMO HARRIS BANK, N.A., as a Lender

By: /s/ Ashley Bake
Name: Ashley Bake
Title: Director

BMO HARRIS BANK, N.A., as a Lender

[OTHER LENDERS]

By: /s/ Timothy N. Naylon

Name: Timothy N. Naylon
Title: Senior Vice President

Signature page to Loan and Security Agreement

Certification by the Chief Executive Officer pursuant to Securities Exchange Act Rules 13a-14(a) and 15d-14(a) as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

I, Amber Mason, certify that:

Date: August 10, 2023

- 1. I have reviewed this Quarterly Report on Form 10-Q of MarketWise, Inc. (the "registrant") for the fiscal quarter ended June 30, 2023;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a–15(e) and 15d–15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a–15(f) and 15d–15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Name:

MarketWise, Inc.

By: /s/ Amber Mason

Title: Chief Executive Officer

Amber Mason

Certification by the Chief Financial Officer pursuant to Securities Exchange Act Rules 13a-14(a) and 15d-14(a) as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

I, Stephen Park, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of MarketWise, Inc. (the "registrant") for the fiscal quarter ended June 30, 2023;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a–15(e) and 15d–15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a–15(f) and 15d–15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Title:

MarketWise, Inc.

Interim Chief Financial Officer

Date: August 10, 2023

By: /s/ Stephen Park

Name: Stephen Park

Certification of Chief Executive Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

I, Amber Mason, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report on Form 10-Q of MarketWise, Inc. for the fiscal quarter ended June 30, 2023 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended, and that information contained in such Quarterly Report on Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of MarketWise, Inc.

MarketWise, Inc.

Date: August 10, 2023 By: /s/ Amber Mason

Name: Amber Mason

Title: Chief Executive Officer

Certification of Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

I, Stephen Park, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report on Form 10-Q of MarketWise, Inc. for the fiscal quarter ended June 30, 2023 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended, and that information contained in such Quarterly Report on Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of MarketWise, Inc.

MarketWise, Inc.

Date: August 10, 2023 By: /s/ Stephen Park

Name: Stephen Park

Title: Interim Chief Financial Officer